SECURITIES AND EXCHANGE COMMISSION

Washington, D.C. 20549

FORM 10-Q

(Mark One)

 \times Quarterly report pursuant to Section 13 or 15(d) of the Securities Exchange Act of 1934 for the quarterly period ended March 31, 2023

Ωr

Transition report pursuant to Section 13 or 15(d) of the Securities Exchange Act of 1934

for the transition period from

Commission file number 001-14431

American States Water Company

(Exact Name of Registrant as Specified in Its Charter)

California 95-4676679

(State or Other Jurisdiction of Incorporation or Organization)

(IRS Employer Identification No.)

630 E. Foothill Blvd

San Dimas

CA

91773-1212

(Address of Principal Executive Offices)

(Zip Code)

(909) 394-3600

(Registrant's Telephone Number, Including Area Code)

Not Applicable

(Former Name, Former Address and Former Fiscal Year, if Changed Since Last Report)

Commission file number 001-12008

Securities registered pursuant to Section 12(b) of the Securities Exchange Act of 1934:

Title of each class Trading symbol Name of each exchange on which registered

Common shares

AWR

New York Stock Exchange

Golden State Water Company

(Exact Name of Registrant as Specified in Its Charter)

California 95-1243678

(State or Other Jurisdiction of Incorporation or Organization)

(IRS Employer Identification No.) 91773-1212

630 E. Foothill Blvd San Dimas CA

(Zip Code)

(Address of Principal Executive Offices)

(909) 394-3600

(Registrant's Telephone Number, Including Area Code)

Not Applicable

(Former Name, Former Address and Former Fiscal Year, if Changed Since Last Report)

Indicate by check mark whether Registrant (1) has filed all reports required to be filed by Section 13 or 15(d) of the Securities Exchange Act of 1934 during the preceding 12 months (or for such shorter period that Registrant was required to file such reports), and (2) has been subject to such filing requirements for the past 90 days.

		States Water Company tate Water Company			Yes Yes			No ⊔ No □	
be submitted and pos	sted pur	•	egulation	2 1			-	Interactive Data File requir s (or such shorter period th	
		States Water Company tate Water Company			Yes Yes	X X	No No		
				celerated filer, an acceled smaller reporting com				smaller reporting company. ct. (Check one):	See
American States Wa	ater Co	mpany							
Large accelerated filer	X	Accelerated filer		Non-accelerated filer		Smaller reporting company		Emerging growth company	
Golden State Water	Comp	any							
Large accelerated filer		Accelerated filer		Non-accelerated filer	X S	Smaller reporting company		Emerging growth company	
or revised financial a	ccounti	ng standards provided	pursuant	the registrant has elected to Section 13(a) of the mpany (as defined in Ru	Exchange	e Act.□	on perio	od for complying with any	new
America	an State	s Water Company		Yes				No 🗵	
Golden	n State	Water Company		Yes				No ⊠	
171 outstanding Com Golden State Water C	nmon Sl Compan	hares of Golden State	Water Co s set forth	mpany were owned by in General Instruction	American	States Water Company	7.	s. As of May 9, 2023, all of nerefore filing this Form, in	

AMERICAN STATES WATER COMPANY and GOLDEN STATE WATER COMPANY FORM 10-Q

INDEX

Part I	Financial Information	
Item 1:	Financial Statements	<u>1</u>
	Consolidated Balance Sheets of American States Water Company as of March 31, 2023 and December 31, 2022	<u>2</u>
	Consolidated Statements of Income of American States Water Company for the Three Months Ended March 31, 2023 and 2022	<u>4</u>
	Consolidated Statements of Changes in Common Shareholders' Equity for the Three Months Ended March 31, 2023 and 2022	<u>5</u>
	Consolidated Statements of Cash Flows of American States Water Company for the Three Months Ended March 31, 2023 and 2022	<u>6</u>
	Balance Sheets of Golden State Water Company as of March 31, 2023 and December 31, 2022	7
	Statements of Income of Golden State Water Company for the Three Months Ended March 31, 2023 and 2022	9
	Statements of Changes in Common Shareholder's Equity for the Three Months Ended March 31, 2023 and 2022	<u>10</u>
	Statements of Cash Flows of Golden State Water Company for the Three Months Ended March 31, 2023 and 2022	<u>11</u>
	Notes to Consolidated Financial Statements	<u>12</u>
Item 2:	Management's Discussion and Analysis of Financial Condition and Results of Operations	<u>24</u>
Item 3:	Quantitative and Qualitative Disclosures About Market Risk	<u>48</u>
Item 4:	Controls and Procedures	<u>48</u>
Part II	Other Information	
Item 1:	<u>Legal Proceedings</u>	<u>49</u>
Item 1A:	Risk Factors	<u>49</u>
Item 2:	<u>Unregistered Sales of Equity Securities and Use of Proceeds</u>	<u>50</u>
Item 3:	<u>Defaults Upon Senior Securities</u>	<u>50</u>
Item 4:	Mine Safety Disclosure	<u>50</u>
Item 5:	Other Information	<u>50</u>
Item 6:	<u>Exhibits</u>	<u>51</u>
	Signatures	53

PART I

Item 1. Financial Statements

General

The basic financial statements included herein have been prepared by Registrant, without audit, pursuant to the rules and regulations of the Securities and Exchange Commission.

Certain information and footnote disclosures normally included in financial statements prepared in accordance with accounting principles generally accepted in the United States of America have been condensed or omitted pursuant to such rules and regulations. In the opinion of management, all adjustments consisting of normal recurring items and estimates necessary for a fair statement of results for the interim period have been made.

It is suggested that these financial statements be read in conjunction with the financial statements and notes thereto in the latest Annual Report on Form 10-K of American States Water Company and its wholly owned subsidiary, Golden State Water Company.

Filing Format

American States Water Company ("AWR") is the parent company of Golden State Water Company ("GSWC"), Bear Valley Electric Service, Inc. ("BVES"), and American States Utility Services, Inc. and its subsidiaries ("ASUS").

This quarterly report on Form 10-Q is a combined report being filed by two separate Registrants: AWR and GSWC. For more information, please see Note 1 of the Notes to Consolidated Financial Statements and the heading entitled "General" in "Item 2. Management's Discussion and Analysis of Financial Condition and Results of Operations." References in this report to "Registrant" are to AWR and GSWC collectively, unless otherwise specified. GSWC makes no representations as to the information contained in this report other than with respect to itself.

Forward-Looking Information

This Form 10-Q and the documents incorporated herein contain forward-looking statements intended to qualify for the "safe harbor" from liability established by the Private Securities Litigation Reform Act of 1995. Forward-looking statements are based on current estimates, expectations and projections about future events and assumptions regarding these events and include statements regarding management's goals, beliefs, plans or current expectations, taking into account the information currently available to management. Forward-looking statements are not statements of historical facts. For example, when we use words such as "anticipate," "believe," "plan," "estimate," "expect," "intend," "may" and other words that convey uncertainty of future events or outcomes, we are making forward-looking statements. We are not able to predict all the factors that may affect future results. We caution you that any forward-looking statements made by us are not guarantees of future performance and the actual results may differ materially from those in our forward-looking statements.

Factors affecting our financial performance are summarized under *Forward-Looking Information* and under "Risk Factors" in our Form 10-K for the period ended December 31, 2022 filed with the SEC. Please consider our forward-looking statements in light of these risks as you read this Form 10-Q. We qualify all of our forward-looking statements by these cautionary statements.

AMERICAN STATES WATER COMPANY CONSOLIDATED BALANCE SHEETS ASSETS (Unaudited)

(in thousands)	March 31, 2023	D	December 31, 2022
Property, Plant and Equipment			
Regulated utility plant, at cost	\$ 2,357,966	\$	2,321,712
Non-utility property, at cost	38,898		38,285
Total	 2,396,864		2,359,997
Less - Accumulated depreciation	(616,403)		(606,231)
Net property, plant and equipment	1,780,461		1,753,766
Other Property and Investments			
Goodwill	1,116		1,116
Other property and investments	38,408		36,907
Total other property and investments	39,524		38,023
Current Assets			
Cash and cash equivalents	2,126		5,997
Accounts receivable — customers (less allowance for doubtful accounts of \$4,364 in 2023 and \$4,387 in 2022)	21,293		26,206
Unbilled receivable	22,444		20,663
Receivable from the U.S. government (Note 2)	41,091		34,974
Other accounts receivable (less allowance for doubtful accounts of \$53 in 2023 and \$53 in 2022)	3,391		4,215
Income taxes receivable	35		3,901
Materials and supplies, at weighted average cost	16,282		14,623
Regulatory assets — current	11,032		14,028
Prepayments and other current assets	11,690		5,450
Purchase power contract derivative at fair value (Note 5)	6,669		11,847
Contract assets (Note 2)	3,637		9,390
Total current assets	139,690		151,294
Other Assets			
Unbilled revenue — receivable from the U.S. government (Note 2)	7,790		6,456
Receivable from the U.S. government (Note 2)	50,207		50,482
Contract assets (Note 2)	9,836		5,592
Operating lease right-of-use assets	8,990		9,535
Regulatory assets	10,474		5,694
Other	13,400		13,532
Total other assets	100,697		91,291
Total Assets	\$ 2,060,372	\$	2,034,374

AMERICAN STATES WATER COMPANY CONSOLIDATED BALANCE SHEETS CAPITALIZATION AND LIABILITIES (Unaudited)

(in thousands, except number of shares)	March 31, 2023	December 31, 2022
Capitalization		
Common shares, no par value		
Authorized: 60,000,000 shares		
Outstanding: 36,976,284 shares in 2023 and 36,962,241 shares in 2022	\$ 261,792	\$ 260,158
Retained earnings	469,056	449,391
Total common shareholders' equity	730,848	709,549
Long-term debt	576,431	446,547
Total capitalization	1,307,279	1,156,096
Current Liabilities		
Notes payable to banks	175,500	255,500
Long-term debt — current	404	399
Accounts payable	65,705	84,849
Income taxes payable	10,855	1,848
Accrued other taxes	14,257	16,257
Accrued employee expenses	17,390	13,996
Accrued interest	8,411	5,308
Regulatory liabilities	2,097	4,574
Contract liabilities (Note 2)	560	903
Operating lease liabilities	1,906	1,892
Other	10,605	10,996
Total current liabilities	307,690	396,522
Other Credits		
Notes payable to banks	25,000	22,000
Advances for construction	64,097	64,351
Contributions in aid of construction - net	148,456	147,918
Deferred income taxes	148,138	149,677
Regulatory liabilities	1,783	40,602
Unamortized investment tax credits	1,064	1,082
Accrued pension and other postretirement benefits	34,796	33,636
Operating lease liabilities	7,559	8,090
Other	14,510	14,400
Total other credits	445,403	481,756
Commitments and Contingencies (Note 9)		
Trad Contablacton and Linkillian	¢ 0.000.070	¢ 2.024.274
Total Capitalization and Liabilities	\$ 2,060,372	\$ 2,034,374

AMERICAN STATES WATER COMPANY CONSOLIDATED STATEMENTS OF INCOME FOR THE THREE MONTHS ENDED MARCH 31, 2023 AND 2022 (Unaudited)

	Three Months Ended March 31,					
(in thousands, except per share amounts)		2023		2022		
Operating Revenues						
Water	\$	112,712	\$	73,906		
Electric		12,904		11,892		
Contracted services		35,807		22,772		
Total operating revenues		161,423		108,570		
Operating Expenses						
Water purchased		14,304		17,848		
Power purchased for pumping		2,354		2,374		
Groundwater production assessment		3,833		4,211		
Power purchased for resale		4,986		5,166		
Supply cost balancing accounts		11,566		(6,343)		
Other operation		10,116		8,667		
Administrative and general		23,547		22,972		
Depreciation and amortization		11,203		10,114		
Maintenance		3,150		3,140		
Property and other taxes		6,295		5,853		
ASUS construction		18,904		10,203		
Total operating expenses		110,258		84,205		
Operating Income		51,165		24,365		
Other Income and Expenses						
Interest expense		(9,481)		(5,606)		
Interest income		1,864		283		
Other, net		1,611		(419)		
Total other income and expenses, net		(6,006)		(5,742)		
Income before income tax expense		45,159		18,623		
Income tax expense		10,752		4,461		
Net Income	\$	34,407	\$	14,162		
Weighted Average Number of Common Shares Outstanding		36,968		36,944		
Basic Earnings Per Common Share	\$	0.93	\$	0.38		
Weighted Average Number of Diluted Shares		37,047		37,019		
Fully Diluted Earnings Per Common Share	\$	0.93	\$	0.38		
Dividends Paid Per Common Share	\$	0.3975	\$	0.365		

AMERICAN STATES WATER COMPANY CONSOLIDATED STATEMENTS OF CHANGES IN COMMON SHAREHOLDERS' EQUITY (Unaudited)

	Three Months Ended March 31, 2023					
	Common Shares					
	Number		_			
	of			Retained		
(in thousands)	Shares	Amount		Earnings		Total
Balances at December 31, 2022	36,962	\$ 260,158	\$	449,391	\$	709,549
Add:						
Net income				34,407		34,407
Exercise of stock options and other issuances of Common Shares	14					
Stock-based compensation, net of taxes paid from shares withheld from employees related to net share settlements (Note 4)		1,587				1,587
Dividend equivalent rights on stock-based awards not paid in cash		47				47
Deduct:						
Dividends on Common Shares				14,695		14,695
Dividend equivalent rights on stock-based awards not paid in cash				47		47
Balances at March 31, 2023	36,976	\$ 261,792	\$	469,056	\$	730,848

	Three Months Ended March 31, 2022				
	Common S	Shares			
(in thousands)	Number of Shares	Amount	Retained Earnings	Total	
Balances at December 31, 2021	36,936	\$ 258,442	\$ 427,505	\$ 685,947	
Add:					
Net income			14,162	14,162	
Exercise of stock options and other issuances of Common Shares	20	_		_	
Stock-based compensation, net of taxes paid from shares withheld from employees related to net share settlements (Note 4)		801		801	
Dividend equivalent rights on stock-based awards not paid in cash		41		41	
Deduct:					
Dividends on Common Shares			13,485	13,485	
Dividend equivalent rights on stock-based awards not paid in cash			41	41	
Balances at March 31, 2022	36,956	\$ 259,284	\$ 428,141	\$ 687,425	

AMERICAN STATES WATER COMPANY CONSOLIDATED STATEMENTS OF CASH FLOWS FOR THE THREE MONTHS ENDED MARCH 31, 2023 AND 2022 (Unaudited)

	Three Months Ended March 31,		
(in thousands)	 2023	2	022
Cash Flows From Operating Activities:			
Net income	\$ 34,407	\$	14,162
Adjustments to reconcile net income to net cash provided by operating activities:			
Depreciation and amortization	11,572		10,208
Provision for doubtful accounts	458		258
Deferred income taxes and investment tax credits	(3,286)		1,552
Stock-based compensation expense	2,254		1,905
(Gain) loss on investments held in a trust	(1,630)		1,653
Other — net	(71)		83
Changes in assets and liabilities:			
Accounts receivable — customers	4,455		7,627
Unbilled receivable	(3,115)		2,451
Other accounts receivable	824		3,467
Receivables from the U.S. government	(5,842)		4,589
Materials and supplies	(1,659)		220
Prepayments and other assets	(5,656)		(4,584)
Contract assets	1,509		(199)
Regulatory assets/liabilities	(35,863)		(5,713)
Accounts payable	(8,542)		341
Income taxes receivable/payable	12,873		(1,592)
Contract liabilities	(343)		(47)
Accrued pension and other postretirement benefits	1,020		71
Other liabilities	 3,599		1,574
Net cash provided by	 6,964		38,026
Cash Flows From Investing Activities:			
Capital expenditures	(49,337)		(35,170)
Other investing activities	172		121
Net cash used in	(49,165)		(35,049)
Cash Flows From Financing Activities:			
Receipt of advances for and contributions in aid of construction	2,064		1,795
Refunds on advances for construction	(712)		(833)
Repayments of long-term debt	(109)		(103)
Proceeds from the issuance of long-term debt, net of issuance costs	129,665		_
Net change in notes payable to banks	(77,000)		16,000
Dividends paid	(14,695)		(13,485)
Other financing activities	(883)		(1,188)
Net cash provided by	38,330		2,186
Net change in cash and cash equivalents	(3,871)		5,163
Cash and cash equivalents, beginning of period	5,997		4,963
Cash and cash equivalents, end of period	\$	\$	10,126
Non-cash transactions:			
Accrued payables for investment in utility plant	\$ 29,746	\$	34,101
Property installed by developers and conveyed	\$ 	\$	130

GOLDEN STATE WATER COMPANY BALANCE SHEETS ASSETS (Unaudited)

(in thousands)	March 31, 2023	December 31, 2022
Utility Plant		
Utility plant, at cost	\$ 2,177,606	\$ 2,147,643
Less - Accumulated depreciation	(539,575)	(530,925)
Net utility plant	 1,638,031	1,616,718
Other Property and Investments	36,159	34,655
·		
Current Assets		
Cash and cash equivalents	813	370
Accounts receivable — customers (less allowance for doubtful accounts of \$4,111 in 2023 and \$4,143 in 2022)	18,697	23,107
Unbilled receivable	13,399	15,006
Other accounts receivable (less allowance for doubtful accounts of \$53 in 2023 and \$53 in 2022)	2,316	2,721
Intercompany receivable	_	621
Income taxes receivable from Parent	_	1,692
Materials and supplies, at average cost	6,338	6,120
Regulatory assets — current	11,032	14,028
Prepayments and other current assets	8,347	4,464
Total current assets	60,942	68,129
	_	
Other Assets		
Operating lease right-of-use assets	8,703	9,208
Other	12,535	12,598
Total other assets	21,238	21,806
Total Assets	\$ 1,756,370	\$ 1,741,308

GOLDEN STATE WATER COMPANY BALANCE SHEETS CAPITALIZATION AND LIABILITIES (Unaudited)

(in thousands, except number of shares)	March 31, 2023	ember 31, 2022	
Capitalization			
Common Shares, no par value:			
Authorized: 1,000 shares			
Outstanding: 171 shares in 2023 and 170 in 2022	\$ 369,770	\$ 358,123	
Retained earnings	288,502	285,783	
Total common shareholder's equity	658,272	643,900	
Long-term debt	541,627	411,748	
Total capitalization	1,199,899	1,055,654	
Current Liabilities			
Intercompany payable	428	_	
Intercompany note payable	45,000	_	
Long-term debt — current	404	39	
Accounts payable	47,860	65,94	
Accrued other taxes	12,248	14,50	
Accrued employee expenses	13,841	11,23	
Accrued interest	7,144	4,36	
Income taxes payable to Parent	10,394	_	
Operating lease liabilities	1,794	1,78	
Other	9,754	10,15	
Total current liabilities	148,867	108,38	
Other Credits			
Intercompany note payable		129,000	
Advances for construction	64,077	64,33	
Contributions in aid of construction — net	148,456	147,91	
Deferred income taxes	135,921	138,78	
Regulatory liabilities	1,783	40,602	
Unamortized investment tax credits	1,064	1,08	
Accrued pension and other postretirement benefits	34,556	33,42	
Operating lease liabilities	7,397	7,87	
Other	14,350	14,25	
Total other credits	407,604	 577,273	
Commitments and Contingencies (Note 9)			
Total Capitalization and Liabilities	\$ 1,756,370	\$ 1,741,308	

GOLDEN STATE WATER COMPANY STATEMENTS OF INCOME FOR THE THREE MONTHS ENDED MARCH 31, 2023 AND 2022 (Unaudited)

	Three Months Ended March 31,					
(in thousands)	20	23	2022			
Operating Revenues						
Water	\$	112,712 \$	73,906			
Total operating revenues		112,712	73,906			
Operating Expenses						
Water purchased		14,304	17,848			
Power purchased for pumping		2,354	2,374			
Groundwater production assessment		3,833	4,211			
Supply cost balancing accounts		12,625	(5,067)			
Other operation		7,271	6,354			
Administrative and general		15,381	15,596			
Depreciation and amortization		9,606	8,545			
Maintenance		1,960	2,156			
Property and other taxes		5,139	4,890			
Total operating expenses		72,473	56,907			
Operating Income		40,239	16,999			
Other Income and Expenses						
Interest expense		(6,922)	(5,236)			
Interest income		1,428	91			
Other, net		1,628	(598)			
Total other income and expenses, net		(3,866)	(5,743)			
Income before income tax expense		36,373	11,256			
Income tax expense		8,910	2,689			
Net Income	\$	27,463 \$	8,567			

GOLDEN STATE WATER COMPANY STATEMENTS OF CHANGES IN COMMON SHAREHOLDER'S EQUITY (Unaudited)

	Three Months Ended March 31, 2023			
	Common S	hares		
	Number	_		
	of		Retained	
(in thousands, except number of shares)	Shares	Amount	Earnings	Total
Balances at December 31, 2022	170 \$	358,123	\$ 285,783	\$ 643,906
Add:				
Net income			27,463	27,463
Issuance of Common Share to Parent	1	10,000		10,000
Stock-based compensation, net of taxes paid from shares withheld from employees related to net share settlements (Note 4)		1,603		1,603
Dividend equivalent rights on stock-based awards not paid in cash		44		44
Deduct:				
Dividends on Common Shares			24,700	24,700
Dividend equivalent rights on stock-based awards not paid in cash			44	44
Balances at March 31, 2023	171 \$	369,770	\$ 288,502	\$ 658,272

	Three Months Ended March 31, 2022						
	Common S	hares					
	Number						
	of		Retained				
(in thousands, except number of shares)	Shares	Amount	Earnings		Total		
Balances at December 31, 2021	170 \$	356,530	\$ 259,15	6 \$	615,686		
Add:							
Net income			8,56	7	8,567		
Stock-based compensation, net of taxes paid from shares withheld from							
employees related to net share settlements (Note 4)		742			742		
Dividend equivalent rights on stock-based awards not paid in cash		39			39		
Deduct:							
Dividends on Common Shares			13,50	0	13,500		
Dividend equivalent rights on stock-based awards not paid in cash			3	9	39		
Balances at March 31, 2022	170 \$	357,311	\$ 254,18	4 \$	611,495		

GOLDEN STATE WATER COMPANY STATEMENTS OF CASH FLOWS FOR THE THREE MONTHS ENDED MARCH 31, 2023 AND 2022 (Unaudited)

		Three Mor Mar			
(in thousands)		2023		2022	
Cash Flows From Operating Activities:					
Net income	\$	27,463	\$	8,567	
Adjustments to reconcile net income to net cash provided by operating activities:					
Depreciation and amortization		9,947		8,610	
Provision for doubtful accounts		420		222	
Deferred income taxes and investment tax credits		(4,348)		1,305	
Stock-based compensation expense		2,190		1,751	
(Gain) loss on investments held in a trust		(1,630)		1,653	
Other — net		(105)		84	
Changes in assets and liabilities:					
Accounts receivable — customers		3,990		7,499	
Unbilled receivable		1,607		4,183	
Other accounts receivable		405		1,944	
Materials and supplies		(218)		797	
Prepayments and other assets		(3,380)		(3,268)	
Regulatory assets/liabilities		(34,059)		(5,135)	
Accounts payable		(7,831)		2,886	
Intercompany receivable/payable		1,077		428	
Income taxes receivable/payable from/to Parent		12,086		(1,791)	
Accrued pension and other postretirement benefits		1,004		36	
Other liabilities		2,259		1,494	
Net cash provided by		10,877		31,265	
Cash Flows From Investing Activities:					
Capital expenditures		(42,005)		(31,465)	
Other investing activities		171		117	
Net cash used in		(41,834)		(31,348)	
Cash Flows From Financing Activities:					
Proceeds from issuance of Common Shares to Parent		10,000		_	
Receipt of advances for and contributions in aid of construction		2,064		1,759	
Refunds on advances for construction		(712)		(833)	
Repayments of long-term debt		(109)		(103)	
Proceeds from the issuance of long-term debt, net of issuance costs		129,665		_	
Net change in intercompany borrowings		(84,000)		18,000	
Dividends paid		(24,700)		(13,500)	
Other financing activities		(808)		(1,088)	
Net cash provided by		31,400		4,235	
Net change in cash and cash equivalents		443		4,152	
Cash and cash equivalents, beginning of period		370			
Cash and cash equivalents, beginning of period	\$	813	\$	525 4,677	
	<u>-</u>			, .	
Non-cash transactions:					
Accrued payables for investment in utility plant	\$	28,363	\$	32,439	
Property installed by developers and conveyed	\$	364	\$	130	

AMERICAN STATES WATER COMPANY AND SUBSIDIARIES AND

GOLDEN STATE WATER COMPANY NOTES TO CONSOLIDATED FINANCIAL STATEMENTS (Unaudited)

Note 1 — Summary of Significant Accounting Policies

Nature of Operations: American States Water Company ("AWR") is the parent company of Golden State Water Company ("GSWC"), Bear Valley Electric Service, Inc. ("BVES"), and American States Utility Services, Inc. ("ASUS") (and its wholly owned subsidiaries: Fort Bliss Water Services Company ("FBWS"), Terrapin Utility Services, Inc. ("TUS"), Old Dominion Utility Services, Inc. ("ODUS"), Palmetto State Utility Services, Inc. ("PSUS"), Old North Utility Services, Inc. ("ONUS"), Emerald Coast Utility Services, Inc. ("ECUS"), and Fort Riley Utility Services, Inc. ("FRUS")). The subsidiaries of ASUS are collectively referred to as the "Military Utility Privatization Subsidiaries". AWR, through its wholly owned subsidiaries, serves over one million people in nine states.

GSWC and BVES are both California public utilities. GSWC is engaged in the purchase, production, distribution and sale of water throughout California serving approximately 263,400 customer connections. BVES distributes electricity in several San Bernardino County mountain communities in California serving approximately 24,700 customer connections. The California Public Utilities Commission ("CPUC") regulates GSWC's and BVES's businesses in matters including properties, rates, services, facilities, and transactions between GSWC, BVES, and their affiliates.

ASUS, through its wholly owned subsidiaries, operates, maintains and performs construction activities (including renewal and replacement capital work) on water and/or wastewater systems at various U.S. military bases pursuant to an initial 50-year firm fixed-price contracts with the U.S. government. These contracts are subject to annual economic price adjustments and modifications for changes in circumstances, changes in laws and regulations, and additions to the contract value for new construction of facilities at the military bases.

There is no direct regulatory oversight by the CPUC over AWR or the operations, rates or services provided by ASUS or the Military Utility Privatization Subsidiaries.

Basis of Presentation: The consolidated financial statements and notes hereto are presented in a combined report filed by two separate Registrants: AWR and GSWC. References in this report to "Registrant" are to AWR and GSWC, collectively, unless otherwise specified. AWR owns all of the outstanding common shares of GSWC, BVES and ASUS. ASUS owns all of the outstanding common stock of the Military Utility Privatization Subsidiaries. The consolidated financial statements of AWR include the accounts of AWR and its subsidiaries. These financial statements are prepared in conformity with accounting principles generally accepted in the United States of America ("GAAP"). Intercompany transactions and balances have been eliminated in the AWR consolidated financial statements.

The consolidated financial statements included herein have been prepared by Registrant, without audit, pursuant to the rules and regulations of the Securities and Exchange Commission ("SEC"). The December 31, 2022 condensed consolidated balance sheet data was derived from audited financial statements but does not include all disclosures required by GAAP. The preparation of the consolidated financial statements requires management to make estimates and assumptions that affect the reported amounts of assets and liabilities, and disclosure of contingent assets and liabilities, at the date of the financial statements, and the reported amounts of revenues and expenses during the reporting period. Actual results could differ from these estimates. In the opinion of management, all adjustments consisting of normal, recurring items, and estimates necessary for a fair statement of the results for the interim periods have been made. It is suggested that these consolidated financial statements be read in conjunction with the consolidated financial statements and the notes thereto included in the Form 10-K for the year ended December 31, 2022 filed with the SEC.

Related Party and Intercompany Transactions: As discussed below under *Liquidity and Financing Plans*, AWR currently borrows under a credit facility and provides funds to GSWC and ASUS in support of their operations. Furthermore, GSWC, BVES and ASUS provide and/ or receive various support service to and from their parent, AWR, and among themselves. GSWC also allocates certain corporate office administrative and general costs to its affiliates, BVES and ASUS, using allocation factors approved by the CPUC. During the three months ended March 31, 2023 and 2022, GSWC allocated corporate office administrative and general costs to BVES of approximately \$1.3 million and \$794,000, respectively. During the three months ended March 31, 2023 and 2022, GSWC allocated corporate office administrative and general costs to ASUS of approximately \$1.5 million and \$1.6 million, respectively.

In January 2023, the Board of Directors approved the issuance of one GSWC Common Share to AWR for \$10.0 million. GSWC used the proceeds from the issuance of equity to AWR and from the issuance of \$130.0 million in unsecured long-term notes on January 13, 2023 to pay-off all intercompany borrowings from AWR. The CPUC requires GSWC to fully pay-off all intercompany borrowings it has from AWR within a 24-month period.

Liquidity and Financing Plans: AWR borrows under a revolving credit facility with a current borrowing capacity of \$280.0 million and provides funds to GSWC and ASUS in support of their operations through intercompany borrowing agreements on terms that are similar to that of the credit facility. The interest rate charged to GSWC and ASUS is comparable to the interest rate AWR pays under the credit facility. AWR's credit agreement was set to expire on May 23, 2023. On May 8, 2023, the credit facility was amended to extend the maturity date by two-months to provide adequate time to put in place a new credit agreement. The amendment extends the maturity date of the existing credit agreement to July 23, 2023 or an earlier date on which the credit agreement is either terminated or cancelled when superseded by a new agreement. All intercompany borrowing agreements will expire concurrent with the expiration of AWR's credit facility. Therefore, the outstanding borrowings under the credit facility of \$175.5 million as of March 31, 2023 have been classified as current liabilities on AWR's Consolidated Balance Sheet, thus creating a negative working-capital condition for AWR of \$168.0 million. Additionally, as of March 31, 2023, the \$45.0 million of outstanding intercompany borrowings of GSWC from AWR have been classified as current liabilities on GSWC's Balance Sheet, also creating a negative working-capital condition for GSWC of \$87.9 million. As of May 10, 2023, neither AWR nor GSWC have sufficient liquidity or capital resources to repay its credit facility or intercompany borrowings, respectively, without either extending its existing credit facility, entering into a new credit facility, or issuing new debt or equity.

AWR is confident and believes it is probable that it will be able to execute a new credit facility agreement with the needed borrowing capacities required to repay its existing credit facility and to run its operations given Registrant's ability to generate consistent cash flows, its A+ credit ratings, and its history in obtaining revolving credit facilities to meet its working-capital needs, as well as its history of successfully raising debt as recently done with GSWC's issuance of \$130.0 million in unsecured long-term notes on January 13, 2023. In addition, management is considering a separate credit facility for GSWC to support its standalone water utility operations. Alternatively, AWR may enter into a new intercompany borrowing agreement with GSWC. Accordingly, management has concluded that Registrant will be able to satisfy its obligations, including those under its current credit facility, for at least the next twelve months from the issuance date of these financial statements. However, Registrant's ability to access the capital markets or to otherwise obtain sufficient financing may be affected by future conditions and, accordingly, no assurances can be made that Registrant will be successful in implementing its financing plans.

BVES has a separate \$35.0 million revolving credit facility without a parent guaranty that matures on July 1, 2024. As of March 31, 2023, there was \$25.0 million outstanding borrowing under this credit facility. Under the terms of the credit agreement, BVES has the option to increase the facility by an additional \$15.0 million, subject to lender approval. Interest rates under this facility are currently based on LIBOR. Effective July 1, 2023, all new borrowings under this credit agreement will be based on the Secured Overnight Financing Rate ("SOFR"). BVES does not believe the change from LIBOR to a new benchmark rate such as SOFR will have a material impact on its financing costs.

COVID-19 Impact: AWR and its subsidiaries continue to monitor the guidance provided by federal, state, and local health authorities and other government officials. On April 10, 2023, the Biden Administration terminated the COVID-19 national emergency. The COVID-19 emergency-related memorandum accounts for GSWC and BVES expired when the COVID-19 national emergency ended. Thus far, the COVID-19 pandemic has not had a material impact on ASUS's current operations.

During the first quarter of 2023, GSWC and BVES continue to incur some incremental costs in excess of their revenue requirements due to the lingering effects of the pandemic that are being tracked in COVID-19-related memorandum accounts and recorded as regulatory assets (Note 3).

Note 2 — Revenues

Most of Registrant's revenues are derived from contracts with customers, including tariff-based revenues from its regulated utilities at GSWC and BVES. ASUS's initial 50-year firm fixed-price contracts with the U.S. government are considered service concession arrangements under ASC 853, Service Concession Arrangements. Accordingly, the services under these contracts are accounted for under Topic 606—Revenue from Contracts with Customers, and the water and/or wastewater systems are not recorded as Property, Plant and Equipment on the Registrant's balance sheets.

Although GSWC and BVES have a diversified customer base of residential, commercial, industrial, and other customers, revenues derived from residential and commercial customers generally account for approximately 90% of total water and electric revenues. Most of ASUS's revenues are derived from contracts with the U.S. government. For the three months ended March 31, 2023 and 2022, disaggregated revenues from contracts with customers by segment were as follows:

	Three Months Ended March 31,					
(dollars in thousands)		2023		2022		
Water:						
Tariff-based revenues	\$	100,541	\$	72,498		
Surcharges (cost-recovery activities)		317		549		
Other		737		518		
Water revenues from contracts with customers		101,595		73,565		
WRAM under/(over) collection (alternative revenue program)		11,117		341		
Total water revenues (1)		112,712		73,906		
Electric:						
Tariff-based revenues		13,063		12,552		
Surcharges (cost-recovery activities)		149		27		
Electric revenues from contracts with customers		13,212		12,579		
BRRAM under/(over) collection (alternative revenue program)		(308)		(687)		
Total electric revenues		12,904		11,892		
Contracted services:						
Water		22,488		13,546		
Wastewater		13,319		9,226		
Contracted services revenues from contracts with customers		35,807		22,772		
Total AWR revenues	\$	161,423	\$	108,570		

⁽¹⁾ Water revenues for the three months ended March 31, 2023 includes approximately \$30 million, which represents the impact of the retroactive new rates for the full year of 2022 as a result of a proposed decision issued by the CPUC in April 2023 on GSWC's general rate case (Note 3).

The opening and closing balances of unbilled receivables, receivable from the U.S. government, contract assets and contract liabilities from contracts with customers, which are related entirely to ASUS, were as follows:

(dollars in thousands)	Marc	ch 31, 2023	De	ecember 31, 2022
Unbilled receivables	\$	15,167	\$	10,125
Receivable from the U.S. government	\$	91,298	\$	85,456
Contract assets	\$	13,473	\$	14,982
Contract liabilities	\$	560	\$	903

Unbilled receivables and Receivable from the U.S. government represent receivables where the right to payment is conditional only by the passage of time.

Contract Assets - Contract assets are assets of ASUS and consist of unbilled revenues recognized from work-in-progress construction projects, where the right to payment is conditional on something other than the passage of time. The classification of this asset as current or noncurrent is based on the timing of when ASUS expects to bill these amounts.

Contract Liabilities - Contract liabilities are liabilities of ASUS and consist of billings in excess of revenue recognized. The classification of this liability as current or noncurrent is based on the timing of when ASUS expects to recognize revenue.

Revenues for the three months ended March 31, 2023, which were included in contract liabilities at the beginning of the period were not material. Contracted services revenues recognized during the three months ended March 31, 2023 from performance obligations satisfied in previous periods were not material.

As of March 31, 2023, AWR's aggregate remaining performance obligations, which are entirely for the contracted services segment, were \$3.5 billion. AWR expects to recognize revenue on these remaining performance obligations over the remaining term of each of the 50-year contracts, which range from 32 to 45 years. Each of the contracts with the U.S. government is subject to termination, in whole or in part, prior to the end of its 50-year term for convenience of the U.S. government.

Note 3 — Regulatory Matters

In accordance with accounting principles for rate-regulated enterprises, GSWC and BVES record regulatory assets, which represent probable future recovery of costs from customers through the rate making process, and regulatory liabilities, which represent probable future refunds that are to be credited to customers through the rate making process. At March 31, 2023, GSWC and BVES had approximately \$77.6 million of regulatory liabilities, net of regulatory assets, not accruing carrying costs. Of this amount, (i) \$75.2 million of regulatory liabilities are excess deferred income taxes arising from the lower federal income tax rate under the Tax Cuts and Jobs Act enacted in December 2017 that are being refunded to customers, (ii) \$2.3 million of regulatory assets relates to the underfunded position in Registrant's pension and other retirement obligations (not including the two-way pension balancing accounts), and (iii) \$6.7 million regulatory liability related to a memorandum account authorized by the CPUC to track unrealized gains and losses on BVES's purchase power contracts over the term of the contracts. The remainder relates to other items that do not provide for or incur carrying costs including flowed-through deferred income taxes.

Regulatory assets represent costs incurred by GSWC and/or BVES for which they have received or expect to receive rate recovery in the future. In determining the probability of costs being recognized in other periods, GSWC and BVES consider regulatory rules and decisions, past practices, and other facts or circumstances that would indicate if recovery is probable. If the CPUC determines that a portion of either GSWC's or BVES's assets are not recoverable in customer rates, the applicable utility must determine if it has suffered an asset impairment that requires it to write down the asset's value. Regulatory assets are offset against regulatory liabilities within each rate making area. Amounts expected to be collected or refunded in the next twelve months have been classified as current assets and current liabilities by rate making area. Regulatory liabilities, less regulatory assets, included in the consolidated balance sheets are as follows:

ollars in thousands)		March 31, 2023		December 31, 2022
GSWC				
2022/2023 general rate case memorandum accounts (unbilled revenue)	\$	38,419	\$	_
Water revenue adjustment mechanism, net of the modified cost balancing account		29,069		31,803
COVID-19 memorandum account		3,540		3,478
Excess deferred income taxes		(71,418)		(71,870)
Flowed-through income taxes		447		(1,134)
Other regulatory assets		22,390		19,964
Other regulatory liabilities		(13,198)		(8,815)
Total GSWC	\$	9,249	\$	(26,574)
BVES				
Derivative instrument memorandum account (Note 5)		(6,669)		(11,847)
Wildfire mitigation and other fire prevention related costs memorandum accounts		14,131		13,007
Other regulatory assets		9,480		7,965
Other regulatory liabilities		(8,565)		(8,005)
Total AWR	\$	17,626	\$	(25,454)

Regulatory matters are discussed in the consolidated financial statements and the notes thereto included in the Company's Form 10-K for the year ended December 31, 2022 filed with the SEC. The discussion below focuses on significant matters and developments since December 31, 2022.

Pending Water General Rate Case and the 2022/2023 General Rate Case Memorandum Accounts:

In July 2020, GSWC filed a general rate case application for all of its water regions and its general office. This general rate case determines new water rates for the years 2022–2024. On April 13, 2023, GSWC received a proposed decision on this application from the assigned administrative law judge at the CPUC. Among other things, the proposed decision approves and adopts in its entirety the settlement agreement between GSWC and the Public Advocates Office at the CPUC ("Public Advocates") that had been filed with the CPUC in November 2021, and resolves all issues related to the 2022 annual revenue requirement in the general rate case application and allows for additional increases in adopted revenues for 2023. The new rates

for 2022 and 2023 are effective and retroactive to January 1, 2022 and January 1, 2023, respectively. As a result of receiving a proposed decision that approves the settlement agreement in its entirety, the impact of retroactive rates for the full year of 2022 and the estimated second-year rate increases for the three months ended March 31, 2023 have been reflected in the 2023 first quarter results as it became probable that the approved retroactive rates for the full year of 2022 and the first three months of 2023 would be permitted to be billed to customers in the future. GSWC expects to receive a final decision during the second quarter of 2023.

Due to the delay in finalizing the water general rate case, water revenues billed to customers for the year ended December 31, 2022 and for the three months ended March 31, 2023 were based on 2021 adopted rates, pending a final decision by the CPUC. GSWC has been authorized to create general rate case memorandum accounts to track the revenue differences between the 2021 adopted rates and the new 2022 and 2023 rates authorized by the CPUC. As of March 31, 2023, there is an aggregate cumulative amount of \$38.4 million in the general rate case memorandum accounts that relates to water revenues recorded during the three months ended March 31, 2023, and which represent the difference between the 2021 adopted rates billed to customers and the rates authorized in the proposed decision for the full year of 2022 and the estimated increases for the period ended March 31, 2023. Additional increases in adopted revenues for 2023 are subject to an earnings test and changes to the forecasted inflationary index values. The best estimate of 2023 rate increases have been computed at this time using inflationary index values as of March 31, 2023. Actual increases for 2023 will be determined when the filings to implement the new rate increases are approved by the CPUC, and will be calculated using the inflationary index values at that time. GSWC will file for the 2023 increases once the CPUC approves the final decision. Once a final decision is issued by the CPUC, GSWC will also request recovery through a surcharge of the cumulative amounts included in the general rate case memorandum accounts.

Furthermore, the proposed decision addressed the three remaining unresolved issues related to GSWC's requests for: (i) a medical insurance cost balancing account, (ii) a general liability insurance cost balancing account, and (iii) the consolidation of two of GSWC's customer service areas. The proposed decision approved both balancing accounts and denied GSWC's consolidation of its two customer service areas. The proposed decision also approved the recovery of previously incurred costs that were being tracked in other CPUC-authorized memorandum accounts. GSWC recorded the amounts tracked in the balancing and memorandum accounts that are being approved in the proposed decision, the net impact of which was not material to 2023 first quarter results.

Alternative-Revenue Programs:

GSWC currently records the difference between what it bills its water customers and what is authorized by the CPUC using the Water Revenue Adjustment Mechanism ("WRAM") and the Modified Cost Balancing Account ("MCBA") approved by the CPUC. The over- or under-collection of the WRAM is aggregated with the MCBA over- or under-collection for the corresponding rate making area and bears interest at the current 90-day commercial-paper rate.

During the three months ended March 31, 2023, GSWC recorded net under-collections in the WRAM/MCBA accounts of approximately \$9.5 million due to lower-than-adopted water usage and to reflect the authorized 2023 amounts. In addition, GSWC recorded a net reduction of \$9.8 million of under-collection during the first quarter of 2023 to reflect the cumulative full-year impact of 2022 based on authorized 2022 amounts for both WRAM/MCBA accounts as a result of receiving the proposed water general rate case decision. Surcharges for the 2022 WRAM/ MCBA balance is expected to be requested after the final CPUC-decision is received on GSWC's general rate case. Surcharges and surcredits have been implemented for all pre-2022 WRAM/MCBA balances. As of March 31, 2023, GSWC had an aggregated regulatory asset of \$29.1 million, which is comprised of a \$28.9 million under-collection in the WRAM accounts and a \$189,000 under-collection in the MCBA accounts.

As required by the accounting guidance for alternative revenue programs, GSWC is required to collect its WRAM balances within 24 months following the year in which an under-collection is recorded. As of March 31, 2023, there were no material WRAM under-collections that were estimated to be collected over more than 24 months.

Cost of Capital Proceeding:

GSWC filed a cost of capital application in May 2021 currently pending CPUC approval. Hearings on this proceeding occurred in May 2022 and briefs were filed in June 2022. On May 9, 2023, GSWC received a proposed decision from the assigned administrative law judge at the CPUC on the cost of capital proceeding. Among other things, the proposed decision (i) adopts GSWC's requested capital structure and cost of debt filed in the application; (ii) adopts a return on equity of 8.85% for GSWC as compared to 8.9% previously authorized; (iii) allows for the continuation of the Water Cost of Capital Mechanism ("WCCM"); and (iv) adopts the new cost of capital for the three-year period commencing January 1, 2022, through December 31, 2024. Comments on the proposed decision are due on May 30. In March 2023, the CPUC issued a decision that approved an extension of the statutory deadline for a final decision in the cost of capital proceeding to August 10, 2023.

Based on management's analysis of this regulatory proceeding and the associated accounting to date, for the three months ended March 31, 2023 and 2022, GSWC reduced revenues by \$1.8 million, and \$1.4 million, respectively, and recorded a corresponding regulatory liability for revenues subject to refund based on its best estimate, which relates to the impact of GSWC's lower cost of debt requested in its application and adopted in the proposed decision. Also, an additional reduction to revenues of \$1.1 million was recorded during the first quarter of 2023 to reflect the incremental impact of revenues subject to refund from the new 2022 rates in the proposed water general rate case decision that results from the lower cost of debt in the

pending cost of capital proceeding. As of March 31, 2023, GSWC had an aggregated regulatory liability of \$9.3 million for the estimated revenues subject to refund from the pending cost of capital proceeding. However, at this time, management cannot predict the ultimate outcome and any changes that may be made to the final decision in the cost of capital application, and the associated impact on 2022 and 2023 revenues. Changes in estimates will be made, if necessary, as more information in this proceeding becomes available.

Furthermore, the proposed decision continues the WCCM for the years 2023 and 2024, which adjusts the return on equity and rate of return on rate base between the three-year cost of capital proceedings only if there is a positive or negative change of more than 100 basis points in the average of the Moody's Aa utility bond rate as measured over the period October 1 through September 30. If there is a positive or negative change of more than 100 basis points, the return on equity is adjusted by one half of the difference. For the period from October 1, 2021 through September 30, 2022, the Moody's rate increased by 103 basis points from the benchmark, which triggers the WCCM adjustment. GSWC recognized revenues for the first quarter of 2023 and all of 2022 based on the previously authorized return of equity of 8.9% that is presently being billed to water customers pending a final decision in the cost of capital proceeding.

COVID-19 Emergency Memorandum Accounts:

The CPUC has authorized GSWC and BVES to track incremental costs, including bad debt expense, in excess of what is included in their respective revenue requirements incurred as a result of the pandemic in COVID-19 emergency-related memorandum accounts. During the first quarter of 2023, GSWC and BVES incurred some incremental costs in excess of their revenue requirements due to the lingering effects of the pandemic that are being tracked in COVID-19-related memorandum accounts and recorded as regulatory assets, which GSWC and BVES intend to file with the CPUC for future recovery. As of March 31, 2023, GSWC and BVES had approximately \$3.5 million and \$500,000, respectively, in regulatory asset accounts related to bad debt expense in excess of their revenue requirements, the purchase of personal protective equipment, additional incurred printing costs, and other incremental COVID-19-related costs. Emergency-related memorandum accounts are well-established cost recovery mechanisms authorized as a result of a state/federal declared emergency, and are therefore recognized as regulatory assets for future recovery. As a result, the amounts recorded in the COVID-19 emergency-related memorandum accounts have not impacted GSWC's or BVES's earnings. On April 10, 2023, the Biden Administration terminated the COVID-19 national emergency. The COVID-19 emergency-related memorandum accounts for GSWC and BVES expired when the COVID-19 national emergency ended and no additional amounts will be included in these memorandum accounts.

The CPUC requires that amounts tracked in GSWC's and BVES's COVID-19 memorandum accounts for unpaid customer bills be first offset by any (i) federal and state relief for water or electric utility bill debt, and (ii) customer payments through payment plan arrangements, prior to receiving recovery from customers at large. After these offsets are made, GSWC will file with the CPUC for recovery of the remaining balance. BVES intends to include the remaining balance in its COVID-19 memorandum account for recovery once all alternative sources of funding have been exhausted and credited to eligible customer accounts.

The CPUC's moratoriums on service disconnections for nonpayment for water and electric customers have ended. As a result, service disconnections due to nonpayment from delinquent residential customers resumed in June 2022.

Other BVES Regulatory Assets:

Wildfire Mitigation and Other Fire Prevention Related Costs Memorandum Accounts

The CPUC adopted regulations intended to enhance the fire safety of overhead electric power lines. Those regulations included increased minimum clearances around electric power lines. BVES was authorized to track incremental costs incurred to implement the regulations in a fire hazard prevention memorandum account for the purpose of obtaining cost recovery in a future general rate case. In August 2019, the CPUC issued a final decision on the electric general rate case, which set new rates through the year 2022. Among other things, the decision authorized BVES to record incremental costs related to vegetation management, such as costs for increased minimum clearances around electric power lines, in the CPUC-approved memorandum account for future recovery. As of March 31, 2023, BVES had approximately \$9.3 million in incremental vegetation management costs recorded as a regulatory asset, which has been included in the new general rate case application filed with the CPUC in August 2022 for future recovery. The incremental costs related to vegetation management included in the memorandum account will be subject to review during the general rate case proceeding.

California legislation enacted in September 2018 requires all investor-owned electric utilities to submit an annual wildfire mitigation plan ("WMP") to the CPUC for approval. The WMP must include a utility's plans on constructing, maintaining and operating its electrical lines and equipment to minimize the risk of catastrophic wildfire. In December 2022, the Office of Energy Infrastructure Safety under the California Natural Resources Agency approved BVES's 2022 WMP update. In February 2023, the CPUC ratified BVES's current WMP. As of March 31, 2023, BVES has approximately \$4.8 million related to expenses accumulated in its WMP memorandum accounts that have been recognized as regulatory assets for future recovery.

All capital expenditures and other costs incurred through March 31, 2023 as a result of BVES's WMPs are not currently in rates and have been filed for future recovery in BVES's general rate case application. These costs will be subject to review during BVES's general rate case proceeding.

2023 Winter Storm Other Regulatory Asset

BVES activated a catastrophic emergency memorandum account ("CEMA") to track the incremental costs incurred in response to a severe winter storm that occurred during the first quarter of 2023, which resulted in the declaration of an emergency by the governor of California. Incremental costs of approximately \$810,000 were incurred and included in the CEMA account, which has been recorded as a regulatory asset as of March 31, 2023 for future recovery. The incremental costs included in the CEMA account will be subject to review and approval by the CPUC. CEMA accounts are well-established cost recovery mechanisms authorized as a result of a state/federal declared emergency, and are therefore recognized as regulatory assets for future recovery. As a result, the amounts recorded in this CEMA account did not impact BVES's earnings.

Other Regulatory Assets:

Other regulatory assets represent costs incurred by GSWC or BVES for which they have received or expect to receive rate recovery in the future. Registrant believes that these regulatory assets are supported by regulatory rules and decisions, past practices, and other facts or circumstances that indicate recovery is probable. If the CPUC determines that a portion of either GSWC's or BVES's assets are not recoverable in customer rates, the applicable entity must determine if it has suffered an asset impairment that requires it to write down the regulatory asset to the amount that is probable of recovery.

Note 4 — Earnings per Share/Capital Stock

In accordance with the accounting guidance for participating securities and earnings per share ("EPS"), Registrant uses the "two-class" method of computing EPS. The "two-class" method is an earnings allocation formula that determines EPS for each class of common stock and participating security. AWR has participating securities related to restricted stock units that earn dividend equivalents on an equal basis with AWR's Common Shares, and that have been issued under AWR's stock incentive plans for employees and the non-employee directors stock plans. In applying the "two-class" method, undistributed earnings are allocated to both common shares and participating securities.

The following is a reconciliation of Registrant's net income and weighted average Common Shares outstanding used for calculating basic net income per share:

Basic:	For The Three Months Ended March 31,								
(in thousands, except per share amounts)		2023		2022					
Net income	\$	34,407	\$	14,162					
Less: (a) Distributed earnings to common shareholders		14,695		13,485					
Distributed earnings to participating securities		37		31					
Undistributed earnings		19,675		646					
(b) Undistributed earnings allocated to common shareholders		19,625		644					
Undistributed earnings allocated to participating securities		50		2					
Total income available to common shareholders, basic (a)+(b)	\$	34,320	\$	14,129					
Weighted average Common Shares outstanding, basic		36,968		36,944					
Basic earnings per Common Share	\$	0.93	\$	0.38					

Diluted EPS is based upon the weighted average number of Common Shares, including both outstanding shares and shares potentially issuable in connection with restricted stock units granted under AWR's stock incentive plans for employees and the non-employee directors stock plans, and net income. There were no options outstanding as of March 31, 2023 and 2022 under these plans. At March 31, 2023 and 2022, there were 106,817 and 96,586 restricted stock units outstanding, respectively, including performance shares awarded to officers of the Registrant.

The following is a reconciliation of Registrant's net income and weighted average Common Shares outstanding for calculating diluted net income per share:

For The Three Months Ended

Diluted:	March 31,								
(in thousands, except per share amounts)			2022						
Common shareholders earnings, basic	\$	34,320	\$	14,129					
Undistributed earnings for dilutive stock-based awards		50		2					
Total common shareholders earnings, diluted	\$	34,370	\$	14,131					
Weighted average common shares outstanding, basic		36,968		36,944					
Stock-based compensation (1)		79		75					
Weighted average common shares outstanding, diluted		37,047		37,019					
Diluted earnings per Common Share	\$	0.93	\$	0.38					

(1) All of the 106,817 and 96,586 restricted stock units at March 31, 2023 and 2022, respectively, were included in the calculation of diluted EPS for the three months ended March 31, 2023 and 2022.

During the three months ended March 31, 2023 and 2022, AWR issued 14,043 and 19,348 of common shares related to restricted stock units, respectively.

During the three months ended March 31, 2023 and 2022, AWR paid \$883,000 and \$1.2 million, respectively, to taxing authorities on employees' behalf for shares withheld related to net share settlements. During the three months ended March 31, 2023 and 2022, GSWC paid \$808,000 and \$1.1 million, respectively, to taxing authorities on employees' behalf for shares withheld related to net share settlements. These payments are included in the stock-based compensation caption of the statements of equity.

During the three months ended March 31, 2023, GSWC issued one Common Share to AWR for \$10.0 million. Proceeds from the stock issuance were used to pay down a portion of intercompany borrowings owed to AWR.

During the three months ended March 31, 2023 and 2022, AWR paid quarterly dividends of approximately \$14.7 million, or \$0.3975 per share, and \$13.5 million, or \$0.365 per share, respectively. During the three months ended March 31, 2023 and 2022, GSWC paid dividends of \$24.7 million and \$13.5 million, respectively, to AWR during these periods.

Note 5 — **Derivative Instruments**

BVES has purchased power under long-term contracts at a fixed cost over three- and five-year terms depending on the amount of power and period during which the power is purchased under the contracts. These long-term contracts are subject to the accounting guidance for derivatives and require mark-to-market derivative accounting.

Among other things, the CPUC authorized the use of a regulatory asset and liability memorandum account to offset the mark-to-market entries required by the accounting guidance. Accordingly, all unrealized gains and losses generated from the purchased power contracts are deferred on a monthly basis into a non-interest bearing regulatory memorandum account that tracks the changes in fair value of the derivative throughout the terms of the contracts. As a result, these unrealized gains and losses do not impact Registrant's earnings. As of March 31, 2023, there was a \$6.7 million derivative asset with a corresponding regulatory liability in the derivative instrument memorandum account for the purchased power contract as a result of fixed prices being lower than future energy prices. The notional volume of derivatives remaining under these long-term contracts as of March 31, 2023 was 184,979 megawatt hours.

The accounting guidance for fair value measurements applies to all financial assets and financial liabilities that are measured and reported on a fair value basis. Under the accounting guidance, BVES has made fair value measurements that are classified and disclosed in one of the following three categories:

- Level 1: Unadjusted quoted prices in active markets that are accessible at the measurement date for identical, unrestricted assets or liabilities;
- Level 2: Quoted prices in markets that are not active or inputs which are observable, either directly or indirectly, for substantially the full term of the asset or liability; or
- Level 3: Prices or valuation techniques that require inputs that are both significant to the fair value measurement and unobservable (i.e., supported by little or no market activity).

To value the contracts, Registrant utilizes various inputs that include quoted market prices for energy over the duration of the contracts. The market prices used to determine the fair value for this derivative instrument were estimated based on

independent sources such as broker quotes and publications that are not observable in or corroborated by the market. When such inputs have a significant impact on the measurement of fair value, the instruments are categorized as Level 3. Accordingly, the valuation of the derivatives on Registrant's purchased power contract has been classified as Level 3 for all periods presented.

The following table presents changes in the fair value of the Level 3 derivatives for the three months ended March 31, 2023 and 2022. The change in fair value was due to the change in market energy prices during the three months ended March 31, 2023 and 2022.

	For The Three M March							
(dollars in thousands)		2023		2022				
Fair value at beginning of the period	\$	11,847	\$	4,441				
Unrealized (losses) gains on purchased power contracts		(5,178)		2,579				
Fair value at end of the period	\$	6,669	\$	7,020				

Note 6 — Fair Value of Financial Instruments

For cash and cash equivalents, accounts receivable, accounts payable and short-term debt, the carrying amount is assumed to approximate fair value due to the short-term nature of these items.

Investments held in a Rabbi Trust for the supplemental executive retirement plan ("SERP") are measured at fair value and totaled \$29.2 million as of March 31, 2023 and \$27.5 million as of December 31, 2022. All equity investments in the Rabbi Trust are Level 1 investments in mutual funds. The investments held in the Rabbi Trust are included in "Other Property and Investments" on Registrant's balance sheets.

The table below estimates the fair value of long-term debt held by AWR and GSWC, respectively. The fair values as of March 31, 2023 and December 31, 2022 were determined using rates for similar financial instruments of the same duration utilizing Level 2 methods and assumptions. Changes in the assumptions will produce different results.

		March		December 31, 2022				
(dollars in thousands)	Carı	ying Amount		Fair Value	Ca	Carrying Amount Fair Va		
Financial liabilities:								
Long-term debt—AWR (1)	\$	580,264	\$	564,605	\$	450,373	\$	424,151
	=							
		March	31, 2023			Decembe	r 31,	2022
(dollars in thousands)	Carr	March rying Amount	31, 2023	Fair Value	Ca	Decembe	r 31,	2022 Fair Value
(dollars in thousands) Financial liabilities:	Carı		31, 2023		Ca		er 31, i	
	Cari		\$		Ca \$		\$	

⁽¹⁾ Excludes debt issuance costs of approximately \$3.4 million as of March 31, 2023 and December 31, 2022.

Note 7 — Income Taxes

AWR's effective income tax rate ("ETR") was 23.8% and 24.0% for the three months ended March 31, 2023 and 2022, respectively. GSWC's ETR was 24.5% and 23.9% for the three months ended March 31, 2023 and 2022, respectively.

The AWR and GSWC ETRs differed from the federal corporate statutory tax rate of 21% primarily due to (i) state taxes; (ii) permanent differences, including certain tax effects from stock compensation; (iii) the ongoing amortization of the excess deferred income tax liability; and (iv) differences between book and taxable income that are treated as flowed-through adjustments in accordance with regulatory requirements (principally from plant, rate-case, and compensation-related items). As a regulated utility, GSWC treats certain temporary differences as being flowed-through in computing its income tax expense consistent with the income tax method used in its CPUC-jurisdiction rate making. Flowed-through items either increase or decrease tax expense and thus impact the ETR.

⁽²⁾ Excludes debt issuance costs of approximately \$3.2 million as of March 31, 2023 and December 31, 2022.

Note 8 — Employee Benefit Plans

The components of net periodic benefit costs for Registrant's pension plan, postretirement medical benefit plan and SERP for the three months ended March 31, 2023 and 2022 were as follows:

For The Three Months Ended March 31, Other Postretirement Benefits SERP Pension Benefits (dollars in thousands) 2023 2022 2023 2022 2023 2022 **Components of Net Periodic Benefits Cost:** Service cost 846 1,480 33 33 312 298 25 Interest cost 2,513 1,844 16 411 256 (2,623)(3,292)(120)Expected return on plan assets (147)Amortization of prior service cost 108 109 Amortization of actuarial (gain) loss (8) 145 (240)(412)Net periodic benefits costs under accounting standards 844 141 (302)(510)715 699 Regulatory adjustment - deferred cost (92)Total expense (benefit) recognized, before surcharges 752 141 (302)\$ (510)715 699 and allocation to overhead pool

In 2023, Registrant expects to contribute approximately \$3.0 million to its pension plan.

As authorized by the CPUC in the water and electric general rate case decisions, GSWC and BVES each utilize two-way balancing accounts to track differences between the forecasted annual pension expenses in rates, or expected to be in rates, and the actual annual expense recorded in accordance with the accounting guidance for pension costs. GSWC's actual pension expense was higher than the amounts included in water customer rates by \$92,000 for the three months ended March 31, 2023. During the three months ended March 31, 2022, GSWC's actual pension expense was lower than the amounts included in water customer rates. BVES's actual expense was lower than the amounts included in electric customer rates for all periods presented. As of March 31, 2023, GSWC and BVES had net over-collections in their two-way pension balancing accounts of \$1,227,000 and \$574,000, respectively, included as part of regulatory assets and liabilities (Note 3).

Note 9 — Contingencies

Environmental Clean-Up and Remediation at GSWC:

GSWC has been involved in environmental remediation and cleanup at one of its plant sites that contained an underground storage tank which was used to store gasoline for its vehicles. This tank was removed from the ground in July 1990 along with the dispenser and ancillary piping. Since then, GSWC has been involved in various remediation activities at this site. Analysis indicates that off-site monitoring wells may be necessary to document effectiveness of remediation.

As of March 31, 2023, the total amount spent to clean up and remediate GSWC's plant facility was approximately \$6.2 million, of which \$1.5 million has been paid by the State of California Underground Storage Tank Fund. Amounts paid by GSWC have been included in rate base and approved by the CPUC for recovery. As of March 31, 2023, GSWC has a regulatory asset and an accrued liability for the estimated additional cost of \$1.3 million to complete the cleanup at the site. The estimate includes costs for continued activities of groundwater cleanup and monitoring, future soil treatment and site-closure-related activities. The ultimate cost may vary as there are many unknowns in remediation of underground gasoline spills and this is an estimate based on currently available information. Management also believes it is probable that the estimated additional costs will be approved in rate base by the CPUC.

Contracted Services:

ASUS's utility privatization contract services are provided to the U.S. government pursuant to the terms of the initial 50-year firm, fixed-price contract and additional firm, fixed-price contracts subject to annual economic price adjustments. Entering into contracts with the U.S. government subjects ASUS to potential government audits or investigations of its business practices and compliance with government procurement statutes and regulations. ASUS is currently under a civil government investigation over bidding and estimating practices used in certain capital upgrade projects. ASUS is cooperating fully with the investigation and management does not currently believe that the investigation will have a material adverse effect on its consolidated results of operations, financial condition, or liquidity. However, at this time, management cannot predict the final outcome or recommendations that may result from the investigation or determine the amount, if any, of penalties and damages that may be assessed.

Other Litigation:

Registrant is also subject to other ordinary routine litigation incidental to its business, some of which may include claims for compensatory and punitive damages. Management believes that rate recovery, proper insurance coverage and reserves are in place to insure against, among other things, property, general liability, employment, and workers' compensation claims incurred in the ordinary course of business. Insurance coverage may not cover certain claims involving punitive damages. Registrant does not believe the outcome from any pending suits or administrative proceedings will have a material effect on Registrant's consolidated results of operations, financial position, or cash flows.

Note 10 — Business Segments

AWR has three reportable segments: water, electric and contracted services. GSWC has one segment, water. On a stand-alone basis, AWR has no material assets or liabilities other than its equity investments in its subsidiaries, notes payable to bank, deferred taxes and intercompany note receivables with its subsidiaries.

All of GSWC's and BVES's business activities are conducted in California. Activities of ASUS and the Military Utility Privatization Subsidiaries are conducted in California, Florida, Georgia, Kansas, Maryland, New Mexico, North Carolina, South Carolina, Texas and Virginia. Some of ASUS's wholly owned subsidiaries are regulated by the state in which the subsidiary primarily conducts water and/or wastewater operations. Fees charged for operations and maintenance and renewal and replacement services are based upon the terms of the contracts with the U.S. government, which have been filed, as appropriate, with the commissions in the states in which ASUS's subsidiaries are incorporated.

The tables below set forth information relating to AWR's operating segments and AWR Parent. The utility plant balance are net of respective accumulated provisions for depreciation. Capital additions reflect capital expenditures paid in cash and exclude U.S. government-funded and third-party prime funded capital expenditures for ASUS and property installed by developers and conveyed to GSWC and BVES.

	As Of And For The Three Months Ended March 31, 2023									
						Contracted		AWR	C	onsolidated
(dollars in thousands)		Water		Electric		Services		Parent		AWR
Operating revenues	\$	112,712	\$	12,904	\$	35,807	\$		\$	161,423
Operating income (loss)		40,239		3,631		7,296		(1)		51,165
Interest expense, net		5,494		573		227		1,323		7,617
Net property, plant and equipment		1,638,031		125,093		17,337		_		1,780,461
Depreciation and amortization expense (1)		9,606		748		849		_		11,203
Income tax expense (benefit)		8,910		701		1,685		(544)		10,752
Capital additions		42,005		6,652		680		_		49,337

	As Of And For The Three Months Ended March 31, 2022									
						Contracted		AWR	C	onsolidated
(dollars in thousands)		Water		Electric		Services		Parent		AWR
Operating revenues	\$	73,906	\$	11,892	\$	22,772	\$		\$	108,570
Operating income (loss)		16,999		3,598		3,770		(2)		24,365
Interest expense (income), net		5,145		113		(135)		200		5,323
Net property, plant and equipment		1,523,665		107,114		19,080		_		1,649,859
Depreciation and amortization expense (1)		8,545		654		915		_		10,114
Income tax expense (benefit)		2,689		952		944		(124)		4,461
Capital additions		31,465		3,468		237		_		35,170

⁽¹⁾ Depreciation computed on GSWC's and BVES's transportation equipment is recorded in other operation expenses and totaled \$368,000 and \$94,000 for the three months ended March 31, 2023 and 2022, respectively. For the three months ended March 31, 2023, approximately \$212,000 of additional depreciation expense on GSWC's transportation equipment was recorded that relates to the cumulative retroactive impact for the full year of 2022 approved in the CPUC proposed decision that resulted from an increase to the transportation equipment composite depreciation rates that are retroactive to January 1, 2022.

The following table reconciles total net property, plant and equipment (a key figure for rate making) to total consolidated assets (in thousands):

	Marc	ch 31,	
	 2023		2022
Total net property, plant and equipment	\$ 1,780,461	\$	1,649,859
Other assets	279,911		264,073
Total consolidated assets	\$ 2,060,372	\$	1,913,932

Item 2. Management's Discussion and Analysis of Financial Condition and Results of Operations

General

The following discussion and analysis provides information on AWR's consolidated operations and assets, and includes specific references to AWR's individual segments and its subsidiaries (GSWC, BVES, and ASUS and its subsidiaries), and AWR (parent) where applicable. The subsidiaries of ASUS are collectively referred to as the "Military Utility Privatization Subsidiaries."

Included in the following analysis is a discussion of AWR's operations in terms of earnings per share by business segment and AWR (parent), which equals each business segment's earnings divided by AWR's weighted average number of diluted common shares. Furthermore, the gains and losses generated on the investments held to fund one of the Company's retirement plans during the three months ended March 31, 2023 and 2022 have been excluded when communicating the results to help facilitate comparisons of AWR's performance from period to period. Also, the impact of retroactive new rates related to the full year 2022 recorded in the three months ended March 31, 2023 resulting from the proposed decision on the water general rate case has been excluded when communicating AWR's consolidated and water segment's results for the three months ended March 31, 2023 and 2022 to help facilitate comparisons of AWR's performance from period to period. Diluted earnings per share by business segment and adjusted diluted earnings per share constitute "non-GAAP financial measures" under the Securities and Exchange Commission rules, which supplement our GAAP disclosures but should not be considered as an alternative to the respective GAAP measures. Furthermore, the non-GAAP financial measures may not be comparable to similarly titled non-GAAP financial measures of other registrants. All of these measures are derived from consolidated financial information of the Registrant, but are not presented in our financial statements that are prepared in accordance with GAAP.

AWR uses earnings per share by business segment as an important measure in evaluating its operating results and believes it provides investors with clarity surrounding the performance of its segments. AWR reviews this measurement regularly and compares it to historical periods and to its operating budget. A reconciliation to AWR's consolidated diluted earnings per share prepared in accordance with GAAP is included in the discussion under the section titled "Summary of First Ouarter Results by Segment."

Overview

Factors affecting our financial performance are summarized under "Risk Factors" in our Form 10-K for the period ended December 31, 2022 filed with the SEC.

Water and Electric Segments:

GSWC's and BVES's revenues, operating income, and cash flows are earned primarily through delivering potable water to homes and businesses in California and electricity in the Big Bear area of San Bernardino County, California, respectively. Rates charged to GSWC and BVES customers are determined by the CPUC. These rates are intended to allow recovery of operating costs and a reasonable rate of return on capital. GSWC and BVES plan to continue seeking additional rate increases in future years from the CPUC to recover operating and supply costs, and receive reasonable returns on invested capital. Capital expenditures in future years at GSWC and BVES are expected to remain at substantially higher levels than depreciation expense. When necessary, GSWC and BVES may obtain funds from external sources in the capital markets and through bank borrowings.

Pending General Rate Case Filings and Other Matters:

Water General Rate Case for the Years 2022–2024:

In July 2020, GSWC filed a general rate case application for all of its water regions and its general office. This general rate case determines new water rates for the years 2022 – 2024. On April 13, 2023, GSWC received a proposed decision from the assigned administrative law judge at the CPUC on GSWC's water general rate case with rates retroactive to January 1, 2022. Among other items, the proposed decision approves and adopts in its entirety the settlement agreement between GSWC and the Public Advocates Office at the CPUC ("Public Advocates") that had been filed with the CPUC in November 2021, and resolves all issues related to the 2022 annual revenue requirement in the general rate case application and allows for additional increases in adopted revenues for 2023. The new rates for 2022 and 2023 are effective and retroactive to January 1, 2022 and 2023, respectively. As a result of receiving a proposed decision that approves the settlement agreement in its entirety, the impact of retroactive new rates for the full year of 2022 and the estimated second-year rate increases for the three months ended March 31, 2023 have been reflected in the 2023 first quarter results as it became probable that the approved retroactive rates for the full year of 2022 and first three months of 2023 would be permitted to be billed to customers in the future. GSWC expects to receive a final decision during the second quarter of 2023.

The settlement agreement approved in the proposed decision (i) authorizes GSWC to invest approximately \$404.8 million in capital infrastructure over the three-year cycle (excluding advice letter projects); (ii) increases the 2022 adopted revenues (excluding the advice letter project revenues) by approximately \$30.3 million, or \$0.59 per share, as compared to the 2021 adopted revenues, and increases the 2022 adopted supply costs by \$9.6 million, or \$0.19 per share, as compared to the 2021

adopted supply costs, which combined is an increase of \$0.40 per share; (iii) adopts new operating expense levels for 2022 including a higher depreciation expense resulting from overall higher composite depreciation rates based on a new depreciation study adopted in the proposed decision; and (iv) allows for additional increases in adopted revenues for 2023 and 2024 subject to an earnings test and changes to the forecasted inflationary index values.

Due to the delay in finalizing the water general rate case, water revenues billed to customers for the year ended December 31, 2022 and for the three months ended March 31, 2023 were based on 2021 adopted rates, pending a final decision by the CPUC. As a result of receiving a proposed decision that approves the settlement agreement in its entirety, the impact of retroactive new rates for the full year of 2022 of \$0.36 per share has been reflected in the 2023 first quarter results and included primarily (i) the increase in 2022's adopted revenues and supply costs that is consistent with the settlement agreement, or \$0.40 per share as discussed above; (ii) a reduction to revenues of \$1.1 million, or \$0.02 per share, to reflect the incremental impact of revenues subject to refund from the new 2022 rates as a result of the lower cost of debt in the pending cost of capital proceeding; and (iii) higher overall depreciation expense for 2022 of approximately \$790,000, or \$0.02 per share, resulting from updated composite depreciation rates adopted in the proposed decision and which are reflected in the 2022 adopted revenue requirement.

The estimated second-year increases for 2023 have also been reflected in the 2023 first quarter results for the three months ended March 31, 2023. This included increases in revenues of approximately \$8.7 million, or \$0.17 per share, compared to the adopted 2021 rates currently being billed, and increases in supply costs of approximately \$1.6 million, or \$0.03 per share, which combined is an increase of \$0.14 per share for the three months ended March 31, 2023. The best estimate of 2023 rates have been computed at this time using inflationary index values as of March 31, 2023. Actual increases for 2023 will be determined when the filings to implement the new rate increases are approved by the CPUC, and will be calculated using the inflationary index values at that time. GSWC will file for the 2023 increases once the CPUC approves the final decision.

As of March 31, 2023, there is an aggregate cumulative amount of \$38.4 million in CPUC-approved general rate case memorandum accounts that relates to water revenues recorded during the three months ended March 31, 2023, and which represent the difference between the 2021 adopted rates billed to customers and the rates authorized in the proposed decision for the full year of 2022 and estimated increases for the three months ended March 31, 2023. Once a final decision is issued by the CPUC, GSWC will request recovery through a surcharge of the cumulative amounts included in the general rate case memorandum accounts.

Furthermore, the proposed decision addressed the three remaining unresolved issues related to GSWC's requests for: (i) a medical insurance cost balancing account, (ii) a general liability insurance cost balancing account, and (iii) the consolidation of two of GSWC's customer service areas. The proposed decision approved both balancing accounts and denied GSWC's consolidation of its two customer service areas. The proposed decision also approved the recovery of previously incurred costs that were being tracked in other CPUC-authorized memorandum accounts. As a result, GSWC recorded the cumulative amounts tracked in these balancing and memorandum accounts that are being approved in the proposed decision, the net impact of which was not material to 2023 first quarter results.

Cost of Capital Proceeding:

Investor-owned water utilities serving California are required to file their cost of capital applications on a triennial basis. GSWC filed a cost of capital application with the CPUC in May 2021 currently pending final approval, which requested a capital structure of 57% equity and 43% debt, a return on equity of 10.5%, an embedded cost of debt of 5.1%, and a return on rate base of 8.18%. Hearings on this proceeding occurred in May 2022 and briefs were filed in June 2022. On May 9, 2023, GSWC received a proposed decision from the assigned administrative law judge at the CPUC on the cost of capital proceeding. Among other things, the proposed decision (i) adopts GSWC's requested capital structure and cost of debt filed in the application; (ii) adopts a return on equity of 8.85% for GSWC as compared to 8.9% previously authorized; (iii) allows for the continuation of the Water Cost of Capital Mechanism ("WCCM"); and (iv) adopts the new cost of capital for the three-year period commencing January 1, 2022, through December 31, 2024. Comments on the proposed decision are due on May 30. In March 2023, the CPUC issued a decision that approved an extension of the statutory deadline for a final decision in the cost of capital proceeding to August 10, 2023.

The 5.1% cost of debt adopted in the proposed decision is lower than the previously authorized amount of 6.6%. The new cost of debt is expected to lower 2023 and 2022 adopted water revenues by approximately \$8.2 million, or \$0.16 per share, and \$7.5 million, or \$0.15 per share, respectively, as compared to 2021 adopted water revenues at the currently authorized cost of debt of 6.6% that is presently being billed to water customers until a final decision is issued in this proceeding. Based on management's analysis of this regulatory proceeding and the associated accounting in 2022 and through March 31, 2023, GSWC has reduced revenues and recorded a corresponding regulatory liability for revenues subject to refund based on its best estimate at this time, which includes the impact of GSWC's lower cost of debt adopted in the proposed decision. For the three months ended March 31, 2023 and 2022, GSWC reduced revenues by \$1.8 million and \$1.4 million, respectively, that are subject to refund. Furthermore, an additional reduction to revenues of \$1.1 million, or \$0.02 per share, related to the full year of 2022 was recorded during the first quarter of 2023 to reflect the incremental impact of revenues subject to refund from the new 2022 rates approved in the proposed water general rate case decision. As of March 31, 2023, GSWC had a cumulative regulatory liability of \$9.3

million for the estimated revenues subject to refund from the pending cost of capital proceeding covering the period January 1, 2022 through March 31, 2023. However, management cannot predict the ultimate outcome and any changes that may be made to the final decision in the cost of capital application, and the associated impact on 2023 and 2022 revenues. Changes in estimates will be made, if necessary, as more information in this proceeding becomes available.

Furthermore, the proposed decision continues the WCCM for the years 2023 and 2024, which adjusts the return on equity and rate of return on rate base between the three-year cost of capital proceedings only if there is a positive or negative change of more than 100 basis points in the average of the Moody's Aa utility bond rate as measured over the period October 1 through September 30. If there is a positive or negative change of more than 100 basis points, the return on equity is adjusted by one half of the difference. For the period from October 1, 2021 through September 30, 2022, the Moody's rate increased by 103 basis points from the benchmark, which triggers the WCCM adjustment. GSWC recognized revenues for the first quarter of 2023 and all of 2022 based on the previously authorized return of equity of 8.9% that is presently being billed to water customers pending a final decision in the cost of capital proceeding.

Final Decision in the First Phase of the Low-Income Affordability Rulemaking:

In August 2020, the CPUC issued a final decision in the first phase of the CPUC's Order Instituting Rulemaking evaluating the low income ratepayer assistance and affordability objectives contained in the CPUC's 2010 Water Action Plan. This decision also addressed other issues, including the discontinued use of the Water Revenue Adjustment Mechanism ("WRAM") and the Modified Cost Balancing Account ("MCBA"). The MCBA is a full-cost balancing account used to track the difference between adopted and actual water supply costs (including the effects of changes in both rates and volume). Based on the final decision, any general rate case application filed by GSWC and the other California water utilities after August 27, 2020 may not include a proposal to continue the use of the WRAM or MCBA, but may instead include a proposal to use a limited price adjustment mechanism and an incremental supply cost balancing account. Since its implementation in 2008, the WRAM and MCBA have helped mitigate fluctuations in GSWC's earnings due to changes in water consumption by its customers or changes in water supply mix. Replacing them with mechanisms recommended in the final decision will likely result in more volatility in GSWC's future earnings and could result in less than, or more than, full recovery of its authorized revenue and supply costs.

The August 2020 decision provides that the WRAM and MCBA for GSWC would be discontinued after 2024. However, on September 30, 2022, the governor of California signed Senate Bill ("SB") 1469. Effective January 1, 2023, SB 1469 allows Class A water utilities, including GSWC, to continue requesting the use of the WRAM in their next general rate case. With the passage of SB 1469, GSWC will be able to request the continued use of the WRAM in its next general rate case to be filed in 2023 that will establish new rates for the years 2025 – 2027. GSWC's request to continue using the WRAM in its next general rate case will be subject to CPUC approval.

In October 2020, GSWC, three other investor-owned water utilities ("IOWUs") operating in California, and the California Water Association ("CWA") filed applications with the CPUC for rehearing on the discontinuation of the WRAM and MCBA, which the CPUC denied in September 2021. GSWC, the three other IOWUs and CWA each separately filed a petition with the California Supreme Court ("Court") to review the CPUC's decision revoking prior authorization of the WRAM and MCBA. In May 2022, the Court granted the petition for writ of review. The Court ordered GSWC, along with the other IOWUs and CWA, to file opening briefs, which were filed on September 1, 2022. The CPUC's answer to the opening brief was originally due on November 15, 2022 and reply briefs were due on December 15, 2022. However, as a result of SB 1469, in October 2022 the CPUC filed a motion to dismiss the IOWUs and CWA's petition with the Court, and also requested that the Court suspend the proceeding schedule until it rules on the motion to dismiss. In November 2022, the Court denied the motion to dismiss resulting in the CPUC filing their answer brief on December 9, 2022 and reply briefs were filed on January 13, 2023. There is no timeline for the Court to complete their review. At this time, management cannot predict the final outcome of this matter.

Electric General Rate Case for the years 2023–2026:

On August 30, 2022, BVES filed a general rate case application that will determine new electric rates for the years 2023 – 2026. In December 2022, a pre-hearing conference was held to discuss the scope of issues and schedule for the proceeding. In February 2023, a scoping memo and ruling that set the final schedule and scope of issues in BVES's general rate case proceeding was issued by the CPUC. Based on the schedule, a proposed decision is expected in the fourth quarter of 2023. On December 15, 2022, the CPUC approved a decision for BVES to establish a general rate case memorandum account that makes the new 2023 rates effective and retroactive to January 1, 2023.

Among other things, BVES requested (i) capital budgets of approximately \$62.0 million for the four-year rate cycle, and another \$6.2 million for a large line replacement capital project to be filed for revenue recovery through an advice letter when the project is completed, and (ii) a capital structure for BVES of 61.8% equity and 38.2% debt, a return on equity of 11.25%, an embedded cost of debt of 5.51%, and a return on rate base of 9.05%. Furthermore, included in the general rate case application is a request for recovery of all capital expenditures and other costs incurred over the last few years in connection with BVES's wildfire mitigation plans that are currently not in customer rates. These costs will be subject to review by the CPUC during the general rate case proceeding.

Contracted Services Segment:

ASUS's revenues, operating income and cash flows are earned by providing water and/or wastewater services, including operation and maintenance services and construction of facilities for the water and/or wastewater systems at various military installations, pursuant to an initial 50-year firm fixed-price contract and additional firm fixed-price contracts. The contract price for each of these contracts is subject to annual economic price adjustments. Additional revenues generated by contract operations are primarily dependent on annual economic price adjustments, and new construction activities under contract modifications with the U.S. government or agreements with other third-party prime contractors.

Entering into contracts with the U.S. government subjects ASUS to potential government audits or investigations of its business practices and compliance with government procurement statutes and regulations. ASUS is currently under a civil government investigation over bidding and estimating practices used in certain capital upgrade projects. ASUS is cooperating fully with the investigation and management does not currently believe that the investigation will have a material adverse effect on its consolidated results of operations, financial condition, or liquidity. However, at this time, management cannot predict the final outcome or recommendations that may result from the investigation or determine the amount, if any, of penalties and damages that may be assessed.

COVID-19:

AWR and its subsidiaries continue to monitor the guidance provided by federal, state, and local health authorities and other government officials. Thus far, the COVID-19 pandemic has not had a material impact on ASUS's current operations.

During 2023, GSWC and BVES incurred some incremental costs in excess of their revenue requirements due to the lingering effects of the pandemic that are being tracked in COVID-19-related memorandum accounts and recorded as regulatory assets. The CPUC has authorized GSWC and BVES to track incremental costs, including bad debt expense, in excess of what is included in their respective revenue requirements incurred as a result of the pandemic in COVID-19 emergency-related memorandum accounts, which GSWC and BVES intend to file with the CPUC for future recovery. As of March 31, 2023, GSWC and BVES had approximately \$3.5 million and \$500,000, respectively, in regulatory asset accounts related to bad debt expense in excess of their revenue requirements, the purchase of personal protective equipment, additional incurred printing costs, and other incremental COVID-19-related costs. Emergency-type memorandum accounts are well-established cost recovery mechanisms authorized as a result of a state/federal declared emergency, and are therefore recognized as regulatory assets for future recovery. As a result, the amounts recorded in the COVID-19 emergency-related memorandum accounts have not impacted GSWC's or BVES's earnings.

The CPUC requires that amounts tracked in GSWC's and BVES's COVID-19 memorandum accounts for unpaid customer bills be first offset by any (i) federal and state relief for water or electric utility bill debt, and (ii) customer payments through payment plan arrangements, prior to receiving recovery from customers at large. After these offsets are made, GSWC will file with the CPUC for recovery of the remaining balance. BVES intends to include the remaining balance in its COVID-19 memorandum account for recovery once all alternative sources of funding have been exhausted and credited to eligible customer accounts.

On April 10, 2023, the Biden Administration terminated the COVID-19 national emergency. The COVID-19 emergency-related memorandum accounts for GSWC and BVES expired when the COVID-19 national emergency ended and no additional amounts will be included in these memorandum accounts.

Summary of First Quarter Results by Segment

The table below sets forth the first quarter diluted earnings per share by business segment and for the parent company:

	Diluted Earnings per Share								
		Three Mon							
	3	3/31/2023		3/31/2022		CHANGE			
Water, adjusted (2023 excludes the impact of retroactive rates related to 2022 from the CPUC proposed decision in the general rate case)	\$	0.38	\$	0.23	\$	0.15			
Electric		0.06		0.07		(0.01)			
Contracted services		0.15		0.08		0.07			
AWR (parent)		(0.02)		_		(0.02)			
Consolidated diluted earnings per share, as adjusted		0.57		0.38		0.19			
Impact of retroactive rates related to the full year of 2022 from the proposed decision in the water general rate case (approximately \$0.08 per share relates to first quarter of 2022)		0.36		_		0.36			
Consolidated diluted earnings per share, as recorded	\$	0.93	\$	0.38	\$	0.55			

For the three months ended March 31, 2023, AWR's recorded consolidated diluted earnings were \$0.93 per share, as compared to \$0.38 per share for the same period in 2022, an increase of \$0.55 per share, which includes the impact of retroactive new rates related to the full 2022 year of \$0.36 per share that was recorded in the first quarter 2023 as a result of receiving the proposed decision in April 2023 on the water general rate case. The impact of retroactive rates related to the full year of 2022 is shown separately in the table above and included primarily: (i) the increase in 2022's adopted revenues and supply costs that is consistent with the settlement agreement, or \$0.40 per share as previously discussed; (ii) a reduction to revenues of \$1.1 million, or \$0.02 per share, to reflect the incremental impact of revenues subject to refund from the new 2022 rates as a result of the lower cost of debt in the pending cost of capital proceeding; and (iii) higher overall depreciation expense for 2022 of approximately \$790,000, or \$0.02 per share, resulting from higher composite depreciation rates adopted in the proposed decision and which are reflected in the 2022 adopted revenue requirement.

Excluding the impact of retroactive rates related to the full 2022 year of \$0.36 per share that was recorded in the first quarter of 2023 and is shown on a separate line in the table above, adjusted consolidated diluted earnings for the three months ended March 31, 2023 were \$0.57 per share. The proposed decision on the water general rate case set new rates for 2022 and 2023, retroactive to January 1, 2022 and January 1, 2023, respectively, and cumulative adjustments were recorded in the first quarter of 2023 to reflect the impact of retroactive rates to the full year 2022 and the three months ended March 31, 2023.

Also included in the results for the first quarter ended March 31, 2023 were gains totaling \$1.6 million, or approximately \$0.03 per share, on investments held to fund one of the Company's retirement plans, as compared to losses of \$1.7 million, or approximately \$0.03 per share, for the same period in 2022, both due to financial market conditions. Excluding the gains and losses on investments from both periods, and excluding the impact of retroactive rates related to the full year of 2022 of \$0.36 per share recorded in the three months ended March 31, 2023, adjusted consolidated diluted earnings for the first quarter of 2023 were \$0.54 per share as compared to adjusted diluted earnings of \$0.41 per share for the same period in 2022, an adjusted increase of \$0.13 per share largely due to the 2023 estimated second-year rate increases recorded as a result of receiving a proposed decision in GSWC's general rate case proceeding.

The following is a computation and reconciliation of recorded diluted earnings per share from the measure of operating income by business segment as disclosed in Note 10 to the Unaudited Consolidated Financial Statements, to AWR's consolidated fully diluted earnings per common share (as recorded), for the three months ended March 31, 2023 and 2022:

	W	ate	r	Electric			Contracted Services				AWR (Parent)					Consolidated (GAAP)			
(in thousands, except per share amounts)	Q1 2023		Q1 2022		Q1 2023		Q1 2022		Q1 2023		Q1 2022		Q1 2023		Q1 2022		Q1 2023		Q1 2022
Operating income (loss) (Note 10)	\$ 40,239	\$	16,999	\$	3,631	\$	3,598	\$	7,296	\$	3,770	\$	(1)	\$	(2)	\$	51,165	\$	24,365
Other (income) and expense	3,866		5,743		560		(30)		257		(171)		1,323		200		6,006		5,742
Income tax expense (benefit)	8,910		2,689		701		952		1,685		944		(544)		(124)		10,752		4,461
Net income (loss)	\$ 27,463	\$	8,567	\$	2,370	\$	2,676	\$	5,354	\$	2,997	\$	(780)	\$	(78)	\$	34,407	\$	14,162
Weighted Average Number of Diluted Shares	37,047		37,019		37,047		37,019		37,047		37,019		37,047		37,019		37,047		37,019
Diluted earnings per share	\$ 0.74	\$	0.23	\$	0.06	\$	0.07	\$	0.15	\$	0.08	\$	(0.02)	\$		\$	0.93	\$	0.38

Water Segment:

For the three months ended March 31, 2023, recorded diluted earnings from the water utility segment were \$0.74 per share, as compared to \$0.23 per share for the same period in 2022, an increase of \$0.51 per share, which includes (i) the impact of retroactive new rates for the full year of 2022 of \$0.36 per share that was recorded in the first quarter 2023 and is shown separately in the table above, and (ii) gains totaling \$1.6 million, or approximately \$0.03 per share, incurred during the first quarter of 2023 on investments held to fund one of the Company's retirement plans, as compared to losses of \$1.7 million, or approximately \$0.03 per share, recorded for the same period in 2022.

Excluding the gains and losses on investments from both periods, and excluding the impact of retroactive rates related to the full year of 2022 recorded in the first quarter of 2023, adjusted diluted earnings for the first quarter of 2023 at the water segment were \$0.35 per share as compared to adjusted diluted earnings of \$0.26 per share for the same period in 2022, an adjusted increase at the water segment of \$0.09 per share due primarily to the following items:

- An increase in water operating revenues of approximately \$9.0 million largely as a result of the estimated second-year rate increases for 2023 that will be effective as of January 1, 2023 and have been reflected in the 2023 first quarter results. Approved 2023 rates will be subject to an earnings test and changes to inflationary index values. Because water revenues recorded during the three months ended March 31, 2022 were based on 2021 adopted rates, the increase in water revenues during the first quarter of 2023 represents the difference from the 2021 adopted rates and the 2023 estimated second-year rate increases for the three months period ended March 31, 2023.
- An increase in water supply costs of \$1.6 million, which consist of purchased water, purchased power for pumping, groundwater production assessments and changes in the water supply cost balancing accounts. Adopted supply costs for the first quarter of 2023 were based on 2023 authorized amounts, pending a final decision by the CPUC in the water general rate case application. Actual water supply costs are tracked and passed through to customers on a dollar-for-dollar basis by way of the CPUC-approved water supply cost balancing accounts. The increase in water supply costs results in a corresponding increase in water operating revenues and has no net impact on the water segment's profitability.
- An overall increase in operating expenses of \$1.3 million (excluding supply costs) mainly due to increases in (i) overall labor costs, (ii) other operation expenses resulting primarily from higher water treatment and transportation costs, (iii) administrative and general expenses largely from higher employee-related expenses and outside-service costs, and (iv) depreciation and amortization expenses resulting from additions to utility plant and the higher composite depreciation rates based on a revised depreciation study approved in the proposed decision on the water general rate case.
- An increase in interest expense (net of interest income) of \$1.1 million resulting primarily from an increase in interest rates, as well as an overall increase in total borrowing levels to support, among other things, the capital expenditures program at GSWC, partially offset by higher interest income earned on regulatory assets bearing interest at the current 90-day commercial-paper rate, which increased compared to 2022's rates, as well as an increase in the level of regulatory assets recorded resulting, in large part, from the proposed decision on the water general rate case.
- An overall increase in other expenses (net of other income) of \$1.1 million due primarily to an increase in the non-service cost components related to GSWC's benefit plans resulting from changes in actuarial assumptions including expected returns on plan assets. However, as a result of GSWC's two-way pension balancing accounts authorized by the CPUC, changes in total net periodic benefit costs related to the pension plan have no material impact to earnings.
- Changes in certain flowed-through taxes and permanent items included in GSWC's income tax expense for the three months ended March 31, 2023 as compared to the same period in 2022 that favorably impacted water earnings. As a regulated utility, GSWC treats certain temporary differences as being flowed-through in computing its income tax expense consistent with the income tax method used in its CPUC-jurisdiction rate making. Changes in the magnitude of flowed-through items either increase or decrease tax expense, thereby affecting diluted earnings per share.

Electric Segment:

Diluted earnings from the electric utility segment decreased by \$0.01 per share for the three months ended March 31, 2023 as compared to the same period in 2022, largely resulting from not having new rates in 2023 while awaiting the processing of the pending electric general rate case that will set new rates for 2023 - 2026, while also experiencing continued increases in overall operating expenses and interest costs. When a decision is issued in the electric general rate case, new rates are expected to be retroactive to January 1, 2023 and cumulative adjustments will be recorded at that time.

Contracted Services Segment:

Diluted earnings from the contracted services segment increased by \$0.07 per share for the three months ended March 31, 2023 as compared to the same period in 2022, largely due to an increase in construction activity due to timing differences of when construction work was performed in 2023 as compared to the same period in 2022, and an increase in management fee revenue resulting from the resolution of various economic price adjustments, partially offset by higher overall operating expenses and interest costs as compared to the same period of 2022. The contracted services segment is expected to contribute \$0.45 to \$0.49 per share for the full 2023 year.

AWR (Parent):

For the three months ended March 31, 2023, diluted earnings from AWR (parent) decreased \$0.02 per share compared to the same period in 2022 due primarily to an increase in interest expense resulting from higher short-term interest rates on borrowings made under AWR's revolving credit facility, as well as changes in state unitary taxes.

The following discussion and analysis for the three months ended March 31, 2023 and 2022 provide information on AWR's consolidated operations and assets and, where necessary, includes specific references to AWR's individual segments and subsidiaries: GSWC, BVES and ASUS and its subsidiaries.

Consolidated Results of Operations — Three Months Ended March 31, 2023 and 2022 (amounts in thousands, except per share amounts):

	Three Months Ended March 31, 2023		Three Months Ended March 31, 2022		\$ CHANGE	% CHANGE	
OPERATING REVENUES							
Water	\$ 112,712	2	\$ 73,906	\$	38,806	52.5 %	
Electric	12,904	4	11,892		1,012	8.5 %	
Contracted services	35,80°	7	22,772		13,035	57.2 %	
Total operating revenues	161,423	3	108,570		52,853	48.7 %	
OPERATING EXPENSES							
Water purchased	14,304	4	17,848		(3,544)	(19.9)%	
Power purchased for pumping	2,354		2,374		(20)	(0.8)%	
Groundwater production assessment	3,833		4,211		(378)	(9.0)%	
Power purchased for resale	4,980		5,166		(180)	(3.5)%	
Supply cost balancing accounts	11,560	6	(6,343)		17,909	*	
Other operation	10,110		8,667		1,449	16.7 %	
Administrative and general	23,54	7	22,972		575	2.5 %	
Depreciation and amortization	11,20		10,114		1,089	10.8 %	
Maintenance	3,150	0	3,140		10	0.3 %	
Property and other taxes	6,29	5	5,853		442	7.6 %	
ASUS construction	18,904	4	10,203		8,701	85.3 %	
Total operating expenses	110,258	8	84,205		26,053	30.9 %	
OPERATING INCOME	51,16:	5	24,365		26,800	110.0 %	
OTHER INCOME AND EXPENSES							
Interest expense	(9,48)	1)	(5,606)		(3,875)	69.1 %	
Interest income	1,864	4	283		1,581	558.7 %	
Other, net	1,61	1	(419)		2,030	*	
	(6,000	6)	(5,742)		(264)	4.6 %	
INCOME BEFORE INCOME TAX EXPENSE	45,159	9	18,623		26,536	142.5 %	
Income tax expense	10,752	2	4,461	_	6,291	141.0 %	
NET INCOME	\$ 34,40	7	\$ 14,162	\$	20,245	143.0 %	
Basic earnings per Common Share	\$ 0.93	3	\$ 0.38	\$	0.55	144.7 %	
Fully diluted earnings per Common Share	\$ 0.93	3	\$ 0.38	\$	0.55	144.7 %	

^{*} not meaningful

Operating Revenues:

General

GSWC and BVES rely upon approvals by the CPUC for rate increases to recover operating expenses and to provide for a return on invested and borrowed capital used to fund utility plant. ASUS relies on economic price and equitable adjustments by the U.S. government in order to recover operating expenses and provide a profit margin for ASUS. Current operating revenues and earnings may be negatively impacted if the Military Utility Privatization Subsidiaries do not receive adequate rate relief or adjustments in a timely manner. ASUS's earnings are also impacted by the level of additional construction projects at the Military Utility Privatization Subsidiaries, which may or may not continue at current levels in future periods.

Water

For the three months ended March 31, 2023, revenues from water operations increased by \$38.8 million to \$112.7 million as compared to the same period in 2022. The increase in water revenues was largely as a result of receiving in April 2023 a proposed decision in the pending general rate case application. As a result of receiving a proposed decision that approves the November 2021 settlement agreement in its entirety, the impact of retroactive new adopted rates for the full year of 2022 of \$30.3 million and the estimated 2023 revenue increases for the three months ended March 31, 2023 of \$8.7 million were recorded during the first quarter of 2023.

Billed water consumption for the first quarter of 2023 was lower by approximately 17% as compared to the same period in 2022 due primarily to above average rainfall in California in the first quarter of 2023 compared to the same period in 2022, which was the driest on record for this three month period. Currently, changes in consumption generally do not have a significant impact on recorded revenues due to the CPUC-approved WRAM that is in place in all but one small rate making area. GSWC records the difference between what it bills its water customers and what is authorized by the CPUC in the WRAM accounts as regulatory assets or liabilities.

Electric

Electric revenues for the three months ended March 31, 2023 increased by \$1.0 million to \$12.9 million due, in large part, to the proposed decision issued in the water general rate case proceeding that updates the costs allocated from the general corporate office to the electric segment. The proposed decision authorizes an increase in the allocation ratio to the electric segment. The increase in general corporate office expenses allocated to the electric segment, as discussed later, also includes a corresponding and offsetting increase in adopted electric revenues as provided in BVES's last general rate case proceeding, resulting in no impact to earnings. There was also an increase in electric revenues from an advice letter filing related to a completed capital project.

Electric usage for the first quarter of 2023 was lower by 2.5% as compared to the same period in 2022. Due to the CPUC-approved Base Revenue Requirement Adjustment Mechanism, which adjusts certain revenues to adopted levels authorized by the CPUC, changes in usage do not have a significant impact on earnings.

Contracted Services

Revenues from contracted services are composed of construction revenues (including renewal and replacements) and management fees for operating and maintaining water and/or wastewater systems at various military bases. For the three months ended March 31, 2023, revenues from contracted services increased \$13.0 million to \$35.8 million as compared to \$22.8 million for the same period in 2022. The increase was largely due to higher construction activity and an increase in management fee revenue from annual economic price adjustments as compared to the same period of 2022.

ASUS's subsidiaries continue to enter into U.S. government-awarded contract modifications and agreements with third-party prime contractors for new construction projects at the military bases served. During the three months ended March 31, 2023, ASUS has been awarded approximately \$4.6 million in new construction projects for completion in 2023 through 2025. This is in addition to \$34.4 million of new construction projects awarded in 2022, which are being completed from late 2022 through 2025. Earnings and cash flows from modifications to the initial 50-year contracts with the U.S. government and agreements with third-party prime contractors for additional construction projects may or may not continue in future periods.

Operating Expenses:

Supply Costs

Total supply costs at the regulated utilities comprise the largest segment of total consolidated operating expenses. Supply costs accounted for approximately 33.6% and 27.6% of total operating expenses for the three months ended March 31, 2023 and 2022, respectively.

Water segment supply costs

Two of the principal factors affecting water supply costs are the amount of water produced and the source of the water. Generally, the variable cost of producing water from wells is less than the cost of water purchased from wholesale suppliers. The actual percentages of purchased water for the three months ended March 31, 2023 and 2022 were approximately 42% and 48%, respectively, as compared to the authorized adopted percentages of 38% and 30% for the three months ended March 31, 2023 and 2022, respectively.

Under the current CPUC-approved Modified Cost Balancing Account ("MCBA"), GSWC tracks adopted and actual expense levels for purchased water, power purchased for pumping and pump taxes. GSWC records the variances (which include the effects of changes in both rate and volume) between adopted and actual purchased water, purchased power and pump tax expenses. GSWC recovers from, or refunds to, customers the amount of such variances. GSWC tracks these variances individually for each water rate making area.

Supply costs for the water segment consist of purchased water, purchased power for pumping, groundwater production assessments and changes in the water supply cost balancing accounts. For the three months ended March 31, 2023 and 2022, water supply costs consisted of the following amounts (in thousands):

	e Months Ended Iarch 31, 2023	Three Months Ended March 31, 2022	\$ CHANGE	% CHANGE		
Water purchased	\$ 14,304	\$ 17,848	\$ (3,544)	-19.9 %		
Power purchased for pumping	2,354	2,374	(20)	-0.8 %		
Groundwater production assessment	3,833	4,211	(378)	-9.0 %		
Water supply cost balancing accounts *	12,625	(5,067)	17,692	**		
Total water supply costs	\$ 33,116	\$ 19,366	\$ 13,750	71.0 %		

^{*} The sum of the water and electric supply-cost balancing accounts are shown on AWR's Consolidated Statements of Income and totaled \$11,566,000 and \$(6,343,000) for the three months ended March 31, 2023 and 2022, respectively.

Purchased water costs for the first quarter of 2023 decreased to \$14.3 million as compared to \$17.8 million for the same period in 2022 primarily due to decreases in water consumption and production that was driven by above-average rainfall in 2023 and drought conditions experienced in 2022 that resulted in related water use restrictions, as well as a lower mix of purchased water compared to the same period in 2022, all partially offset by increases in wholesale water costs. Groundwater production assessments decreased largely due to a decrease in water consumption and production compared to the three months ended March 31, 2022 due primarily to above-average rainfall in 2023.

For the three months ended March 31, 2023, the water supply cost balancing account had a \$12.6 million over-collection as compared to a \$5.1 million under-collection during the same period in 2022. The increase in the over-collection was primarily due to updated adopted supply costs from the proposed decision received in April 2023 in the water general rate case proceeding. This increase includes largely the full year impact of 2022 to reflect new adopted supply costs retroactive to January 1, 2022, with a corresponding and offsetting increase in adopted water revenues, resulting in no impact to earnings.

Electric segment supply costs

Supply costs for the electric segment consist primarily of purchased power for resale, the cost of natural gas used by BVES's generating unit, the cost of renewable energy credits and changes in the electric supply cost balancing account. For the three months ended March 31, 2023 and 2022, electric supply costs consisted of the following amounts (in thousands):

	 Months Ended arch 31, 2023	Three Months Ended March 31, 2022	\$ CHANGE	% CHANGE
Power purchased for resale	\$ 4,986	\$ 5,166	\$ (180)	-3.5 %
Electric supply cost balancing account *	(1,059)	(1,276)	217	-17.0 %
Total electric supply costs	\$ 3,927	\$ 3,890	\$ 37	1.0 %

^{*} The sum of the water and electric supply-cost balancing accounts are shown on AWR's Consolidated Statements of Income and totaled \$11,566,000 and \$(6,343,000) for the three months ended March 31, 2023 and 2022, respectively.

For the three months ended March 31, 2023, the cost of power purchased for resale to BVES's electric customers decreased by \$180,000 to \$5.0 million as compared to \$5.2 million during the same period in 2022 due to a decrease in customer usage and lower average prices per megawatt-hour that include all fixed costs. The average price per megawatt-hour, including fixed costs, decreased from \$141.21 for the three months ended March 31, 2022 to \$94.73 for the same period in 2023. The decrease in average price per megawatt-hour resulted in a decrease in the under-collection in the electric supply cost balancing account, which decreased by \$217,000 as compared to the three months ended March 31, 2022.

^{**} not meaningful

Other Operation

The primary components of other operation expenses include payroll costs, materials and supplies, chemicals and water treatment costs and outside-service costs of operating the regulated water systems, including the costs associated with water transmission and distribution, pumping, water quality, meter reading, billing, and operations of district offices and the electric system. Registrant's contracted services operations incur many of the same types of expenses. For the three months ended March 31, 2023 and 2022, other operation expenses by business segment consisted of the following (dollar amounts in thousands):

	7	Three Months Ended March 31, 2023	,	Three Months Ended March 31, 2022	\$ CHANGE	% CHANGE
Water Services	\$	7,271	\$	6,354	\$ 917	14.4 %
Electric Services		1,059		876	183	20.9 %
Contracted Services		1,786		1,437	349	24.3 %
Total other operation	\$	10,116	\$	8,667	\$ 1,449	16.7 %

For the three months ended March 31, 2023, the increase in other operation expenses at the water segment was due primarily to higher operation-related labor, water treatment, and vehicle and equipment costs. As a result of receiving the proposed decision in the water general rate case, the increase at the water segment also included a cumulative depreciation adjustment for 2022 of \$212,000 on GSWC's transportation equipment, which is recorded in other operation expenses.

The increase in electric segment was due primarily to operation-related labor and transportation expense. The increase at the contracted services segment was due primarily to higher operation-related labor and outside services. Transportation costs were higher due, in part, to increases in fuel and maintenance costs compared to the same period in 2022.

Administrative and General

Administrative and general expenses include payroll costs related to administrative and general functions, all employee-related benefits, insurance expenses, outside legal and consulting fees, regulatory-utility-commission expenses, expenses associated with being a public company and general corporate expenses charged to expense accounts. For the three months ended March 31, 2023 and 2022, administrative and general expenses by business segment, including AWR (parent), consisted of the following (dollar amounts in thousands):

	1	Three Months Ended March 31, 2023	7	Three Months Ended March 31, 2022	\$ CHANGE	% CHANGE
Water Services	\$	15,381	\$	15,596	\$ (215)	(1.4)%
Electric Services		2,673		2,166	507	23.4 %
Contracted Services		5,492		5,208	284	5.5 %
AWR (parent)		1		2	(1)	(50.0)%
Total administrative and general	\$	23,547	\$	22,972	\$ 575	2.5 %

Administrative and general expenses decreased for the water segment due, in large part, to a reduction of approximately \$447,000 to reflect the April 2023 proposed decision on the water general rate case for recovery of previously incurred costs that were tracked in CPUC-authorized memorandum accounts. This was offset by an increase in outside-service costs, labor and employee-related expenses, partially offset by a decrease in the service cost component of GSWC's defined-benefit pension plan. Due to GSWC's two-way pension balancing accounts authorized by the CPUC, changes in total net periodic benefit costs related to the pension plan have no material impact to earnings.

Administrative and general expenses increased at the electric segment primarily due to an increase in costs allocated from the general corporate office as a result of the allocation ratio update authorized in the proposed decision on the water general rate case. The increase in general corporate office expenses allocated to the electric segment also includes a corresponding and offsetting increase in adopted electric revenues, resulting in no impact to earnings.

Administrative and general expenses increased at the contracted services segment mainly due to higher labor and outside services.

Depreciation and Amortization

For the three months ended March 31, 2023 and 2022, depreciation and amortization by business segment consisted of the following (dollar amounts in thousands):

	Three Months Ended March 31, 2023	Three Months Ended March 31, 2022	\$ CHANGE	% CHANGE
Water Services	\$ 9,606	\$ 8,545	\$ 1,061	12.4 %
Electric Services	748	654	94	14.4 %
Contracted Services	849	915	(66)	(7.2)%
Total depreciation and amortization	\$ 11,203	\$ 10,114	\$ 1,089	10.8 %

The water general rate case proposed decision issued in April 2023 approves overall higher composite depreciation rates based on a revised depreciation study. The increase in composite depreciation rates increases the adopted water revenue requirement, with a corresponding increase in adopted depreciation expense, resulting in no impact to net earnings. Therefore, the overall increase in depreciation and amortization expenses at the water segment included the retroactive impact for the full year of 2022 of approximately \$576,000. There was also an increase in depreciation at the water and electric segments due to additions to utility plant and other fixed assets since the first quarter of 2022.

Maintenance

For the three months ended March 31, 2023 and 2022, maintenance expense by business segment consisted of the following (dollar amounts in thousands):

	7	Three Months Ended March 31, 2023	,	Three Months Ended March 31, 2022	\$ CHANGE	% CHANGE
Water Services	\$	1,960	\$	2,156	\$ (196)	(9.1)%
Electric Services		321		250	71	28.4 %
Contracted Services		869		734	135	18.4 %
Total maintenance	\$	3,150	\$	3,140	\$ 10	0.3 %

Maintenance expense decreased for the water segment due, in part, to a reduction of approximately \$98,000 to reflect the April 2023 proposed decision on the water general rate case for recovery of previously incurred costs that were tracked in CPUC-authorized memorandum accounts. Maintenance expense also decreased at the water segment due to timing of maintenance activities compared to the same period in 2022.

Maintenance expense increased at the electric and contracted services segments due to higher maintenance planned and unplanned activities as compared to the same period in 2022.

Property and Other Taxes

For the three months ended March 31, 2023 and 2022, property and other taxes by business segment consisted of the following (dollar amounts in thousands):

	7	Three Months Ended March 31, 2023	-	Three Months Ended March 31, 2022	\$ CHANGE	% CHANGE
Water Services	\$	5,139	\$	4,890	\$ 249	5.1 %
Electric Services		545		458	87	19.0 %
Contracted Services		611		505	106	21.0 %
Total property and other taxes	\$	6,295	\$	5,853	\$ 442	7.6 %

The increase to property and other taxes was primarily attributed to an increase in franchise fees at the water segment resulting from higher water revenues recognized in 2023 compared to 2022.

ASUS Construction

For the three months ended March 31, 2023, construction expenses for contracted services were \$18.9 million, increasing \$8.7 million compared to the same period in 2022 primarily due to an increase in construction activity resulting from timing differences of when such work was performed in 2023 as compared to the same period of 2022.

Interest Expense

For the three months ended March 31, 2023 and 2022, interest expense by business segment, including AWR (parent), consisted of the following (dollar amounts in thousands):

	Three Months Ended March 31, 2023	7	Three Months Ended March 31, 2022	\$ CHANGE	% CHANGE
Water Services	\$ 6,922	\$	5,236	\$ 1,686	32.2 %
Electric Services	834		112	722	644.6 %
Contracted Services	416		62	354	571.0 %
AWR (parent)	1,309		196	1,113	567.9 %
Total interest expense	\$ 9,481	\$	5,606	\$ 3,875	69.1 %

AWR's borrowings consist of bank debt under revolving credit facilities and long-term debt issuances at GSWC and BVES. Consolidated interest expense increased as compared to the same period in 2022 resulting primarily from an increase in total borrowing levels to support, among other things, the capital expenditures program at the regulated utilities, as well as an overall increase in average interest rates both short- and long-term. On January 13, 2023, GSWC issued \$130.0 million unsecured private-placement notes consisting of: \$100.0 million in aggregate notes at a coupon rate of 5.12% due January 31, 2033, and \$30.0 million in aggregate notes at a coupon rate of 5.22% due January 31, 2038. In April 2022, BVES issued \$35.0 million in unsecured private-placement notes consisting of 10 and 15 year term notes with interest rates at 4.548% and 4.949%, respectively.

Interest Income

For the three months ended March 31, 2023 and 2022, interest income by business segment, including AWR (parent), consisted of the following (dollar amounts in thousands):

	7	Three Months Ended March 31, 2023	7	Three Months Ended March 31, 2022	\$ CHANGE	% CHANGE	
Water Services	\$	1,428	\$	91	\$ 1,337	*	
Electric Services		261		(1)	262	*	
Contracted Services		189		197	(8)	(4.1)%	6
AWR (parent)		(14)		(4)	(10)	*	
Total interest income	\$	1,864	\$	283	\$ 1,581	*	

^{*}not meaningful

For the three months ended March 31, 2023, overall interest income increased by \$1.6 million as compared to the same period in 2022 due primarily to higher interest income earned on regulatory assets at the water segment bearing interest at the current 90-day commercial-paper rate, which have increased compared to 2022's rates, as well as an overall increase in recorded regulatory assets as a result of the proposed decision in the water general rate case received in April 2023.

Other Income and (Expense), net

For the three months ended March 31, 2023 and 2022, other income and (expense), net by business segment, consisted of the following (dollar amounts in thousands):

	Three Months Ended March 31, 2023	Three Months Ended March 31, 2022	\$ CHANGE	% CHANGE
Water Services	\$ 1,628	\$ (598)	\$ 2,226	*
Electric Services	13	143	(130)	(90.9)%
Contracted Services	(30)	36	(66)	*
Total other income and (expense), net	\$ 1,611	\$ (419)	\$ 2,030	*

^{*} not meaningful

For the three months ended March 31, 2023, other income (net of other expense) increased mostly as a result of gains of \$1.6 million recorded on investments held to fund one of the Company's retirement plans, as compared to losses of \$1.7 million generated during the same period in 2022, both due to financial market conditions. This was partially offset by an increase in the non-service cost components of net periodic benefit costs related to the Company's defined-benefit pension plan and other retirement benefits. However, as a result of GSWC's and BVES's two-way pension balancing accounts authorized by the CPUC, changes in total net periodic benefit costs related to the pension plan have no material impact to earnings.

Income Tax Expense

For the three months ended March 31, 2023 and 2022, income tax expense by business segment, including AWR (parent), consisted of the following (dollar amounts in thousands):

	Three Months Ended March 31, 2023	Three Months Ended March 31, 2022	\$ CHANGE	% CHANGE
Water Services	\$ 8,910	\$ 2,689	\$ 6,221	231.3 %
Electric Services	701	952	(251)	(26.4)%
Contracted Services	1,685	944	741	78.5 %
AWR (parent)	 (544)	(124)	(420)	338.7 %
Total income tax expense	\$ 10,752	\$ 4,461	\$ 6,291	141.0 %

Consolidated income tax expense for the three months ended March 31, 2023 increased by \$6.3 million primarily due to an increase in pretax income as compared to the same period in 2022, partially offset by a decrease in the consolidated effective income tax rate ("ETR"). AWR's ETR was 23.8% and 24.0% for the three months ended March 31, 2023 and 2022, respectively. GSWC's ETR was 24.5% and 23.9% for the three months ended March 31, 2023 and 2022, respectively. GSWC's income tax expense was affected by net changes in certain flowed-through and permanent items. The increase in the tax benefit recorded at AWR (parent) during the three months ended March 31, 2023 compared to the same period in 2022 was primarily due to an increase in pretax loss at AWR (parent) resulting from higher interest expense, as well as changes in state unitary taxes.

For a comparison of the financial results for the first quarter of 2022 to 2021, see "Consolidated Results of Operations-Three Months Ended March 31, 2022 and March 31, 2021" in Registrant's Form 10-Q for the period ended March 31, 2022 filed with the SEC.

Critical Accounting Policies and Estimates

Critical accounting policies and estimates are those that are important to the portrayal of AWR's financial condition, results of operations and cash flows and require the most difficult, subjective or complex judgments of AWR's management. The need to make estimates about the effect of items that are uncertain is what makes these judgments difficult, subjective and/or complex. Management makes subjective judgments about the accounting and regulatory treatment of many items. These judgments are based on AWR's historical experience, terms of existing contracts, AWR's observance of trends in the industry, and information available from other outside sources, as appropriate. Actual results may differ from these estimates under different assumptions or conditions.

The critical accounting policies used in the preparation of Registrant's financial statements that it believes affect the more significant judgments and estimates used in the preparation of its consolidated financial statements presented in this report are described in "Management's Discussion and Analysis of Financial Condition and Results of Operations" included in Registrant's Annual Report on Form 10-K for the year ended December 31, 2022 filed with the SEC. There have been no material changes to Registrant's critical accounting policies.

Liquidity and Capital Resources

AWR

AWR's regulated business is capital intensive and requires considerable capital resources. A portion of these capital resources is provided by internally generated cash flows from operations. AWR anticipates that interest expense will increase in future periods due to the need for additional external capital to fund construction programs at its regulated utilities and as market interest rates increase. In addition, as the capital investment program continues to increase, coupled with the elimination of bonus depreciation for regulated utilities due to tax reform enacted in 2017, AWR and its subsidiaries anticipate they will need to access external financing more often. AWR believes that costs associated with capital used to fund construction at GSWC and BVES will continue to be recovered through water and electric rates charged to customers.

AWR funds its operating expenses and pays dividends on its outstanding Common Shares primarily through dividends from its wholly owned subsidiaries. The ability of GSWC and BVES to pay dividends to AWR is restricted by California law. Under these restrictions, approximately \$658.3 million was available for GSWC to pay dividends to AWR on March 31, 2023. Approximately \$67.2 million was available for BVES to pay dividends to AWR as of March 31, 2023. ASUS's ability to pay dividends to AWR is dependent upon state laws in which each Military Utility Privatization Subsidiary operates, as well as ASUS's ability to pay dividends under California law.

When necessary, AWR obtains funds from external sources through the capital markets and from bank borrowings. Access to external financing on reasonable terms depends on the credit ratings of AWR and GSWC and current business conditions, including that of the water utility industry in general as well as conditions in the debt or equity capital markets.

AWR currently borrows under a revolving credit facility with a current borrowing capacity of \$280.0 million and provides funds to GSWC and ASUS in support of their operations through intercompany borrowing agreements on terms that are similar to that of the credit facility. The interest rate charged to GSWC and ASUS is comparable to the interest rate AWR pays under the credit facility. AWR's credit agreement was set to expire on May 23, 2023. On May 8, 2023, the credit facility was amended to extend the maturity date by two-months to provide adequate time to put in place a new credit agreement. The amendment extends the maturity date of the existing credit agreement to July 23, 2023 or an earlier date on which the credit agreement is either terminated or cancelled when superseded by a new agreement. All intercompany borrowing agreements will expire concurrent with the expiration of AWR's credit facility. Therefore, the outstanding borrowings under the credit facility of \$175.5 million as of March 31, 2023 have been classified as current liabilities on AWR's Consolidated Balance Sheet, thus creating a negative working-capital condition for AWR of \$168.0 million. Additionally, as of March 31, 2023, the \$45.0 million of outstanding intercompany borrowings of GSWC from AWR have been classified as current liabilities on GSWC's Balance Sheet, also creating a negative working-capital condition for GSWC of \$87.9 million. As of May 10, 2023, neither AWR nor GSWC have sufficient liquidity or capital resources to repay its credit facility or intercompany borrowings, respectively, without either extending its existing credit facility, entering into a new credit facility, or issuing new debt or equity.

AWR is confident and believes it is probable that it will be able to execute a new credit facility agreement with the needed borrowing capacities required to repay its existing credit facility and to run its operations given Registrant's ability to generate consistent cash flows, its A+ credit ratings, and its history in obtaining revolving credit facilities to meet its working-capital needs, as well as its history of successfully raising debt as recently done with GSWC's issuance of \$130 million in unsecured long-term notes on January 13, 2023. In addition, management is considering a separate credit facility for GSWC to support its standalone water utility operations. Alternatively, AWR may enter into a new intercompany borrowing agreement with GSWC. Accordingly, management has concluded that Registrant will be able to satisfy its obligations, including those under its current credit facility, for at least the next twelve months from the issuance date of these financial statements. However, Registrant's ability to access the capital markets or to otherwise obtain sufficient financing may be affected by future conditions and, accordingly, no assurances can be made that Registrant will be successful in implementing its financing plans.

BVES has a separate \$35.0 million revolving credit facility without a parent guaranty that matures on July 1, 2024. As of March 31, 2023, there were \$25.0 million of outstanding borrowings under this facility. Under the terms of the credit agreement, BVES has the option to increase the facility by an additional \$15.0 million, subject to lender approval. Interest rates under this facility are currently based on LIBOR. Effective July 1, 2023, all new borrowings under this credit agreement will be based on SOFR. BVES does not believe the change from LIBOR to a new benchmark rate such as SOFR will have a material impact on its financing costs.

In June 2022, Standard and Poor's Global Ratings ("S&P") affirmed an A+ credit rating for both AWR and GSWC. S&P also affirmed its negative outlook for both companies. S&P's debt ratings range from AAA (highest possible) to D (obligation is in default). In January 2023, Moody's Investors Service ("Moody's") affirmed its A2 rating with a stable outlook for GSWC. Securities ratings are not recommendations to buy, sell or hold a security, and are subject to change or withdrawal at any time by the rating agencies. Management believes that AWR's and GSWC's sound capital structures and A+ credit ratings, combined with its financial discipline, will enable AWR to access the debt and equity markets. However, unpredictable financial market conditions in the future may limit its access or impact the timing of when to access the market, in which case AWR may choose to temporarily reduce its capital spending.

AWR's ability to pay cash dividends on its Common Shares outstanding depends primarily upon cash flows from its subsidiaries. AWR intends to continue paying quarterly cash dividends on or about March 1, June 1, September 1 and December 1, subject to earnings and financial conditions, regulatory requirements and such other factors as the Board of Directors may deem relevant. On May 8, 2023, AWR's Board of Directors approved a second quarter dividend of \$0.3975 per share on AWR's Common Shares. Dividends on the Common Shares will be paid on June 1, 2023 to shareholders of record at the close of business on May 19, 2023. AWR has paid common dividends every year since 1931, and has increased the dividends received by shareholders each calendar year for 68 consecutive years, which places it in an exclusive group of companies on the New York Stock Exchange that have achieved that result. AWR has achieved a 9.2% compound annual growth rate in its annual dividend payments from 2012 – 2022. AWR's current policy is to achieve a compound annual growth rate in the dividend of more than 7% over the long-term.

Cash Flows from Operating Activities:

Cash flows from operating activities have generally provided sufficient cash to fund operating requirements, including a portion of construction expenditures at GSWC and BVES, and construction expenses at ASUS, and to pay dividends. AWR's future cash flows from operating activities are expected to be affected by a number of factors, including utility regulation; changes in tax law; maintenance expenses; inflation; compliance with environmental, health and safety standards; production costs; customer growth; per-customer usage of water and electricity; weather and seasonality; conservation efforts; compliance with local governmental requirements, including mandatory restrictions on water use; the lingering effects of the COVID-19 pandemic on its customers' ability to pay utility bills; and required cash contributions to pension and post-retirement plans. Future cash flows from contracted services subsidiaries will depend on new business activities, existing operations, the construction of new and/or replacement infrastructure at military bases, timely economic price and equitable adjustment of prices, and timely collection of payments from the U.S. government and other prime contractors operating at the military bases, and any adjustments arising out of an audit or investigation by federal governmental agencies.

ASUS funds its operating expenses primarily through internal operating sources, which include U.S. government funding under contracts with the U.S. government for operations and maintenance costs and construction activities, as well as investments by, or loans from, AWR. ASUS, in turn, provides funding to its subsidiaries. ASUS's subsidiaries may also from time to time provide funding to ASUS or other subsidiaries of ASUS.

Cash flows from operating activities are primarily generated by net income, adjusted for non-cash expenses such as depreciation and amortization, and deferred income taxes. Cash generated by operations varies during the year. Net cash provided by operating activities of AWR was \$7.0 million for the three months ended March 31, 2023 as compared to \$38.0 million for the same period in 2022. During the first quarter of 2022, GSWC and BVES received \$9.5 million and \$321,000, respectively, in COVID-19 relief funds from the state of California to provide assistance to customers for delinquent water and electric customer bills incurred during the pandemic. There were no relief funds received during the first quarter of 2023.

The decrease in operating cash flows was also due to a 17% decrease in billed water consumption, as well as the continued delay in receiving the water general rate case final decision as billed water revenues in 2022 and 2023 were based on 2021 adopted rates pending a final CPUC decision, while operating expenses continued to rise due to inflation. On April 13, 2023, GSWC received a proposed decision on its water general rate case with 2022 and 2023 rates retroactive to January 1, 2022 and 2023, respectively. A final decision by the CPUC is expected during the second quarter of 2023. Once a final decision is received, GSWC will request recovery through a surcharge of all revenues accumulated since 2022 with retroactive new rates, and will also file for the 2023 second-year rate increases. Furthermore, decreases in operating cash flows was also due to differences in the timing of vendor payments, and the timing of billing of and cash receipts for construction work at military bases. The billings (and cash receipts) for this construction work generally occur at completion of the work or in accordance with a billing schedule contractually agreed to with the U.S. government and/or other prime contractors. Thus, cash flow from construction-

related activities may fluctuate from period to period with such fluctuations representing timing differences of when the work is being performed and when the cash is received for payment of the work. The timing of cash receipts and disbursements related to other working capital items also affected the change in net cash provided by operating activities.

The timing of cash receipts and disbursements related to other working capital items also affected the change in net cash provided by operating activities.

Cash Flows from Investing Activities:

Net cash used in investing activities was \$49.2 million for the three months ended March 31, 2023 as compared to \$35.0 million for the same period in 2022, which is mostly related to capital expenditures at the regulated utilities for both periods. AWR invests capital to provide essential services to its regulated customer base, while working with the CPUC to have the opportunity to earn a fair rate of return on investment. AWR's infrastructure investment plan consists of both infrastructure renewal programs (to replace infrastructure, including those to mitigate wildfire risk) and major capital investment projects (to construct new water treatment, supply and delivery facilities). The regulated utilities may also be required from time to relocate existing infrastructure in order to accommodate local infrastructure improvement projects. Projected capital expenditures and other investments are subject to periodic review and revision.

For the year 2023, the regulated utilities' company-funded capital expenditures are expected to be between \$140 million and \$160 million, barring any delays resulting from changes in capital improvement schedules due to supply-chain issues.

Cash Flows from Financing Activities:

AWR's financing activities include primarily: (i) the issuance and repayment of long-term debt and notes payable to banks, and (ii) the payment of dividends on Common Shares. In order to finance new infrastructure, GSWC also receives customer advances (net of refunds) for, and contributions in aid of, construction. Borrowings on AWR's and BVES's credit facilities are used to fund GSWC and BVES capital expenditures, respectively, until long-term financing is arranged. Overall debt levels are expected to increase to fund a portion of the costs of the capital expenditures that will be made by the regulated utilities.

Net cash provided by financing activities was \$38.3 million for the three months ended March 31, 2023 as compared to \$2.2 million during the same period in 2022. The increase in net cash provided by financing activities in 2023 was due primarily to an increase in total borrowing levels necessary to support operations affected by a significant decrease in cash flows from operating activities and to support, among other things, the capital expenditures program at the regulated utilities. In January 2023, GSWC issued \$130.0 million of unsecured private-placement notes and used the proceeds to pay down the majority of its outstanding intercompany borrowings from AWR, which in turn used the proceeds to pay down outstanding borrowings under the AWR credit facility. During the three months ended March 31, 2023, AWR had net payments on its credit facilities of \$77.0 million, while during the three months ended March 31, 2022, AWR had a net increase in borrowings on its credit facilities of \$16.0 million to support operations and capital expenditures.

GSWC

GSWC funds its operating expenses, payments on its debt, dividends on its outstanding common shares, and a portion of its construction expenditures through internal sources. Internal sources of cash flow are provided primarily by retention of a portion of earnings from operating activities. Internal cash generation is influenced by factors such as weather patterns, conservation efforts, environmental regulation, litigation, changes in tax law and deferred taxes, changes in supply costs and regulatory decisions affecting GSWC's ability to recover these supply costs, timing of rate relief, increases in maintenance expenses and capital expenditures, surcharges authorized by the CPUC to enable GSWC to recover expenses previously incurred from customers, and CPUC requirements to refund amounts previously charged to customers. Internal cash flows may also be impacted by delays in receiving payments from GSWC customers due to the lingering effects of the COVID-19 pandemic.

GSWC may, at times, utilize external sources for long-term financing, as well as obtain funds from equity investments and intercompany borrowings from its parent, AWR, to help fund a portion of its operations and construction expenditures. AWR borrows under a revolving credit facility and provides funds to GSWC in support of its operations under intercompany borrowing arrangements. On April 22, 2022, the AWR credit facility was amended to increase the borrowing capacity from \$200.0 million to \$280.0 million, which provided an increase in GSWC's borrowing capacity under its intercompany borrowing agreement. All intercompany borrowing agreements expire concurrently with the expiration of AWR's credit facility in July 2023. Upon the expiration of GSWC's intercompany borrowing agreement with AWR, GSWC intends to either execute a new intercompany borrowing agreement consistent with a new credit facility at AWR, or enter into its own separate credit facility. As of March 31, 2023, GSWC had \$45.0 million outstanding under its intercompany borrowing arrangement with AWR that has been classified as a current liability on GSWC's Balance Sheet, creating a negative working-capital condition for GSWC of \$87.9 million. As of May 10, 2023, GSWC does not have sufficient liquidity or capital resources to repay its intercompany borrowings from AWR without either issuing new long-term debt, or entering into a new short-term borrowing arrangement, which is confident it will be

able to achieve. On January 13, 2023, GSWC issued unsecured private placement notes totaling \$130.0 million. GSWC used the proceeds to pay off the majority of outstanding borrowings under the intercompany borrowing arrangement with AWR.

In addition, GSWC receives advances and contributions from customers, home builders and real estate developers to fund construction necessary to extend service to new areas. Advances for construction are generally refundable at a rate of 2.5% in equal annual installments over 40 years. Utility plant funded by advances and contributions is excluded from rate base. GSWC amortizes contributions in aid of construction at the same composite rate of depreciation for the related property.

Cash Flows from Operating Activities:

Net cash provided by operating activities was \$10.9 million for the three months ended March 31, 2023 as compared to \$31.3 million for the same period in 2022. During the first quarter of 2022, GSWC received \$9.5 million in COVID-19 relief funds from the state of California to provide assistance to customers for delinquent water bills incurred during the pandemic. There were no relief funds received during the first quarter of 2023. The decrease in operating cash flow was also due to a 17% decrease in billed water consumption, as well as the continued delay in receiving the final water general rate case decision as billed water revenues in 2022 and 2023 were based on 2021 adopted rates pending a final CPUC decision, while operating expenses continued to rise due to inflation. On April 13, 2023, GSWC received a proposed decision on its water general rate case with 2022 and 2023 rates retroactive to January 1, 2022 and 2023, respectively. A final decision by the CPUC is expected during the second quarter of 2023. Once a final decision is received, GSWC will request recovery through a surcharge of all revenues accumulated since 2022 with retroactive new rates, and will also file for 2023 second-year rate increases. Furthermore, decreases in operating cash flows was also due to differences in the timing of vendor payments.

The timing of cash receipts and disbursements related to other working capital items also affected the change in net cash provided by operating activities.

Cash Flows from Investing Activities:

Net cash used in investing activities was \$41.8 million for the three months ended March 31, 2023 as compared to \$31.3 million for the same period in 2022, which is mostly related to spending under GSWC's infrastructure investment plans that are consistent with capital budgets authorized in its general rate cases.

Cash Flows from Financing Activities:

Net cash provided by financing activities was \$31.4 million for the three months ended March 31, 2023 as compared to \$4.2 million for the same period in 2022. In January 2023, GSWC issued \$130.0 million of unsecured private-placement notes and also issued \$10 million of equity to AWR. GSWC used the proceeds from its issuance of private-placement notes and the issuance of equity to pay-off all its outstanding intercompany borrowings from AWR. The CPUC requires GSWC to fully pay-off all intercompany borrowings it has from AWR within a 24-month period. As a result, during the three months ended March 31, 2023, GSWC had net payments on intercompany borrowings of \$84.0 million. During the three months ended March 31, 2022, GSWC had an increase in net intercompany borrowings of \$18.0 million from AWR parent.

Contractual Obligations and Other Commitments

Registrant has various contractual obligations, which are recorded as liabilities in the consolidated financial statements. Other items, such as certain purchase commitments, are not recognized as liabilities in the consolidated financial statements but are required to be disclosed. In addition to contractual maturities, Registrant has certain debt instruments that contain an annual sinking fund or other principal payments. Registrant believes that it will be able to refinance debt instruments at their maturity through public issuance, or private placement, of debt or equity. Annual payments to service debt are generally made from cash flows from operations.

See "Management's Discussion and Analysis of Financial Condition and Results of Operations—Contractual Obligations, Commitments and Off-Balance Sheet Arrangements" section of the Registrant's Form 10-K for the year ended December 31, 2022 filed with the SEC for a detailed discussion of contractual obligations and other commitments.

Contracted Services

Under the terms of utility privatization contracts with the U.S. government, each contract's price is subject to an economic price adjustment ("EPA") on an annual basis. In the event that ASUS (i) is managing more assets at specific military bases than were included in the U.S. government's request for proposal, (ii) is managing assets that are in substandard condition as compared to what was disclosed in the request for proposal, (iii) prudently incurs costs not contemplated under the terms of the utility privatization contract, and/or (iv) becomes subject to new regulatory requirements, such as more stringent water-quality standards, ASUS is permitted to file, and has filed, requests for equitable adjustment ("REAs"). The timely filing for and receipt of EPAs and/or REAs continues to be critical in order for the Military Utility Privatization Subsidiaries to recover increasing costs of operating, maintaining, renewing and replacing the water and/or wastewater systems at the military bases it serves.

During sequestration or automatic spending cuts, the Military Utility Privatization Subsidiaries did not experience any earnings impact to their existing operations and maintenance and renewal and replacement services, as utility privatization contracts are an "excepted service." With the expiration of sequestration, similar issues including further sequestration pursuant to the Balanced Budget and Emergency Deficit Control Act may arise as part of the fiscal uncertainty and/or future debt-ceiling limits imposed by Congress. Any future impact on ASUS and its operations through the Military Utility Privatization Subsidiaries will likely be limited to (a) the timing of funding to pay for services rendered, (b) delays in the processing of EPAs and/or REAs, (c) the timing of the issuance of contract modifications for new construction work not already funded by the U.S. Government, and/or (d) delays in solicitation for and/or awarding of new contracts under the Department of Defense utility privatization program.

At times, the DCAA and/or the DCMA may, at the request of a contracting officer, perform audits/reviews of contractors for compliance with certain government guidance and regulations, such as the Federal Acquisition Regulations and Defense Federal Acquisition Regulation Supplements. Certain audit/review findings, such as system deficiencies for government-contract-business-system requirements, may result in delays in the resolution of filings submitted to and/or the ability to file new proposals with the U.S. government.

Regulatory Matters

An update on various regulatory matters is included in the discussion under the section titled "Overview" in this Form 10-Q's "Management's Discussion and Analysis of Financial Condition and Results of Operations". The discussion below focuses on other regulatory matters and developments.

Water Segment:

Recent Changes in Rates

Rates that GSWC is authorized to charge are determined by the CPUC in general rate cases. Water revenues billed to customers for the three months ended March 31, 2023 and 2022 were based on 2021 adopted rates, pending a final decision by the CPUC. GSWC has a pending general rate case that will determine new water rates for the years 2022–2024. On April 13, 2023, GSWC received a proposed decision from the assigned administrative law judge at the CPUC on GSWC's water general rate case with rates retroactive to January 1, 2022. Among other items, the proposed decision approves and adopts in its entirety the settlement agreement between GSWC and Public Advocates in November 2021 that had been filed with the CPUC, and resolves all issues related to the 2022 annual revenue requirement in the general rate case application. The settlement agreement also allows for additional increases in adopted revenues for 2023 and 2024 subject to an earnings test and changes to the forecasted inflationary index values. As a result of receiving a proposed decision that approves the settlement agreement in its entirety, the impact of retroactive new rates for the full year of 2022 and the estimated second-year rate increases for the three months ended March 31, 2023 have been reflected in the 2023 first quarter results.

GSWC expects to receive a final decision during the second quarter of 2023. Once a final decision is issued, GSWC will request recovery of the cumulative amounts related to 2022 rate increases through a surcharge. Actual increases for 2023 will be determined when the filings to implement the new rate increases are approved by the CPUC, and will be calculated using the inflationary index values at that time. GSWC will file for the 2023 increases once the CPUC approves the final decision.

Cost of Capital Proceeding

GSWC also has a pending cost of capital proceeding that will determine a new return on rate base for the years 2022 – 2024. On May 9, 2023, GSWC received a proposed decision from the assigned administrative law judge at the CPUC on the cost of capital proceeding. Among other things, the proposed decision (i) adopts GSWC's requested capital structure and cost of debt filed in the application; (ii) adopts a return on equity of 8.85% for GSWC as compared to 8.9% previously authorized; (iii) allows for the continuation of the Water Cost of Capital Mechanism ("WCCM"); and (iv) adopts the new cost of capital for the three-year period commencing January 1, 2022, through December 31, 2024. Comments on the proposed decision are due on May 30. In March 2023, the CPUC issued a decision that approved an extension of the statutory deadline for a final decision in the cost of capital proceeding to August 10, 2023.

While this proceeding is pending, the previously authorized return, which includes a cost of debt of 6.6%, has been and is presently being billed to water customers during 2022 and 2023 until a final decision is issued in this proceeding. In the pending cost of capital proceeding, GSWC requested an updated cost of debt of 5.1%, which is being adopted in the proposed decision. Based on management's analysis of this regulatory proceeding and the associated accounting in 2022 and through March 31, 2023, GSWC has reduced revenues and recorded a corresponding regulatory liability for revenues subject to refund based on its best estimate at this time, which includes the impact of GSWC's lower cost of debt adopted in the proposed decision. For the three months ended March 31, 2023 and 2022, GSWC reduced revenues by \$1.8 million and \$1.4 million, respectively, that are subject to refund. An additional reduction to revenues of \$1.1 million was recorded during the first quarter of 2023 to reflect the incremental impact of revenues subject to refund from the new 2022 rates recorded as a result of the proposed water general rate case decision. As of March 31, 2023, GSWC had an aggregated regulatory liability of \$9.3 million for the estimated revenues subject to refund from the pending cost of capital proceeding. However, at this time, management cannot predict the ultimate outcome and any changes that may be made to the final decision in the cost of capital application, and the associated impact on 2022 and 2023 revenues. Changes in estimates will be made, if necessary, as more information in this proceeding becomes available.

Furthermore, the proposed decision continues the WCCM for the years 2023 and 2024, which adjusts the return on equity and rate of return on rate base between the three-year cost of capital proceedings only if there is a positive or negative change of more than 100 basis points in the average of the Moody's Aa utility bond rate as measured over the period October 1 through September 30. If there is a positive or negative change of more than 100 basis points, the return on equity is adjusted by one half of the difference. For the period from October 1, 2021 through September 30, 2022, the Moody's rate increased by 103 basis points from the benchmark, which triggers the WCCM adjustment. GSWC recognized revenues for the first quarter of 2023 and all of 2022 based on the previously authorized return of equity of 8.9% that is presently being billed to water customers pending a final decision in the cost of capital proceeding.

Electric Segment:

Recent Changes in Rates

On August 30, 2022, BVES filed a new general rate case application with the CPUC to determine new rates for the years 2023–2026. On December 15, 2022, the CPUC approved a decision for BVES to establish a general rate case memorandum account that makes the new 2023 rates effective and retroactive to January 1, 2023. Based on the established schedule in this proceeding, a proposed decision is expected in the fourth quarter of 2023.

Vegetation Management, Wildfire Mitigation Plans and Legislation

The CPUC adopted regulations intended to enhance the fire safety of overhead electric power lines. Those regulations included increased minimum clearances around electric power lines. BVES was authorized to track incremental costs incurred to implement the regulations in a fire hazard prevention memorandum account for the purpose of obtaining cost recovery in a future general rate case. The August 2019 final decision also authorized BVES to record incremental costs related to vegetation management, such as costs for increased minimum clearances around electric power lines, in the CPUC-approved memorandum account for future recovery. As of March 31, 2023, BVES had approximately \$9.3 million in incremental vegetation management costs recorded as a regulatory asset. As part of its general rate case application filing with the CPUC in August 2022, BVES requested recovery of the costs accumulated in this memorandum account, which will be subject to CPUC review.

California legislation enacted in September 2018 requires all investor-owned electric utilities to have a wildfire mitigation plan ("WMP") approved by the Office of Energy Infrastructure Safety ("OEIS") and ratified by the CPUC. The WMP must include a utility's plans on constructing, maintaining, and operating its electrical lines and equipment to minimize the risk of catastrophic wildfire. BVES submitted an update to its WMP in May 2022 to OEIS for approval prior to going to the

CPUC for ratification. In December 2022, OEIS issued a final decision of approval to BVES for its 2022 WMP update. In February 2023, the CPUC ratified BVES's current WMP. As of March 31, 2023, BVES has approximately \$4.8 million related to expenses accumulated in its WMP memorandum accounts that have been recognized as regulatory assets for future recovery. All capital expenditures and other costs incurred through March 31, 2023 as a result of BVES's WMPs are not currently in rates and have been filed for future recovery in BVES's general rate case application in August 2022, which will be subject to CPUC review.

Additionally, the governor of California approved Assembly Bill ("AB") 1054 in July 2019 that, among other things, changed the burden of proof applicable in CPUC proceedings in which an electric utility with a valid safety certification seeks to recover wildfire costs. Previously, an electric utility seeking to recover costs had the burden to prove that it acted reasonably. Under AB 1054, if an electric utility has a valid safety certification, it will be presumed to have acted reasonably unless a party to the relevant proceeding creates a "serious doubt" as to the reasonableness of the utility's conduct. In September 2021, OEIS under the California Natural Resources Agency approved BVES's latest safety certification filing. In December 2022, OEIS issued a renewal of its safety certification to BVES.

2023 Winter Storm Other Regulatory Asset

BVES activated a catastrophic emergency memorandum account ("CEMA") to track the incremental costs incurred in response to a severe winter storm that occurred during the first quarter of 2023, which resulted in the declaration of an emergency by the governor of California. Incremental costs of approximately \$810,000 were incurred and included in the CEMA account, which has been recorded as a regulatory asset as of March 31, 2023 for future recovery. The incremental costs included in the CEMA account will be subject to review and approval by the CPUC. CEMA accounts are well-established cost recovery mechanisms authorized as a result of a state/federal declared emergency, and are therefore recognized as regulatory assets for future recovery. As a result, the amounts recorded in this CEMA account did not impact BVES's earnings.

See also "Management's Discussion and Analysis of Financial Condition and Results of Operations—Regulatory Matters" section of the Registrant's Form 10-K for the year-ended December 31, 2022 filed with the SEC for a discussion of other regulatory matters.

Environmental Matters

AWR's subsidiaries are subject to stringent environmental regulations. GSWC is required to comply with the safe drinking water standards established by the U.S. Environmental Protection Agency ("U.S. EPA") and the Division of Drinking Water ("DDW"), under the State Water Resources Control Board ("SWRCB"). The U.S. EPA regulates contaminants that may have adverse health effects that are known or likely to occur at levels of public health concern, and the regulation of which will provide a meaningful opportunity for health risk reduction. The DDW, acting on behalf of the U.S. EPA, administers the U.S. EPA's program in California. Similar state agencies administer these rules in the other states in which Registrant operates.

GSWC currently tests its water supplies and water systems according to, among other things, requirements listed in the Federal Safe Drinking Water Act ("SDWA"). GSWC works proactively with third parties and governmental agencies to address issues relating to known contamination threatening GSWC water sources. GSWC also incurs operating costs for testing to determine the levels, if any, of the constituents in its sources of supply, and additional expense to treat contaminants in order to meet the federal and state maximum contaminant level standards and consumer demands. GSWC expects to incur additional capital costs as well as increased operating costs to maintain or improve the quality of water delivered to its customers in light of anticipated stress on water resources associated with watershed and aquifer pollution, drought impacts, as well as to meet future water quality standards and consumer expectations. The CPUC rate making process provides GSWC with the opportunity to recover prudently incurred capital and operating costs in future filings associated with achieving water quality standards. Management believes that such incurred and expected future costs should be authorized for recovery by the CPUC.

<u>Drinking Water Notification Levels and proposed Maximum Contamination Levels:</u>

In July 2018, DDW issued drinking water notification levels for certain fluorinated organic chemicals used to make certain fabrics and other materials, and used in various industrial processes. These chemicals were also present in certain fire suppression agents. These chemicals are referred to as perfluoroalkyl substances ("PFAS"). Notification levels are health-based advisory levels established for contaminants in drinking water for which maximum contaminant levels have not been established. The U.S. EPA has also established health advisory levels for these compounds. Notification to consumers and stakeholders is required when the advisory levels or notification levels are exceeded. Assembly Bill 756, signed into law in July 2019 and effective in January 2020, requires, among other things, additional notifications by water systems when they detect levels of PFAS above response levels.

GSWC is in the process of collecting and analyzing samples for PFAS under the direction of DDW. GSWC has removed some wells from service, installed treatment in some wells and expects to incur additional costs to treat impacted wells. GSWC has provided customers with information regarding PFAS detection, and provided updated information via its website.

To date, DDW has established advisory drinking water notification and response levels for four individual PFAS chemicals namely, perfluorooctanoic acid ("PFOA"), perfluorooctane sulfonate ("PFOS"), perfluorobutane sulfonate ("PFBS"), and perfluorohexane sulfonate ("PFHXS").

In March 2023, the U.S. EPA proposed maximum contaminant levels ("MCLs") for six specific PFAS compounds in drinking water. The proposed rule addresses PFOS and PFOA as individual constituents and addresses the other four PFAS constituents as a mixture of chemicals. The proposed MCLs are 4 parts per trillion for PFOA and PFOS and a hazard index of 1 for PFBS, PFHxS, perfluorononanoic acid ("PFNA") and GenX chemicals.

When finalized, the proposed regulation will require public water systems to monitor and treat for these chemicals. It will also require water systems to notify the customers and reduce the levels if it exceeds the regulatory standards. The U.S. EPA anticipates finalizing and adopting this rule by the end of 2023. Once the rule is finalized, water systems will be required to comply with the MCLs after a specified implementation period, which is currently anticipated to be three years from the rule-adoption date. These proposed MCLs, once finalized, are expected to increase GSWC's water treatment and other operating costs.

Lead and Copper Rule Revisions:

On December 16, 2021, the U.S. EPA announced the lead and copper rule revisions under an executive order with a compliance date of October 16, 2024. Additionally, the U.S. EPA announced its intention to develop a new proposed rule, the Lead and Copper Rule Improvements ("LCRI") that will further strengthen the regulatory framework prior to the October 2024 compliance date. There are still many unknowns regarding the implementation of the rule. The details of the requirements will be better understood over the next year once the LCRI is published and a final rule is approved.

See "Management's Discussion and Analysis of Financial Condition and Results of Operations—Environmental Matters" section of the Registrant's Form 10-K for the year-ended December 31, 2022 filed with the SEC for a discussion of environmental matters applicable to AWR and its subsidiaries.

Water Supply

See "Management's Discussion and Analysis of Financial Condition and Results of Operations—California Drought" section of the Registrant's Form 10-K for the year-ended December 31, 2022 filed with the SEC for a discussion of water supply issues. The discussion below focuses on significant matters and changes since December 31, 2022.

Drought Impact:

In May 2018, the California Legislature passed two bills that provide a framework for long-term water-use efficiency standards and drought planning and resiliency. The initial steps in implementation of this legislation have been laid out in a summary document by the California Department of Water Resources ("DWR") and State Water Resources Control Board ("SWRCB"). A notable milestone is the establishment of an indoor water use standard of 55 gallons per capita per day ("gpcd") until 2025. Legislation signed by the Governor into law in September 2022 has set more stringent indoor standard targets than initially set forth in the 2018 legislation. The indoor standard will now be set at 47 gpcd in 2025 and then reduced to 42 gpcd in 2030 (previously had been set at 52.5 gpcd and 50 gpcd, respectively). The SWRCB will begin a formal rulemaking process in the next several months and consider adoption of a final regulation for overall conservation standards by the end of this year that will include the indoor standard as well as outdoor use standards.

California started the 2023 water year, beginning on October 1, 2022, as a potentially fourth driest consecutive year of drought. However, a series of nine atmospheric storm events between January and March 2023 have delivered a promising outlook to the State's supply conditions. As of May 2, 2023, the U.S. Drought Monitor reported that none of California was in "Extreme Drought" as compared to 41% one year ago, and none of California is in "Severe Drought" as compared to 33% just three months ago and 95% one year ago. On April 27, 2023, the DWR reported that the statewide snowpack is at 211% of average for April 1 and 54 inches of snow water equivalent making this year's snowpack one of the largest since the 1950's.

At the start of 2023, DWR initially set the California State Water Project ("SWP") allocations at 5%. However, due to improved precipitation and snow levels experienced state-wide, DWR increased the SWP allocation to 75%, and as a result, the Metropolitan Water District of Southern California ("MWD") lifted restrictions that had impacted SWP dependent service areas of Simi Valley and Claremont that had been in place since mid-2022. DWR again increased the SWP allocation to 100% on April 20, 2023, which is the first time the full contracted SWP allocation has been at this level since 2006.

On March 15, 2023, the governor of California issued an executive order modifying the drought restrictions that were issued in March 2022. The order ended the voluntary water use requirement as well as the required stage 2 implementation of the water supply contingency plans but did not end all water use restrictions. GSWC will continue to work with its local suppliers to assess water supply conditions and water-use restrictions in its service areas and make appropriate adjustments as needed. In response to improving supply conditions in much of the State, GSWC has made an advice letter filing with the CPUC to move from stage 2 to stage 1 effective on May 14, 2023 in all of GSWC's service areas except for three coastal systems that are still experiencing depressed groundwater water levels.

Prolonged drought conditions still exist on the Colorado River System, which is experiencing historically low reservoir levels in Lake Mead and Lake Powell. Urgent action to reduce water demand on the lower river by 2 to 4 million acre feet annually has been requested by the US Bureau of Reclamation (the "Bureau"). The Bureau prepared a supplemental environmental impact statement with options that may result in modifications to current agreements. This may result in water delivery cuts by all of the lower states including California.

Other Climate Change Matters

Climate change is one area that we focus on as we develop and execute our business strategy and financial planning, both in the short- and long-term. The risks posed by climate variability increase the need for us to plan for and address supply resiliency. Climate change has also impacted electric utilities in California increasing wildfire risks and requiring the need to develop robust wildfire mitigation plans. We address these and other climate change risks by planning, assessing, mitigating, and investing in our infrastructure for the long-term benefit of our communities. See "Item 1. Business Overview" section of Registrant's Form 10-K for the year-ended December 31, 2022 filed with the SEC for a discussion of climate change planning, risks and opportunities.

Cybersecurity Matters

The increase in cyberattacks results in a greater threat to water, wastewater and electric utility systems and thereby the safety and security of our communities. We continue to increase our investments in information technology to monitor and address these threats and attempted cyber-attacks, and to improve our posture in addressing security vulnerabilities. See "*Item 1. Business Overview*" section of Registrant's Form 10-K for the year-ended December 31, 2022 filed with the SEC for a discussion of cybersecurity matters.

New Accounting Pronouncements

Registrant is subject to newly issued requirements as well as changes in existing requirements issued by the Financial Accounting Standards Board. There are no current accounting pronouncements that Registrant believes will significantly impact it. See Note 1 of the Unaudited Notes to Consolidated Financial Statements.

Item 3. Quantitative and Qualitative Disclosures About Market Risk

Registrant is exposed to certain market risks, including fluctuations in interest rates, commodity price risk primarily relating to changes in the market price of electricity at BVES, and other economic conditions. Market risk is the potential loss arising from adverse changes in prevailing market rates and prices.

The quantitative and qualitative disclosures about market risk are discussed in *Item 7A-Quantitative and Qualitative Disclosures About Market Risk*, contained in Registrant's Annual Report on Form 10-K for the year ended December 31, 2022 filed with the SEC.

Item 4. Controls and Procedures

(a) Evaluation of Disclosure Controls and Procedures

As required by Rule 13a-15(b) under the Securities and Exchange Act of 1934 (the "Exchange Act"), we have carried out an evaluation, under the supervision and with the participation of our management, including our Chief Executive Officer ("CEO") and our Chief Financial Officer ("CFO"), of the effectiveness, as of the end of the fiscal quarter covered by this report, of the design and operation of our "disclosure controls and procedures" as defined in Rule 13a-15(e) and 15d-15(e) promulgated by the SEC under the Exchange Act. Based upon that evaluation, the CEO and the CFO concluded that disclosure controls and procedures, as of the end of such fiscal quarter, were adequate and effective to ensure that information required to be disclosed by us in the reports that we file or submit under the Exchange Act is recorded, processed, summarized and reported within the time periods specified in the SEC's rules and forms, and that such information is accumulated and communicated to our management, including our CEO and CFO, as appropriate to allow timely decisions regarding required disclosure.

(b) Changes in Internal Controls over Financial Reporting

There has been no change in our internal control over financial reporting during the quarter ended March 31, 2023, that has materially affected or is reasonably likely to materially affect our internal control over financial reporting.

PART II

Item 1. Legal Proceedings

Registrant is subject to ordinary routine litigation incidental to its business, some of which may include claims for compensatory and punitive damages. No legal proceedings are pending, which are believed to be material. Management believes that rate recovery, proper insurance coverage and reserves are in place to insure against, among other things, property, general liability, employment, and workers' compensation claims incurred in the ordinary course of business. Insurance coverage may not cover certain claims involving punitive damages.

Item 1A. Risk Factors

There have been no significant changes in the risk factors disclosed in our 2022 Annual Report on Form 10-K filed with the SEC.

Item 2. Unregistered Sales of Equity Securities and Use of Proceeds

The shareholders of AWR have approved the material features of all equity compensation plans under which AWR issues equity securities. The following table provides information about repurchases of Common Shares by AWR during the first quarter of 2023:

Period	Total Number of Shares Purchased	Average Price Paid per Share	Total Number of Shares Purchased as Part of Publicly Announced Plans or Programs (1)	Maximum Number of Shares That May Yet Be Purchased under the Plans or Programs (1)(3)
January 1 – 31, 2023	296	\$ 93.57	_	_
February 1 – 28, 2023	227	\$ 94.17	_	_
March 1 – 31, 2023	8,857	\$ 86.05	_	_
Total	9,380 (2)	\$ 86.48		

- (1) None of the Common Shares were purchased pursuant to any publicly announced stock repurchase program.
- (2) Of these amounts, 6,204 Common Shares were acquired on the open market for employees pursuant to the 401(k) Plan. The remainder of the shares were acquired on the open market for participants in the Common Share Purchase and Dividend Reinvestment Plan.
- (3) Neither the 401(k) plan nor the Common Share Purchase and Dividend Reinvestment Plan contain a maximum number of Common Shares that may be purchased in the open market.

Item 3. Defaults Upon Senior Securities

None

Item 4. Mine Safety Disclosure

Not applicable

Item 5. Other Information

- (a) On May 8, 2023, AWR's Board of Directors approved a second quarter dividend of \$0.3975 per share on AWR's Common Shares. Dividends on the Common Shares will be paid on June 1, 2023 to shareholders of record at the close of business on May 19, 2023.
- (b) There have been no material changes during the first quarter of 2023 to the procedures by which shareholders may nominate persons to the Board of Directors of AWR.

Item 6. Exhibits

(a) The following documents are filed as Exhibits to this rep	port:
---	-------

ne monov	ving documents are fried as Exhibits to this report.
3.1	By-Laws of American States Water Company incorporated by reference to Exhibit 3.1 of Registrant's Form 10-Q, filed August 6, 2012 (File No. 1-14431)
3.2	By-laws of Golden State Water Company incorporated by reference to Exhibit 3.2 of Registrant's Form 8-K filed May 13, 2011 (File No. 1-14431)
3.3	Amended and Restated Articles of Incorporation of American States Water Company, as amended, incorporated by reference to Exhibit 3.1 of Registrant's Form 8-K filed June 19, 2013
3.4	Restated Articles of Incorporation of Golden State Water Company, as amended, incorporated herein by reference to Exhibit 3.1 of Registrant's Form 10-Q for the quarter ended September 30, 2005 (File No. 1-14431)
4.1	Indenture, dated September 1, 1993 between Golden State Water Company and The Bank of New York Mellon Trust Company, N.A., as successor trustee, as supplemented, incorporated herein by reference to Exhibit 4.01 of Golden State Water Company Form S-3 filed December 12, 2008 (File No. 333-156112)
4.2	Note Purchase Agreement dated as of October 11, 2005 between Golden State Water Company and Co-Bank, ACB incorporated by reference to Exhibit 4.1 of Registrant's Form 8-K filed October 13, 2005 (File No. 1-14431)
4.3	Description of Common Shares incorporated by reference to Exhibit 4.3 to Registrant's Form 10-K for the year ended December 31, 201
4.4	Description of Debt Securities incorporated by reference to Exhibit 4.4 to Registrant's Form 10-K for the year ended December 31,2021
10.1	Second Sublease dated October 5, 1984 between Golden State Water Company and Three Valleys Municipal Water District incorporated herein by reference to Registrant's Registration Statement on Form S-2, Registration No. 33-5151
10.2	Loan Agreement between California Pollution Control Financing Authority and Golden State Water Company, dated as of December 1, 1996 incorporated by reference to Exhibit 10.7 of Registrant's Form 10-K for the year ended December 31, 1998 (File No. 1-14431)
10.3	Water Supply Agreement dated as of June 1, 1994 between Golden State Water Company and Central Coast Water Authority incorporated herein by reference to Exhibit 10.15 of Registrant's Form 10-K with respect to the year ended December 31, 1994 (File No. 1-14431)
10.4	2003 Non-Employee Directors Stock Purchase Plan, as amended, incorporated herein by reference to Exhibit 10.4 to Registrant's Form 8 K filed on May 20, 2015 (File No. 1-14431) (2)
10.5	<u>Dividend Reinvestment and Common Share Purchase Plan incorporated herein by reference to American States Water Company Registrant's Form S-3D filed November 12, 2008 (File No. 1-14431)</u>
10.6	<u>Change in Control Agreement between American States Water Company or a subsidiary and certain executives incorporated herein by reference to Exhibit 10.6 to Registrant's Form 10-K for the year ended December 31, 2022 (2)</u>
10.7	Golden State Water Company Supplemental Executive Retirement Plan, amended and restated, incorporated herein by reference to Exhibit 10.7 to Registrant's Form 10-Q filed on May 2, 2022 (2)
10.8	Amended and Restated Credit Agreement between American States Water Company dated June 3, 2005 with Wells Fargo Bank, N.A., a Administrative Agent, as amended, incorporated by reference herein to Exhibit 10.8 to Registrant's Form 10-Q for the quarter ended March 31, 2023 (1)
10.0	Delicy Pagarding the Pagarament of Cartain Parformance Pagad Companyation Payments incorporated barein by reference to

- 10.9 Policy Regarding the Recoupment of Certain Performance-Based Compensation Payments incorporated herein by reference to Exhibit 10.3 to the Registrant's Form 8-K filed on April 2, 2014 (2)
- 10.10 Officer Relocation Policy incorporated herein by reference to Exhibit 10.5 to the Registrant's Form 8-K filed on July 31, 2009 (2)
- Form of Indemnification Agreement for directors incorporated by reference herein to Exhibit 10.35 to the Registrant's Form 10-K for the period ended December 31, 2012 (2)
- 10.12 2016 Stock Incentive Plan incorporated by reference to Exhibit 10.1 to Registrant's Form 8-K filed on May 19, 2016 (2)
- 10.13 2013 Non-Employee Directors Plan incorporated by reference herein to Exhibit 10.2 to the Registrant's Form 8-K filed on March 25, 2016 (2)

10.14	Form of Restricted Stock Award Agreement for officers with respect to time-vested restricted stock awards under the 2016 Stock Incentive Plan after December 31, 2017 incorporated by reference to Exhibit 10.1 of Form 8-K filed on November 3, 2017 (2)
10.15	Note Purchase Agreement dated July 8, 2020 incorporated by reference to Exhibit 10.1 to Registrant's Form 8-K filed July 14, 2020
10.16	Form of 2020 Performance Award Agreement incorporated by reference to Exhibit 10.1 of Registrant's Form 8-K filed January 31, 2020 (2)
10.17	Form of 2021 Performance Award Agreement incorporated by reference to Exhibit 10.1 of Registrant's Form 8-K filed February 5, 2021 (2)
10.18	Separation Agreement and General Release of Claims dated August 10, 2021 incorporated by reference to Exhibit 10.1 to Registrant's Form 8-K filed on August 13, 2021 (2)
10.19	Retirement Agreement and General Release of Claims effective January 14, 2022, incorporated by reference to Exhibit 10.1 to Registrant's Form 8-K filed on January 21, 2022 (2)
10.20	Contract for Professional Services effective January 15, 2022, incorporated by reference to Exhibit 10.2 to Registrant's Form 8-K filed on January 21, 2022 (2)
10.21	Form of 2022 Performance Award Agreement incorporated by reference to Exhibit 10.1 of Registrant's Form 8-K filed February 4, 2022 (2)
10.22	Form of 2023 Performance Award Agreement incorporated by reference to Exhibit 10.1 of Registrant's Form 8-K filed February 10, 2023 (2)
10.23	Form of Indemnification Agreement for directors and officers incorporated by reference herein to Exhibit 10.24 to the Registrant's Form 10-K for the year ended December 31, 2022 (2)
10.24	Short-Term Incentive Program incorporated by reference to Exhibit 10.1 of Registrant's Form 8-K filed on March 31, 2023 (2)
10.25	Form of 2023 Short-Term Incentive Program Award Agreement incorporated by reference to Exhibit 10.2 of Registrant's Form 8-K filed on March 31, 2023 (2)
31.1	Certification of Chief Executive Officer pursuant to Section 302 of the Sarbanes-Oxley Act of 2002 for AWR (1)
31.1.1	Certification of Chief Executive Officer pursuant to Section 302 of the Sarbanes-Oxley Act of 2002 for GSWC (1)
31.2	Certification of Chief Financial Officer pursuant to Section 302 of the Sarbanes-Oxley Act of 2002 for AWR (1)
31.2.1	Certification of Chief Financial Officer pursuant to Section 302 of the Sarbanes-Oxley Act of 2002 for GSWC (1)
32.1	Certification of Chief Executive Officer pursuant to Section 906 of the Sarbanes-Oxley Act of 2002 (3)
32.2	Certification of Chief Financial Officer pursuant to Section 906 of the Sarbanes-Oxley Act of 2002 (3)
101.INS	XBRL Instance Document - the instance document does not appear in the Interactive Data File because its XBRL tags are embedded within the Inline XBRL document.
01.SCH	XBRL Taxonomy Extension Schema (3)
01.CAL	XBRL Taxonomy Extension Calculation Linkbase (3)
01.DEF	XBRL Taxonomy Extension Definition Linkbase (3)
01.LAB	XBRL Taxonomy Extension Label Linkbase (3)
01.PRE	XBRL Taxonomy Extension Presentation Linkbase (3)
104	Cover Page Interactive Data File (formatted as Inline XBRL and contained in Exhibit 101)

Filed concurrently herewith
 Management contract or compensatory arrangement
 Furnished concurrently herewith

SIGNATURE

Pursuant to the requirements of the Securities Exchange Act of 1934, the Registrant has duly caused this report to be signed on its behalf by the undersigned thereunto duly authorized and as its principal financial officer.

AMERICAN STATES WATER COMPANY ("AWR"):

By: /s/ EVA G. TANG

Eva G. Tang

Senior Vice President - Finance, Chief Financial Officer, Corporate Secretary and Treasurer

GOLDEN STATE WATER COMPANY ("GSWC"):

By: /s/ EVA G. TANG

Eva G. Tang

Senior Vice President - Finance, Chief Financial

Officer and Secretary

Date: May 10, 2023

AMENDED AND RESTATED CREDIT AGREEMENT

Dated as of June 3, 2005

among

AMERICAN STATES WATER COMPANY,

as Borrower,

THE LENDERS NAMED HEREIN

and

WELLS FARGO BANK, NATIONAL ASSOCIATION,

as Administrative Agent and Lead Arranger

TABLE OF CONTENTS

		Page
Article 1. DEF	INITIONS AND ACCOUNTING TERMS	1
1.1	Defined Terms	1
1.2	Use of Defined Terms	16
1.3	Accounting Terms; Covenant Calculations	17
1.4	Rounding	17
1.5	Exhibits and Schedules	17
1.6	References to "Borrower and its Subsidiaries"	17
1.7	Miscellaneous Terms	17
Article 2. ADV	ANCES AND LETTERS OF CREDIT	17
2.1	Advances-General	17
2.2	Alternate Base Rate Advances	18
2.3	Eurodollar Rate Advances	18
2.4	Conversion and Continuation of Advances	18
2.5	Letters of Credit	19
2.6	Termination or Reduction of the Commitments	22
2.7	Administrative Agent's Right to Assume Funds Available for Advances	22
2.8	Swing Line	22
2.9	Adjusting Purchase Payments	23
Article 3. PAY	MENTS AND FEES	23
3.1	Principal and Interest.	23
3.2	Unused Revolving Facility Commitment Fee	24
3.3	Closing Fees; Arrangement Fee; Agency Fee etc.	24
3.4	Letter of Credit Fees	25
3.5	Increased Commitment Costs	25
3.6	Eurodollar Costs and Related Matters	25
3.7	Late Payments and Default Rate	27
3.8	Computation of Interest and Fees	27
3.9	Non-Banking Days	28
3.10	Manner and Treatment of Payments	28
3.11	Funding Sources	28
3.12	Failure to Charge Not Subsequent Waiver	28
3.13	Administrative Agent's Right to Assume Payments Will be Made	29
3.14	Fee Determination Detail	29
3.15	Survivability	29
Article 4. REP	RESENTATIONS AND WARRANTIES	29
4.1	Existence and Qualification; Power; Compliance With Laws	29
4.2	Authority; Compliance With Other Agreements and Instruments and Government Regulations	29
4.3	No Governmental Approvals Required	30
11	Subsidiaries	30

4.5	Financial Statements	30		
4.6	No Other Liabilities; No Material Adverse Changes			
4.7	Title to and Location of Property	31		
4.8	Intangible Assets	31		
4.9	Litigation	31		
4.10	Binding Obligations	31		
4.11	No Default	31		
4.12	ERISA	31		
4.13	Regulation U; Investment Company Act	32		
4.14	Disclosure	32		
4.15	Tax Liability	32		
4.16	Projections	32		
4.17	Hazardous Materials	32		
4.18	Employee Matters	32		
4.19	Fiscal Year	32		
4.20	Solvency	32		
Article 5. AFF	IRMATIVE COVENANTS (OTHER THAN INFORMATION AND REPORTING REQUIREMENTS)	32		
5.1	Payment of Taxes and Other Potential Liens	32		
5.2	Preservation of Existence	33		
5.3	Maintenance of Properties	33		
5.4	Maintenance of Insurance	33		
5.5	Compliance With Laws	33		
5.6	Inspection Rights	33		
5.7	Keeping of Records and Books of Account	33		
5.8	Compliance With Agreements	33		
5.9	Use of Proceeds	33		
5.10	Hazardous Materials Laws	33		
5.11	Minimum Debt Rating	34		
5.12	Syndication Process	34		
Article 6. NEG	ATIVE COVENANTS	34		
6.1	Drangument of Indebtedness	2.4		
6.2	Prepayment of Indebtedness Prepayment of Subordinated Obligations	34 34		
6.3	Disposition of Property	34		
6.4		34		
6.5	Mergers Hostile Tender Offers	34		
6.6	Distributions	34		
6.7	ERISA	35		
6.8	Change in Nature of Business	35		
6.9	Liens and Negative Pledges	35		
6.10	Indebtedness and Guaranty Obligations	35		
6.10	Transactions with Affiliates	35		
6.12	Total Funded Debt Ratio	35		
6.12	Interest Coverage Ratio	36		
6.13	Investments and Acquisitions	36		
6.14	Operating Leases	36		
6.16	Amendments	36		
6.17	Use of Lender's Name	36		
0.1/	OSE OF DEFINED S INGINE	30		

Article 7. INFORMATION AND REPORTING REQUIREMENTS 36 7.1 Financial and Business Information 36 7.2 Compliance Certificates 38 Article 8. CONDITIONS 38 8.1 Closing Date Advances 38 8.2 Any Advance 39 Article 9. EVENTS OF DEFAULT AND REMEDIES UPON EVENT OF DEFAULT 40 9.1 Events of Default 40 9.2 Remedies Upon Event of Default 41 Article 10. THE ADMINISTRATIVE AGENT 42 10.1 Appointment and Authorization 42 10.2 Administrative Agent and Affiliates 42 10.3 Proportionate Interest in any Collateral 42 10.4 Lenders' Credit Decisions 43 10.5 Action by Administrative Agent 43 10.6 Liability of Administrative Agent 43 10.7 Indemnification 44 10.8 Successor Administrative Agent 45 11.1 Multiclation 45 11.2 Amendments, Consents 45 11.1 Cumulative Remedies; No Waiver	6.18	Change of Fiscal Periods	36
7.2 Compliance Certificates 38 Article 8. CONDITIONS 38 8.1 Closing Date Advances 39 8.2 Any Advance 39 Article 9. EVENTS OF DEFAULT AND REMEDIES UPON EVENT OF DEFAULT 40 9.1 Events of Default 40 9.2 Remedies Upon Event of Default 41 Article 10. THE ADMINISTRATIVE AGENT 42 10.1 Appointment and Authorization 42 10.2 Administrative Agent and Affiliates 42 10.3 Proportionate Interes in any Collateral 42 10.4 Lenders' Credit Decisions 43 10.5 Action by Administrative Agent 43 10.6 Liability of Administrative Agent 43 10.8 Successor Administrative Agent 44 10.9 No Obligations of Borrower 45 Article 11. MISCELLANEOUS 45 Article 11. MISCELLANEOUS 45 11.1 Cumulative Remedies; No Waiver 45 11.2 Amendments; Consents 45 11.1 Notices 46 11.6	Article 7. INF	ORMATION AND REPORTING REQUIREMENTS	36
Article 8. CONDITIONS 38 8.1 Closing Date Advances 39 8.2 Any Advance 39 Article 9. EVENTS OF DEFAULT AND REMEDIES UPON EVENT OF DEFAULT 40 9.1 Events of Default 40 9.2 Remedies Upon Event of Default 41 Article 10. THE ADMINISTRATIVE AGENT 42 10.1 Appointment and Authorization 42 10.2 Administrative Agent and Affiliates 42 10.3 Proportionate Interest in any Collateral 42 10.4 Lenders' Credit Decisions 43 10.5 Action by Administrative Agent 43 10.6 Liability of Administrative Agent 43 10.7 Indemnification 44 10.8 Successor Administrative Agent 44 10.9 No Obligations of Borrower 45 Article 11. MISCELLANEOUS 45 Article 11. Miscret Laneous Consents 45 11.1 Cumulative Remedies; No Waiver 45 11.2 Amendments; Consents 45 11.1. Survival of Representations and Warranties 46	7.1	Financial and Business Information	36
8.1 Closing Date Advance 38 8.2 Any Advance 39 Article 9. EVENTS OF DEFAULT AND REMEDIES UPON EVENT OF DEFAULT 40 9.1 Events of Default 40 9.2 Remedies Upon Event of Default 41 Article 10. THE ADMINISTRATIVE AGENT 42 10.1 Appointment and Authorization 42 10.2 Administrative Agent and Affiliates 42 10.3 Proportionate Interest in any Collateral 42 10.4 Lenders' Credit Decisions 43 10.5 Action by Administrative Agent 43 10.6 Liability of Administrative Agent 43 10.7 Indeminification 44 10.8 Successor Administrative Agent 43 10.9 No Obligations of Borrower 45 Article 11. MISCELLANEOUS 45 11.1 Cumulative Remedies; No Waiver 45 11.2 Amendments; Consents 45 11.3 Costs and Expenses 45 11.4 Nature of Lenders' Obligations 46 11.6 Notices 46			
8.2 Any Advance 39 Article 9. EVENTS OF DEFAULT AND REMEDIES UPON EVENT OF DEFAULT 40 9.1 Events of Default 40 9.2 Remedies Upon Event of Default 41 Article 10. THE ADMINISTRATIVE AGENT 42 10.1 Appointment and Authorization 42 10.2 Administrative Agent and Affiliates 42 10.3 Proportionate Interest in any Collateral 42 10.4 Lenders' Credit Decisions 43 10.5 Action by Administrative Agent 43 10.6 Liability of Administrative Agent 43 10.7 Indemnification 44 10.9 No Obligations of Borrower 45 Article 11. MISCELLANEOUS 45 11.1 Cumulative Remedies; No Waiver 45 11.2 Amendments; Consents 45 11.3 Costs and Expenses 45 11.4 Nature of Lenders' Obligations 46 11.5 Survival of Representations and Warranties 46 11.6 Notices 46 11.7 Execution of Loan Documents	Article 8. CON	NDITIONS	38
8.2 Any Advance 39 Article 9. EVENTS OF DEFAULT AND REMEDIES UPON EVENT OF DEFAULT 40 9.1 Events of Default 40 9.2 Remedies Upon Event of Default 41 Article 10. THE ADMINISTRATIVE AGENT 42 10.1 Appointment and Authorization 42 10.2 Administrative Agent and Affiliates 42 10.3 Proportionate Interest in any Collateral 42 10.4 Lenders' Credit Decisions 43 10.5 Action by Administrative Agent 43 10.6 Liability of Administrative Agent 43 10.7 Indemnification 44 10.9 No Obligations of Borrower 45 Article 11. MISCELLANEOUS 45 11.1 Cumulative Remedies; No Waiver 45 11.2 Amendments; Consents 45 11.3 Costs and Expenses 45 11.4 Nature of Lenders' Obligations 46 11.5 Survival of Representations and Warranties 46 11.6 Notices 46 11.7 Execution of Loan Documents	8.1	Closing Date Advances	38
9.1 Events of Default 40 9.2 Remedies Upon Event of Default 41 Article 10. THE ADMINISTRATIVE AGENT 42 10.1 Appointment and Authorization 42 10.2 Administrative Agent and Affiliates 42 10.3 Proportionate Interest in any Collateral 42 10.4 Lenders' Credit Decisions 43 10.5 Action by Administrative Agent 43 10.6 Liability of Administrative Agent 43 10.7 Indemnification 44 10.8 Successor Administrative Agent 44 10.9 No Obligations of Borrower 45 Article 11. MISCELLANEOUS 45 11.1 Cumulative Remedies; No Waiver 45 11.2 Amendments; Consents 45 11.3 Costs and Expenses 45 11.4 Nature of Lenders' Obligations 46 11.5 Survival of Representations and Warranties 46 11.6 Notices 46 11.7 Execution of Loan Documents 47 11.8 Binding Effect; Assignment	8.2		39
9.2 Remedies Upon Event of Default 41 Article 10. THE ADMINISTRATIVE AGENT 42 10.1 Appointment and Authorization 42 10.2 Administrative Agent and Affiliates 42 10.3 Proportionate Interest in any Collateral 42 10.4 Lenders' Credit Decisions 43 10.5 Action by Administrative Agent 43 10.6 Liability of Administrative Agent 43 10.7 Indemnification 44 10.8 Successor Administrative Agent 44 10.9 No Obligations of Borrower 45 Article 11. MISCELLANEOUS 45 11.1 Cumulative Remedies; No Waiver 45 11.2 Amendments; Consents 45 11.3 Costs and Expenses 45 11.4 Nature of Lenders' Obligations 46 11.5 Survival of Representations and Warranties 46 11.6 Notices 46 11.7 Execution of Loan Documents 47 11.8 Binding Effect; Assignment 47 11.9 Right of Setoffs	Article 9. EVE	ENTS OF DEFAULT AND REMEDIES UPON EVENT OF DEFAULT	40
Article 10. THE ADMINISTRATIVE AGENT 10.1 Appointment and Authorization 42 10.2 Administrative Agent and Affiliates 42 10.3 Proportionate Interest in any Collateral 42 10.4 Lenders' Credit Decisions 43 10.5 Action by Administrative Agent 43 10.6 Liability of Administrative Agent 43 10.7 Indemnification 44 10.8 Successor Administrative Agent 40.9 No Obligations of Borrower 45 45 Article 11. MISCELLANEOUS 45 45 47 41.1 Cumulative Remedies; No Waiver 45 11.2 Amendments; Consents 45 11.3 Costs and Expenses 45 11.4 Nature of Lenders' Obligations 45 11.5 Survival of Representations and Warranties 46 11.5 Survival of Representations and Warranties 46 11.6 Notices 46 11.7 Execution of Loan Documents 46 11.8 Binding Effect; Assignment 47 11.9 Right of Setoff 48 11.10 Sharing of Setoffs 49 11.11 Indemnity by Borrower 49 11.12 Nonliability of the Lenders 50 11.13 No Third Parties Benefited	9.1	Events of Default	40
10.1 Appointment and Authorization 42 10.2 Administrative Agent and Affiliates 42 10.3 Proportionate Interest in any Collateral 42 10.4 Lenders' Credit Decisions 43 10.5 Action by Administrative Agent 43 10.6 Liability of Administrative Agent 43 10.7 Indemnification 44 10.8 Successor Administrative Agent 44 10.9 No Obligations of Borrower 45 Article 11. MISCELLANEOUS 45 11.1 Cumulative Remedies; No Waiver 45 11.2 Amendments; Consents 45 11.3 Costs and Expenses 45 11.4 Nature of Lenders' Obligations 46 11.5 Survival of Representations and Warranties 46 11.6 Notices 46 11.7 Execution of Loan Documents 47 11.8 Binding Effect; Assignment 47 11.9 Right of Setoff 48 11.10 Sharing of Setoffs 49 11.11 Indemnity by Borrower	9.2	Remedies Upon Event of Default	41
10.2 Administrative Agent and Affiliates 42 10.3 Proportionate Interest in any Collateral 42 10.4 Lenders' Credit Decisions 43 10.5 Action by Administrative Agent 43 10.6 Liability of Administrative Agent 43 10.7 Indemnification 44 10.8 Successor Administrative Agent 44 10.9 No Obligations of Borrower 45 Article 11. MISCELLANEOUS 45 11.1 Cumulative Remedies; No Waiver 45 11.2 Amendments; Consents 45 11.3 Costs and Expenses 45 11.4 Nature of Lenders' Obligations 46 11.5 Survival of Representations and Warranties 46 11.6 Notices 46 11.7 Execution of Loan Documents 47 11.8 Binding Effect; Assignment 47 11.9 Right of Setoff 48 11.10 Sharing of Setoffs 49 11.11 Indemnify by Borrower 49 11.12 Nonliability of the Lenders	Article 10. TH	IE ADMINISTRATIVE AGENT	42
10.3 Proportionate Interest in any Collateral 42 10.4 Lenders' Credit Decisions 43 10.5 Action by Administrative Agent 43 10.6 Liability of Administrative Agent 43 10.7 Indemnification 44 10.8 Successor Administrative Agent 44 10.9 No Obligations of Borrower 45 Article 11. MISCELLANEOUS 45 11.1 Cumulative Remedies; No Waiver 45 11.2 Amendments; Consents 45 11.3 Costs and Expenses 45 11.4 Nature of Lenders' Obligations 46 11.5 Survival of Representations and Warranties 46 11.6 Notices 46 11.7 Execution of Loan Documents 47 11.8 Binding Effect; Assignment 47 11.9 Right of Setoff 48 11.10 Sharing of Setoffs 49 11.11 Indemnity by Borrower 49 11.12 Nonliability of the Lenders 50 11.13 No Third Parties Benefited	10.1	Appointment and Authorization	42
10.4 Lenders' Credit Decisions 43 10.5 Action by Administrative Agent 43 10.6 Liability of Administrative Agent 43 10.7 Indemnification 44 10.8 Successor Administrative Agent 44 10.9 No Obligations of Borrower 45 Article 11. MISCELLANEOUS 45 11.1 Cumulative Remedies; No Waiver 45 11.2 Amendments; Consents 45 11.3 Costs and Expenses 45 11.4 Nature of Lenders' Obligations 46 11.5 Survival of Representations and Warranties 46 11.6 Notices 46 11.7 Execution of Loan Documents 47 11.8 Binding Effect; Assignment 47 11.9 Right of Setoff 48 11.10 Sharing of Setoffs 49 11.11 Indemnity by Borrower 49 11.12 Nonliability of the Lenders 50 11.13 No Third Parties Benefited 50	10.2	Administrative Agent and Affiliates	42
10.5 Action by Administrative Agent 43 10.6 Liability of Administrative Agent 43 10.7 Indemnification 44 10.8 Successor Administrative Agent 44 10.9 No Obligations of Borrower 45 Article 11. MISCELLANEOUS 45 11.1 Cumulative Remedies; No Waiver 45 11.2 Amendments; Consents 45 11.3 Costs and Expenses 45 11.4 Nature of Lenders' Obligations 46 11.5 Survival of Representations and Warranties 46 11.6 Notices 46 11.7 Execution of Loan Documents 47 11.8 Binding Effect; Assignment 47 11.9 Right of Setoff 48 11.10 Sharing of Setoffs 49 11.11 Indemnity by Borrower 49 11.12 Nonliability of the Lenders 50 11.13 No Third Parties Benefited 50	10.3	Proportionate Interest in any Collateral	42
10.6 Liability of Administrative Agent 43 10.7 Indemnification 44 10.8 Successor Administrative Agent 44 10.9 No Obligations of Borrower 45 Article 11. MISCELLANEOUS 45 11.1 Cumulative Remedies; No Waiver 45 11.2 Amendments; Consents 45 11.3 Costs and Expenses 45 11.4 Nature of Lenders' Obligations 46 11.5 Survival of Representations and Warranties 46 11.6 Notices 46 11.7 Execution of Loan Documents 47 11.8 Binding Effect; Assignment 47 11.9 Right of Setoff 48 11.11 Indemnity by Borrower 49 11.12 Nonliability of the Lenders 50 11.13 No Third Parties Benefited 50	10.4	Lenders' Credit Decisions	43
10.7 Indemnification 44 10.8 Successor Administrative Agent 44 10.9 No Obligations of Borrower 45 Article 11. MISCELLANEOUS 11.1 Cumulative Remedies; No Waiver 45 11.2 Amendments; Consents 45 11.3 Costs and Expenses 45 11.4 Nature of Lenders' Obligations 46 11.5 Survival of Representations and Warranties 46 11.6 Notices 46 11.7 Execution of Loan Documents 47 11.8 Binding Effect; Assignment 47 11.9 Right of Setoff 48 11.10 Sharing of Setoffs 49 11.11 Indemnity by Borrower 49 11.12 Nonliability of the Lenders 50 11.13 No Third Parties Benefited 50	10.5	Action by Administrative Agent	43
10.8 Successor Administrative Agent 44 10.9 No Obligations of Borrower 45 Article 11. MISCELLANEOUS 45 11.1 Cumulative Remedies; No Waiver 45 11.2 Amendments; Consents 45 11.3 Costs and Expenses 45 11.4 Nature of Lenders' Obligations 46 11.5 Survival of Representations and Warranties 46 11.6 Notices 46 11.7 Execution of Loan Documents 47 11.8 Binding Effect; Assignment 47 11.9 Right of Setoff 48 11.10 Sharing of Setoffs 49 11.11 Indemnity by Borrower 49 11.12 Nonliability of the Lenders 50 11.13 No Third Parties Benefited 50	10.6	Liability of Administrative Agent	43
10.9 No Obligations of Borrower 45 Article 11. MISCELLANEOUS 45 11.1 Cumulative Remedies; No Waiver 45 11.2 Amendments; Consents 45 11.3 Costs and Expenses 45 11.4 Nature of Lenders' Obligations 46 11.5 Survival of Representations and Warranties 46 11.6 Notices 46 11.7 Execution of Loan Documents 47 11.8 Binding Effect; Assignment 47 11.9 Right of Setoff 48 11.10 Sharing of Setoffs 49 11.11 Indemnity by Borrower 49 11.12 Nonliability of the Lenders 50 11.13 No Third Parties Benefited 50	10.7	Indemnification	44
Article 11. MISCELLANEOUS 11.1 Cumulative Remedies; No Waiver 11.2 Amendments; Consents 11.3 Costs and Expenses 11.4 Nature of Lenders' Obligations 11.5 Survival of Representations and Warranties 11.6 Notices 11.7 Execution of Loan Documents 11.8 Binding Effect; Assignment 11.9 Right of Setoff 11.10 Sharing of Setoffs 11.11 Indemnity by Borrower 11.12 Nonliability of the Lenders 11.13 No Third Parties Benefited 45 45 45 46 47 48 49 49 49 49 40 40 40 40 40 40 40 40 40 40 40 40 40	10.8	Successor Administrative Agent	44
11.1 Cumulative Remedies; No Waiver 45 11.2 Amendments; Consents 45 11.3 Costs and Expenses 45 11.4 Nature of Lenders' Obligations 46 11.5 Survival of Representations and Warranties 46 11.6 Notices 46 11.7 Execution of Loan Documents 47 11.8 Binding Effect; Assignment 47 11.9 Right of Setoff 48 11.10 Sharing of Setoffs 49 11.11 Indemnity by Borrower 49 11.12 Nonliability of the Lenders 50 11.13 No Third Parties Benefited 50	10.9	No Obligations of Borrower	45
11.2 Amendments; Consents 45 11.3 Costs and Expenses 45 11.4 Nature of Lenders' Obligations 46 11.5 Survival of Representations and Warranties 46 11.6 Notices 46 11.7 Execution of Loan Documents 47 11.8 Binding Effect; Assignment 47 11.9 Right of Setoff 48 11.10 Sharing of Setoffs 49 11.11 Indemnity by Borrower 49 11.12 Nonliability of the Lenders 50 11.13 No Third Parties Benefited 50	Article 11. MI	SCELLANEOUS	45
11.3 Costs and Expenses 45 11.4 Nature of Lenders' Obligations 46 11.5 Survival of Representations and Warranties 46 11.6 Notices 46 11.7 Execution of Loan Documents 47 11.8 Binding Effect; Assignment 47 11.9 Right of Setoff 48 11.10 Sharing of Setoffs 49 11.11 Indemnity by Borrower 49 11.12 Nonliability of the Lenders 50 11.13 No Third Parties Benefited 50	11.1	Cumulative Remedies; No Waiver	45
11.4 Nature of Lenders' Obligations 46 11.5 Survival of Representations and Warranties 46 11.6 Notices 46 11.7 Execution of Loan Documents 47 11.8 Binding Effect; Assignment 47 11.9 Right of Setoff 48 11.10 Sharing of Setoffs 49 11.11 Indemnity by Borrower 49 11.12 Nonliability of the Lenders 50 11.13 No Third Parties Benefited 50	11.2	Amendments; Consents	45
11.4 Nature of Lenders' Obligations 46 11.5 Survival of Representations and Warranties 46 11.6 Notices 46 11.7 Execution of Loan Documents 47 11.8 Binding Effect; Assignment 47 11.9 Right of Setoff 48 11.10 Sharing of Setoffs 49 11.11 Indemnity by Borrower 49 11.12 Nonliability of the Lenders 50 11.13 No Third Parties Benefited 50	11.3	Costs and Expenses	45
11.6 Notices 46 11.7 Execution of Loan Documents 47 11.8 Binding Effect; Assignment 47 11.9 Right of Setoff 48 11.10 Sharing of Setoffs 49 11.11 Indemnity by Borrower 49 11.12 Nonliability of the Lenders 50 11.13 No Third Parties Benefited 50	11.4		46
11.6 Notices 46 11.7 Execution of Loan Documents 47 11.8 Binding Effect; Assignment 47 11.9 Right of Setoff 48 11.10 Sharing of Setoffs 49 11.11 Indemnity by Borrower 49 11.12 Nonliability of the Lenders 50 11.13 No Third Parties Benefited 50	11.5	Survival of Representations and Warranties	46
11.8 Binding Effect; Assignment 47 11.9 Right of Setoff 48 11.10 Sharing of Setoffs 49 11.11 Indemnity by Borrower 49 11.12 Nonliability of the Lenders 50 11.13 No Third Parties Benefited 50	11.6	Notices	46
11.9 Right of Setoff 48 11.10 Sharing of Setoffs 49 11.11 Indemnity by Borrower 49 11.12 Nonliability of the Lenders 50 11.13 No Third Parties Benefited 50	11.7	Execution of Loan Documents	47
11.9 Right of Setoff 48 11.10 Sharing of Setoffs 49 11.11 Indemnity by Borrower 49 11.12 Nonliability of the Lenders 50 11.13 No Third Parties Benefited 50	11.8	Binding Effect; Assignment	47
11.10Sharing of Setoffs4911.11Indemnity by Borrower4911.12Nonliability of the Lenders5011.13No Third Parties Benefited50	11.9		48
11.11Indemnity by Borrower4911.12Nonliability of the Lenders5011.13No Third Parties Benefited50	11.10		49
11.12Nonliability of the Lenders5011.13No Third Parties Benefited50	11.11		49
11.13 No Third Parties Benefited 50	11.12		
	11.13		

11.15	Further Assurances	51
11.16	Integration	51
11.17	Governing Law	51
11.18	Severability of Provisions	51
11.19	Headings	51
11.20	Time of the Essence	51
11.21	Foreign Lenders and Participants	51
11.22	Waiver of Right to Trial by Jury	52
11.23	Purported Oral Amendments	52
11.24	Replacement of Lender	52
11.25	USA Patriot Act Notice	52

Exhibits

A - Assignment and Acceptance
B - Compliance Certificate
C - Letter of Credit Agreement

D - Note

E - Opinion of Counsel F - Request for Borrowing

G - Request for Continuation/Conversion

H - Request for Letter of CreditI - Adjusting Purchase Payments

Schedules

1.1	.]	Lender	Comm	itments	/Pro	Rata	Shares
-----	-----	--------	------	---------	------	------	--------

- 1.3 Material Contracts
- 4.4 Subsidiaries
- 4.7 Existing Liens, Negative Pledges and Rights of Others
- 4.8 Intangible Assets
- 4.9 Material Litigation
- 4.17 Hazardous Materials Matters
- 6.14 Existing Investments

AMENDED AND RESTATED CREDIT AGREEMENT

Dated as of June 3, 2005

This AMENDED AND RESTATED CREDIT AGREEMENT (as amended, supplemented or otherwise modified from time to time, this "<u>Agreement</u>") is entered into by and among AMERICAN STATES WATER COMPANY, a California corporation ("<u>Borrower</u>"), each lender whose name is set forth on the signature pages of this Agreement and each lender that may hereafter become a party to this Agreement pursuant to <u>Section 11.8</u> (each a "<u>Lender</u>" and collectively, "<u>Lenders</u>"), and WELLS FARGO BANK, NATIONAL ASSOCIATION, as Administrative Agent and Lead Arranger.

In consideration of the mutual covenants and agreements herein contained, the parties hereto covenant and agree as follows:

Article 1. DEFINITIONS AND ACCOUNTING TERMS

- 1.1 <u>Defined Terms</u>. As used in this Agreement, the following terms shall have the meanings set forth below:
- "Acquired Person" means (a) any Person that is the subject of an Acquisition after the Closing Date and (b) any assets constituting a discrete business or operation unit that is the subject of an Acquisition after the Closing Date.
- "Acquisition" means any transaction, or any series of related transactions, consummated after the Closing Date, by which Borrower or any of its Subsidiaries directly or indirectly (a) acquires any ongoing business or all or substantially all of the assets of any firm, partnership, joint venture, limited liability company, corporation or division thereof, whether through purchase of assets, merger or otherwise, (b) acquires in one transaction or as the most recent transaction in a series of transactions control of Securities of a Person engaged in an ongoing business representing more than 50% of the ordinary voting power for the election of directors or other governing position if the business affairs of such Person are managed by a board of directors or other governing body or (c) acquires control of more than 50% of the ownership interest in any partnership, joint venture, limited liability company, business trust or other Person that is not managed by a board of directors or other governing body.
- "Administrative Agent" means Wells Fargo when acting in its capacity as the Administrative Agent under any of the Loan Documents, or any successor Administrative Agent.
- "Administrative Agent's Office" means the Administrative Agent's address as set forth on the signature pages of this Agreement, or such other address as the Administrative Agent hereafter may designate by written notice to Borrower and the Lenders.
 - "Advance" means any advance made or to be made by any Lender to Borrower as provided in Section 2.1(a).
- "Advances for Construction" means funds advanced by any person in connection with the addition of utility plant which funds are subject to refund and, in accordance with GAAP as in effect on the date hereof, are reflected as "Other Credits" in the financial statements of Borrower and its Subsidiaries, until refunded.
- "Aerojet Write-Off" means any write-off of regulatory assets that may be required as a result of the failure of the California Public Utilities Commission to approve all, or in part, the request of SCW to amortize the remainder of the costs in the memorandum account established by SCW to record the costs incurred in connection with prosecuting the water contamination lawsuits filed by SCW against the State of California and Aerojet General Corporation.

"Affiliate" means, as to any Person, any other Person which directly or indirectly controls, or is under common control with, or is controlled by, such Person. As used in this definition, "control" (and the correlative terms, "controlled by" and "under common control with") shall mean possession, directly or indirectly, of power to direct or cause the direction of management or policies (whether through ownership of Securities or partnership or other ownership interests, by contract or otherwise); provided that, in any event, any Person that owns, directly or indirectly, 10% or more of the Securities having ordinary voting power for the election of directors or other governing body of a corporation that has more than 100 record holders of such Securities, or 10% or more of the partnership or other ownership interests of any other Person that has more than 100 record holders of such interests, will be deemed to be an Affiliate of such corporation, partnership or other Person.

"Aggregate Effective Amount" means, as of any date of determination and with respect to all Letters of Credit then outstanding, the <u>sum of (a)</u> the aggregate effective face amounts of all such Letters of Credit not then paid by Issuing Lender <u>plus</u> (b) the aggregate amounts paid by Issuing Lender under such Letters of Credit not then reimbursed to Issuing Lender by Borrower pursuant to <u>Section 2.5(d)</u> and not the subject of one or more Advances made pursuant to <u>Section 2.5(e)</u> or <u>(f)</u>.

"Alternate Base Rate" means, as of any date of determination, the rate per annum (rounded upwards, if necessary, to the next 1/100 of 1%) equal to the higher of (a) the Prime Rate in effect on such date and (b) the Federal Funds Rate in effect on such date plus ½ of 1% (50 basis points).

"Alternate Base Rate Advance" means an Advance that bears interest in relation to the Alternate Base Rate as provided in Section 3.1(b).

"Applicable Alternate Base Rate Margin" means, with respect to any Alternate Base Rate Advance, for each Pricing Period, the interest rate margin set forth below

(expressed in basis points per annum) opposite the Applicable Pricing Level for that Pricing Period:

Applicable Pricing Level	Margin
I	0
II	0
III	0
IV	0
V	0
III IV	(

"Applicable Commitment Fee Margin" means, for each Pricing Period, the margin set forth below (expressed in basis points per annum) opposite the Applicable Pricing Level for that Pricing Period:

Applicable Pricing Level			
Level	Margin		
I	7.5		
II	10.0		
III	12.5		
IV	15.0		
V	17.5		

"Applicable Eurodollar Rate Margin" means, with respect to any Eurodollar Rate Advance, for each Pricing Period, the interest rate margin set forth below (expressed in basis points per annum) opposite the Applicable Pricing Level for that Pricing Period:

Applicable Pricing	
Level	Margin
I	50.0
II	62.5
III	75.0
IV	87.5
V	100.0

[&]quot;Applicable Letter of Credit Fee Rate" means, as of any date of determination, the then effective Applicable Eurodollar Rate Margin.

"Applicable Pricing Level" means, for each Pricing Period the pricing level set forth below opposite the Debt Rating achieved by Borrower as of the first day of that Pricing Period:

ricing Level	Debt Rating		
I	Greater than or equal to A1 / A+		
II	Less than A1 / A+ but greater than or equal to A2 / A		
III	Less than A2 / A but greater than or equal to A3 / A-		
IV	Less than A3 / A- but greater than or equal to Baa2/BBB		
V	Less than Baa2/BBB		

provided that in the event that the then prevailing Debt Ratings are "split ratings", Borrower will receive the benefit of the higher Debt Rating, unless the split is a "double split rating" (in which case the pricing level applicable to the middle Debt Rating will apply) or a "triple split rating" (in which case the pricing level applicable to the Debt Rating applicable to the lowest pricing level will apply). For purposes hereof, a Debt Rating is only a "split rating" if the Debt Rating applies to a different pricing level.

"Assignment and Acceptance" means an assignment and acceptance agreement substantially in the form of Exhibit A.

"Banking Day" means any Monday, Tuesday, Wednesday, Thursday or Friday, other than a day on which banks are authorized or required to be closed in California or New York.

"Borrowing" means a borrowing consisting of simultaneous Advances of the same type.

"Capital Expenditure" means any expenditure that is treated as a capital expenditure under GAAP, including any expenditure that is required to be capitalized in accordance with GAAP that relates to an asset subject to a Capital Lease.

"<u>Capital Lease</u>" means, as to any Person, a lease of any Property by that Person as lessee that is, or should be in accordance with GAAP (<u>including</u> Financial Accounting Standards Board Statement No. 13, as amended or superseded from time to time), recorded as a "capital lease" on the balance sheet of that Person prepared in accordance with GAAP.

"Capital Lease Obligations" means all monetary obligations of a Person under any Capital Lease.

"Cash" means, when used in connection with any Person, all monetary and non-monetary items owned by that Person that are treated as cash in accordance with GAAP.

"Cash Equivalents" means, when used in connection with any Person, that Person's Investments in:

- (a) Government Securities due within one year after the date of the making of the Investment;
- (b) readily marketable direct obligations of any State of the United States of America or any political subdivision of any such State or any public agency or instrumentality thereof given on the date of such Investment a credit rating of at least Aa by Moody's or AA by S&P, in each case due within one year from the making of the Investment;
- (c) certificates of deposit issued by, bank deposits in, eurodollar deposits through, bankers' acceptances of, and repurchase agreements covering Government Securities executed by any Lender or any bank incorporated under the Laws of the United States of America, any State thereof or the District of Columbia and having on the

date of such Investment combined capital, surplus and undivided profits of at least \$250,000,000, or total assets of at least \$5,000,000,000, in each case due within one year after the date of the making of the Investment;

(d) certificates of deposit issued by, bank deposits in, eurodollar deposits through, bankers' acceptances of, and repurchase agreements covering Government Securities executed by any Lender or any branch or office located in the United States of America of a bank incorporated under the Laws of any jurisdiction outside the United States of America having on the date of such Investment combined capital, surplus and undivided profits of at least \$500,000,000, or total assets of at least \$15,000,000,000, in each case due within one year after the date of the making of the Investment;

- (e) repurchase agreements covering Government Securities executed by a broker or dealer registered under Section 15(b) of the Securities Exchange Act of 1934, as amended, having on the date of the Investment capital of at least \$50,000,000, due within 90 days after the date of the making of the Investment; provided that the maker of the Investment receives written confirmation of the transfer to it of record ownership of the Government Securities on the books of a "primary dealer" in such Government Securities or on the books of such registered broker or dealer, as soon as practicable after the making of the Investment;
- (f) readily marketable commercial paper or other debt Securities issued by corporations doing business in and incorporated under the Laws of the United States of America or any State thereof or of any corporation that is the holding company for a bank described in clause (c) or (d) above given on the date of such Investment a credit rating of at least P-1 by Moody's or A-1 by S&P, in each case due within one year after the date of the making of the Investment;
- (g) "money market preferred stock" issued by a corporation incorporated under the Laws of the United States of America or any State thereof (i) given on the date of such Investment a credit rating of at least Aa by Moody's and AA by S&P, in each case having an investment period not exceeding 50 days or (ii) to the extent that investors therein have the benefit of a standby letter of credit issued by Lender or a bank described in clauses (c) or (d) above; provided that (y) the amount of all such Investments issued by the same issuer does not exceed \$5,000,000 and (z) the aggregate amount of all such Investments does not exceed \$10,000,000;
- (h) a readily redeemable "money market mutual fund" sponsored by a bank described in clause (c) or (d) hereof, or a registered broker or dealer described in clause (e) hereof, that has and maintains an investment policy limiting its investments primarily to instruments of the types described in clauses (a) through (g) hereof and given on the date of such Investment a credit rating of at least Aa by Moody's and AA by S&P; and
- (i) corporate notes or bonds having an original term to maturity of not more than one year issued by a corporation incorporated under the Laws of the United States of America, or a participation interest therein; <u>provided</u> that (i) commercial paper issued by such corporation is given on the date of such Investment a credit rating of at least Aa by Moody's and AA by S&P, (ii) the amount of all such Investments issued by the same issuer does not exceed \$5,000,000 and (iii) the aggregate amount of all such Investments does not exceed \$10,000,000.

"Certificate" means a certificate signed by a Senior Officer or Responsible Official (as applicable) of the Person providing the certificate.

"Change in Control" means any of the following events: (a) the sale, lease, transfer or other disposition (other than by way of merger or consolidation), in one or a series of related transactions, of all or substantially all of the assets of Borrower and its Subsidiaries taken as a whole to any "person" or "group" (within the meaning of Sections 13(d) and 14(d)(2) of the Securities Exchange Act), (b) Borrower shall fail to own, directly or indirectly, 100% of the outstanding capital stock or other equity interests of any Closing Date Subsidiary, (c) any Person or two or more Persons acting in concert shall have acquired beneficial ownership, directly or indirectly, of, or shall have acquired by contract or otherwise, or shall have entered into a contract or arrangement that, upon consummation, will result in its or their acquisition of, control over, 20% or more of the capital stock or other equity interests of Borrower, (d) during any period of up to 24 consecutive months, commencing after the Closing Date, individuals who at the beginning of such 24-month period were directors of Borrower (together with any new director whose election by Borrower's board of directors or whose nomination for election by Borrower's shareholders was approved by a vote of at least two-thirds of the directors then still in office who either were directors at the beginning of such period or whose election or nomination for election was previously so approved) cease for any reason to constitute a majority of the directors of Borrower then in office or (e) any transaction or series of related transactions constituting a "change in control" or similar occurrence under documentation evidencing or governing Indebtedness of Borrower and/or any of its Subsidiaries of \$1,000,000 or more, which gives the holder(s) of such Indebtedness the right to accelerate or otherwise require payment of such Indebtedness prior to the maturity date thereof. As

used herein, "beneficial ownership" shall have the meaning provided in Rule 13d-3 of the Securities and Exchange Commission under the Securities Exchange Act.

"Closing Date" means the time and Banking Day on which the conditions set forth in <u>Section 8.1</u> are satisfied or waived. The Administrative Agent shall notify Borrower and the Closing Date Lenders of the date that is the Closing Date.

"Closing Date Lenders" means Wells Fargo, CoBank, ACB, Union Bank of California, N.A., Comerica Bank, The Northern Trust Company, and any other lender party to this Agreement as of the Closing Date.

"Closing Date Subsidiaries" means SCW, American States Utility Services, Inc., a California corporation, Chaparral City Water Company, an Arizona corporation, California Cities Water Company, Inc., a California corporation, AWR Merger Company, a California corporation, and Fort Bliss Water Services Company, a Texas corporation.

"Code" means the Internal Revenue Code of 1986, as amended or replaced and as in effect from time to time.

"Commitment" means with respect to each Lender, the commitment of such Lender to make Advances (expressed as the maximum aggregate amount of the Advances to be made by such Lender hereunder), as such commitment may be (a) reduced from time to time pursuant to Section 2.6 and (b) reduced or increased from time to time pursuant to assignments by or to such Lender pursuant to Section 11.8. The initial amount of each Lender's Commitment is set forth on Schedule 1.1 or in the Assignment and Acceptance pursuant to which such Lender shall have assumed its Commitment, as applicable. The initial aggregate amount of the Lenders' Commitments is \$85,000,000.

"Compliance Certificate" means a certificate in the form of Exhibit B, properly completed and signed by the president or chief financial officer of Borrower.

"Continuation," "Continue" and "Continued" each refers to a continuation of Eurodollar Rate Advances from one Eurodollar Period to the next Eurodollar Period pursuant to Section 2.4(c).

"Contractual Obligation" means, as to any Person, any provision of any outstanding Security issued by that Person or of any material agreement, instrument or undertaking to which that Person is a party or by which it or any of its Property is bound.

"Conversion," "Converted" each refers to a conversion of Advances of one Type into Advances of the other Type pursuant to Section 2.4(a) or 2.4(b).

"Debtor Relief Laws" means the Bankruptcy Code of the United States of America, as amended from time to time, and all other applicable liquidation, conservatorship, bankruptcy, moratorium, rearrangement, receivership, insolvency, reorganization, or similar debtor relief Laws from time to time in effect affecting the rights of creditors generally.

"Debt Ratings" means, as of each date of determination, (a) in the event such a credit rating is issued by either Moody's or S&P, the bank debt credit rating assigned to the Indebtedness evidenced by this Agreement by that credit reporting agency, or (b) if no bank debt credit rating is assigned, the most creditworthy credit rating, actual or implicit, assigned to senior unsecured Indebtedness of Borrower by that credit rating agency.

"<u>Default</u>" means any event that, with the giving of any applicable notice or passage of time specified in <u>Section 9.1</u>, or both, would be an Event of Default.

"Default Rate" means the interest rate prescribed in Section 3.7.

"<u>Designated Deposit Account</u>" means a deposit account to be maintained by Borrower with Wells Fargo or one of its Affiliates, as from time to time designated by Borrower by written notification to the Administrative Agent.

"Designated Eurodollar Market" means, with respect to any Eurodollar Rate Advance, the London Eurodollar Market.

"<u>Disqualified Stock</u>" means any capital stock, warrants, options or other rights to acquire capital stock (but excluding any debt Security which is convertible, or exchangeable, for capital stock), which, by its terms (or by the terms of any Security into which it is convertible or for which it is exchangeable), or upon the happening of any event, matures or is mandatorily redeemable, pursuant to a sinking fund obligation or otherwise, or is redeemable at the option of the holder thereof, in whole or in part, on or prior to the Maturity Date.

"<u>Disposition</u>" means the sale, transfer or other disposition (each, a "<u>Transfer</u>") in any single transaction or series of related transactions of any asset, or group of related assets, of Borrower or any Subsidiary <u>other than</u> (a) a Transfer of Cash, Cash Equivalents, Investments (<u>other than</u> Investments in a Subsidiary), Inventory or other assets sold or otherwise disposed of in the ordinary course of business of Borrower or any Subsidiary, (b) a Transfer of equipment sold or otherwise disposed of where substantially similar equipment in replacement thereof has theretofore been acquired, or thereafter within 90 days is acquired, by Borrower or any Subsidiary and (c) a Transfer of obsolete assets no longer useful in the business of Borrower or any Subsidiary whose carrying value on the books of Borrower or such Subsidiary is less than \$1,000,000 and (d) a Transfer to Borrower or a wholly-owned Subsidiary of Borrower.

"<u>Distribution</u>" means, with respect to any equity interest or Security issued by a Person, or any warrant or right to acquire any equity interest or Security of a Person, (a) the retirement, redemption, purchase, or other acquisition for value by such Person of any such equity interest or Security, (b) the declaration or (without duplication) payment by such Person of any dividend in Cash or in Property (<u>other than</u> in common stock or an equivalent equity interest of such Person) on or with respect to any such equity interest or Security, (c) any Investment by such Person in the holder of any such equity interest or Security, and (d) any other payment by such Person constituting a distribution under applicable Laws with respect to such equity interest or Security.

"Dollars" or "\$" means United States of America dollars.

"EBITDA" means, with respect to any fiscal period, the <u>sum of</u> (a) Net Income for that period, <u>plus</u> (b) any extraordinary loss reflected in such Net Income, <u>minus</u> (c) any extraordinary gain reflected in such Net Income, <u>plus</u> (d) Interest Expense of Borrower and its Subsidiaries for that period, <u>plus</u> (e) the aggregate amount of federal and state taxes on or measured by income of Borrower and its Subsidiaries for that period (whether or not payable during that period), <u>plus</u> (f) depreciation and amortization expense of Borrower and its Subsidiaries for that period, <u>plus</u> (g) (i) the actual amount of the Aerojet Write-Off taken during such period (provided that for purposes of this calculation, such amount shall not exceed \$16,000,000) and (ii) all other non-cash, extraordinary expenses of Borrower and its Subsidiaries for that period acceptable, in the case of <u>clause (ii)</u>, to the Requisite Lenders, in each case as determined in accordance with GAAP, consistently applied and, in the case of items (<u>d</u>), (<u>e</u>), (<u>f</u>), and (g) only to the extent reflected in the determination of Net Income for that period.

"Eligible Assignee" means (a) another Lender, (b) with respect to any Lender, any Affiliate of that Lender, (c) any commercial bank having total assets of \$250,000,000 or more, (d) any (i) savings bank, savings and loan association, finance company or similar financial institution or entity or (ii) insurance company engaged in the business of writing insurance which, in either case (A) has total assets of \$250,000,000 or more, (B) is engaged in the business of lending money and extending credit under credit facilities similar to those extended under this Agreement and (C) is operationally and procedurally able to meet the obligations of a Lender hereunder to the same degree as a commercial bank (as reasonably determined by the assigning Lender) and (e) any other financial institution (including a mutual fund or other fund) having total assets of \$250,000,000 or more which meets the requirements set forth in subclauses (B) and (C) of clause (d) above;

provided that each Eligible Assignee must either (x) be organized under the Laws of the United States of America, any State thereof or the District of Columbia or (y) be organized under the Laws of the Cayman Islands or any country which is a member of the Organization for Economic Cooperation and Development, or a political subdivision of such a country, and (1) act hereunder through a branch, agency or funding office located in the United States of America and (2) be exempt from withholding of tax on interest and deliver the documents related thereto pursuant to Section 11.21.

"ERISA" means the Employee Retirement Income Security Act of 1974, and any regulations issued pursuant thereto, as amended or replaced and as in effect from time to time.

"ERISA Affiliate" means, with respect to any Person, any Person (or any trade or business, whether or not incorporated) that is under common control with that Person within the meaning of Section 414 of the Code.

"Eurodollar Banking Day" means any Banking Day on which dealings in Dollar deposits are conducted by and among banks in the Designated Eurodollar Market.

"Eurodollar Base Rate" means with respect to any Eurodollar Rate Advance comprising part of the same Borrowing, the interest rate per annum (rounded upward, if necessary, to the nearest 1/100th of 1%) at which deposits in Dollars are offered by the Eurodollar Reference Lender to prime banks in the Designated Eurodollar Market at or about 10:00 a.m. local time in the Designated Eurodollar Market, two (2) Eurodollar Banking Days before the first day of the applicable Eurodollar Period in an aggregate amount approximately equal to the amount of the Advance to be made by the Eurodollar Reference Lender comprising part of such Borrowing and for a period of time comparable to the number of days in the applicable Eurodollar Period. The determination of the Eurodollar Base Rate by the Administrative Agent shall be conclusive in the absence of manifest error.

"<u>Eurodollar Lending Office</u>" means, as to each Lender, its office or branch so designated by written notice to Borrower and the Administrative Agent as its Eurodollar Lending Office. If no Eurodollar Lending Office is designated by a Lender, its Eurodollar Lending Office shall be its office at its address for purposes of notices hereunder.

"Eurodollar Market" means a regular established market located outside the United States of America by and among banks for the solicitation, offer and acceptance of Dollar deposits in such banks.

"Eurodollar Obligations" means eurocurrency liabilities, as defined in Regulation D or any comparable regulation of any Governmental Agency having jurisdiction over any Lender.

"Eurodollar Period" means, as to each Eurodollar Rate Advance comprising part of the same Borrowing, the period commencing on the date specified by Borrower pursuant to Section 2.1(b) and ending 1, 2, 3 or 6 months (or, if available to all Lenders, 9 or 12 months) thereafter, as specified by Borrower in the applicable Request for Borrowing or Request for Continuation/Conversion provided that:

- (a) The first day of any Eurodollar Period shall be a Eurodollar Banking Day;
- (b) Any Eurodollar Period that would otherwise end on a day that is not a Eurodollar Banking Day shall be extended to the immediately succeeding Eurodollar Banking Day unless such Eurodollar Banking Day falls in another calendar month, in which case such Eurodollar Period shall end on the immediately preceding Eurodollar Banking Day; and
 - (c) No Eurodollar Period for any Eurodollar Rate Advance shall extend beyond the Maturity Date.

"<u>Eurodollar Rate</u>" means, with respect to any Eurodollar Rate Advance comprising part of the same Borrowing, an interest rate per annum (rounded upward, if necessary, to the nearest 1/100th of one percent) determined pursuant to the following formula:

Eurodollar Base Rate
Rate = 1.00-Eurodollar Reserve
Percentage

"Eurodollar Rate Advance" means an Advance that bears interest in relation to the Eurodollar Rate as provided in Section 3.1(c).

"Eurodollar Reference Lender" means Wells Fargo or the Administrative Agent if Wells Fargo is no longer the Administrative Agent.

"Eurodollar Reserve Percentage" means, with respect to any Eurodollar Rate Advance comprising part of the same Borrowing, the maximum reserve percentage (expressed as a decimal, rounded upward, if necessary, to the nearest 1/100th of one percent) in effect on the date the Eurodollar Base Rate for the Borrowing of which such Eurodollar Rate Advance is a part is determined (whether or not such reserve percentage is applicable to any Lender) under regulations issued from time to time by the Federal Reserve Board for determining the maximum reserve requirement (including any emergency, supplemental or other marginal reserve requirement) with respect to eurocurrency funding (currently referred to as "eurocurrency liabilities") having a term comparable to the Eurodollar Period for such Eurodollar Rate Advance. The determination by the Administrative Agent of any applicable Eurodollar Reserve Percentage shall be conclusive in the absence of manifest error.

"Event of Default" shall have the meaning provided in Section 9.1.

"Federal Funds Rate" means, as of any date of determination, the rate set forth in the weekly statistical release designated as H.15(519), or any successor publication, published by the Federal Reserve Board (including any such successor, "H.15(519)") for such date opposite the caption "Federal Funds (Effective)". If for any relevant date such rate is not yet published in H.15(519), the rate for such date will be the rate set forth in the daily statistical release designated as the Composite 3:30 p.m. Quotations for U.S. Government Securities, or any successor publication, published by the Federal Reserve Bank of New York (including any such successor, the "Composite 3:30 p.m. Quotation") for such date under the caption "Federal Funds Effective Rate". If on any relevant date the appropriate rate for such date is not yet published in either H.15(519) or the Composite 3:30 p.m. Quotations, the rate for such date will be the arithmetic mean of the rates for the last transaction in overnight Federal funds arranged prior to 9:00 a.m. (New York City time) on that date by each of three leading brokers of Federal funds transactions in New York City selected by the Administrative Agent. For purposes of this Agreement, any change in the Alternate Base Rate due to a change in the Federal Funds Rate shall be effective as of the opening of business on the effective date of such change.

"Fiscal Quarter" means any fiscal quarter of Borrower and its Subsidiaries ending on each March 31, June 30, September 30 and December 31.

"Fiscal Year" means the fiscal year of Borrower and its Subsidiaries ending on each December 31.

"GAAP" means, as of any date of determination, accounting principles (a) set forth as generally accepted in then currently effective Opinions of the Accounting Principles Board of the American Institute of Certified Public Accountants, (b) set forth as generally accepted in then currently effective Statements of the Financial Accounting Standards Board or (c) that are then approved by such other entity as may be approved by a significant segment of the accounting profession in

the United States of America. The term "consistently applied," as used in connection therewith, means that the accounting principles applied are consistent in all material respects with those applied at prior dates or for prior periods.

"Government Securities" means readily marketable (a) direct full faith and credit obligations of the United States of America or obligations guaranteed by the full faith and credit of the United States of America and (b) obligations of an agency or instrumentality of, or corporation owned, controlled or sponsored by, the United States of America that are generally considered in the securities industry to be implicit obligations of the United States of America.

"Governmental Agency" means (a) any international, foreign, federal, state, county or municipal government, or political subdivision thereof, (b) any governmental or quasi-governmental agency, authority, board, bureau, commission, department, instrumentality or public body or (c) any court or administrative tribunal of competent jurisdiction.

"Guaranty Obligation" means, as to any Person, any (a) guarantee by that Person of Indebtedness of, or other obligation performable by, any other Person or (b) assurance given by that Person to an obligee of any other Person with respect to the performance of an obligation by, or the financial condition of, such other Person, whether direct, indirect or contingent, including any purchase or repurchase agreement covering such obligation or any collateral security therefor, any agreement to provide funds (by means of loans, capital contributions or otherwise) to such other Person, any agreement to support the solvency or level of any balance sheet item of such other Person or any "keep-well" or other arrangement of whatever nature given for the purpose of assuring or holding harmless such obligee against loss with respect to any obligation of such other Person; provided, however, that the term Guaranty Obligation shall not include endorsements of instruments for deposit or collection or similar arrangements in the ordinary course of business. The amount of any Guaranty Obligation in respect of Indebtedness shall be deemed to be an amount equal to the stated or determinable amount of the related Indebtedness (unless the Guaranty Obligation is limited by its terms to a lesser amount, in which case to the extent of such amount) or, if not stated or determinable, the maximum reasonably anticipated liability in respect thereof as determined by the Person in good faith. The amount of any other Guaranty Obligation shall be deemed to be zero unless and until the amount thereof has been (or in accordance with Financial Accounting Standards Board Statement No. 5 should be) quantified and reflected or disclosed in the consolidated financial statements (or notes thereto) of Borrower and its Subsidiaries.

"<u>Hazardous Materials</u>" means oil or petrochemical products, poly-chlorinated biphenyls, asbestos, urea formaldehyde, flammable explosives, radioactive materials, hazardous wastes, toxic substances or related materials, including any substances considered "hazardous substances," "hazardous wastes," "hazardous materials," "infectious wastes", "pollutant substances", "solid waste" or "toxic substances" under any Hazardous Materials Laws.

"Hazardous Materials Laws" means all Laws pertaining to the treatment, transportation or disposal of Hazardous Materials on or about any Real Property owned or leased by Borrower or any Subsidiary thereof, or any portion thereof, including without limitation the following: the Federal Water Pollution Control Act (33 U.S.C. § 1251, et seq.), the Federal Resource Conservation and Recovery Act of 1976 (42 U.S.C. § 6901, et seq.), the Comprehensive Environmental Response, Compensation and Liability Act of 1980, as amended (42 U.S.C. § 9601, et seq.) and the Superfund Amendments and Reauthorization Act of 1986, the Hazardous Materials Transportation Act, as amended (44 U.S.C. § 1801, et seq.), the Toxic Substances Control Act, 15 U.S.C. § 2601 et seq., the California Health and Safety Code (Section 25100, et seq.), the California Water Code and the California Administrative Code, in each case as such Laws are amended from time to time.

"Indebtedness" means, as to any Person (without duplication), (a) indebtedness of such Person for borrowed money or for the deferred purchase price of Property (excluding trade and other accounts payable in the ordinary course of business in accordance with ordinary trade terms), including any Guaranty Obligation for any such indebtedness, (b) indebtedness of such Person of the nature described in clause (a) that is non-recourse to the credit of such Person but is secured by assets of such Person, to the extent of the fair market value of such assets as determined in good faith by such Person, (c) Capital Lease Obligations of such Person, (d) indebtedness of such Person arising under bankers' acceptance facilities or under facilities for the discount of accounts receivable of such Person, (e) any direct or contingent obligations of such Person under

letters of credit issued for the account of such Person and (f) any net obligations of such Person under Interest Rate Protection Agreements. For the avoidance of doubt, (i) Advances for Construction of any Subsidiary of the Borrower in the ordinary course of business, and (ii) the purchase price obligation in connection with a Military Utility Privatization, to the extent that such obligation is recorded as a liability offset by a receivable in the same amount on the financial statements of Borrower, will not constitute Indebtedness hereunder.

"Intangible Assets" means assets that are considered intangible assets under GAAP, including customer lists, goodwill, covenants not to compete, copyrights, trade names, trademarks and patents.

"Interest Coverage Ratio" means, as of the last day of any Fiscal Quarter, the ratio of (a) EBITDA for the Rolling Period ending on that date, to (b) Interest Expense of Borrower and its Subsidiaries for such Rolling Period.

"Interest Expense" means, with respect to any Person and as of the last day of any fiscal period, the <u>sum of</u> (a) all interest, fees, charges and related expenses (in each case as such expenses are calculated according to GAAP) paid or payable (without duplication) for that fiscal period by that Person to a lender in connection with borrowed money (<u>including</u> any obligations for fees, charges and related expenses payable to the issuer of any letter of credit) or the deferred purchase price of assets that are considered "interest expense" under GAAP <u>plus</u> (b) the portion of rent paid or payable (without duplication) for that fiscal period by that Person under Capital Lease Obligations that should be treated as interest in accordance with Financial Accounting Standards Board Statement No. 13.

"Interest Rate Protection Agreement" means a written agreement between Borrower and one or more financial institutions providing for "swap", "cap", "collar" or other interest rate protection with respect to any Indebtedness.

"Investment" means, when used in connection with any Person, any investment by or of that Person, whether by means of purchase or other acquisition of stock or other Securities of any other Person or by means of a loan, advance creating a debt, capital contribution, guaranty or other debt or equity participation or interest in any other Person, including any partnership, limited liability company and joint venture interests of such Person. The amount of any Investment shall be the amount actually invested (minus any return of capital with respect to such Investment which has actually been received in Cash or has been converted into Cash), without adjustment for subsequent increases or decreases in the value of such Investment.

"Issuing Lender" means Wells Fargo, when acting in its capacity as Issuing Lender under any of the Loan Documents (including such other Persons that may act as agent for and on behalf of Wells Fargo) or any successor Issuing Lender.

"Laws" means, collectively, all international, foreign, federal, state and local statutes, treaties, rules, regulations, ordinances, codes and administrative or judicial precedents.

"Lead Arranger" means Wells Fargo Bank, National Association.

"Lender" means each Closing Date Lender and each lender that may hereafter become a party to this Agreement pursuant to Section 11.8.

"Letter of Credit" means any of the standby letters of credit issued by the Issuing Lender under the Revolving Facility pursuant to Section 2.5, either as originally issued or as the same may be supplemented, modified, amended, extended, restated or supplanted.

"Letter of Credit Agreement" means the standby letter of credit agreement executed by Borrower in connection with the Original Credit Agreement and attached hereto in the form of $\underline{\text{Exhibit C}}$, either as originally executed or as it may from time to time be supplemented, modified, amended, extended, restated or supplanted.

"<u>Letter of Credit Collateral Account</u>" means a deposit account to be maintained at Wells Fargo in the name of the Administrative Agent, for the benefit of the Lenders, which account shall be held as collateral for any and all Obligations incurred by Borrower or its Subsidiaries in connection with any Letter of Credit outstanding subsequent to the Maturity Date.

"Lien" means any mortgage, deed of trust, pledge, hypothecation, assignment for security, security interest, encumbrance, lien or charge of any kind, whether voluntarily incurred or arising by operation of Law or otherwise, affecting any Property, including any conditional sale or other title retention agreement, any lease in the nature of a security interest, and/or the filing of any financing statement (other than a precautionary financing statement with respect to a lease that is not in the nature of a security interest) under the UCC or comparable Law of any jurisdiction with respect to any Property.

"Loan Documents" means, collectively, this Agreement, the Notes, the Letter of Credit Agreement, any Request for Borrowing, any Request for Letter of Credit (and any corresponding application and/or reimbursement agreement with respect to any Letter of Credit), any Compliance Certificate, and any other agreements of any type or nature hereafter executed and delivered by Borrower or any other Party to the Administrative Agent or to any Lender in any way relating to or in furtherance of this Agreement, in each case either as originally executed or as the same may from time to time be supplemented, modified, amended, restated, extended or supplanted.

"Margin Stock" means "margin stock" as such term is defined in Regulation U.

"Material Adverse Effect" means any set of circumstances or events which (a) has had or would reasonably be expected to have any material adverse effect whatsoever upon the validity or enforceability of any Loan Document, (b) has been or would reasonably be expected to be material and adverse to the business, condition (financial or otherwise), prospects or operations of Borrower and its Subsidiaries, taken as a whole, (c) has materially impaired or would reasonably be expected to materially impair the ability of Borrower to perform the Obligations, or (d) has materially impaired or would reasonably be expected to materially impair the ability of Borrower to perform its Obligations under the Loan Documents.

"Material Contracts" means, collectively, (a) the agreements identified on Schedule 1.3 attached hereto and (b) any other agreement that would, if terminated, materially affect the business, condition (financial or otherwise), prospects or operations of Borrower and its Subsidiaries, taken as a whole.

"Maturity Date" means the earlier of (a) June 3, 2010 and (b) the termination or cancellation of the Revolving Facility (and all of the Commitments pertaining thereto) pursuant to the terms of this Agreement.

"Maximum Revolving Credit Amount" means \$85,000,000.00.

"Military Utility Privatization" means the acquisition of a water or wastewater system or systems from the U.S. Government pursuant to 10 USC §2688 in connection with a contract to provide utility services. It is understood that, in accordance with GAAP as in effect on the date hereof, the water and wastewater systems acquired are not reflected as fixed assets on the financial statements of the Borrower and the purchase price obligation is recorded as a liability offset by a receivable in the same amount, thereby having no effect on the financial position of the Borrower.

"Monthly Payment Date" means the last Banking Day of each calendar month.

"Moody's" means Moody's Investor Service, Inc. and its successors.

"Multiemployer Plan" means any employee benefit plan of the type described in Section 4001(a)(3) of ERISA to which Borrower or any of its ERISA Affiliates contributes, is obligated to contribute or has had an obligation to contribute.

"Negative Pledge" means a Contractual Obligation which contains a covenant binding on Borrower or any Subsidiary that prohibits Liens on any of its Property, other than (a) any such covenant contained in a Contractual Obligation granting or relating to a particular Lien which affects only the Property that is the subject of such Lien and (b) any such covenant that does not apply to Liens securing the Obligations.

"Net Income" means, with respect to any fiscal period, the consolidated net income of Borrower and its Subsidiaries for that period, determined in accordance with GAAP, consistently applied.

"Note" means any of the promissory notes made by Borrower to a Lender evidencing Advances under that Lender's Commitment, substantially in the form of $\underline{\text{Exhibit D}}$, either as originally executed or as the same may from time to time be supplemented, modified, amended, renewed, extended or supplanted.

"Obligations" means all present and future obligations of every kind or nature of Borrower at any time and from time to time owed to the Lenders, the Administrative Agent and/or the Issuing Lender, under any one or more of the Loan Documents, whether due or to become due, matured or unmatured, liquidated or unliquidated, or contingent or noncontingent, including obligations of performance as well as obligations of payment, and including interest that accrues after the commencement of any proceeding under any Debtor Relief Law by or against Borrower.

"Opinion of Counsel" means the favorable written legal opinion of O'Melveny & Myers LLP, special counsel to Borrower, substantially in the form of Exhibit E.

"Original Administrative Agent" means Wells Fargo Bank, National Association, in its capacity as administrative agent under the Original Credit Agreement.

"Original Credit Agreement" means the Credit Agreement, dated as of June 6, 2002 by and among Borrower, each of the Original Lenders from time to time parties thereto, and the Original Administrative Agent, as amended, supplemented or otherwise modified from time to time prior to the date hereof.

"Original Lenders" means each of the Lenders from time to time parties to the Original Credit Agreement.

"Party" means any Person other than Lenders and/or Administrative Agent, which now or hereafter is a party to any of the Loan Documents.

"PBGC" means the Pension Benefit Guaranty Corporation or any successor thereof established under ERISA.

"Pension Plan" means any "pension plan" (as such term is defined in Section 3(2) of ERISA), other than a Multiemployer Plan, which is maintained by Borrower or to which Borrower contributes or has or has had an obligation to contribute.

"<u>Permitted Acquisition</u>" means an Acquisition by Borrower or any wholly-owned Subsidiary of Borrower of all or substantially all of the assets of, or all of the capital stock or other equity interests of, an Acquired Person engaged in similar or related line(s) of business as Borrower or any of its Subsidiaries, <u>provided</u>, that:

- (a) if such Acquisition is of all of the capital stock or other equity interests of the Acquired Person, such Acquired Person is merged with and into Borrower or such Subsidiary substantially simultaneously with such party's acquisition of such capital stock or other equity interests or becomes a wholly-owned Subsidiary of Borrower or such Subsidiary;
- (b) in the case of the Acquisition of the capital stock or other equity interest of an Acquired Person, the board of directors (or comparable governing body) of such Acquired Person shall have duly approved such Acquisition;
- (c) Borrower shall have delivered a pro-forma Compliance Certificate for the most recently completed Rolling Period, demonstrating that, upon giving effect to the proposed Acquisition as of the last day of such Rolling Period, Borrower and its Subsidiaries shall be in compliance with the covenants set forth in <u>Sections 6.12</u> and <u>6.13</u>;
- (d) such Acquired Person shall have had a positive "EBITDA" for the twelve-month fiscal period immediately preceding the date of such Acquisition (with EBITDA calculated for such Acquired Person in a manner consistent with the calculation of EBITDA for Borrower and its Subsidiaries specified herein);

- (e) at the time of such Acquisition, each of the representations and warranties contained in the Loan Documents shall be true and correct in all material respects (except to the extent such representations and warranties expressly relate to an earlier date), no Default or Event of Default shall have occurred and remain in effect and after giving effect to such Acquisition, on a pro forma combined basis, (i) no Default of Event of Default would have occurred at any time during the twelve-month fiscal period immediately preceding the date of such Acquisition assuming that such Acquisition had occurred on the first day of such period and (ii) Borrower and its Subsidiaries, on a projected basis, will be in compliance with Section 6.12 and 6.13, as of each of the four Fiscal Quarters ending after the date of the Acquisition, as reflected in updated projections provided by Borrower to the Administrative Agent and the Lenders prior to the effective date of such Acquisition;
- (f) if such Acquisition involves the purchase of an interest in a partnership between Borrower (or a Subsidiary of Borrower) as a general partner and entities unaffiliated with Borrower or such Subsidiary as the other partners, such transaction shall be effected by having such equity interest acquired by a corporate holding company directly or indirectly wholly-owned by Borrower newly formed for the sole purpose of effecting such transaction;
- (g) the Indebtedness assumed or consideration paid or payable in cash in connection with such Acquisition shall not exceed the lesser of (x) \$15,000,000 and (y) the fair market value thereof; and
- (h) the Indebtedness assumed or consideration paid or payable in cash in connection with such Acquisition, when taken together with each other Permitted Acquisitions consummated since the Closing Date shall not exceed \$25,000,000 in the aggregate.

"<u>Permitted Acquisition Indebtedness</u>" means Indebtedness in existence at the time of, and assumed by Borrower or its Subsidiaries in connection with, a Permitted Acquisition provided that (a) such Indebtedness was not created in contemplation of such Permitted Acquisition and (b) the aggregate amount of all such Indebtedness does not at any time exceed \$10,000,000.

"<u>Permitted Capital Asset Indebtedness</u>" means Indebtedness of Borrower and its Subsidiaries consisting of Capital Lease Obligations, or otherwise incurred to finance the purchase or construction of capital assets (which shall be deemed to exist if the Indebtedness is incurred at or within 90 days before or after the purchase or construction of the capital asset), or to refinance any such Indebtedness; <u>provided</u> that the aggregate principal amount of such Indebtedness shall not exceed \$10,000,000 at any one time outstanding (as determined in accordance with GAAP consistently applied).

"Permitted Encumbrances" means:

- (a) Inchoate Liens incident to construction on or maintenance of Property; or Liens incident to construction on or maintenance of Property now or hereafter filed of record for which adequate reserves have been set aside (or deposits made pursuant to applicable Law) and which are being contested in good faith by appropriate proceedings and have not proceeded to judgment, <u>provided</u> that, by reason of nonpayment of the obligations secured by such Liens, no such Property is subject to an impending risk of loss or forfeiture;
- (b) Liens for taxes and assessments on Property which are not yet past due; or Liens for taxes and assessments on Property for which adequate reserves have been set aside and are being contested in good faith by appropriate proceedings and have not proceeded to judgment, <u>provided</u> that, by reason of nonpayment of the obligations secured by such Liens, no such Property is subject to an impending risk of loss or forfeiture;
- (c) defects and irregularities in title to any Property which in the aggregate do not materially impair the fair market value or use of the Property for the purposes for which it is or may reasonably be expected to be held;

- (d) easements, exceptions, reservations, or other agreements for the purpose of pipelines, conduits, cables, wire communication lines, power lines and substations, streets, trails, walkways, drainage, irrigation, water, and sewerage purposes, dikes, canals, ditches, the removal of oil, gas, coal, or other minerals, and other like purposes affecting Property which in the aggregate do not materially burden or impair the fair market value or use of such Property for the purposes for which it is or may reasonably be expected to be held;
- (e) easements, exceptions, reservations, or other agreements for the purpose of facilitating the joint or common use of Property in or adjacent to a shopping center or similar project affecting Property which in the aggregate do not materially burden or impair the fair market value or use of such Property for the purposes for which it is or may reasonably be expected to be held;
- (f) rights reserved to or vested in any Governmental Agency to control or regulate, or obligations or duties to any Governmental Agency with respect to, the use of any Property;
- (g) rights reserved to or vested in any Governmental Agency to control or regulate, or obligations or duties to any Governmental Agency with respect to, any right, power, franchise, grant, license, or permit;
- (h) present or future zoning laws and ordinances or other laws and ordinances restricting the occupancy, use, or enjoyment of Property;
- (i) statutory Liens, other than those described in clauses (a) or (b) above, arising in the ordinary course of business with respect to obligations which are not delinquent or are being contested in good faith, provided that, if delinquent, adequate reserves have been set aside with respect thereto and, by reason of nonpayment, no Property is subject to an impending risk of loss or forfeiture;
- (j) covenants, conditions, and restrictions affecting the use of Property which in the aggregate do not materially impair the fair market value or use of the Property for the purposes for which it is or may reasonably be expected to be held;
- (k) rights of tenants under leases and rental agreements covering Property entered into in the ordinary course of business of the Person owning such Property;
- (l) Liens consisting of pledges or deposits to secure obligations under workers' compensation laws or similar legislation, including Liens of judgments thereunder which are not currently dischargeable;
- (m) Liens consisting of pledges or deposits of Property to secure performance in connection with operating leases made in the ordinary course of business, <u>provided</u> the aggregate value of all such pledges and deposits (excluding the property subject to such lease) in connection with any such lease does not at any time exceed 10% of the annual fixed rentals payable under such lease;
- (n) Liens consisting of deposits of Property to secure bids made with respect to, or performance of, contracts (other than contracts creating or evidencing an extension of credit to the depositor);
- (o) Liens consisting of any right of offset, or statutory bankers' lien, on bank deposit accounts maintained in the ordinary course of business so long as such bank deposit accounts are not established or maintained for the purpose of providing such right of offset or bankers' lien;
- (p) any (i) interest or title of a lessor or sublessor under any lease not prohibited by this Agreement, but only to the extent such interest or title pertains solely to the property leased, (ii) Lien or restriction that the interest or title

of such lessor or sublessor may be subject to, or (iii) subordination of the interest of the lessee or sublessee under such lease to any Lien or restriction referred to in the preceding clause (ii), so long as the holder of such Lien or restriction agrees to recognize the rights of such lessee or sublessee under such lease;

- (q) Liens consisting of deposits of Property to secure statutory obligations of Borrower; and
- (r) Liens consisting of deposits of Property to secure (or in lieu of) surety, appeal or customs bonds.

"<u>Permitted Right of Others</u>" means a Right of Others consisting of (a) an interest (<u>other than</u> a legal or equitable co-ownership interest, an option or right to acquire a legal or equitable co-ownership interest and any interest of a ground lessor under a ground lease), that does not materially impair the fair market value or use of Property for the purposes for which it is or may reasonably be expected to be held, (b) an option or right to acquire a Lien that would be a Permitted Encumbrance or other encumbrance permitted pursuant to <u>Section 6.9</u>, (c) the subordination of a lease or sublease in favor of a financing entity and (d) a license, or similar right, of or to Intangible Assets or other similar Property granted in the ordinary course of business.

"Person" means any individual or entity, including a trustee, corporation, limited liability company, general partnership, limited partnership, joint stock company, trust, estate, unincorporated organization, business association, firm, joint venture, Governmental Agency, or other entity.

"Platform" means an electronic delivery system (which may be provided by Administrative Agent, an Affiliate of Administrative Agent or any Person that is not an Affiliate of Administrative Agent), such as IntraLinks or a substantially similar electronic system.

"Pricing Occurrence" means with respect to any change in the Debt Rating which results in a change in the Applicable Pricing Level, the date which is five (5) Banking Days after the Administrative Agent has received evidence reasonably satisfactory to it of such change.

"Pricing Period" means (a) the period commencing on the Closing Date and ending on the first Pricing Occurrence to occur thereafter and (b) each subsequent period commencing on the date of a Pricing Occurrence and ending on the next Pricing Occurrence to occur.

"Prime Rate" means the rate of interest most recently announced within Wells Fargo, at its principal office in San Francisco, California, as its "prime rate." The "prime rate" is one of several base rates used by Wells Fargo and serves as the basis upon which effective rates of interest are calculated for loans and other credits making reference thereto. The "prime rate" is evidenced by the recording thereof after its announcement in such internal publication or publications as Wells Fargo may designate. Any change in the Prime Rate shall take effect on the day the change is announced within Wells Fargo.

"Projections" means the financial projections of Borrower and its Subsidiaries heretofore and hereafter distributed by or on behalf of Borrower to the Administrative Agent.

"Property" means any interest in any kind of property or asset, whether real, personal or mixed, or tangible or intangible.

"Pro Rata Share" of any amount means, with respect to any Lender at any time, the <u>product of</u> (a) a fraction the numerator of which is the amount of such Lender's Commitment (or, if such Commitment shall have expired or been terminated, the amount of such Lender's Advances), and the denominator of which is the aggregate Commitments or

Advances, as the case may be, at such time, <u>multiplied</u> by (b) such amount. <u>Schedule 1.1</u> sets forth the Pro Rata Shares of the Closing Date Lenders as of the Closing Date.

"Quarterly Payment Date" means each March 31, June 30, September 30 and December 31.

"Real Property" means, as of any date of determination, all real property then or theretofore owned, leased or occupied by Borrower or any Subsidiary.

"Regulation D" means Regulation D, as at any time amended, of the Board of Governors of the Federal Reserve System, or any other regulation in substance substituted therefor.

"Regulation U" means Regulation U, as at any time amended, of the Board of Governors of the Federal Reserve System, or any other regulation in substance substituted therefor.

"Request for Borrowing" means a written request for a Borrowing substantially in the form of Exhibit F, signed by a Responsible Official of Borrower, and properly completed to provide all information required to be included therein.

"Request for Continuation/Conversion" means a written request to Continue or Convert a Borrowing substantially in the form of Exhibit G, signed by a Responsible Official of Borrower, and properly completed to provide all information required to be included therein.

"Request for Letter of Credit" means a written request for a Letter of Credit substantially in the form of Exhibit H, signed by a Responsible Official of Borrower, and properly completed to provide all information required to be included therein.

"Requirement of Law" means, as to any Person, the articles or certificate of incorporation and by-laws or other organizational or governing documents of such Person, and any Law, or judgment, award, decree, writ or determination of a Governmental Agency, in each case applicable to or binding upon such Person or any of its Property or to which such Person or any of its Property is subject.

"Requisite Lenders" means, (a) as of any date of determination if the Commitments are then in effect, Lenders having in the aggregate more than 50% of such aggregate Commitments, and (b) as of any date of determination if the Commitments have then been terminated and there is then any outstanding Indebtedness evidenced by the Notes, the Swing Line Documents and/or Letters of Credit, Lenders owed or holding in the aggregate more than 50% of then applicable Revolving Credit Facility Usage.

"Responsible Official" means, as to any Person, (a) when used with reference to a Person other than an individual, a corporate officer of such Person, general partner of such Person, corporate officer of a corporate general partner of such Person, corporate officer of a corporate general partner of a partnership that is a general partner of such Person, manager or managing member (in the case of a Person that is a limited liability company), or any other responsible official thereof duly acting on behalf thereof, and (b) when used with reference to a Person who is an individual, such Person. The Lenders shall be entitled to conclusively rely upon any document or certificate that is signed or executed by a Responsible Official of Borrower or any Subsidiary as having been authorized by all necessary corporate, partnership, limited liability company and/or other action on the part of Borrower or such Subsidiary.

"Revolving Credit Facility Usage" means, as of any date of determination, the sum of (a) the aggregate principal amount of funded Indebtedness then outstanding under the Notes <u>plus</u> (b) the Aggregate Effective Amount under all outstanding Letters of Credit <u>plus</u> (c) the Swing Line Outstandings.

"Revolving Facility" means the revolving credit facility provided hereunder in respect of the aggregate Commitments.

"Right of Others" means, as to any Property in which a Person has an interest, any legal or equitable right, title or other interest (other than a Lien) held by any other Person in that Property, and any option or right held by any other Person to acquire any such right, title or other interest in that Property, including any option or right to acquire a Lien; provided, however, that (a) no covenant restricting the use or disposition of Property of such Person contained in any Contractual Obligation of such Person and (b) no provision contained in a contract creating a right of payment or performance in favor of a Person that conditions, limits, restricts, diminishes, transfers or terminates such right shall be deemed to constitute a Right of Others.

"Rolling Period" means any period of four consecutive Fiscal Quarters of Borrower and its Subsidiaries.

"S&P" means Standard & Poor's Ratings Group, a division of McGraw Hill, Inc. and its successors.

"Sale and Leaseback" means, with respect to any Person, the sale of Property owned by that Person (the "Seller") to another Person (the "Buyer"), together with the substantially concurrent leasing of such Property by the Buyer to the Seller.

"SCW" means Southern California Water Company, a California corporation and wholly-owned Subsidiary of Borrower.

"Security" means any capital stock, share, voting trust certificate, bond, debenture, note or other evidence of Indebtedness, limited partnership interest, member interest, or any warrant, option or other right to purchase or acquire any of the foregoing.

"Senior Officer" means (a) the chief executive officer, (b) the president, (c) any executive vice president, (d) the chief financial officer, (e) the treasurer, or (f) any assistant treasurer, in each case of any Person.

"Solvent" means, as of any date of determination, and as to any Person, that on such date: (a) the fair valuation of the assets of such Person is greater than the fair valuation of such Person's probable liability in respect of existing debts; (b) such Person does not intend to, and does not believe that it will, incur debts beyond such Person's ability to pay as such debts mature; (c) such Person is not engaged in a business or transaction, and is not about to engage in a business or transaction, which would leave such Person with assets remaining which would constitute unreasonably small capital after giving effect to the nature of the particular business or transaction (including, in the case of Borrower, the transactions occurring on the Closing Date); and (d) such Person is generally paying its debts as they become due. For purposes of the foregoing (1) the "fair valuation" of any assets means the amount realizable within a reasonable time, either through collection or sale, of such assets at their regular market value, which is the amount obtainable by a capable and diligent businessman from an interested buyer willing to purchase such assets within a reasonable time under ordinary circumstances; and (2) the term "debts" includes any legal liability whether matured or unmatured, liquidated or unliquidated, absolute, fixed, or contingent.

"Special Eurodollar Circumstance" means the application or adoption after the Closing Date of any Law or interpretation, or any change therein or thereof, or any change in the interpretation or administration thereof by any Governmental Agency, central bank or comparable authority charged with the interpretation or administration thereof, or compliance by any Lender or its Eurodollar Lending Office with any request or directive (whether or not having the force of Law) of any such Governmental Agency, central bank or comparable authority.

"Stockholders' Equity" means, as of any date of determination and with respect to Borrower and its Subsidiaries, the consolidated stockholders' equity of Borrower and its Subsidiaries as of that date determined in accordance with GAAP; provided that there shall be excluded from Stockholders' Equity (i) any amount attributable to Disqualified Stock and (ii) any actual reduction attributable to the Aerojet Write-Off in an amount not to exceed \$16,000,000.

"Swing Line" means the revolving line of credit established by the Swing Line Lender in favor of Borrower pursuant to Section 2.8.

"Swing Line Documents" means the promissory note and any other documents executed by Borrower in favor of the Swing Line Lender in connection with the Swing Line.

"Swing Line Lender" means Wells Fargo.

"Swing Line Loans" means loans made by the Swing Line Lender to Borrower pursuant to Section 2.8.

"Swing Line Outstandings" means, as of any date of determination, the aggregate principal Indebtedness of Borrower on all Swing Line Loans then outstanding.

"Subordinated Obligations" means, as of any date of determination (without duplication), any Indebtedness of Borrower or any Subsidiary on that date which has been subordinated in right of payment to the Obligations in a manner reasonably satisfactory to the Administrative Agent and the Requisite Lenders and contains such other protective terms with respect to senior debt (such as payment blockage) as the Administrative Agent and the Requisite Lenders may reasonably require.

"Subsidiary" means, as of any date of determination and with respect to any Person, any corporation, limited liability company or partnership (whether or not, in any case, characterized as such or as a "joint venture"), whether now existing or hereafter organized or acquired:

(a) in the case of a corporation or limited liability company, of which a majority of the Securities having ordinary voting power for the election of directors or other governing body (other than Securities having such power only by reason of the happening of a contingency) are at the time beneficially owned by such Person and/or one or more Subsidiaries of such Person, or (b) in the case of a partnership, of which a majority of the partnership or other ownership interests are at the time beneficially owned by such Person and/or one or more of its Subsidiaries. Any reference to a "Subsidiary" or "Subsidiaries" shall, unless otherwise provided, be deemed to be a reference to a Subsidiary (or Subsidiaries, as the case may be) of Borrower.

"<u>Termination Date</u>" means the date on which the Advances and all other Obligations under this Agreement and the other Loan Documents (including, without limitation, Obligations with respect to any Letters of Credit which remain outstanding subsequent to the Maturity Date pursuant to <u>Section 2.5(a)</u>) are indefeasibly paid in full, in Cash, and Borrower shall have no further right to borrow any moneys or obtain other credit extensions or financial accommodations under this Agreement or any of the other Loan Documents.

"to the best knowledge of" means, when modifying a representation, warranty or other statement of any Person, that the fact or situation described therein is known by the Person (or, in the case of a Person other than a natural Person, known by a Responsible Official of that Person) making the representation, warranty or other statement, or with the exercise of reasonable due diligence under the circumstances (in accordance with the standard of what a reasonable Person in similar circumstances would have done) would have been known by the Person (or, in the case of a Person other than a natural Person, would have been known by a Responsible Official of that Person).

"Total Funded Debt" means, as of any date of determination, without duplication, the <u>sum of</u> (a) all principal Indebtedness of Borrower and its Subsidiaries for borrowed money (<u>including</u> Subordinated Obligations and any other subordinated indebtedness, debt Securities issued by Borrower and any of its Subsidiaries, the aggregate principal Indebtedness outstanding under the Notes and the Aggregate Effective Amount of all outstanding Letters of Credit) on that date <u>plus</u> (b) the aggregate amount of the principal portion of all Capital Lease Obligations of Borrower and its Subsidiaries <u>plus</u> (c) any Guaranty Obligations of Borrower and its Subsidiaries with respect to the Indebtedness of others of the types referred to in (a) and (b) above.

"<u>Total Funded Debt Ratio</u>" means, as of the last day of any Fiscal Quarter, the <u>ratio of</u> (a) Total Funded Debt as of such date <u>to</u> (b) the <u>sum of</u> (i) Total Funded Debt as of such date <u>plus</u> Stockholders' Equity as of such date.

"Type" refers to the distinction between Advances bearing interest at the Alternate Base Rate and Advances bearing interest at the Eurodollar Rate.

"<u>UCC</u>" means the Uniform Commercial Code as the same may from time to time be enacted and in effect in the State of California; provided that, in the event by reason of mandatory provisions of law, any or all of the attachment, perfection or priority of the Administrative Agent's or any Lender's Lien on any collateral is governed by the Uniform Commercial Code as enacted and in effect in a jurisdiction other than the State of California, the term "<u>UCC</u>" shall mean the Uniform Commercial Code as enacted and in effect in such other jurisdiction solely for purposes of the provisions hereof relating to such attachment, perfection or priority and for purposes of definitions related to such provisions.

"Wells Fargo" means Wells Fargo Bank, National Association.

1.2 <u>Use of Defined Terms</u>. Any defined term used in the plural shall refer to all members of the relevant class, and any defined term used in the singular shall refer to any one or more of the members of the relevant class.

- 1.3 Accounting Terms; Covenant Calculations. All accounting terms not specifically defined in this Agreement shall be construed in conformity with, and all financial data required to be submitted by this Agreement shall be prepared in conformity with, GAAP applied on a consistent basis, except as otherwise specifically prescribed herein. In the event that GAAP changes during the term of this Agreement such that the covenants contained in Sections 6.12 and 6.13 would then be calculated in a different manner or with different components, (i) Borrower and the Lenders agree to amend this Agreement in such respects as are necessary to conform those covenants as criteria for evaluating Borrower's financial condition to substantially the same criteria as were effective prior to such change in GAAP and (ii) Borrower shall be deemed to be in compliance with the covenants contained in the aforesaid Sections if and to the extent that Borrower would have been in compliance therewith under GAAP as in effect immediately prior to such change, but shall have the obligation to deliver each of the materials described in Article 7 to the Administrative Agent and the Lenders, on the dates therein specified, with financial data presented in a manner which conforms with GAAP as in effect immediately prior to such change.
- 1.4 <u>Rounding</u>. Any financial ratios required to be maintained by Borrower pursuant to this Agreement shall be calculated by dividing the appropriate component by the other component, carrying the result to one place more than the number of places by which such ratio is expressed in this Agreement and rounding the result up or down to the nearest number (with a round-up if there is no nearest number) to the number of places by which such ratio is expressed in this Agreement.
- 1.5 Exhibits and Schedules. All Exhibits and Schedules to this Agreement, either as originally existing or as the same may from time to time be supplemented, modified or amended, are incorporated herein by this reference. A matter disclosed on any Schedule shall be deemed disclosed on all Schedules.
- 1.6 <u>References to "Borrower and its Subsidiaries"</u>. Any reference herein to "Borrower and its Subsidiaries" or the like shall refer solely to Borrower during such times, if any, as Borrower shall have no Subsidiaries.
- 1.7 <u>Miscellaneous Terms</u>. The term "or" is disjunctive; the term "and" is conjunctive. The term "shall" is mandatory; the term "may" is permissive. Masculine terms also apply to females; feminine terms also apply to males. The term "including" is by way of example and not limitation.

Article 2. ADVANCES AND LETTERS OF CREDIT

2.1 Advances-General

- (a) Subject to the terms and conditions set forth in this Agreement, from time to time on any Banking Day during the period from the Closing Date to but not including the Maturity Date, each Lender severally agrees to make Advances ("Advances") to Borrower under the Revolving Facility in such amounts as Borrower may request provided that, after giving effect to such Advances, (i) Revolving Credit Facility Usage does not exceed the Maximum Revolving Credit Amount and (ii) as to each Lender, such Lender's Pro Rata Share of Revolving Credit Facility Usage does not exceed such Lender's Commitment. All Advances shall be made by the Lenders ratably according to their respective Commitments. Within the limits of each Lender's Commitment in effect from time to time and subject to the foregoing, Borrower may borrow under this Section 2.1(a), prepay Advances pursuant to Section 3.1 and reborrow under this Section 2.1(a).
- (b) Subject to the next sentence, each Borrowing shall be made pursuant to a Request for Borrowing which shall specify (i) the date of such requested Borrowing, (ii) the Type of Advances comprising such Borrowing, (iii) the amount of such Borrowing, and (iv) in the case of a Borrowing consisting of Eurodollar Rate Advances, the Eurodollar Period therefor. Unless the Administrative Agent has notified, in its sole and absolute discretion, Borrower to the contrary not less than three (3) days prior to the date of any Borrowing, a Borrowing may be requested by telephone by a Responsible Official of Borrower, in which case Borrower shall confirm such request by promptly delivering a Request for Borrowing (conforming to the preceding sentence) in person or by telecopier to the Administrative Agent. The Administrative Agent shall incur no liability whatsoever hereunder in acting upon any telephonic request for a Borrowing purportedly made by a

Responsible Official of Borrower, and Borrower hereby agrees to indemnify the Administrative Agent from any loss, cost, expense or liability as a result of so acting.

- (c) Promptly following receipt of a Request for Borrowing, the Administrative Agent shall notify each Lender by telephone or telecopier (and if by telephone, promptly confirmed by telecopier) of the date of the requested Borrowing, the Type of Advances comprising such Borrowing, the Eurodollar Period (if applicable), and the amount corresponding to that Lender's ratable share of the Borrowing. Not later than 1:00 p.m., California time, on the date specified for any Borrowing (which must be a Banking Day), each Lender shall make its ratable share of the Borrowing in immediately available funds available to the Administrative Agent at the Administrative Agent's Office. Upon satisfaction or waiver of the applicable conditions set forth in Article 8, all Advances shall be credited on that date in immediately available funds to the Designated Deposit Account.
- (d) Anything in paragraph (b) above to the contrary notwithstanding, Borrower may not (a) request Alternate Base Rate Advances for any Borrowing if the aggregate amount of such Borrowing is less than \$500,000 (and any such Borrowing exceeding such minimum amount shall be in an integral multiple of \$100,000), provided that the foregoing minimum amount shall not apply to an Alternate Base Rate Advance that causes the aggregate amount borrowed under the Revolving Facility to equal the full amount available for Advances hereunder or Advances pursuant to Section 2.5, or (b) elect Eurodollar Rate Advances for any Borrowing (i) if the aggregate amount of such Borrowing is less than \$1,000,000 (and any such Borrowing exceeding such minimum amount shall be in an integral multiple of \$500,000) or (ii) if the obligation of the Lenders to make Eurodollar Rate Advances shall then be suspended pursuant to Section 2.4, 3.5 or 3.6.
 - (e) The Advances made by each Lender under its Commitment shall be evidenced by that Lender's Note.
 - (f) A Request for Borrowing shall be irrevocable upon the Administrative Agent's first notification thereof.
- (g) The Administrative Agent, on behalf of the Lenders, is hereby authorized to make Borrowings available to Borrower upon fulfillment of the applicable conditions set forth in Article-8. Upon fulfillment of such applicable conditions, the proceeds of Borrowings shall either be credited in immediately available funds to the Designated Deposit Account or remitted directly to one or more third parties, as directed by Borrower and approved by the Administrative Agent. The proceeds of any Borrowing consisting of Eurodollar Rate Advances shall be so credited or remitted on the first day of the applicable Eurodollar Period for such Borrowings.
- 2.2 <u>Alternate Base Rate Advances</u>. Each request by Borrower for a Borrowing comprised of Alternate Base Rate Advances shall be made pursuant to a Request for Borrowing (or telephonic or other request for Borrowing referred to in the second sentence of <u>Section 2.1(b)</u>, if applicable) received by the Administrative Agent, at the Administrative Agent's Office, not later than 11:00 a.m. California time, at least one (1) Banking Day before the requested Borrowing. All Advances shall constitute Alternate Base Rate Advances unless properly designated as a Eurodollar Rate Advance pursuant to <u>Section 2.3</u> or <u>2.4</u>.

2.3 Eurodollar Rate Advances.

- (a) Each request by Borrower for a Borrowing comprised of Eurodollar Rate Advances shall be made pursuant to a Request for Borrowing (or telephonic or other request for Borrowing referred to in the second sentence of Section 2.1(b), if applicable) received by the Administrative Agent, at the Administrative Agent's Office, not later than 9:00 a.m., California time, at least three (3) Eurodollar Banking Days before the first day of the applicable Eurodollar Period.
- (b) On the date which is two (2) Eurodollar Banking Days before the first day of the applicable Eurodollar Period, the Administrative Agent shall confirm its determination of the applicable Eurodollar Rate (which

determination shall be conclusive in the absence of manifest error) and promptly shall give notice of the same to Borrower and the Lenders by telephone or telecopier (and if by telephone, promptly confirmed by telecopier).

- (c) Unless the Administrative Agent and the Requisite Lenders otherwise consent, Eurodollar Rate Advances may not be outstanding under more than five (5) separate Eurodollar Periods at any one time.
- (d) No Borrowing comprised of Eurodollar Rate Advances may be requested during the continuation of a Default or Event of Default.
- (e) Nothing contained herein shall require any Lender to fund any Eurodollar Rate Advance in the Designated Eurodollar Market.

2.4 Conversion and Continuation of Advances.

(a) Optional Conversion. Borrower may on any Banking Day, upon notice given to the Administrative Agent not later than 9:00 a.m. (California time) on the third Eurodollar Banking Day prior to the date of a proposed Conversion if the Conversion is into Eurodollar Rate Advances, or one Banking Day prior to the date of a proposed Conversion if the Conversion is into Alternate Base Rate Advances, and subject to the provisions of Sections 3.5 and 3.6, Convert all or any portion of the Advances of one Type outstanding under the Revolving Facility; provided that any Conversion of Eurodollar Rate Advances into Alternate Base Rate Advances on other than the last day of a Eurodollar Period for such Eurodollar Rate Advances shall be subject to Section 3.6(e), any Conversion of Alternate Base Rate Advances into Eurodollar Rate Advances shall be in an amount not less than \$1,000,000 or integral multiples of \$500,000 in excess thereof and no Conversion of any Advances shall result in more than five (5) separate Eurodollar Periods being outstanding under the Revolving Facility. Each such notice of Conversion shall be made pursuant to a Request for Continuation/Conversion and shall, within the restrictions specified above, specify (i) the date of such Conversion, (ii) the aggregate amount and Type of the Advances (and, in the case of Eurodollar Rate Advances, the Eurodollar Period therefor) to be Converted and (iii) if such Conversion is into Eurodollar Rate Advances, the duration of the initial Eurodollar Period for such Advances. Each request for Conversion shall be irrevocable and binding on Borrower.

(b) Certain Mandatory Conversions.

- (i) On the date on which the aggregate unpaid principal amount of Eurodollar Rate Advances comprising any Borrowing shall be reduced, by payment or prepayment or otherwise, to less than \$1,000,000 such Advances shall automatically Convert into Alternate Base Rate Advances.
- (ii) If Borrower shall fail to select the duration of any Eurodollar Period for any outstanding Eurodollar Rate Advances in accordance with the provisions contained in <u>Section 2.1(b)</u> and in <u>clause (a)</u> or <u>(c)</u> of this <u>Section 2.4</u>, each such Eurodollar Rate Advance will automatically, on the last day of the then existing Eurodollar Period therefor, Convert into an Alternate Base Rate Advance.
- (iii) Upon the occurrence and during the continuance of any Event of Default and upon notice from the Administrative Agent to Borrower at the request of the Requisite Lenders, (x) each Eurodollar Rate Advance will automatically, on the last day of the then existing Eurodollar Period therefor, Convert into an Alternate Base Rate Advance and (y) the obligation of the Lenders to make, or to Convert Advances into, or to Continue, Eurodollar Rate Advances shall be suspended.
- (c) <u>Continuations</u>. Borrower may, on any Eurodollar Banking Day, upon notice given to the Administrative Agent not later than 9:00 a.m. (California time) on the third Eurodollar Banking Day prior to the date of the

proposed Continuation and subject to the provisions of Sections 3.5 and 3.6, Continue all or any portion of the Eurodollar Rate Advances outstanding under a Facility having the same Eurodollar Period; provided that any such Continuation shall be made only on the last day of a Eurodollar Period for such Eurodollar Rate Advances, no Continuation of Eurodollar Rate Advances shall be in an amount less than \$1,000,000 and no Continuation of any Eurodollar Rate Advances shall result in more than five (5) separate Eurodollar Periods being outstanding under the Revolving Facility. Each such notice of Continuation shall be made pursuant to a Request for Continuation/Conversion and shall, within the restrictions specified above, specify (i) the date of such Continuation, (ii) the aggregate amount and category of, and the Eurodollar Period for, the Advances being Continued and (iii) the duration of the initial Eurodollar Period for the Eurodollar Rate Advances subject to such Continuation. Each notice of Continuation shall be irrevocable and binding on Borrower.

2.5 Letters of Credit.

(a) Subject to the terms and conditions hereof, at any time and from time to time from the Closing Date to but not including the Maturity Date, the Issuing Lender shall issue such Letters of Credit under the Revolving Facility for the benefit of Borrower and/or its whollyowned Subsidiaries as Borrower may request by a Request for Letter of Credit; provided that giving effect to all such Letters of Credit, (i) Revolving Credit Facility Usage does not exceed the Maximum Revolving Credit Amount and (ii) the Aggregate Effective Amount under all outstanding Letters of Credit shall not exceed \$20,000,000. Each Letter of Credit shall be in a form reasonably acceptable to the Issuing Lender. Unless the Issuing Lender and the Requisite Lenders otherwise consent, the term of any Letter of Credit shall not exceed three years. Unless all the Lenders otherwise consent in a writing delivered to the Administrative Agent, the term of any Letter of Credit shall not extend beyond the date that is one year after the Maturity Date; provided, however, that no Letter of Credit shall extend beyond the Maturity Date unless, on or prior to the Maturity Date, Borrower shall have (a) deposited into the Letter of Credit Collateral Account an amount equal to one hundred five percent (105%) of the Aggregate Effective Amount of all Letters of Credit outstanding on the Maturity Date and (b) executed and delivered to the Administrative Agent such security agreements or other similar agreement as shall be reasonably required to provide the Administrative Agent, for the benefit of the Lenders, with a first priority perfected security interest in such Letter of Credit Collateral Account, Notwithstanding such deposit into the Letter of Credit Collateral Account, in the event that any Letter of Credit remains outstanding subsequent to the Maturity Date, (a) the Termination Date shall be extended until such time as all Obligations in connection with such Letters of Credit shall have been fully performed and paid in full and (b) each of the Lenders agrees to continue to comply with the provisions of this Section 2.5 until the Termination Date, as so extended. A Request for Letter of Credit shall be irrevocable absent the consent of the Issuing Lender.

(b) Each Request for Letter of Credit shall be submitted to the Issuing Lender, with a copy to the Administrative Agent, at least three (3) Banking Days prior to the date upon which the related Letter of Credit is proposed to be issued. The Administrative Agent shall promptly notify the Issuing Lender whether such request, and the issuance of a Letter of Credit pursuant thereto, conforms to the requirements of this Agreement. Upon issuance of a Letter of Credit, the Issuing Lender shall promptly notify the Administrative Agent of the amount and terms thereof. Unless the Issuing Lender has notified, in its sole and absolute discretion, Borrower to the contrary not less than three (3) days prior to the date of any Request for Letter of Credit, a Request for Letter of Credit may be delivered to the Issuing Lender by facsimile by a Responsible Official of Borrower, in which case Borrower shall confirm such request by promptly delivering a Request for Letter of Credit (conforming to the preceding sentence) in person to the Issuing Lender. The Issuing Lender shall incur no liability whatsoever hereunder in acting upon any Request for Letter of Credit received by facsimile purportedly made by a Responsible Official of Borrower, and Borrower hereby agrees to indemnify the Issuing Lender from any loss, cost, expense or liability as a result of so acting.

(c) Upon issuance of a Letter of Credit, each Lender shall be deemed to have purchased a pro rata participation in such Letter of Credit from the Issuing Lender in proportion to that Lender's Pro Rata Share of the Revolving Facility. Without limiting the scope and nature of each Lender's participation in any Letter of Credit, to the extent that the Issuing Lender has not been reimbursed by Borrower for any payment required to be made by the Issuing Lender under any

Letter of Credit, each Lender shall, pro rata according to its Pro Rata Share of the Revolving Facility, reimburse the Issuing Lender through the Administrative Agent promptly upon demand for the amount of such payment. The obligation of each Lender to so reimburse the Issuing Lender shall be absolute and unconditional and shall not be affected by the occurrence of an Event of Default or any other occurrence or event. Any such reimbursement shall not relieve or otherwise impair the obligation of Borrower to reimburse the Issuing Lender for the amount of any payment made by the Issuing Lender under any Letter of Credit together with interest as hereinafter provided.

- (d) Borrower agrees to pay to the Issuing Lender through the Administrative Agent an amount equal to any payment made by the Issuing Lender with respect to each Letter of Credit within one (1) Banking Day after demand made by the Issuing Lender therefor, together with interest on such amount from the date of any payment made by the Issuing Lender at the rate applicable to Alternate Base Rate Advances under the Revolving Facility for the period commencing on the date of any such payment and continuing through the first Banking Day following such demand and thereafter at the Default Rate. The principal amount of any such payment shall be used to reimburse the Issuing Lender for the payment made by it under the Letter of Credit. Each Lender that has reimbursed the Issuing Lender pursuant to Section 2.5(c) for its Pro Rata Share of any payment made by the Issuing Lender under a Letter of Credit shall thereupon acquire a pro rata participation, to the extent of such reimbursement, in the claim of the Issuing Lender against Borrower under this Section 2.5(d) and shall share, in accordance with that pro rata participation, in any payment made by Borrower with respect to such claim. Upon receipt of any such reimbursement from Borrower, the Issuing Lender shall pay to the Administrative Agent, for the ratable benefit of those Lenders that had reimbursed the Issuing Lender pursuant to Section 2.5(c) for their respective Pro Rata Shares of any payment made by the Issuing Lender under a Letter of Credit to which such reimbursement applies, the amount of such reimbursement.
- (e) Borrower may, pursuant to a Request for Borrowing, request that Advances be made pursuant to <u>Section 2.1(a)</u> to provide funds for the payment required by <u>Section 2.5(d)</u>. The proceeds of such Advances shall be paid directly to the Issuing Lender to reimburse it for the payment made by it under the Letter of Credit.
- (f) If Borrower fails to make the payment required by Section 2.5(d) within the time period therein set forth, in lieu of the reimbursement to the Issuing Lender under Section 2.5(c) the Issuing Lender may (but is not required to), without notice to or the consent of Borrower, instruct the Administrative Agent to cause Advances to be made by the Lenders under the Revolving Facility in an aggregate amount equal to the amount paid by the Issuing Lender with respect to that Letter of Credit and, for this purpose, the conditions precedent set forth in Article 8 shall not apply. The proceeds of such Advances shall be paid to the Issuing Lender to reimburse it for the payment made by it under the Letter of Credit.
- (g) The issuance of any supplement, modification, amendment, renewal, or extension to or of any Letter of Credit shall be treated in all respects the same as the issuance of a new Letter of Credit.
- (h) The obligation of Borrower to pay to the Issuing Lender the amount of any payment made by the Issuing Lender under any Letter of Credit shall be absolute, unconditional, and irrevocable, subject only to performance by the Issuing Lender of its obligations to Borrower under Section 5108 of the UCC. Without limiting the foregoing, Borrower's obligations shall not be affected by any of the following circumstances:
 - (i) any lack of validity or enforceability of the Letter of Credit, this Agreement, or any other agreement or instrument relating thereto;
 - (ii) any amendment or waiver of or any consent to departure from the Letter of Credit, this Agreement, or any other agreement or instrument relating thereto, with the written consent of Borrower executed by a Responsible Official of Borrower;

- (iii) the existence of any claim, setoff, defense, or other rights that Borrower may have at any time against the Issuing Lender, the Administrative Agent or any Lender, any beneficiary of the Letter of Credit (or any Persons for whom any such beneficiary may be acting) or any other Person, whether in connection with the Letter of Credit, this Agreement, or any other agreement or instrument relating thereto, or any unrelated transactions;
- (iv) any demand, statement, or any other document presented under the Letter of Credit proving to be forged, fraudulent, invalid, or insufficient in any respect or any statement therein being untrue or inaccurate in any respect whatsoever so long as any such document reasonably appeared to comply with the terms of the Letter of Credit;
- (v) payment by the Issuing Lender in good faith under the Letter of Credit against presentation of a draft or any accompanying document which does not strictly comply with the terms of the Letter of Credit;
- (vi) the existence, character, quality, quantity, condition, packing, value or delivery of any Property purported to be represented by documents presented in connection with any Letter of Credit or for any difference between any such Property and the character, quality, quantity, condition, or value of such Property as described in such documents;
- (vii) the time, place, manner, order or contents of shipments or deliveries of Property as described in documents presented in connection with any Letter of Credit or the existence, nature and extent of any insurance relative thereto;
 - (viii) the solvency or financial responsibility of any party issuing any documents in connection with a Letter of Credit;
 - (ix) any failure or delay in notice of shipments or arrival of any Property;
- (x) any error in the transmission of any message relating to a Letter of Credit not caused by the Issuing Lender, or any delay or interruption in any such message;
 - (xi) any error, neglect or default of any correspondent of the Issuing Lender in connection with a Letter of Credit;
- (xii) any consequence arising from acts of God, war, insurrection, civil unrest, disturbances, labor disputes, emergency conditions or other causes beyond the control of the Issuing Lender; and
- (xiii) so long as the Issuing Lender in good faith determines that the contract or document appears to comply with the terms of the Letter of Credit, the form, accuracy, genuineness or legal effect of any contract or document referred to in any document submitted to the Issuing Lender in connection with a Letter of Credit.

Notwithstanding anything to the contrary contained in this <u>Section 2.5(h)</u>, Borrower shall retain any and all rights it may have against the Issuing Lender for any liability arising solely out of the gross negligence or willful misconduct of the Issuing Lender, as determined by a final judgment of a court of competent jurisdiction.

(i) The Issuing Lender shall be entitled to the protection accorded to the Administrative Agent pursuant to <u>Section 10.6</u> (subject to the standards set forth therein), <u>mutatis mutandis</u>.

(j) The Uniform Customs and Practice for Documentary Credits, as published in its most current version by the International Chamber of Commerce, shall be deemed a part of this Section and shall apply to all Letters of Credit to the extent not inconsistent with applicable Law.

2.6 Termination or Reduction of the Commitments

- (a) Optional. Borrower may at any time or from time to time, upon not less than three (3) Banking Days' notice to the Administrative Agent, terminate in whole or reduce in part the Commitments, provided that each partial reduction of the Commitments shall be in an aggregate amount of \$1,000,000 or an integral multiple of \$500,000 in excess thereof.
 - (b) Mandatory. The Commitments shall be automatically and permanently reduced to zero on the Maturity Date.
- (c) <u>Reduction Pro Rata; No Reinstatements</u>. Each reduction of the Commitments shall be applied to the respective Commitments of the Lenders according to their respective Pro Rata Shares. Commitments once terminated or reduced may not be reinstated.
- 2.7 Administrative Agent's Right to Assume Funds Available for Advances. Unless the Administrative Agent shall have been notified by any relevant Lender no later than 10:00 a.m., California time, on the Banking Day of the proposed funding by the Administrative Agent of any Borrowing that such Lender does not intend to make available to the Administrative Agent such Lender's portion of the total amount of such Borrowing, the Administrative Agent may assume that such Lender has made such amount available to the Administrative Agent on the date of the Borrowing and the Administrative Agent may, in reliance upon such assumption, make available to Borrower a corresponding amount. If the Administrative Agent has made funds available to Borrower based on such assumption and such corresponding amount is not in fact made available to the Administrative Agent by such Lender, the Administrative Agent shall be entitled to recover such corresponding amount on demand from such Lender. If such Lender does not pay such corresponding amount forthwith upon the Administrative Agent's demand therefor, the Administrative Agent promptly shall notify Borrower and Borrower shall pay such corresponding amount to the Administrative Agent. The Administrative Agent also shall be entitled to recover from such Lender interest on such corresponding amount in respect of each day from the date such corresponding amount was made available by the Administrative Agent to Borrower to the date such corresponding amount is recovered by the Administrative Agent, at a rate per annum equal to the daily Federal Funds Rate. Nothing herein shall be deemed to relieve any Lender from its obligation to fulfill its Commitments or to prejudice any rights which the Administrative Agent or Borrower may have against any Lender as a result of any default by such Lender hereunder.

2.8 Swing Line.

(a) The Swing Line Lender shall from time to time from the Closing Date through the day prior to the Maturity Date make Swing Line Loans to Borrower in such amounts as Borrower may request, provided that (a) after giving effect to such Swing Line Loan, Revolving Credit Facility Usage does not exceed the Maximum Revolving Credit Amount, (b) after giving effect to such Swing Line Loan, the Swing Line Outstandings do not exceed \$10,000,000 and/or, (c) without the consent of all of the Lenders, no Swing Line Loan may be made during the continuation of an Event of Default if written notice of such Event of Default shall have been provided to Swing Line Lender by the Administrative Agent or a Lender sufficiently in advance of the making of such Swing Line Loan. Borrower may borrow, repay and reborrow under this Section. Borrowings under the Swing Line may be made in amounts which are integral multiples of \$100,000 (or the remaining availability under the Swing Line) upon telephonic request by a Responsible Official of Borrower made to the Administrative Agent not later than 2:00 p.m., California time, on the Banking Day of the requested borrowing (which telephonic request shall be promptly confirmed in writing by telecopier or electronic mail). Promptly after receipt of such a request for borrowing, the Administrative Agent shall provide telephonic verification to the Swing Line Lender that, after giving effect to such request, availability for Loans will exist under Section 2.1(a) (and such verification shall be promptly confirmed in writing by telecopier or electronic mail). Each repayment of a Swing Line Loan shall be in an amount which is an integral multiple of \$100,000 (or the Swing Line Outstandings). Borrower shall notify the Swing Line Lender of its intention to make a repayment of a Swing Line Loan not later than 1:00 p.m. California time on the date of repayment. If Borrower instructs the Swing Line Lender to debit its demand deposit account at the Swing Line Lender in the amount of any

payment with respect to a Swing Line Loan, or the Swing Line Lender otherwise receives repayment, after 3:00 p.m., California time, on a Banking Day, such payment shall be deemed received on the next Banking Day. The Swing Line Lender shall promptly notify the Administrative Agent of the Swing Loan Outstandings each time there is a change therein.

- (b) Swing Line Loans shall bear interest at a fluctuating rate per annum equal to the Alternate Base Rate <u>plus</u> the Applicable Alternate Base Rate Margin. Interest shall be payable on such dates, not more frequent than monthly, as may be specified by the Swing Line Lender and in any event on the Maturity Date. The Swing Line Lender shall be responsible for invoicing Borrower for such interest. The interest payable on Swing Line Loans is solely for the account of the Swing Line Lender (subject to <u>clause (d)</u> below).
- (c) Subject to <u>subsection (e)</u> below, the principal amount of all Swing Line Loans shall be due and payable on the earlier of (i) the maturity date agreed to by the Swing Line Lender and Borrower with respect to such loan (which maturity date shall not be a date more than ten (10) consecutive Banking Days from the date of advance thereof) or (ii) the Maturity Date.
- (d) Upon the making of a Swing Line Loan, each Lender shall be deemed to have purchased from the Swing Line Lender a participation therein in an amount equal to that Lender's Pro Rata Share of the Revolving Facility times the amount of the Swing Line Loan. Upon demand made by the Swing Line Lender, each Lender shall, according to its Pro Rata Share of the Revolving Facility, promptly provide to the Swing Line Lender its purchase price therefor in an amount equal to its participation therein. The obligation of each Lender to so provide its purchase price to the Swing Line Lender shall be absolute and unconditional (except only demand made by the Swing Line Lender) and shall not be affected by the occurrence of a Default or Event of Default; provided that no Lender shall be obligated to purchase its Pro Rata Share of (i) Swing Line Loans to the extent that, after giving effect to such Swing Line Loan, Revolving Credit Facility Usage exceeds the Maximum Revolving Credit Amount, (ii) Swing Line Loans to the extent that, after giving effect to such Swing Line Loan, Swing Line Outstandings exceed \$5,000,000 and (iii) any Swing Line Loan made (absent the consent of all of the Lenders) during the continuation of an Event of Default if written notice of such Event of Default shall have been provided to Swing Line Lender by the Administrative Agent or a Lender sufficiently in advance of the making of such Swing Line Loan. Each Lender that has provided to the Swing Line Lender the purchase price due for its participation in Swing Line Loans shall thereupon acquire a pro rata participation, to the extent of such payment, in the claim of the Swing Line Lender against Borrower for principal and interest and shall share, in accordance with that pro rata participation, in any principal payment made by Borrower with respect to such claim and in any interest payment made by Borrower (but only with respect to periods subsequent to the date such Lender paid the Swing Line Lender its purchase price) with respect to such claim.
- (e) In the event that any Swing Line Loan remains outstanding for ten (10) consecutive Banking Days, then on the next Banking Day (unless Borrower has made other arrangements acceptable to the Swing Line Lender to repay such Swing Line Loan, in full), Borrower shall request a Loan pursuant to Section 2.1(a) sufficient to repay the aggregate principal amount of such Swing Line Loan together with any and all accrued and unpaid interest with respect thereto. In addition, the Swing Line Lender may, at any time, in its sole discretion, by written notice to Borrower and the Lenders, demand payment of the Swing Line Loans by way of a Advance in the full amount or any portion of the Swing Line Outstandings. In each case, the Administrative Agent shall automatically provide the responsive Advances made by each Lender to the Swing Line Lender (which the Swing Line Lender shall then apply to the Swing Line Outstandings). In the event that Borrower fails to request a Loan within the time specified by Section 2.2 on any such date, the Administrative Agent may, but is not required to, without notice to or the consent of Borrower, cause Alternate Base Rate Advances to be made by the Lenders under the Revolving Facility in amounts which are sufficient to reduce the Swing Line Outstandings as required above. The proceeds of such Advances shall be paid directly to the Swing Line Lender for application to the Swing Line Outstandings.
- 2.9 <u>Adjusting Purchase Payments</u>. Principal amounts outstanding under the Commitment (as defined in the Original Credit Agreement) on the Closing Date (the "Carryover Principal Balance"), shall remain outstanding hereunder. Concurrently with the effectiveness of this Agreement, the Lenders agree to purchase and sell undivided interests in the Carryover

Principal Balance by making or receiving adjusting purchase payments as specified in <u>Exhibit I</u> hereto (the "Adjusting Purchase Payment(s)") so that the Carryover Principal Balance will be properly allocated and owing to the Lenders under the Notes in accordance with the Pro Rata Shares specified in <u>Schedule 1.1</u> hereto. Each Lender making an Adjusting Purchase Payment shall deliver it to the Agent and the Agent shall forward such Adjusting Purchase Payments to the Lenders entitled thereto promptly after receipt in accordance with the allocations specified in <u>Exhibit I</u>. On the Closing Date, in addition to any other Advances that may be made, each Lender shall be deemed as having an Advance outstanding in the amount of its Pro Rata Share of the Carryover Principal Balance. As of the Effective Date, the Lenders shall hold participations in all issued and outstanding Letters of Credit in accordance with their Pro Rata Shares. As of the Closing Date, Revolving Credit Facility Usage is \$60,181,000 which amount consists of (i) \$49,000,000 of outstanding Advances under the Revolving Facility, (ii) issued and outstanding Letters of Credit with an Aggregate Effective Amount of \$11,181,000.00 and (iii) no outstanding Swing Line Loans.

Article 3. PAYMENTS AND FEES

3.1 Principal and Interest.

- (a) Interest shall be payable on the outstanding daily unpaid principal amount of each Advance from the date thereof until payment in full is made and shall accrue and be payable at the rates set forth or provided for herein before and after Default, before and after maturity, before and after judgment, and before and after the commencement of any proceeding under any Debtor Relief Law, with interest on overdue interest at the Default Rate to the fullest extent permitted by applicable Laws.
- (b) Interest accrued on each Alternate Base Rate Advance shall be due and payable on each Monthly Payment Date. Except as otherwise provided in Section 3.7, the unpaid principal amount of any Alternate Base Rate Advance shall bear interest at a fluctuating rate per annum equal to the Alternate Base Rate plus the Applicable Alternate Base Rate Margin. Each change in the interest rate under this Section 3.1(b) due to a change in the Alternate Base Rate shall take effect simultaneously with the corresponding change in the Alternate Base Rate.
- (c) Interest accrued on each Eurodollar Rate Advance which is for a term of three months or less shall be due and payable on the last day of the related Eurodollar Period. Interest accrued on each other Eurodollar Rate Advance shall be due and payable on the date which is three months after the date such Eurodollar Rate Advance was made (and, in the event that all of the Lenders have approved a Eurodollar Period of longer than six months, every three months thereafter through the last day of the Eurodollar Period) and on the last day of the related Eurodollar Period. Except as otherwise provided in Section 3.7, the unpaid principal amount of any Eurodollar Rate Advance shall bear interest at a rate per annum equal to the Eurodollar Rate for that Eurodollar Rate Advance plus the Applicable Eurodollar Rate Margin.
 - (d) If not sooner paid, the principal Indebtedness evidenced by the Notes shall be payable as follows:
 - (i) the amount, if any, by which the principal Indebtedness evidenced by the Notes at any time exceeds the Maximum Revolving Credit Amount shall be payable immediately; and
 - (ii) the principal Indebtedness evidenced by the Notes shall in any event be payable on the Maturity Date.
- (e) The principal Indebtedness evidenced by the Notes may, at any time and from time to time, voluntarily be paid or prepaid in whole or in part without premium or penalty, except that with respect to any voluntary prepayment under this subsection, (i) any partial prepayment shall be not less than \$1,000,000 and shall be an integral multiple of \$500,000, except as provided in Section 2.8(a), (ii) the Administrative Agent shall have received written notice of any prepayment by 9:00 a.m. California time on the date that is (x) in the case of a Eurodollar Rate Advance three (3) Banking Days before the date of prepayment unless the prepayment is of a Eurodollar Rate Advance to be made at

the end of its applicable Eurodollar Period and (y) in the case of an Alternate Base Rate Advance or a prepayment of a Eurodollar Rate Advance made at the end of its applicable Eurodollar Period, one (1) Banking Day before the date of prepayment, which notice shall identify the date and amount of the prepayment and the Advance(s) being prepaid, (iii) each prepayment of principal on any Eurodollar Rate Advance shall be accompanied by payment of interest accrued to the date of payment on the amount of principal paid, and (iv) any payment or prepayment of all or any part of any Eurodollar Rate Advance on a day other than the last day of the applicable Eurodollar Period shall be subject to Section 3.6(e).

3.2 <u>Unused Revolving Facility Commitment Fee</u>. From the Closing Date to but not including the Maturity Date, Borrower shall pay to the Administrative Agent, for the ratable accounts of the applicable Lenders in accordance with their respective Pro Rata Shares, a commitment fee equal to the Applicable Commitment Fee Margin <u>times</u> the average daily amount by which the Maximum Revolving Credit Amount exceeds the <u>sum of</u> (a) the aggregate principal amount of funded Indebtedness then outstanding under the Notes <u>plus</u> (b) the Aggregate Effective Amount under all outstanding Letters of Credit. The commitment fee shall be payable quarterly in arrears on each Quarterly Payment Date.

3.3 Closing Fees; Arrangement Fee; Agency Fee etc.

- (a) On the Closing Date, Borrower shall pay to the Closing Date Lenders, through the Administrative Agent, the closing fees in the amount heretofore agreed upon by letter agreement among Borrower and each Closing Date Lender. All such fees shall be fully earned when paid and shall be non-refundable.
- (b) On the date of the execution hereof, Borrower shall pay to the Administrative Agent, for the sole account of the Lead Arranger, an arrangement fee in the amount heretofore agreed upon by letter agreement between Borrower and the Lead Arranger. Such arrangement fee is for the services of the Lead Arranger in arranging the credit facility under this Agreement, is fully earned as of the date hereof and is nonrefundable.
- (c) Borrower shall pay to the Administrative Agent an annual agency fee in such amounts and at such times as heretofore agreed upon by letter agreement between Borrower and the Administrative Agent. The agency fee is for the services to be performed by the Administrative Agent in acting as Administrative Agent and is fully earned on the date paid. The agency fee paid to the Administrative Agent is solely for its own account and is nonrefundable.

- 3.4 Letter of Credit Fees. With respect to each Letter of Credit, Borrower shall pay the following fees:
- (a) concurrently with the issuance of each Letter of Credit and on each Quarterly Payment Date thereafter so long as such Letter of Credit shall remain outstanding, to the Administrative Agent for the ratable accounts of the Lenders in accordance with their respective Pro Rata Shares, a standby letter of credit fee in an amount equal to the product of the then Applicable Letter of Credit Fee Rate times the then outstanding undrawn amount of such Letter of Credit, for the period commencing on such payment date and ending on the next succeeding Quarterly Payment Date or for the remaining term of such Letter of Credit, whichever is shorter; provided, however, that the applicable standby letter of credit fee payable in connection with the original issuance of any Letter of Credit (and on each anniversary date thereof if such Letter of Credit is renewed or extended) shall be no less than \$410; and
- (b) concurrently with the issuance of each Letter of Credit, and on each Quarterly Payment Date thereafter so long as such Letter of Credit shall remain outstanding, to the Issuing Lender for its own account, a fronting fee equal to 0.125% (12.5 basis points) per annum on the daily average stated amount of such Letter of Credit.

In addition to the foregoing, in connection with a Letter of Credit and activity relating thereto, Borrower also shall pay amendment, transfer, issuance, negotiation and such other fees as the Issuing Lender normally charges, in the amounts set forth from time to time as the Issuing Lender's published scheduled fees for such services. Each of the fees payable with respect to Letters of Credit under this Section is earned when due and is nonrefundable.

3.5 Increased Commitment Costs. If any Lender shall determine in good faith that the introduction after the Closing Date of any applicable law, rule, regulation or guideline regarding capital adequacy, or any change therein or any change in the interpretation or administration thereof by any central bank or other Governmental Agency charged with the interpretation or administration thereof, or compliance by such Lender (or its Eurodollar Lending Office) or any corporation controlling such Lender, with any request, guideline or directive regarding capital adequacy (whether or not having the force of Law) of any such central bank or other authority not imposed as a result of such Lender's or such corporation's failure to comply with any other Laws, affects or would affect the amount of capital required or expected to be maintained by such Lender or any corporation controlling such Lender and (taking into consideration such Lender's or such corporation's policies with respect to capital adequacy and such Lender's desired return on capital) determines in good faith that the amount of such capital is increased, or the rate of return on capital is reduced, as a consequence of its obligations under this Agreement, then, within five (5) days after demand of such Lender, Borrower shall pay to such Lender, from time to time as specified in good faith by such Lender, additional amounts sufficient to compensate such Lender in light of such circumstances, to the extent reasonably allocable to such obligations under this Agreement, provided that Borrower shall not be obligated to pay any such amount which arose prior to the date which is 180 days preceding the date of such demand or is attributable to periods prior to the date which is 180 days preceding the date of such demand or is attributable to periods prior to the date which is 180 days preceding the date of such demand. Each Lender's determination of such amounts shall be conclusive in the absence of manifest error.

3.6 Eurodollar Costs and Related Matters.

- (a) In the event that any Governmental Agency imposes on any Lender any reserve or comparable requirement (<u>including</u> any emergency, supplemental or other reserve) with respect to the Eurodollar Obligations of that Lender, <u>other than</u> the Eurodollar Reserve Percentage, Borrower shall pay that Lender within five (5) days after demand all amounts necessary to compensate such Lender (determined as though such Lender's Eurodollar Lending Office had funded 100% of its Eurodollar Rate Advances in the Designated Eurodollar Market) in respect of the imposition of such reserve requirements (<u>provided</u> that Borrower shall not be obligated to pay any such amount which arose prior to the date which is 180 days preceding the date of such demand or is attributable to periods prior to the date which is 180 days preceding the date of such demand). Any Lender's determination of such amount shall be conclusive in the absence of manifest error.
 - (b) If, after the date hereof, the existence or occurrence of any Special Eurodollar Circumstance:
 - (i) shall subject any Lender or its Eurodollar Lending Office to any tax, duty or other charge or cost with respect to any Eurodollar Rate Advance, any of its Notes evidencing Eurodollar Rate Advances or its

obligation to make Eurodollar Rate Advances, or shall change the basis of taxation of payments to any Lender attributable to the principal of or interest on any Eurodollar Rate Advance or any other amounts due under this Agreement in respect of any Eurodollar Rate Advance, any of its Notes evidencing Eurodollar Rate Advances or its obligation to make Eurodollar Rate Advances, excluding (A) taxes imposed on or measured in whole or in part by its overall net income by (1) any jurisdiction (or political subdivision thereof) in which it is organized or maintains its principal office or Eurodollar Lending Office or (2) any jurisdiction (or political subdivision thereof) in which it is "doing business" and (B) any withholding taxes or other taxes based on gross income imposed by the United States of America for any period with respect to which it has failed to provide Borrower with the appropriate form or forms required by Section 11.21, to the extent such forms are then required by applicable Laws;

- (ii) shall impose, modify or deem applicable any reserve not applicable or deemed applicable on the date hereof (including any reserve imposed by the Board of Governors of the Federal Reserve System (other than the Eurodollar Reserve Percentage), special deposit, capital or similar requirements against assets of, deposits with or for the account of, or credit extended by, any Lender or its Eurodollar Lending Office); or
- (iii) shall impose on any Lender or its Eurodollar Lending Office or the Designated Eurodollar Market any other condition affecting any Eurodollar Rate Advance, any of its Notes evidencing Eurodollar Rate Advances, its obligation to make Eurodollar Rate Advances or this Agreement, or shall otherwise affect any of the same;

and the result of any of the foregoing, as determined in good faith by such Lender, increases the cost to such Lender or its Eurodollar Lending Office of making or maintaining any Eurodollar Rate Advance or in respect of any Eurodollar Rate Advance, any of its Notes evidencing Eurodollar Rate Advances or its obligation to make Eurodollar Rate Advances or reduces the amount of any sum received or receivable by such Lender or its Eurodollar Lending Office with respect to any Eurodollar Rate Advance, any of its Notes evidencing Eurodollar Rate Advances or its obligation to make Eurodollar Rate Advances (assuming such Lender's Eurodollar Lending Office had funded 100% of its Eurodollar Rate Advances in the Designated Eurodollar Market), then, within five (5) Banking Days after demand by such Lender (with a copy to the Administrative Agent), Borrower shall pay to such Lender such additional amount or amounts as will compensate such Lender for such increased cost or reduction (determined as though such Lender's Eurodollar Lending Office had funded 100% of its Eurodollar Rate Advances in the Designated Eurodollar Market); provided that Borrower shall not be obligated to pay any such amount which arose prior to the date which is 180 days preceding the date of such demand. A statement of any Lender claiming compensation under this subsection shall be conclusive in the absence of manifest error.

(c) If, after the date hereof, the existence or occurrence of any Special Eurodollar Circumstance shall, in the good faith opinion of any Lender, make it unlawful or impossible for such Lender or its Eurodollar Lending Office to make, maintain or fund its portion of any Borrowing consisting of Eurodollar Rate Advances, or materially restrict the authority of such Lender to purchase or sell, or to take deposits of, Dollars in the Designated Eurodollar Market, or to determine or charge interest rates based upon the Eurodollar Rate, then such Lender's obligation to make Eurodollar Rate Advances shall be suspended for the duration of such illegality or impossibility and the Administrative Agent forthwith shall give notice thereof to the other Lenders and Borrower. Upon receipt of such notice, the outstanding principal amount of such Lender's affected Eurodollar Rate Advances, together with accrued interest thereon, automatically shall be converted to Alternate Base Rate Advances on either (i) the last day of the Eurodollar Period(s) applicable to such Eurodollar Rate Advances if such Lender may lawfully continue to maintain and fund such Eurodollar Rate Advances to such day(s) or (ii) immediately if such Lender may not lawfully continue to fund and maintain such Eurodollar Rate Advances to such day(s). Each Lender agrees to endeavor promptly to notify Borrower of any event of which it has actual knowledge, occurring after the Closing Date, which will cause such Lender to notify Borrower as set forth in the first sentence of this Section, and agrees to designate a different Eurodollar Lending Office if such designation will avoid the need for such notice and will not, in the good faith judgment of such Lender, otherwise be materially disadvantageous to such Lender. In the

event that any Lender is unable, for the reasons set forth above, to make, maintain or fund any Eurodollar Rate Advance, such Lender shall fund such Eurodollar Dollar Rate Advance as an Alternate Base Rate Advance for the same period of time, and such amount shall be treated in all respects as an Alternate Base Rate Advance. In the event that any Lender's obligation to make Eurodollar Rate Advances has been suspended under this Section, such Lender shall promptly notify the Administrative Agent and Borrower of the cessation of the Special Eurodollar Circumstance which gave rise to such suspension.

- (d) If, with respect to any proposed Borrowing comprised of Eurodollar Rate Advances:
- (i) the Administrative Agent reasonably determines that, by reason of circumstances affecting the Designated Eurodollar Market generally that are beyond the reasonable control of the Lenders, deposits in Dollars (in the applicable amounts) are not being offered to any Lender in the Designated Eurodollar Market for the applicable Eurodollar Period; or

(ii) the Requisite Lenders advise the Administrative Agent that the Eurodollar Rate as determined by the Administrative Agent (A) does not represent the effective pricing to such Lenders for deposits in Dollars in the Designated Eurodollar Market in the relevant amount for the applicable Eurodollar Period, or (B) will not adequately and fairly reflect the cost to such Lenders of making the applicable Eurodollar Rate Advances;

then the Administrative Agent forthwith shall give notice thereof to Borrower and the Lenders, whereupon until the Administrative Agent notifies Borrower that the circumstances giving rise to such suspension no longer exist, the obligation of the Lenders to make any future Eurodollar Rate Advances shall be suspended.

- (e) Upon payment or prepayment of any Eurodollar Rate Advance on a day other than the last day in the applicable Eurodollar Period (whether voluntarily, involuntarily, by reason of acceleration, or otherwise), or upon the failure of Borrower (for a reason other than the breach by a Lender of its obligation pursuant to Section 2.1(a)) to borrow on the date or in the amount specified for a Borrowing comprised of Eurodollar Rate Advances in any Request for Borrowing, Borrower shall pay to the appropriate Lender within five (5) Banking Days after demand a prepayment fee or failure to borrow fee, as the case may be (determined as though 100% of the Eurodollar Rate Advance had been funded in the Designated Eurodollar Market) equal to the <u>sum</u> of:
 - (1) the amount, if any, by which (i) the additional interest would have accrued on the amount prepaid or not borrowed at the Eurodollar Rate <u>plus</u> the Applicable Eurodollar Rate Margin if that amount had remained or been outstanding through the last day of the applicable Eurodollar Period <u>exceeds</u> (ii) the interest that such Lender could recover by placing such amount on deposit in the Designated Eurodollar Market for a period beginning on the date of the prepayment or failure to borrow and ending on the last day of the applicable Eurodollar Period (or, if no deposit rate quotation is available for such period, for the most comparable period for which a deposit rate quotation may be obtained); <u>plus</u>
 - (2) all out-of-pocket expenses incurred by such Lender reasonably attributable to such payment, prepayment or failure to borrow.

Each Lender's determination of the amount of any prepayment fee payable under this Section shall be conclusive in the absence of manifest error.

- (f) Each Lender agrees to endeavor promptly to notify Borrower of any event of which it has actual knowledge, occurring after the Closing Date, which will entitle such Lender to compensation pursuant to clause (a) or clause (b) of this Section, and agrees to designate a different Eurodollar Lending Office if such designation will avoid the need for or reduce the amount of such compensation and will not, in the good faith judgment of such Lender, otherwise be materially disadvantageous to such Lender. Any request for compensation by a Lender under this Section shall set forth the basis upon which it has been determined that such an amount is due from Borrower, a calculation of the amount due, and a certification that the corresponding costs have been incurred by such Lender.
- 3.7 <u>Late Payments and Default Rate.</u> If any installment of principal or interest or any fee or cost or other amount payable under any Loan Document to the Administrative Agent or any Lender is not paid when due, it shall thereafter bear interest at a fluctuating interest rate per annum at all times equal to the <u>sum of</u> the interest rate otherwise applicable thereto hereunder (or, if no interest rate is otherwise applicable thereto hereunder, the Alternate Base Rate) <u>plus</u> 2.00% (the "<u>Default Rate</u>"), to the fullest extent permitted by applicable Laws. While any Event of Default exists or after acceleration, at the option of the Requisite Lenders, Borrower shall pay interest (after as well as before entry of judgment thereon to the extent permitted by Law) on the principal amount of all outstanding Obligations, at the Default Rate, to the fullest extent permitted by Law. Accrued and unpaid interest on past due amounts (<u>including</u> interest on past due interest) shall be compounded monthly, on the last day of each calendar month, to the fullest extent permitted by applicable Laws.
- 3.8 <u>Computation of Interest and Fees</u>. Computation of interest on Alternate Base Rate Advances calculated with reference to the Prime Rate shall be calculated on the basis of a year of 365 or 366 days, as the case may be, and the actual

number of days elapsed; computation of interest on Alternate Base Rate Advances calculated by reference to the Federal Funds Rate, and on Eurodollar Rate Advances and all fees under this Agreement shall be calculated on the basis of a year of 360 days and the actual number of days elapsed. Borrower acknowledges that such latter calculation method will result in a higher yield to the Lenders than a method based on a year of 365 or 366 days. Interest shall accrue on each Advance for the day on which the Advance is made; interest shall not accrue on an Advance, or any portion thereof, for the day on which the Advance or such portion is paid. Any Advance that is repaid on the same day on which it is made shall bear interest for one day. Notwithstanding anything in this Agreement to the contrary, interest in excess of the maximum amount permitted by applicable Laws shall not accrue or be payable hereunder or under the Notes, and any amount paid as interest hereunder or under the Notes which would otherwise be in excess of such maximum permitted amount shall instead be treated as a payment of principal.

3.9 <u>Non-Banking Days</u>. If any payment to be made by Borrower or any other Party under any Loan Document shall come due on a day other than a Banking Day, payment shall instead be considered due on the next succeeding Banking Day and the extension of time shall be reflected in computing interest and fees.

3.10 Manner and Treatment of Payments.

- (a) Each payment hereunder (except payments pursuant to Sections 3.4, 3.5, 11.3, 11.11 and 11.21) or on the Notes or under any other Loan Document shall be made to the Administrative Agent at the Administrative Agent's Office, in immediately available funds not later than 11:00 a.m. California time, on the day of payment (which must be a Banking Day). All payments received after such time, on any Banking Day, shall be deemed received on the next succeeding Banking Day. The amount of all payments received by the Administrative Agent for the account of each Lender shall be immediately paid by the Administrative Agent to the applicable Lender in immediately available funds and, if such payment was received by the Administrative Agent by 11:00 a.m., California time, on a Banking Day and not so made available to the account of a Lender on that Banking Day, the Administrative Agent shall reimburse that Lender for the cost to such Lender of funding the amount of such payment at the Federal Funds Rate. All payments shall be made in lawful money of the United States of America.
- (b) Borrower hereby authorizes the Administrative Agent to debit the Designated Deposit Account to effect any payment due to the Lenders or the Administrative Agent pursuant to this Agreement. Any resulting overdraft in the Designated Deposit Account shall be payable by Borrower to the Administrative Agent on the next following Banking Day.
- (c) Each payment or prepayment on account of any Borrowing shall be applied pro rata according to the outstanding Advances made by each Lender comprising such Borrowing.
- (d) Each Lender shall use its best efforts to keep a record (in writing or by an electronic data entry system) of Advances made by it and payments received by it with respect to its Note and, subject to Section 10.6(g), such record shall, as against Borrower, be presumptive evidence of the amounts owing. Notwithstanding the foregoing sentence, the failure by any Lender to keep such a record shall not affect Borrower's obligation to pay the Obligations.
- (e) Each payment of any amount payable by Borrower or any other Party to any Lender under this Agreement or any other Loan Document shall be made free and clear of, and without reduction by reason of, any taxes, assessments or other charges imposed by any Governmental Agency, central bank or comparable authority, excluding (i) taxes imposed on or measured in whole or in part by its overall net income and franchise taxes imposed in lieu of net income taxes by (A) any jurisdiction (or political subdivision thereof) in which it is "doing business" and (ii) any withholding taxes or other taxes based on gross income imposed by the United States of America for any period with respect to which it has failed to provide Borrower with the appropriate form or forms required by Section 11.21, to the extent such forms are then required by applicable Laws (all such non-excluded taxes, assessments or other charges being hereinafter

referred to as "Taxes"). To the extent that Borrower is obligated by applicable Laws to make any deduction or withholding on account of Taxes from any amount payable to any Lender under this Agreement, Borrower shall (1) make such deduction or withholding and pay the same to the relevant Governmental Agency and (2) pay such additional amount to that Lender as is necessary to result in that Lender's receiving a net after-Tax amount equal to the amount to which that Lender would have been entitled under this Agreement absent such deduction or withholding. If and when receipt of such payment results in an excess payment or credit to that Lender on account of such Taxes, that Lender shall promptly refund such excess to Borrower.

- 3.11 <u>Funding Sources</u>. Nothing in this Agreement shall be deemed to obligate any Lender to obtain the funds for any Advance in any particular place or manner or to constitute a representation by any Lender that it has obtained or will obtain the funds for any Advance in any particular place or manner.
- 3.12 Failure to Charge Not Subsequent Waiver. Any decision by the Administrative Agent or any Lender not to require payment of any interest (<u>including</u> interest arising under <u>Section 3.7</u>), fee, cost or other amount payable under any Loan Document, or to calculate any amount payable by a particular method, on any occasion shall in no way limit or be deemed a waiver of the Administrative Agent's or such Lender's right to require full payment of any interest (<u>including</u> interest arising under <u>Section 3.7</u>), fee, cost or other amount payable under any Loan Document, or to calculate an amount payable by another method that is not inconsistent with this Agreement, on any other or subsequent occasion, <u>except</u> as provided in <u>Sections 3.5</u> and <u>3.6</u>.

- 3.13 Administrative Agent's Right to Assume Payments Will be Made. Unless the Administrative Agent shall have been notified by Borrower prior to the date on which any payment to be made by Borrower hereunder is due that Borrower does not intend to remit such payment (or otherwise cause sufficient funds to be available in the Designated Deposit Account for debit pursuant to Section 3.10(b)), the Administrative Agent may, in its discretion, assume that Borrower has remitted such payment (or caused funds sufficient to make such payment to be available) when so due and the Administrative Agent may, in its discretion and in reliance upon such assumption, make available to each Lender on such payment date, an amount equal to such Lender's share of such assumed payment. If Borrower has not in fact remitted such payment (or caused funds sufficient to make such payment to be available) to the Administrative Agent, each Lender shall forthwith on demand repay to the Administrative Agent the amount of such assumed payment made available to such Lender, together with interest thereon in respect of each day from and including the date such amount was made available by the Administrative Agent to such Lender to the date such amount is repaid to the Administrative Agent at the Federal Funds Rate.
- 3.14 <u>Fee Determination Detail</u>. The Administrative Agent, and any Lender, shall provide reasonable detail to Borrower regarding the manner in which the amount of any payment to the Administrative Agent and the Lenders, or that Lender, under <u>Article 3</u> has been determined, concurrently with demand for such payment.
- 3.15 <u>Survivability</u>. All of Borrower's obligations under <u>Sections 3.4</u> and <u>3.5</u> shall survive for the one year period following the Termination Date, and Borrower shall remain obligated thereunder for all claims under such Sections made by any Lender to Borrower prior to the expiration of such period.

Article 4. REPRESENTATIONS AND WARRANTIES

Borrower represents and warrants to the Administrative Agent and each of the Lenders that:

- 4.1 Existence and Qualification; Power; Compliance With Laws. Borrower is a corporation duly formed, validly existing and in good standing under the Laws of the State of California. Borrower is duly qualified or registered to transact business and is in good standing in the State of California, and each other jurisdiction in which the conduct of its business or the ownership or leasing of its Properties makes such qualification or registration necessary, except where the failure so to qualify or register and to be in good standing could not reasonably be expected to have a Material Adverse Effect. Borrower has all requisite power and authority to conduct its business, to own and lease its Properties and to execute and deliver each Loan Document to which it is a Party and to perform its Obligations. The chief executive offices of Borrower are located in San Dimas, California. All outstanding capital stock of Borrower is duly authorized, validly issued, fully paid and non-assessable, and no holder thereof has any enforceable right of rescission under any applicable state or federal securities or other Laws. Borrower is in compliance with all Laws and other legal requirements applicable to its business, has obtained all authorizations, consents, approvals, orders, licenses and permits from, and has accomplished all filings, registrations and qualifications with, or obtained exemptions from any of the foregoing from, any Governmental Agency that are necessary for the transaction of its business, except where the failure so to comply with Laws and other legal requirements applicable to its business, obtain authorizations, etc., file, register, qualify or obtain exemptions could not reasonably be expected to have a Material Adverse Effect.
- 4.2 <u>Authority; Compliance With Other Agreements and Instruments and Government Regulations</u>. The execution and delivery by Borrower of the Loan Documents to which it is a Party and payment of the Obligations have been duly authorized by all necessary corporate or company action, as applicable, and do not and will not:
 - (a) Require any consent or approval not heretofore obtained of any partner, director, stockholder, member, security holder or creditor of Borrower;
 - (b) Violate or conflict with any provision of Borrower's charter, certificate of incorporation, bylaws, or other organizational documents, as applicable;

(c)	Result in or require the creation or imposition of any Lien (other than pursuant to the Loan Documents) or Right of Others
upon or with respect t	o any Property now owned or leased or hereafter acquired by Borrower;

(d) Violate any Requirement of Law applicable to Borrower;

- (e) Result in a breach of or constitute a default under, or cause or permit the acceleration of any obligation owed under, any indenture or loan or credit agreement or any other Contractual Obligation to which Borrower is a party or by which Borrower or any of its Property is bound or affected; and Borrower is not in violation of, or default under, any Requirement of Law or Contractual Obligation, or any indenture, loan or credit agreement described in Section 4.2(e), in any respect that could reasonably be expected to have a Material Adverse Effect
- 4.3 <u>No Governmental Approvals Required</u>. Except as previously obtained or made, no authorization, consent, approval, order, license or permit from, or filing, registration or qualification with, any Governmental Agency is or will be required to authorize or permit under applicable Laws the execution and delivery by Borrower of the Loan Documents to which it is a Party and payment of the Obligations.

4.4 Subsidiaries.

- (a) Schedule 4.4 hereto correctly sets forth, as of the Closing Date, the names, form of legal entity, number of shares of capital stock or membership or other equity interests, as applicable, issued and outstanding, number of shares of capital stock or membership or other equity interests, as applicable, owned by Borrower or any Subsidiary of Borrower (specifying such owner) and jurisdictions of organization of all Subsidiaries of Borrower. Except as described in Schedule 4.4, as of the Closing Date, Borrower does not own any capital stock, membership interest, other equity interest or debt Security which is convertible, or exchangeable, for capital stock, membership interests or other equity interests in any Person. Unless otherwise indicated in Schedule 4.4, as of the Closing Date, all of the outstanding shares of capital stock, all of the outstanding membership interests or all of the units of other equity interest, as the case may be, of each Subsidiary are owned of record and beneficially by Borrower, there are no outstanding options, warrants or other rights to purchase capital stock of any such Subsidiary, and all such shares, membership interests or other equity interests so owned are duly authorized, validly issued, fully paid and non-assessable, and were issued in compliance with all applicable state and federal securities and other Laws, and are free and clear of all Liens, except for Permitted Encumbrances and other encumbrances permitted pursuant to Section 6.9.
- (b) As of the Closing Date, each Subsidiary is a legal entity of the type described in <u>Schedule 4.4</u> duly formed, validly existing and, if such concept is legally recognized in such Subsidiary's jurisdiction of organization, in "good standing" under the Laws of its jurisdiction of organization, is duly qualified to do business as a foreign organization and, if such concept is legally recognized in any applicable jurisdiction, is in "good standing" as such in each jurisdiction in which the conduct of its business or the ownership or leasing of its Properties makes such qualification necessary (except where the failure to be so duly qualified and in good standing could not reasonably be expected to have a Material Adverse Effect), and has all requisite power and authority to conduct its business and to own and lease its Properties.
- (c) Each Subsidiary is in compliance with all Laws and other requirements applicable to its business and has obtained all authorizations, consents, approvals, orders, licenses, and permits from, and each such Subsidiary has accomplished all filings, registrations, and qualifications with, or obtained exemptions from any of the foregoing from, any Governmental Agency that are necessary for the transaction of its business, except where the failure to be in such compliance, obtain such authorizations, consents, approvals, orders, licenses, and permits, accomplish such filings, registrations, and qualifications, or obtain such exemptions, could not reasonably be expected to have a Material Adverse Effect.
- 4.5 <u>Financial Statements</u>. Borrower has furnished to the Lenders (a) the audited consolidated financial statements of Borrower and its Subsidiaries for the Fiscal Year ended December 31, 2004 and (b) the consolidating and consolidated financial statements of Borrower and its Subsidiaries for the Fiscal Quarter ended March 31, 2005. Such financial statements fairly present in all material respects the financial condition, results of operations and changes in financial position as of such dates and for such periods in conformity with GAAP consistently applied.

4.6 No Other Liabilities; No Material Adverse Changes. As of the Closing Date, Borrower and its Subsidiaries do not have any material liability or material contingent liability required under GAAP to be reflected or disclosed, and not reflected or disclosed, in the financial statements described in Section 4.5, other than liabilities and contingent liabilities arising in the ordinary course of business since the date of such financial statements. As of the Closing Date, no circumstance or event has occurred that could reasonably be expected to have a Material Adverse Effect since December 31, 2004.

- 4.7 <u>Title to and Location of Property</u>. As of the Closing Date, <u>except</u> as set forth in <u>Schedule 4.7</u>, Borrower and its Subsidiaries have valid title to the Property (<u>other than</u> assets which are the subject of a Capital Lease Obligation) reflected in the financial statements described in <u>Section 4.5</u>, <u>other than</u> items of Property or exceptions to title which are in each case immaterial and Property subsequently sold or disposed of in the ordinary course of business. Such Property is free and clear of all Liens and Rights of Others, <u>other than</u> Liens or Rights of Others described in <u>Schedule 4.7</u> and Permitted Encumbrances, other encumbrances permitted pursuant to <u>Section 6.9</u>, and Permitted Rights of Others.
- 4.8 <u>Intangible Assets</u>. Borrower and its Subsidiaries own, or possess the right to use to the extent necessary in their respective businesses, all material trademarks, trade names, copyrights, patents, patent rights, computer software, licenses and other Intangible Assets that are used in the conduct of their businesses as now operated, and no such Intangible Asset, to the best knowledge of Borrower, conflicts with the valid trademark, trade name, copyright, patent right or Intangible Asset of any other Person to the extent that such conflict could reasonably be expected to have a Material Adverse Effect. <u>Schedule 4.8</u> sets forth all patents, patent applications, trademarks, trade names and trade styles used by Borrower or any of its Subsidiaries at any time within the five (5) year period ending on the Closing Date.
- 4.9 <u>Litigation</u>. Except for (a) any matter fully covered as to subject matter and amount (subject to applicable deductibles and retentions) by insurance for which the insurance carrier has not asserted lack of subject matter coverage or reserved its right to do so, (b) any matter, or series of related matters, involving a claim against Borrower or any of its Subsidiaries of less than \$250,000, (c) matters of an administrative nature not involving a claim or charge against Borrower or any Subsidiary of Borrower and (d) matters set forth in <u>Schedule 4.9</u>, there are no actions, suits, proceedings or investigations pending as to which Borrower or any of its Subsidiaries have been served or have received notice or, to the best knowledge of Borrower, its Subsidiaries, or, to the best knowledge of Borrower, any executive officer of any such Persons has been indicted or convicted in connection with or is engaging in any criminal conduct which constitutes a felony, or is currently subject to any lawsuit or proceeding or, to the best of Borrower's knowledge, under investigation in connection with any anti-racketeering or criminal conduct or activity which constitutes a felony.
- 4.10 <u>Binding Obligations</u>. Each of the Loan Documents to which Borrower is a Party will, when executed and delivered by Borrower, constitute the legal, valid and binding obligation of Borrower, enforceable against Borrower in accordance with its terms, <u>except</u> as enforcement may be limited by Debtor Relief Laws or equitable principles relating to the granting of specific performance and other equitable remedies as a matter of judicial discretion.
 - 4.11 No Default. No event has occurred and is continuing that is a Default or Event of Default.

4.12 ERISA.

- (a) With respect to each Pension Plan:
- (i) such Pension Plan complies in all material respects with ERISA and any other applicable Laws to the extent that noncompliance could reasonably be expected to have a Material Adverse Effect;
- (ii) such Pension Plan has not incurred any "accumulated funding deficiency" (as defined in Section 302 of ERISA) that could reasonably be expected to have a Material Adverse Effect;
- (iii) no "reportable event" (as defined in Section 4043 of ERISA, but <u>excluding</u> such events as to which the PBGC has by regulation waived the requirement therein contained that it be notified within thirty days of the occurrence of such event) has occurred that could reasonably be expected to have a Material Adverse Effect; and

- (iv) neither Borrower nor any of its Subsidiaries has engaged in any non-exempt "prohibited transaction" (as defined in Section 4975 of the Code) that could reasonably be expected to have a Material Adverse Effect.
- (b) Neither Borrower nor any of its Subsidiaries has incurred or expects to incur any withdrawal liability to any Multiemployer Plan that could reasonably be expected to have a Material Adverse Effect.

- 4.13 <u>Regulation U; Investment Company Act</u>. No part of the proceeds of any Advance hereunder will be used to purchase or carry, or to extend credit to others for the purpose of purchasing or carrying, any Margin Stock in violation of Regulation U. Neither Borrower nor any of its Subsidiaries is or is required to be registered as an "investment company" under the Investment Company Act of 1940.
- 4.14 <u>Disclosure</u>. No written statement made by a Senior Officer of Borrower to the Administrative Agent or any Lender pursuant to this Agreement, or in connection with any Advance, as of the date thereof contained any untrue statement of a material fact or omitted a material fact necessary to make the statement made not misleading in light of all the circumstances existing at the date the statement was made.
- 4.15 <u>Tax Liability</u>. Borrower and its Subsidiaries have filed all tax returns which are required to be filed, and have paid, or made provision for the payment of, all taxes with respect to the periods, Property or transactions covered by said returns, or pursuant to any assessment received by Borrower or any of its Subsidiaries, <u>except</u> (a) such taxes, if any, as are being contested in good faith by appropriate proceedings and as to which adequate reserves have been established and maintained and (b) immaterial taxes so long as no material Property of Borrower or any Subsidiary is at impending risk of being seized, levied upon or forfeited.
- 4.16 <u>Projections</u>. As of the Closing Date, to the best knowledge of Borrower the assumptions set forth in the Projections most recently delivered to the Administrative Agent are reasonable and consistent with each other and with all facts known to Borrower, and the Projections are reasonably based on such assumptions. Nothing in this <u>Section 4.16</u> shall be construed as a representation or covenant that the Projections in fact will be achieved.
- 4.17 <u>Hazardous Materials</u>. Except as described in <u>Schedule 4.17</u>, or as may subsequently be disclosed by Borrower in writing to the Administrative Agent, (a) neither Borrower nor any of its Subsidiaries at any time has disposed of, discharged, released or threatened the release of any Hazardous Materials on, from or under the Real Property in violation of any Hazardous Materials Law that would individually or in the aggregate constitute a Material Adverse Effect, (b) no condition exists that violates any Hazardous Material Law affecting any Real Property except for such violations that would not individually or in the aggregate constitute a Material Adverse Effect, (c) no Real Property or any portion thereof is or has been utilized by Borrower or any Subsidiary as a site for the manufacture of any Hazardous Materials and (d) to the extent that any Hazardous Materials are used, generated or stored by Borrower or any Subsidiary on any Real Property, or transported to or from such Real Property by Borrower, or any Subsidiary, such use, generation, storage and transportation are in compliance in all material respects with all Hazardous Materials Laws.
- 4.18 <u>Employee Matters</u>. There is no strike, work stoppage or labor dispute with any union or group of employees pending or, to the best knowledge of Borrower overtly threatened involving Borrower or any Subsidiary that individually or in the aggregate could reasonably be expected to have a Material Adverse Effect.
 - 4.19 <u>Fiscal Year</u>. Borrower and its Subsidiaries each operate on a fiscal year ending on December 31.
- 4.20 <u>Solvency</u>. After giving effect to this Agreement and the other Loan Documents (including after giving effect to Advances under this Agreement as of the Closing Date), Borrower shall be Solvent.

Article 5. AFFIRMATIVE COVENANTS (OTHER THAN INFORMATION AND REPORTING REQUIREMENTS)

So long as any Advance remains unpaid, or any other Obligation remains unpaid, or any portion of any of the Commitments remains in force, Borrower shall, and shall cause each of the Subsidiaries to, unless the Administrative Agent (with the written approval of the Requisite Lenders) otherwise consents:

5.1 Payment of Taxes and Other Potential Liens. Pay and discharge promptly all taxes, assessments and governmental charges or levies imposed upon any of them, upon their respective Property or any part thereof and upon their respective income or profits or any part thereof, except that Borrower and its Subsidiaries shall not be required to pay or cause to be paid (a) any tax, assessment, charge or levy that is not yet past due, or is being contested in good faith by appropriate proceedings so long as the relevant entity has established and maintains adequate reserves for the payment of the same or (b) any immaterial tax so long as no material Property of Borrower or any Subsidiary is at impending risk of being seized, levied upon or forfeited.

- 5.2 <u>Preservation of Existence</u>. Preserve and maintain their respective existences (<u>except</u> as permitted by <u>Section 6.4</u>) in their respective jurisdictions of formation and all material authorizations, rights, franchises, privileges, consents, approvals, orders, licenses, permits, or registrations from any Governmental Agency that are necessary for the transaction of their respective business and qualify and remain qualified to transact business in each jurisdiction in which such qualification is necessary in view of their respective business or the ownership or leasing of their respective Properties <u>except</u> where the failure to so qualify or remain qualified could not reasonably be expected to have a Material Adverse Effect.
- 5.3 <u>Maintenance of Properties</u>. Maintain, preserve and protect all of their respective Properties in good order and condition, subject to wear and tear in the ordinary course of business, and not permit any waste of their respective Properties, <u>except</u> (a) that the failure to maintain, preserve and protect a particular item of Property that is at the end of its useful life or that is not of significant value, either intrinsically or to the operations of Borrower and Subsidiaries, taken as a whole, shall not constitute a violation of this covenant, and (b) this covenant shall not be construed to prohibit any Disposition otherwise permitted pursuant to <u>Section 6.3</u>.
- 5.4 <u>Maintenance of Insurance</u>. Maintain, or cause to be maintained, liability, casualty and other insurance (subject to customary deductibles and retentions) with responsible insurance companies in such amounts and against such risks as is carried by responsible companies engaged in similar businesses and owning similar assets.
- 5.5 <u>Compliance With Laws</u>. Comply with all Requirements of Law noncompliance with which could reasonably be expected to have a Material Adverse Effect, <u>except</u> that Borrower and its Subsidiaries need not comply with a Requirement of Law then being contested by any of them in good faith by appropriate proceedings.
- 5.6 Inspection Rights. Upon reasonable notice, at any time during regular business hours and, as requested by the Administrative Agent (but not so as to materially interfere with the business of Borrower or any of the Subsidiaries) permit the Administrative Agent, or any Lender, or any authorized employee, agent or representative thereof, to examine, audit and make copies and abstracts from the records and books of account of (including any software or CD Rom files relating thereto), and to visit and inspect the Properties of, Borrower and its Subsidiaries and to discuss the affairs, finances and accounts of Borrower and its Subsidiaries with any of their officers, key employees or accountants and, upon request, furnish promptly to the Administrative Agent or any Lender true copies of all financial information made available to the board of directors or audit committee of the board of directors of Borrower. If any of the Properties, books or records of Borrower or any of the Subsidiaries are in the possession of a third party, Borrower authorizes that third party to permit the Administrative Agent or any Lender or any agents thereof to have access to perform inspections or audits and to respond to the Administrative Agent's or any Lender's request for information concerning such Properties, books and records. Notwithstanding the foregoing, no prior notice of any such examination, audit, visit, inspection or discussion shall be required if an Event of Default has occurred and remains in effect.
- 5.7 <u>Keeping of Records and Books of Account</u>. Keep adequate records and books of account reflecting all financial transactions in conformity with GAAP, consistently applied, and in material conformity with all applicable requirements of any Governmental Agency having regulatory jurisdiction over Borrower and its Subsidiaries.
- 5.8 <u>Compliance With Agreements</u>. Promptly and fully comply with all Contractual Obligations to which any one or more of them is a party, <u>except</u> for any such Contractual Obligations (a) the performance of which would cause a Default or (b) then being contested by any of them in good faith by appropriate proceedings or (c) if the failure to comply could not reasonably be expected to have a Material Adverse Effect.
- 5.9 <u>Use of Proceeds</u>. Use the proceeds of all Advances to (a) refinance certain Indebtedness of Borrower and its Subsidiaries (<u>including</u> SCW), (ii) finance certain acquisitions and (c) provide for the working capital and general corporate purpose need of Borrower and its Subsidiaries.

- Hazardous Materials Laws. Keep and maintain all Real Property and each portion thereof in compliance with all applicable Hazardous Materials Laws, except to the extent that would not individually or in the aggregate constitute a Material Adverse Effect, and promptly notify the Administrative Agent in writing (attaching a copy of any pertinent written material) of (a) any and all material enforcement, cleanup, removal or other governmental or regulatory actions instituted, completed or threatened in writing by a Governmental Agency pursuant to any applicable Hazardous Materials Laws, (b) any and all material claims made or threatened in writing by any Person against Borrower or any Subsidiary relating to damage, contribution, cost recovery, compensation, loss or injury resulting from any Hazardous Materials and (c) discovery by any Senior Officer of Borrower of any material occurrence or condition on any real property adjoining or in the vicinity of such Real Property that could reasonably be expected to cause such Real Property or any part thereof to be subject to any restrictions on the ownership, occupancy, transferability or use of such Real Property under any applicable Hazardous Materials Laws.
- 5.11 <u>Minimum Debt Rating</u>. Maintain at all times a Debt Rating (in the case of Borrower only) equal to (or better than) Baa3 or BBB-. For purposes of this <u>Section 5.11</u>, if the prevailing Debt Ratings as of any date of determination are "split ratings", the lower/lowest Debt Rating shall apply.
- 5.12 <u>Syndication Process</u>. Cooperate in such respects as may be reasonably requested by the Lead Arranger in connection with the syndication of the credit facility under this Agreement, <u>including</u> the provision of information (in form and substance acceptable to the Lead Arranger) for inclusion in written materials furnished to prospective syndicate members and the participation by Senior Officers of Borrower and its Subsidiaries in meetings with prospective syndicate members.

Article 6. NEGATIVE COVENANTS

So long as any Advance remains unpaid, or any other Obligation remains unpaid, or any portion of any of the Commitments remains in force, Borrower shall not, and shall not permit any of its Subsidiaries to, unless the Administrative Agent (with the written approval of the Requisite Lenders or, if required by Section 11.2, all of the Lenders) otherwise consents:

- 6.1 <u>Prepayment of Indebtedness.</u> Prepay any principal or interest on any Indebtedness of Borrower or any Subsidiary prior to the date when due, or make any payment or deposit with any Person that has the effect of providing for the satisfaction of any Indebtedness of Borrower or any Subsidiary prior to the date when due, <u>except</u> (a) Indebtedness to the Administrative Agent and the Lenders under this Agreement and the other Loan Documents, (b) the prepayment or refunding of Indebtedness solely through the issuance of additional Indebtedness that is permitted under this Agreement; <u>provided</u> that such additional Indebtedness has either (i) a longer weighted average life than the Borrowings hereunder, or (ii) a weighted average life that is equal to or longer than the Indebtedness being refunded and (c) Indebtedness to other Persons the prepayment of which is approved in advance by the Requisite Lenders in writing.
- 6.2 <u>Prepayment of Subordinated Obligations</u>. Pay any (a) principal (including sinking fund payments) or any other amount (<u>other than</u> scheduled interest payments) with respect to any Subordinated Obligation, or purchase or redeem any Subordinated Obligation or deposit any monies, Securities or other Property with any trustee or other Person to provide assurance that the principal or any portion thereof of any Subordinated Obligation will be paid when due or otherwise to provide for the defeasance of any Subordinated Obligation or (b) scheduled interest on any Subordinated Obligation if the payment thereof is then prohibited under the terms of the subordination provisions governing such Subordinated Obligations.
- 6.3 <u>Disposition of Property.</u> Make any Disposition of its Property, whether now owned or hereafter acquired, <u>except</u> (a) Dispositions of obsolete Property or Property with no material remaining useful life, (b) Dispositions in an aggregate amount not to exceed \$2,000,000 in any Fiscal Year ending after the Closing Date or \$8,000,000 in the aggregate from and after the Closing Date to the Termination Date; <u>provided</u> that (i) at the time of any such Disposition pursuant to <u>clause (b) only</u>, no Default or Event of Default shall exist or shall result from such Disposition and (ii) the sales price relating to a Disposition (pursuant to <u>clause (a)</u> or <u>(b)</u>) shall be paid in Cash and/or Indebtedness or other evidence of an Investment permitted pursuant to <u>Section 6.14(h)</u>, and

- (c) Dispositions pursuant to any order of any Governmental Agency in an eminent domain proceeding and any settlement of any such proceeding.
- 6.4 <u>Mergers</u>. Merge or consolidate with or into any Person, except mergers and consolidations of a Subsidiary into Borrower (with Borrower as the surviving entity), mergers of Subsidiaries with each other or mergers entered into in connection with Permitted Acquisitions, <u>provided</u> that (a) no Default or Event of Default would result therefrom and (b) any such "surviving" entity shall have executed such amendments to the Loan Documents, if any, as the Administrative Agent may reasonably determine are appropriate as a result of such merger.
- 6.5 <u>Hostile Tender Offers</u>. Make any offer to purchase or acquire, or consummate a purchase or acquisition of, five percent (5%) or more of the voting interest in any corporation or other business entity if the board of directors or management of such corporation or business entity has notified Borrower that it opposes such offer or purchase and such notice has not been withdrawn or superseded.
- 6.6 <u>Distributions</u>. Declare or pay or make any form of Distribution, whether from capital, income or otherwise, and whether in Cash or other Property, except:
 - (a) Distributions by any Subsidiary to Borrower or to any wholly-owned Subsidiary of Borrower;
 - (b) Distributions consisting of dividends payable solely in capital stock or rights to purchase capital stock so long as no Default or Event of Default then exists; and
 - (c) Distributions consisting of (i) repurchases of preferred stock of Borrower in an amount not to exceed \$2,500,000 in the aggregate from and after the Closing Date and (ii) dividends paid in Cash in a manner reasonably consistent with Borrower's past practices made in any Fiscal Year <u>if</u>, in any such case, no Default or Event of Default then exists and, giving effect thereto on a pro-forma estimated basis, Borrower would be in compliance with <u>Section 6.12</u> as of the end of the then current Fiscal Quarter.
- 6.7 <u>ERISA</u>. (a) At any time, permit any Pension Plan to: (i) engage in any non-exempt "prohibited transaction" (as defined in Section 4975 of the Code); (ii) fail to comply with ERISA or any other applicable Laws; (iii) incur any material "accumulated funding deficiency" (as defined in Section 302 of ERISA); or (iv) terminate in any manner, which, with respect to each event listed above, could reasonably be expected to result in a Material Adverse Effect or (b) withdraw, completely or partially, from any Multiemployer Plan if to do so could reasonably be expected to result in a Material Adverse Effect.
- 6.8 <u>Change in Nature of Business</u>. Make any change in the nature of the business of Borrower and its Subsidiaries, taken as a whole, as at present conducted; <u>provided</u> that a Permitted Acquisition shall not be deemed such a change.
- 6.9 <u>Liens and Negative Pledges</u>. Create, incur, assume or suffer to exist any Lien or Negative Pledge of any nature upon or with respect to any of their respective Properties, or engage in any Sale and Leaseback transaction with respect to any of their respective Properties, whether now owned or hereafter acquired, <u>except</u>:
 - (a) Liens and Negative Pledges existing on the Closing Date and disclosed in <u>Schedule 4.7</u> and any renewals/extensions or amendments thereof, <u>provided</u> that the obligations secured or benefited thereby are not increased;
 - (b) Liens and Negative Pledges under the Loan Documents;
 - (c) Permitted Encumbrances;

- (d) Liens which secure Permitted Acquisition Indebtedness which were in existence at the time of the Permitted Acquisition and were not created in contemplation of such Permitted Acquisition;
- (e) Liens securing Permitted Capital Asset Indebtedness on and limited to the capital assets acquired, constructed or financed with the proceeds of such Permitted Capital Asset Indebtedness or with the proceeds of any Indebtedness directly or indirectly refinanced by such Indebtedness; <u>provided</u> that the aggregate principal amount of such Indebtedness secured by such Liens and incurred by Borrower and/or its Subsidiaries after the Closing Date shall not exceed \$10,000,000 at any one time outstanding (as determined in accordance with GAAP consistently applied); and
- (f) any Negative Pledge with respect to the rights of a Subsidiary of Borrower under a Military Utility Privatization entered into by such Subsidiary.
- 6.10 Indebtedness and Guaranty Obligations. Create, incur or assume any Indebtedness or Guaranty Obligation if an Event of Default has occurred and is continuing or if, after giving effect thereto, Borrower would not be in compliance with the provisions of Section 6.12 or Section 6.13 or an Event of Default would otherwise occur. Notwithstanding the foregoing, Borrower shall not permit any Subsidiary to create, incur, assume or suffer to exist any Indebtedness or Guaranty Obligation, except (a) Permitted Acquisition Indebtedness, (b) Permitted Capital Asset Indebtedness, (c) existing Indebtedness set forth on Schedule 6.10(b), (d) Indebtedness owed to Borrower or a wholly-owned Subsidiary, (e) unsecured term Indebtedness (i.e., not revolving credit) that (i) either has a longer weighted average life than the Borrowings hereunder or satisfies the requirements of Section 6.1, (ii) to the extent that a Governmental Agency has regulatory jurisdiction over the issuance of such Indebtedness of such Subsidiary, the issuance of such Indebtedness is permitted by such regulatory jurisdiction, (iii) is incurred in the ordinary course of business of such Subsidiary and is substantially consistent with the prior practices of SCW, and (iv) is provided by any Person or Governmental Agency, other than a commercial bank under a credit agreement or facility substantially similar thereto, and (f) other unsecured Indebtedness in the aggregate principal amount not to exceed \$1,000,000.
- 6.11 <u>Transactions with Affiliates</u>. Enter into any transaction of any kind with any Affiliate of Borrower <u>other than</u> (without duplication): (a) salary, bonus, employee stock option and other compensation arrangements with directors or officers in the ordinary course of business; (b) Investments permitted pursuant to <u>Section 6.14(d)</u>, (c) Distributions permitted pursuant to <u>Section 6.6</u>; (d) transactions with wholly-owned Subsidiaries; and (e) transactions on overall terms at least as favorable to Borrower or its Subsidiaries as would be the case in an arm's-length transaction between unrelated parties of equal bargaining power.
- 6.12 <u>Total Funded Debt Ratio</u>. Permit the Total Funded Debt Ratio, as of the last day of any Fiscal Quarter, to be greater than 0.65 to 1.00.
 - 6.13 Interest Coverage Ratio. Permit the Interest Coverage Ratio, as of the last day of any Fiscal Quarter, to be less than 3.25 to 1.00.
- 6.14 <u>Investments and Acquisitions</u>. Make any Acquisition or enter into any agreement to make any Acquisition unless approved in advance by the Administrative Agent and the Requisite Lenders in writing, or make or suffer to exist any Investment, <u>other than</u>:
 - (a) Permitted Acquisitions;
 - (b) Investments in existence on the Closing Date and disclosed on Schedule 6.14;
 - (c) Investments consisting of Cash Equivalents;
 - (d) Investments consisting of loans and advances to officers, directors and employees of Borrower and its Subsidiaries for travel, entertainment, relocation, anticipated bonus, exercise of stock options and analogous ordinary

business purposes provided that the aggregate amount of such Investments does not exceed \$1,000,000 at any time outstanding;

- (e) Investments in a Subsidiary that is a wholly-owned Subsidiary of Borrower;
- (f) Investments consisting of the extension of credit to customers or suppliers of Borrower and its Subsidiaries in the ordinary course of business and any Investments received in satisfaction or partial satisfaction thereof;
- (g) Investments received in connection with the settlement of a bona fide dispute with another Person <u>provided</u> that the aggregate amount of such Investments does not exceed \$500,000 at any time outstanding;
- (h) Investments representing all or a portion of the sales price of Property sold or services provided to another Person <u>provided</u> that the aggregate amount of such Investments does not exceed \$500,000 at any time outstanding; and
 - (i) Investments consisting of Indebtedness and Guaranty Obligations owed to Borrower or any of its Subsidiaries.
- 6.15 Operating Leases. Incur any obligation to pay rent under an operating lease in any Fiscal Year if to do so would result in the aggregate obligation of Borrower and its Subsidiaries to pay rent under all operating leases in that Fiscal Year to exceed \$4,000,000.
- 6.16 <u>Amendments</u>. Amend or modify any term or provision of (a) any indenture, agreement or instrument evidencing or governing any Subordinated Obligation or (b) any material provision of any Material Contract, if in any such case such amendment or modification in any respect will or may adversely affect the interest of the Lenders.
- 6.17 <u>Use of Lender's Name</u>. Use any Lender's name (or the name of any of any Lender's Affiliates) in connection with any of their business operations <u>except</u> to identify the existence of the Revolving Facility and the names of the Lenders in the ordinary course of Borrower's business or to comply with Borrower's obligations under Law. Nothing contained in this Agreement is intended to permit or authorize Borrower to make any commitment or contract on behalf of any Lender or the Administrative Agent.
- 6.18 <u>Change of Fiscal Periods</u>. Change its Fiscal Year or any other fiscal period with respect to which it reports financial results hereunder or otherwise.

Article 7. INFORMATION AND REPORTING REQUIREMENTS

- 7.1 <u>Financial and Business Information</u>. So long as any Advance remains unpaid, or any other Obligation remains unpaid, or any portion of any of the Commitments remains in force, Borrower shall, unless the Administrative Agent (with the written approval of the Requisite Lenders) otherwise consents, at Borrower's sole expense, deliver to the Administrative Agent for distribution by it to the Lenders, a sufficient number of copies for all of the Lenders of the following:
 - (a) (i) As soon as practicable, and in any event within fifty (50) days after the end of each Fiscal Quarter ending March 31, June 30 and September 30 (commencing with the Fiscal Quarter ending June 30, 2005), the consolidating and consolidated balance sheet of Borrower and its Subsidiaries as at the end of such Fiscal Quarter and the consolidating and consolidated statements of income, operations and cash flows for such Fiscal Quarter, and the portion of the Fiscal Year ended with such Fiscal Quarter, together with a statement of Stockholders' Equity as of the last day of such Fiscal Quarter, all in reasonable detail, (ii) such financial statements shall be certified by the president or chief financial

officer of Borrower as fairly presenting the financial condition, results of operations and cash flows of Borrower and its Subsidiaries in accordance with GAAP (other than footnote disclosures), consistently applied, as at such date and for such periods, subject only to normal year-end accruals and audit adjustments.

- (i) As soon as practicable, and in any event within one hundred (100) days after the end of each Fiscal Year, the consolidating and consolidated balance sheet of Borrower and its Subsidiaries as at the end of such Fiscal Year and the consolidating and consolidated statements of income, operations, stockholders' equity and cash flows, in each case of Borrower and its Subsidiaries for such Fiscal Year, all in reasonable detail. Such financial statements shall be prepared in accordance with GAAP, consistently applied, and such consolidated financial statements shall be accompanied by a report of PricewaterhouseCoopers LLP or other independent public accountants of recognized standing selected by Borrower and reasonably satisfactory to the Requisite Lenders, which report shall be prepared in accordance with generally accepted auditing standards as at such date, and shall not be subject to any qualifications or exceptions, and which report shall specifically disclose any changes discovered by such accountants in Borrower's or its Subsidiaries' applicable process of management of accounts, (ii) such accountants' report shall be accompanied by a certificate stating that, in making the examination pursuant to generally accepted auditing standards necessary for the certification of such financial statements and such report, such accountants have obtained no knowledge of any Default then existing or, if, in the opinion of such accountants, any such Default shall exist, stating the nature and status of such Default, and stating that such accountants have reviewed Borrower's financial calculations as at the end of such Fiscal Year (which shall accompany such certificate) under Sections 6.12 and 6.13, have read such Sections (including the definitions of all defined terms used therein) and that nothing has come to the attention of such accountants in the course of such examination that would cause them to believe that the same were not calculated by Borrower in the manner prescribed by this Agreement, and (iii) in addition, Borrower shall deliver to the Administrative Agent a copy of (A) any "management letter" prepared by such accountants in conjunction with preparation of the foregoing report and (B) a separate report prepared by such accountants in conjunction with preparation of the foregoing report, pursuant to which separate report such accountants shall be required to disclose any material changes discovered by such accountants in the then current account management process (including the determination of returns and reserves, inventory management practices, and accounts receivable management practices).
- (c) As soon as practicable, and in any event not later than sixty (60) days subsequent to the commencement of each Fiscal Year, a budget and projection of Borrower and its Subsidiaries setting forth (i) by Fiscal Quarter for the four (4) Fiscal Quarters of that Fiscal Year and (ii) on an annual basis for each succeeding Fiscal Year thereafter through the Maturity Date, projected balance sheets, statements of operations and statements of cash flow, all in reasonable detail;
- (d) Promptly after request by the Administrative Agent or any Lender, copies of any detailed audit reports, management letters or recommendations submitted to Borrower (or its board of directors) by independent accountants in connection with the accounts or books of Borrower, or any of its Subsidiaries, or any audit of any of them;
- (e) Promptly after the same are available, and in any event within five (5) Banking Days after filing with the Securities and Exchange Commission, copies of each annual report, proxy or financial statement or other report or communication sent to the stockholders of Borrower or any of its Subsidiaries, and copies of all annual, regular, periodic and special reports and registration statements which Borrower or any of its Subsidiaries may file or be required to file with the Securities and Exchange Commission under Section 13 or 15(d) of the Securities Exchange Act of 1934, as amended, and not otherwise required to be delivered to the Lenders pursuant to other provisions of this Section 7.1:
- (f) Promptly after request by Lender, subject to confidentiality requirements of any Governmental Agency, copies of any other report or other document that was filed by Borrower, with any Governmental Agency;
- (g) Promptly upon a Senior Officer of Borrower, becoming aware, and in any event within five (5) Banking Days after becoming aware, of the occurrence of any (i) "reportable event" (as such term is defined in Section 4043

of ERISA, but <u>excluding</u> such events as to which the PBGC has by regulation waived the requirement therein contained that it be notified within thirty days of the occurrence of such event) or (ii) non-exempt "prohibited transaction" (as such term is defined in Section 406 of ERISA or Section 4975 of the Code) involving any Pension Plan or any trust created thereunder, telephonic notice specifying the nature thereof, and, no more than two (2) Banking Days after such telephonic notice, written notice again specifying the nature thereof and specifying what action Borrower is taking or proposes to take with respect thereto, and, when known, any action taken by the Internal Revenue Service with respect thereto; <u>provided</u> that no such notice shall be required pursuant to this <u>Section 7.1(g)</u> if the anticipated liability is less than \$100,000;

- (h) As soon as practicable, and in any event within two (2) Banking Days after a Senior Officer of Borrower becomes aware of the existence of any condition or event which constitutes a Default or Event of Default, telephonic notice specifying the nature and period of existence thereof, and, no more than two (2) Banking Days after such telephonic notice, written notice again specifying the nature and period of existence thereof and specifying what action Borrower is taking or proposes to take with respect thereto;
- (i) Promptly upon a Senior Officer of Borrower becoming aware that (i) any Person has commenced a legal proceeding with respect to a claim against Borrower or any of its Subsidiaries that is \$250,000 or more in excess of the amount thereof that is fully covered by insurance, (ii) any creditor under a credit agreement involving Indebtedness of \$250,000 or more or any lessor under a lease involving aggregate rent of \$250,000 or more has asserted a default thereunder on the part of Borrower or any of its Subsidiaries or, (iii) any Person has commenced a legal proceeding with respect to a claim against Borrower or any of its Subsidiaries under a contract that is not a credit agreement or material lease with respect to a claim of in excess of \$250,000 or which otherwise may reasonably be expected to result in a Material Adverse Effect, a written notice describing the pertinent facts relating thereto and what action Borrower and/or its applicable Subsidiaries are taking or propose to take with respect thereto; and
- (j) Such other data and information as from time to time may be reasonably requested by the Administrative Agent or the Requisite Lenders.

Information required to be delivered pursuant to this <u>Section 7.1</u> (to the extent any such documents are included in materials otherwise filed with the Securities and Exchange Commission) shall be deemed to have been delivered on the date (a) on which Borrower provides notice to Lenders that such information has been posted on Borrower's Internet website at the website address listed on the signature page hereof or at another website identified in such notice and accessible to Lenders without charge, or (b) on which documents are posted on Borrower's behalf on the Platform; <u>provided</u> that (i) Company shall notify the Administrative Agent and each Lender (by telecopier or electronic mail) of the posting of any such documents and provide to Administrative Agent by electronic mail electronic versions (i.e., soft copies) of such documents.

7.2 <u>Compliance Certificates</u>. So long as any Advance remains unpaid, or any other Obligation remains unpaid or unperformed, or any portion of any of the Commitments remain outstanding, Borrower shall, at Borrower's sole expense, deliver to the Administrative Agent for distribution by it to the Lenders concurrently with the financial statements required pursuant to <u>Sections 7.1(a)</u> and <u>7.1(b)</u>, a Compliance Certificate signed by the president or chief financial officer of Borrower.

Article 8. CONDITIONS

- 8.1 <u>Closing Date Advances</u>. The obligation of each Closing Date Lender to make Advances on and after the Closing Date, and the obligation of the Issuing Lender to issue additional Letters of Credit (as applicable), is subject to the following conditions precedent, each of which shall be satisfied prior to the making of any further Advances or the issuance of the additional Letters of Credit (as applicable) (unless all of the Closing Date Lenders, in their sole and absolute discretion, shall agree otherwise):
 - (a) The Administrative Agent shall have received all of the following, each of which shall be originals unless otherwise specified, each properly executed by a Responsible Official of each party thereto, each dated as of the

Closing Date and each in form and substance satisfactory to the Administrative Agent and its legal counsel (unless otherwise specified or, in the case of the date of any of the following, unless the Administrative Agent otherwise agrees or directs):

- (1) at least one (1) executed counterpart of this Agreement, together with arrangements satisfactory to the Administrative Agent for additional executed counterparts, sufficient in number for distribution to the Closing Date Lenders and Borrower;
- (2) Notes executed by Borrower in favor of each Closing Date Lender, each in a principal amount equal to that Lender's Commitment;
 - (3) the Swing Line Documents executed by Borrower;
- (4) with respect to Borrower, such documentation as the Administrative Agent may reasonably require to establish the due organization, valid existence and good standing of Borrower, its qualification to engage in business in each material jurisdiction in which it is engaged in business or required to be so qualified, its authority to execute, deliver and perform the Loan Documents to which it is a Party, the identity, authority and capacity of each Responsible Official thereof authorized to act on its behalf, including certified copies of articles of incorporation and amendments thereto, bylaws and amendments thereto, certificates of good standing and/or qualification to engage in business, tax clearance certificates, certificates of corporate resolutions or other applicable authorization documents, incumbency certificates, Certificates of Responsible Officials, and the like;
 - (5) the Opinion of Counsel;
- (6) one or more Requests for Borrowing, Requests for Letters of Credit or Requests for Continuation/Conversion, as applicable;
- (7) a Certificate signed by a Senior Officer of Borrower certifying that the conditions specified in $\underbrace{\text{Sections } 8.1(\underline{d})}_{\text{Sections } 8.1(\underline{e})}$ have been satisfied; and
- (8) such other assurances, certificates, documents, consents or opinions as the Administrative Agent and/or any Closing Date Lender reasonably may require.
 - (b) The fees payable on or before the Closing Date pursuant to Section 3.3 shall have been paid.
- (c) The reasonable costs and expenses of the Administrative Agent in connection with the preparation of this Agreement and each of the other Loan Documents prepared in connection therewith payable pursuant to <u>Section 11.3</u>, and invoiced to Borrower prior to the Closing Date, shall have been paid.
 - (d) The representations and warranties of Borrower contained in Article 4 shall be true and correct in all material respects.
- (e) Borrower and any other Parties shall be in compliance with all the terms and provisions of the Loan Documents, and giving effect to the any Advances made, or Letters of Credits issued, on the Closing Date, no Default or Event of Default shall have occurred and be continuing.
- (f) All legal matters relating to the Loan Documents shall be reasonably satisfactory to Sheppard, Mullin, Richter & Hampton LLP, special counsel to Lender.
 - (g) The Closing Date shall have occurred on or before June 6, 2005.

- 8.2 <u>Any Advance</u>. The obligation of each Lender to make any Advance, and the obligation of the Issuing Lender to issue any Letter of Credit, is subject to the following conditions precedent (unless the Requisite Lenders or, in any case where the approval of all of the Lenders is required pursuant to <u>Section 11.2</u>, all of the Lenders, in their sole and absolute discretion, shall agree otherwise):
 - (a) except (i) for representations and warranties which expressly speak as of a particular date or are no longer true and correct as a result of a change which is permitted by this Agreement or (ii) as disclosed by Borrower and approved in writing by the Requisite Lenders, the representations and warranties contained in Article 4 (other than Sections 4.4, 4.6 (first sentence), 4.9 and 4.16) shall be true and correct in all material respects on and as of the date of the Advance or the Letter of Credit as though made on that date;
 - (b) no circumstance or event shall have occurred that constitutes a Material Adverse Effect since the Closing Date;
 - (c) <u>other</u> than matters described in <u>Schedule 4.9</u> or not required as of the Closing Date to be therein described, there shall not be then pending or threatened any action, suit, proceeding or investigation against or affecting Borrower or any Subsidiary of Borrower or any Property of any of them before any Governmental Agency that constitutes a Material Adverse Effect;
 - (d) the Administrative Agent shall have timely received a Request for Borrowing (or telephonic or other request for Borrowing referred to in the second sentence of Section 2.1(b), if applicable), or the Issuing Lender shall have received a Request for Letter of Credit, as the case may be, in compliance with Article 2; and
 - (e) the Administrative Agent shall have received, in form and substance reasonably satisfactory to the Administrative Agent, such other certificates, documents or consents related to the foregoing as the Administrative Agent or Requisite Lenders reasonably may require.

Article 9. EVENTS OF DEFAULT AND REMEDIES UPON EVENT OF DEFAULT

- 9.1 Events of Default. The existence or occurrence of any one or more of the following events, whatever the reason therefor and under any circumstances whatsoever, shall constitute an Event of Default:
 - (a) Borrower fails to pay any principal on any of the Notes, or any portion thereof, on the date when due; or
 - (b) Borrower fails to pay any interest on any of the Notes, or any fees under Sections 3.2 or 3.4, or any portion thereof, within five (5) Banking Days after the date when due; or fails to pay any other fee or amount payable to the Lenders or the Administrative Agent under any Loan Document, or any portion thereof, within five (5) Banking Days after written demand therefor; or
 - (c) Borrower fails to comply with, or cause or permit any of its Subsidiaries to fail to comply with, any of the covenants contained in Article 6; or
 - (d) (i) Borrower fails to comply with <u>Section 7.1(i)</u> in the manner stated therein or (ii) Borrower fails to perform any other reporting requirement set forth in <u>Article 7</u> within five (5) Banking Days of the date specified for performance therein; or
 - (e) Borrower or any other Party fails to perform or observe any other covenant or agreement (not specified in clause (a), (b), (c) or (d) above) contained in any Loan Document on its part to be performed or observed and such default shall continue unremedied for twenty (20) days after the giving of notice by the Administrative Agent on behalf of the Requisite Lenders of such Default; or
 - (f) Any representation or warranty of Borrower or any other Party made in any Loan Document, or in any certificate or other writing delivered by Borrower or such other Party pursuant to any Loan Document, proves to have been incorrect when made or reaffirmed in any material respect; or
 - (g) Borrower or any of its Subsidiaries (i) fails to pay the principal, or any principal installment, of any present or future Indebtedness of \$2,500,000 or more, or any guaranty of present or future Indebtedness of \$2,500,000 or more, on its part to be paid, when due (or within any stated grace period), whether at the stated maturity, upon acceleration, by reason of required prepayment or otherwise or (ii) fails to perform or observe any other term, covenant or agreement on its part to be performed or observed, or suffers any event of default to occur, in connection with any present or future Indebtedness of \$2,500,000 or more, or of any guaranty of present or future Indebtedness of \$2,500,000 or more, if as a result of such failure or sufferance any holder or holders thereof (or an agent or trustee on its or their behalf) has the right to declare such Indebtedness due before the date on which it otherwise would become due or the right to require Borrower or any such Subsidiary to redeem or purchase, or offer to redeem or purchase, all or any portion of such Indebtedness; or
 - (h) Any Loan Document, at any time after its execution and delivery and for any reason other than the agreement or action (or omission to act) of the Administrative Agent or satisfaction in full of all the Obligations, ceases to be in full force and effect or is declared by a court of competent jurisdiction to be null and void, invalid or unenforceable in any respect which is materially adverse to the interests of the Lenders; or any Party thereto denies in writing that it has any or further liability or obligation under any Loan Document, or purports to revoke, terminate or rescind same; or
 - (i) A final judgment against Borrower or any of its Subsidiaries is entered for the payment of money in excess of \$2,500,000 (not covered by insurance or for which an insurer has reserved its rights) and, absent procurement of a stay of execution, such judgment remains unsatisfied for thirty (30) calendar days after the date of entry of judgment, or in any event later than five (5) days prior to the date of any proposed sale thereunder; or any writ or warrant of attachment or execution or similar process is issued or levied against all or any material part of the Property of Borrower or any of its Subsidiaries and is not released, vacated or fully bonded within thirty (30) calendar days after its issue or levy; or

(j) Borrower or any of its Subsidiaries institutes or consents to the institution of any proceeding under a Debtor Relief Law relating to it or to all or any material part of its Property, or is unable or admits in writing its inability to pay its debts as they mature, or makes an assignment for the benefit of creditors; or applies for or consents to the appointment of any receiver, trustee, custodian, conservator, liquidator, rehabilitator or similar officer for it or for all or any material part of its Property; or any receiver, trustee, custodian, conservator, liquidator, rehabilitator or similar officer is appointed without the application or consent of that Person and the appointment continues undischarged or unstayed for sixty (60) calendar days; or any proceeding under a Debtor Relief Law relating to any such Person or to all or any part of its Property is instituted without the consent of that Person and continues undismissed or unstayed for sixty (60) calendar days; or

- (k) A Change in Control occurs; or
- (l) The dissolution or liquidation of Borrower, any of SCW or Chapparal City Water Company, or any other Subsidiary that has, immediately prior to the commencement of such dissolution or liquidation, assets having a fair market value of more than \$20,000,000 or Borrower or any such Subsidiary, or any of their respective partners, members, directors or stockholders, as the case may be, shall take action seeking to effect the dissolution or liquidation of Borrower or such Subsidiary; or
- (m) The occurrence of an Event of Default (as such term is or may hereafter be specifically defined in any other Loan Document) under any other Loan Document; or
- (n) Any Pension Plan maintained by Borrower is finally determined by the PBGC to have a material "accumulated funding deficiency" as that term is defined in Section 302 of ERISA in excess of an amount equal to 5% of the consolidated total assets of Borrower as of the most-recently ended Fiscal Quarter; or
- (o) Any holder of a Subordinated Obligation asserts in writing that such Subordinated Obligation is not subordinated to the Obligations in accordance with its terms and Borrower does not promptly deny in writing such assertion and contest any attempt by such holder to take action based on such assertion; or
- (p) Any event occurs which gives the holder or holders of any Subordinated Obligation (or an agent or trustee on its or their behalf) the right to declare such Subordinated Obligation due before the date on which it otherwise would become due, or the right (other than by reason of a Change in Control) to require the issuer thereof, to redeem or purchase, or offer to redeem or purchase, all or any portion of any Subordinated Obligation, or a final judgment is entered by a court of competent jurisdiction that any Subordinated Obligation is not subordinated in accordance with its terms to the Obligations.
- 9.2 <u>Remedies Upon Event of Default.</u> Without limiting any other rights or remedies of the Administrative Agent or the Lenders provided for elsewhere in this Agreement, or the other Loan Documents, or by applicable Law, or in equity, or otherwise:
 - (a) Upon the occurrence, and during the continuance, of any Event of Default other than an Event of Default described in Section 9.1(j):
 - (1) the commitments to make Advances and all other obligations of the Administrative Agent or the Lenders and all rights of Borrower and any other Parties under the Loan Documents shall be suspended without notice to or demand upon Borrower, which are expressly waived by Borrower except that all of the Lenders or the Requisite Lenders (as the case may be, in accordance with Section 11.2) may waive an Event of Default or, without waiving, determine, upon terms and conditions satisfactory to the Lenders or Requisite Lenders, as the case may be, to reinstate the Commitments and such other obligations and rights and make further Advances, which waiver or determination shall apply equally to, and shall be binding upon, all the Lenders;
 - (2) the Issuing Lender may demand immediate payment by Borrower of an amount equal to the Aggregate Effective Amount of all outstanding Letters of Credit to be held by the Administrative Agent, on behalf of the Lenders, in an interest-bearing cash collateral account as collateral for all of the Obligations; and
 - (3) the Requisite Lenders may request the Administrative Agent to, and the Administrative Agent thereupon shall, terminate the Commitments and/or declare all or any part of the unpaid principal of all Notes, all interest accrued and unpaid thereon and all other amounts payable under the Loan Documents to be forthwith due

and payable, whereupon the same shall become and be forthwith due and payable, without protest, presentment, notice of dishonor, demand or further notice of any kind, all of which are expressly waived by Borrower.

(b) Upon the occurrence of any Event of Default described in Section 9.1(j):

(1) the Commitments shall terminate without notice to or demand upon Borrower, which are expressly waived by Borrower, except that all of the Lenders may waive the Event of Default or, without waiving, determine, upon terms and conditions satisfactory to all the Lenders, to reinstate the Commitments and make further Advances, which determination shall apply equally to, and shall be binding upon, all the Lenders;

- (2) an amount equal to the Aggregate Effective Amount of all outstanding Letters of Credit shall be immediately due and payable to the Issuing Lender without notice to or demand upon Borrower, which are expressly waived by Borrower, to be held by the Administrative Agent, on behalf of the Lenders, in an interest-bearing cash collateral account as collateral for all of the Obligations; and
- (3) the unpaid principal of all Notes, all interest accrued and unpaid thereon and all other amounts payable under the Loan Documents shall be forthwith due and payable, without protest, presentment, notice of dishonor, demand or further notice of any kind, all of which are expressly waived by Borrower.
- (c) Upon the occurrence of any Event of Default, the Lenders and the Administrative Agent, or any of them, without notice to (except as expressly provided for in any Loan Document) or demand upon Borrower, which are expressly waived by Borrower (except as to notices expressly provided for in any Loan Document), may proceed (but only with the consent of the Requisite Lenders) to protect, exercise and enforce their rights and remedies under the Loan Documents against Borrower and any other Party and such other rights and remedies as are provided by Law or equity.
- (d) The order and manner in which the Lenders' rights and remedies are to be exercised shall be determined by the Requisite Lenders in their sole discretion, and all payments received by the Administrative Agent and the Lenders, or any of them, shall be applied first to the costs and expenses (including reasonable attorneys' fees and disbursements and the reasonably allocated costs of attorneys employed by the Administrative Agent or by any Lender) of the Administrative Agent and of the Lenders, and thereafter paid pro rata to the Lenders in the same proportions that the aggregate Obligations owed to each Lender under the Loan Documents bear to the aggregate Obligations owed under the Loan Documents to all the Lenders, without priority or preference among the Lenders. Regardless of how each Lender may treat payments for the purpose of its own accounting, for the purpose of computing Borrower's Obligations hereunder and under the Notes, payments shall be applied first, to the costs and expenses of the Administrative Agent and the Lenders, as set forth above, second, to the payment of accrued and unpaid interest due under any Loan Documents to and including the date of such application (ratably, and without duplication, according to the accrued and unpaid interest due under each of the Loan Documents), and third, to the payment of all other amounts (including principal and fees) then owing to the Administrative Agent or the Lenders under the Loan Documents. No application of payments will cure any Event of Default, or prevent acceleration, or continued acceleration, of amounts payable under the Loan Documents, or prevent the exercise, or continued exercise, of rights or remedies of the Lenders hereunder or thereunder or at Law or in equity.

Article 10. THE ADMINISTRATIVE AGENT

- Administrative Agent to take such action as agent on its behalf and to exercise such powers under the Loan Documents as are delegated to the Administrative Agent by the terms thereof or are reasonably incidental, as determined by the Administrative Agent, thereto. This appointment and authorization is intended solely for the purpose of facilitating the servicing of the Revolving Facility and does not constitute appointment of the Administrative Agent as trustee for any Lender or as representative of any Lender for any other purpose and, except as specifically set forth in the Loan Documents to the contrary, the Administrative Agent shall take such action and exercise such powers only in an administrative and ministerial capacity.
- 10.2 <u>Administrative Agent and Affiliates</u>. Wells Fargo (and each successor Administrative Agent) has the same rights and powers under the Loan Documents as any other Lender and may exercise the same as though it were not the Administrative Agent, and the term "Lender" or "Lenders" includes Wells Fargo in its individual capacity. Wells Fargo (and each successor Administrative Agent) and its Affiliates may accept deposits from, lend money to and generally engage in any kind of banking, trust or other business with Borrower, any Subsidiary thereof, or any Affiliate of Borrower or any Subsidiary thereof, as if it were not the Administrative Agent and without any duty to account therefor to the Lenders. Wells Fargo (and each successor Administrative Agent) need not account to any other Lender for any monies received by it for reimbursement of its costs and expenses as Administrative Agent hereunder, or (subject to Section 11.10) for any monies received by it in its capacity as a Lender hereunder. The

Administrative Agent shall not be deemed to hold a fiduciary relationship with any Lender and no implied covenants, functions, responsibilities, duties, obligations or liabilities shall be read into this Agreement or otherwise exist against the Administrative Agent.

- 10.3 Proportionate Interest in any Collateral. The Administrative Agent, on behalf of all the Lenders, shall hold in accordance with the Loan Documents all items of collateral (if any) or interests therein received or held by the Administrative Agent. Subject to the Administrative Agent's and the Lenders' rights to reimbursement for their costs and expenses hereunder (including reasonable attorneys' fees and disbursements and other professional services and the reasonably allocated costs of attorneys employed by the Administrative Agent or a Lender) and subject to the application of payments in accordance with Section 9.2(d), each Lender shall have an interest in the Lenders' interest in such collateral or interests therein in the same proportions that the aggregate Obligations owed such Lender under the Loan Documents bear to the aggregate Obligations owed under the Loan Documents to all the Lenders, without priority or preference among the Lenders.
- 10.4 <u>Lenders' Credit Decisions</u>. Each Lender agrees that it has, independently and without reliance upon the Administrative Agent, any other Lender or the directors, officers, agents, employees or attorneys of the Administrative Agent or of any other Lender, and instead in reliance upon information supplied to it by or on behalf of Borrower and upon such other information as it has deemed appropriate, made its own independent credit analysis and decision to enter into this Agreement. Each Lender also agrees that it shall, independently and without reliance upon the Administrative Agent, any other Lender or the directors, officers, agents, employees or attorneys of the Administrative Agent or of any other Lender, continue to make its own independent credit analyses and decisions in acting or not acting under the Loan Documents.

10.5 Action by Administrative Agent.

- (a) The Administrative Agent may assume that no Default has occurred and is continuing, unless the Administrative Agent (or the Lender that is then the Administrative Agent) has received notice from Borrower stating the nature of the Default or has received notice from a Lender stating the nature of the Default and that such Lender considers the Default to have occurred and to be continuing.
 - (b) The Administrative Agent has only those obligations under the Loan Documents as are expressly set forth therein.
- (c) Except for any obligation expressly set forth in the Loan Documents and as long as the Administrative Agent may assume that no Event of Default has occurred and is continuing, the Administrative Agent may, but shall not be required to, exercise its discretion to act or not act, except that the Administrative Agent shall be required to act or not act upon the instructions of the Requisite Lenders (or of all the Lenders, to the extent required by Section 11.2) and those instructions shall be binding upon the Administrative Agent and all the Lenders, provided that the Administrative Agent shall not be required to act or not act if to do so would be contrary to any Loan Document or to applicable Law or would result, in the reasonable judgment of the Administrative Agent, in substantial risk of liability to the Administrative Agent.
- (d) If the Administrative Agent has received a notice specified in <u>clause (a)</u>, the Administrative Agent shall immediately give notice thereof to the Lenders and shall act or not act upon the instructions of the Requisite Lenders (or of all the Lenders, to the extent required by <u>Section 11.2</u>), <u>provided</u> that the Administrative Agent shall not be required to act or not act if to do so would be contrary to any Loan Document or to applicable Law or would result, in the reasonable judgment of the Administrative Agent, in substantial risk of liability to the Administrative Agent, and <u>except</u> that if the Requisite Lenders (or all the Lenders, if required under <u>Section 11.2</u>) fail, for five (5) Banking Days after the receipt of notice from the Administrative Agent, to instruct the Administrative Agent, then the Administrative Agent, in its sole discretion, may act or not act as it deems advisable for the protection of the interests of the Lenders.

- (e) The Administrative Agent shall have no liability to any Lender for acting, or not acting, as instructed by the Requisite Lenders (or all the Lenders, if required under Section 11.2), notwithstanding any other provision hereof.
- 10.6 <u>Liability of Administrative Agent</u>. Neither the Administrative Agent nor any of its directors, officers, agents, employees or attorneys shall be liable for any action taken or not taken by them under or in connection with the Loan Documents, <u>except</u> for their own gross negligence or willful misconduct. Without limitation on the foregoing, the Administrative Agent and its directors, officers, agents, employees and attorneys:
 - (a) May treat the payee of any Note as the holder thereof until the Administrative Agent receives notice of the assignment or transfer thereof, in form satisfactory to the Administrative Agent, signed by the payee, and may treat each Lender as the owner of that Lender's interest in the Obligations for all purposes of this Agreement until the Administrative Agent receives notice of the assignment or transfer thereof, in form satisfactory to the Administrative Agent, signed by that Lender;
 - (b) May consult with legal counsel (<u>including</u> in-house legal counsel), accountants (including in-house accountants) and other professionals or experts selected by it, or with legal counsel, accountants or other professionals or experts for Borrower or any Subsidiary of Borrower and/or any of their Affiliates or the Lenders, and shall not be liable for any action taken or not taken by it in good faith in accordance with any advice of such legal counsel, accountants or other professionals or experts;
 - (c) Shall not be responsible to any Lender for any statement, warranty or representation made in any of the Loan Documents or in any notice, certificate, report, request or other statement (written or oral) given or made in connection with any of the Loan Documents;
 - (d) <u>Except</u> to the extent expressly set forth in the Loan Documents, shall have no duty to ask or inquire as to the performance or observance by Borrower of any of the terms, conditions or covenants of any of the Loan Documents or to inspect any Property, books or records of Borrower or any Subsidiary of Borrower;
 - (e) Will not be responsible to any Lender for the due execution, legality, validity, enforceability, genuineness, effectiveness, sufficiency or value of any Loan Document, any other instrument or writing furnished pursuant thereto or in connection therewith;
 - (f) Will not incur any liability by acting or not acting in reliance upon any Loan Document, notice, consent, certificate, statement, request or other instrument or writing believed in good faith by it to be genuine and signed or sent by the proper party or parties; and
 - (g) Will not incur any liability for any arithmetical error in computing any amount paid or payable by Borrower or paid or payable to or received or receivable from any Lender under any Loan Document, including principal, interest, commitment fees, Advances and other amounts; provided that, promptly upon discovery of such an error in computation, the Administrative Agent, the Lenders and (to the extent applicable) Borrower shall make such adjustments as are necessary to correct such error and to restore the parties to the position that they would have occupied had the error not occurred.
- 10.7 <u>Indemnification</u>. Each Lender shall, ratably in accordance with its Pro Rata Share of all of the then applicable Commitments (if any of the Commitments are then in effect) and/or in accordance with its proportion of the aggregate Indebtedness then evidenced by the Notes (if all of the Commitments have then been terminated), indemnify and hold the Administrative Agent and its directors, officers, agents, employees and attorneys harmless against any and all liabilities, obligations, losses, damages, penalties, actions, judgments, suits, costs, expenses or disbursements of any kind or nature whatsoever (<u>including</u>

reasonable attorneys' fees and disbursements and allocated costs of attorneys employed by the Administrative Agent) that may be imposed on, incurred by or asserted against it or them in any way relating to or arising out of the Loan Documents (other than losses incurred by reason of the failure of Borrower to pay the Indebtedness represented by the Notes) or any action taken or not taken by it as Administrative Agent thereunder, except such as result from its own gross negligence or willful misconduct. Without limitation on the foregoing, each Lender shall reimburse the Administrative Agent upon demand for that Lender's Pro Rata Share of any out-of-pocket cost or expense incurred by the Administrative Agent in connection with the negotiation, preparation, execution, delivery, amendment, waiver, restructuring, reorganization (including a bankruptcy reorganization), enforcement or attempted enforcement of the Loan Documents, to the extent that Borrower or any other Party are required by Section 11.3 to pay that cost or expense but fails to do so upon demand. Nothing in this Section 10.7 shall entitle the Administrative Agent or any indemnitee referred to above to recover any amount from the Lenders if and to the extent that such amount has theretofore been recovered from Borrower. To the extent that the Administrative Agent or any indemnitee referred to above is later reimbursed such amount by Borrower, it shall return the amounts paid to it by the Lenders in respect of such amount.

Administrative Agent upon reasonable notice to the Lenders and Borrower effective upon acceptance of appointment by a successor Administrative Agent. If the Administrative Agent shall resign as Administrative Agent under this Agreement, the Requisite Lenders shall appoint from among the Lenders a successor Administrative Agent for the Lenders, which successor Administrative Agent shall be approved by Borrower (and such approval shall not be unreasonably withheld or delayed). If no successor Administrative Agent is appointed prior to the effective date of the resignation of the Administrative Agent, the Administrative Agent may appoint, after consulting with the Lenders and Borrower, a successor Administrative Agent from among the Lenders. Upon the acceptance of its appointment as successor Administrative Agent hereunder, such successor Administrative Agent shall succeed to all the rights, powers and duties of the retiring Administrative Agent and the term "Administrative Agent" shall mean such successor Administrative Agent and the retiring Administrative Agent and the term "Administrative Agent shall be terminated. After any retiring Administrative Agent's resignation hereunder as Administrative Agent, the provisions of this Article 10, and Sections 11.3, 11.11 and 11.21, shall inure to its benefit as to any actions taken or omitted to be taken by it while it was Administrative Agent under this Agreement. Notwithstanding the foregoing, if no successor Administrative Agent has accepted appointment as Administrative Agent by the date which is thirty (30) days following a retiring Administrative Agent's notice of resignation, the retiring Administrative Agent's resignation shall nevertheless thereupon become effective and the Lenders shall perform all of the duties of the Administrative Agent hereunder until such time, if any, as the Requisite Lenders appoint a successor Administrative Agent as provided for above.

10.9 No Obligations of Borrower. Nothing contained in this Article 10 shall be deemed to impose upon Borrower any obligation in respect of the due and punctual performance by the Administrative Agent of its obligations to the Lenders under any provision of this Agreement, and Borrower shall have no liability to the Administrative Agent or any of the Lenders in respect of any failure by the Administrative Agent or any Lender to perform any of its obligations to the Administrative Agent or the Lenders under this Agreement. Without limiting the generality of the foregoing, where any provision of this Agreement relating to the payment of any amounts due and owing under the Loan Documents provides that such payments shall be made by Borrower to the Administrative Agent for the account of the Lenders, Borrower's obligations to the Lenders in respect of such payments shall be deemed to be satisfied upon the making of such payments to the Administrative Agent in the manner provided by this Agreement. In addition, Borrower may rely on a written statement by the Administrative Agent to the effect that it has obtained the written consent of the Requisite Lenders or all of the Lenders, as applicable under Section 11.2, in connection with a waiver, amendment, consent, approval or other action by the Lenders hereunder, and shall have no obligation to verify or confirm the same.

Article 11. MISCELLANEOUS

- 11.1 <u>Cumulative Remedies; No Waiver</u>. The rights, powers, privileges and remedies of the Administrative Agent and the Lenders provided herein or in any Note or other Loan Document are cumulative and not exclusive of any right, power, privilege or remedy provided by Law or equity. No failure or delay on the part of the Administrative Agent or any Lender in exercising any right, power, privilege or remedy may be, or may be deemed to be, a waiver thereof; nor may any single or partial exercise of any right, power, privilege or remedy preclude any other or further exercise of the same or any other right, power, privilege or remedy. The terms and conditions of <u>Article 8</u> hereof are inserted for the sole benefit of the Administrative Agent and the Lenders; the same may be waived in whole or in part, with or without terms or conditions, in respect of any Borrowing without prejudicing the Administrative Agent's or the Lenders' rights to assert them in whole or in part in respect of any other Borrowing.
- Amendments; Consents. No amendment, modification, supplement, extension, termination or waiver of any provision of this Agreement or any other Loan Document, no approval or consent thereunder, and no consent to any departure by Borrower or any other Party therefrom, may in any event be effective unless in writing signed by the Administrative Agent with the written approval of the Requisite Lenders (and, in the case of any amendment, modification or supplement of or to any Loan Document to which Borrower is a Party, signed by Borrower, and, in the case of any amendment, modification or supplement to Article 10, signed by the Administrative Agent), and then only in the specific instance and for the specific purpose given; and, without the approval in writing of all the Lenders, no amendment, modification, supplement, termination, waiver or consent may be effective:
 - (a) To amend or modify the principal of, or the amount of principal, principal prepayments or the rate of interest payable on, any Note, or the amount of the Revolving Facility, or the Pro Rata Share of any Lender or the amount of any commitment fee payable to any Lender, or any other fee or amount payable to any Lender (in its capacity as a Lender) under the Loan Documents or to waive an Event of Default consisting of the failure of Borrower to pay when due principal, interest or any fee, or to provide for additional extensions of credit to Borrower by the Lenders pursuant to the Loan Documents;
 - (b) To postpone any date fixed for any payment of principal of, prepayment of principal of or any installment of interest on, any Note or any installment of any fee, or to extend the term of the Revolving Facility;
 - (c) To amend the provisions of the definition of "Requisite Lenders" or "Maturity Date";
 - (d) To amend or waive <u>Article 8</u> or this <u>Section 11.2</u>; or
 - (e) To amend any provision of this Agreement that expressly requires the consent or approval of all the Lenders.

Any amendment, modification, supplement, termination, waiver or consent pursuant to this <u>Section 11.2</u> shall apply equally to, and shall be binding upon, all the Lenders and the Administrative Agent.

- 11.3 <u>Costs and Expenses</u>. Borrower agrees to pay within five (5) Banking Days after demand, accompanied by an invoice therefor, all reasonable, out-of-pocket expenses (<u>except</u> in the case of the Administrative Agent's allocated in-house counsel costs described below, which shall not be required to be "out-of-pocket") of the Administrative Agent (<u>including</u> the reasonable fees and out-of-pocket expenses of counsel to the Administrative Agent (including reasonable allocated costs of in-house counsel employed by the Administrative Agent) and of local counsel, if any, who may be retained by counsel to the Administrative Agent) in connection with:
 - (a) The negotiation, preparation, execution and delivery of this Agreement and of each other Loan Document, including schedules and exhibits, and any amendments, waivers, consents, supplements or other modifications to this Agreement or any other Loan Document as may from time to time hereafter be required, whether or not the transactions contemplated hereby (or thereby) are consummated, provided that, Borrower's maximum liability for fees and expenses of counsel to the Administrative Agent in connection with the Closing Date shall not exceed \$50,000;
 - (b) The preparation and review of the form of any document or instrument relevant to this Agreement or any other Loan Document; and
 - (c) The preparation of any information or response required with respect to any investigative request or inquiry, approval, findings of suitability or any other response or communication involving a Governmental Agency arising out of this Agreement, any other Loan Document or any Obligation evidenced by the Loan Documents or the participation in any public or investigatory hearing or meeting.

Borrower further agrees to pay, and to save the Administrative Agent, the Issuing Lender and the Lenders harmless from all liability for, any stamp and similar taxes that may be payable in connection with the execution or delivery of this Agreement, the credit extensions made hereunder, or the issuance of the Notes, the Letters of Credit or any other Loan Document. Borrower also agrees to reimburse the Administrative Agent and, after the occurrence and during the continuance of an Event of Default, the Issuing Lender and each Lender upon demand for all reasonable out-of-pocket expenses (<u>including</u> reasonable attorneys' fees and legal expenses of counsel and fees and expenses of consultants to the Administrative Agent, the Issuing Lender and the Lenders) incurred by the Administrative Agent, the Issuing Lender or such Lenders in connection with (i) the negotiation of any restructuring or "work-out" with Borrower whether or not consummated, of any Obligations, (ii) the enforcement or attempted enforcement of any Obligations and any matter related thereto and (iii) any bankruptcy of Borrower or any of its Subsidiaries. Any amount payable to the Administrative Agent or any Lender under this Section 11.3 shall bear interest from the fifth Banking Day following the date of demand, if not then paid, for payment at the Default Rate.

- 11.4 <u>Nature of Lenders' Obligations</u>. The obligations of the Lenders hereunder are several and not joint or joint and several. Nothing contained in this Agreement or any other Loan Document and no action taken by the Administrative Agent or the Lenders or any of them pursuant hereto or thereto may, or may be deemed to, make the Lenders a partnership, an association, a joint venture or other entity, either among themselves or with Borrower or any Subsidiary or Affiliate of Borrower. A default by any Lender will not increase the Commitment of any other Lender or the Pro Rata Share of the Revolving Facility attributable to any other Lender. Any Lender not in default may, if it desires, assume (in such proportion as the nondefaulting Lenders agree) the obligations of any Lender in default, but no Lender is obligated to do so.
- 11.5 <u>Survival of Representations and Warranties.</u> All representations and warranties contained herein or in any other Loan Document, or in any certificate or other writing delivered by or on behalf of any one or more of the Parties to any Loan Document, will survive the making of the Advances hereunder and the execution and delivery of the Notes, and have been or will be relied upon by the Administrative Agent and each Lender, notwithstanding any investigation made by the Administrative Agent or any Lender or on their behalf.

- 11.6 Notices. (a) Except as otherwise expressly provided in the Loan Documents, all notices, requests, demands, directions and other communications provided for hereunder or under any other Loan Document must be in writing and must be mailed, telecopied, dispatched by commercial courier or delivered to the appropriate party at the address set forth on the signature pages of this Agreement or other applicable Loan Document or, as to any party to any Loan Document, at any other address as may be designated by it in a written notice sent to all other parties to such Loan Document in accordance with this Section. Except as otherwise expressly provided in any Loan Document, if any notice, request, demand, direction or other communication required or permitted by any Loan Document is given by mail it will be effective on the earlier of receipt or the fourth Banking Day after deposit in the United States mail with first class or airmail postage prepaid; if given by telecopier, when sent; if dispatched by commercial courier, on the scheduled delivery date; or if given by personal delivery, when delivered.
- (b) Notwithstanding the foregoing, Borrower agrees that Administrative Agent may make any material delivered by Borrower to Administrative Agent, as well as any amendments, waivers, consents and other written information, documents, instruments and other materials relating to Borrower, any of its Subsidiaries, or any other materials or matters relating to the Loan Documents or any of the transactions contemplated hereby that Administrative Agent is required or authorized pursuant to the terms hereof or of any Loan Document to provide to Lenders (collectively, the "Communications"), available to Lenders by posting such notices on a Platform. Borrower acknowledges that (i) the distribution of material through an electronic medium is not necessarily secure and that there are confidentiality and other risks associated with such distribution, (ii) a Platform is provided "as is" and "as available" and (iii) neither Administrative Agent nor any of its Affiliates warrants the accuracy, completeness, timeliness, sufficiency, or sequencing of the Communications posted on a Platform. Administrative Agent and its Affiliates expressly disclaim with respect to a Platform any liability for errors in transmission, incorrect or incomplete downloading, delays in posting or delivery, or problems accessing the Communications posted on such Platform and any liability for any losses, costs, expenses or liabilities that may be suffered or incurred in connection with such Platform. No warranty of any kind, express, implied or statutory, including, without limitation, any warranty of merchantability, fitness for a particular purpose, non-infringement of third party rights or freedom from viruses or other code defects, is made by Administrative Agent or any of its Affiliates in connection with any Platform.
- (c) Each Lender agrees that notice to it (as provided in the next sentence) specifying that any Communication has been posted to a Platform shall for purposes of this Agreement constitute effective delivery to such Lender of such information, documents or other materials comprising such Communication. Each Lender agrees (i) to notify, on or before the date such Lender becomes a party to this Agreement, Administrative Agent in writing of such Lender's e-mail address to which a notice may be sent (and from time to time thereafter to ensure that Administrative Agent has on record an effective e-mail address for such Lender) and (ii) that any notice may be sent to such e-mail address.
- 11.7 Execution of Loan Documents. Unless the Administrative Agent otherwise specifies with respect to any Loan Document, (a) this Agreement and any other Loan Document may be executed in any number of counterparts and any party hereto or thereto may execute any counterpart, each of which when executed and delivered will be deemed to be an original and all of which counterparts of this Agreement or any other Loan Document, as the case may be, when taken together will be deemed to be but one and the same instrument and (b) execution of any such counterpart may be evidenced by a telecopier transmission of the signature of such party. The execution of this Agreement or any other Loan Document by any party hereto or thereto will not become effective until counterparts hereof or thereof, as the case may be, have been executed by all the parties hereto or thereto.

11.8 Binding Effect; Assignment.

(a) This Agreement and the other Loan Documents to which Borrower is a Party will be binding upon and inure to the benefit of Borrower, the Administrative Agent, each of the Lenders, and their respective successors and assigns, except that Borrower may not assign its rights hereunder or thereunder or any interest herein or therein without the prior written consent of all the Lenders. Each Lender represents that it is not acquiring its Note with a view to the distribution thereof within the meaning of the Securities Act of 1933, as amended (subject to any requirement that disposition of such Note must be within the control of such Lender). Any Lender may at any time pledge its Note or any other instrument

evidencing its rights as a Lender under this Agreement to a Federal Reserve Bank, but no such pledge shall release that Lender from its obligations hereunder or grant to such Federal Reserve Bank the rights of a Lender hereunder absent foreclosure of such pledge.

- (b) From time to time following the Closing Date, each Lender may assign to one or more Eligible Assignees all or any portion of its rights and obligations under this Agreement (including all or a portion of its Commitment, the Advances owing to it and the Note or Notes held by it); provided that, subject to subsection (f) below, (i) such Eligible Assignee, if not then a Lender or an Affiliate of the assigning Lender, shall be approved by the Administrative Agent and Borrower (neither of which approvals shall be unreasonably withheld or delayed), (ii) such assignment shall be evidenced by an Assignment and Acceptance, a copy of which shall be furnished to the Administrative Agent as hereinbelow provided, (iii) except in the case of an assignment to an Affiliate of the assigning Lender, to another Lender or of the entire remaining rights and obligations of the assigning Lender under this Agreement, the assignment shall not assign a portion of such assigning Lender's Commitments and/or Advances owing to such assigning Lender that is equivalent to less than \$3,000,000, and (iv) the effective date of any such assignment shall be as specified in the Assignment and Acceptance, but not earlier than the date which is five (5) Banking Days after the date the Administrative Agent has received the Assignment and Acceptance. Upon the effective date of such Assignment and Acceptance, the Eligible Assignee named therein shall be a Lender for all purposes of this Agreement, with the Commitments and/or Advances therein set forth and, to the extent of such Commitments and/or Advances, the assigning Lender shall be released from its further obligations under this Agreement. Borrower agrees that it shall execute and deliver (against delivery by the assigning Lender to Borrower of such Lender's Notes) to such assignee Lender, Notes evidencing the remaining balance of the Commitments and/or Advances retained by the assigning Lender.
- (c) By executing and delivering an Assignment and Acceptance, the Eligible Assignee thereunder acknowledges and agrees that: (i) other than the representation and warranty that it is the legal and beneficial owner of the rights and obligations hereunder being assigned thereby free and clear of any adverse claim, the assigning Lender has made no representation or warranty and assumes no responsibility with respect to any statements, warranties or representations made in or in connection with this Agreement or the execution, legality, validity, enforceability, genuineness or sufficiency of this Agreement or any other Loan Document; (ii) the assigning Lender has made no representation or warranty and assumes no responsibility with respect to the financial condition of Borrower or the performance by Borrower of the Obligations; (iii) it has received a copy of this Agreement, together with copies of the most recent financial statements delivered pursuant to Section 7.1 and such other documents and information as it has deemed appropriate to make its own credit analysis and decision to enter into such Assignment and Acceptance; (iv) it will, independently and without reliance upon the Administrative Agent or any Lender and based on such documents and information as it shall deem appropriate at the time, continue to make its own credit decisions in taking or not taking action under this Agreement; (v) it appoints and authorizes the Administrative Agent to take such action and to exercise such powers under this Agreement as are delegated to the Administrative Agent by this Agreement; and (vi) it will perform in accordance with their terms all of the obligations which by the terms of this Agreement are required to be performed by it as a Lender.
- (d) The Administrative Agent shall maintain at the Administrative Agent's Office a copy of each Assignment and Acceptance delivered to it and a register (the "Register") of the names and address of each of the Lenders and the Pro Rata Share of the Commitments held by each Lender, giving effect to each Assignment and Acceptance. The Register shall be available during normal business hours for inspection by Borrower or any Lender upon reasonable prior notice to the Administrative Agent. After receipt of a completed Assignment and Acceptance executed by any Lender and an Eligible Assignee, and receipt of an assignment fee of \$3,500 from such Lender or Eligible Assignee, the Administrative Agent shall, promptly following the effective date thereof, provide to Borrower and the Lenders a revised Schedule 1.1 giving effect thereto. Borrower, the Administrative Agent and the Lenders shall deem and treat the Persons listed as Lenders in the Register as the holders and owners of the Pro Rata Shares of the Revolving Facility listed therein for all purposes hereof, and no assignment or transfer of any Lender's rights and obligations hereunder shall be effective, in each case unless

and until an Assignment and Acceptance effecting the assignment or transfer thereof shall have been accepted by the Administrative Agent and recorded in the Register as provided above. Prior to such recordation, all amounts owed with respect to the applicable Pro Rata Share of the Revolving Facility shall be owed to the Lender listed in the Register as the owner thereof, and any request, authority or consent of any Person who, at the time of making such request or giving such authority or consent, is listed in the Register as a Lender shall be conclusive and binding on any subsequent holder, assignee or transferee of the corresponding Pro Rata Share of the Revolving Facility.

- (e) Each Lender may from time to time grant participations to one or more banks or other financial institutions in or to all or a portion of its rights and/or obligations under this Agreement; provided, however, that (i) such Lender's obligations under this Agreement shall remain unchanged, (ii) such Lender shall remain solely responsible to the other parties hereto for the performance of such obligations, (iii) the participating banks or other financial institutions shall not be a Lender hereunder for any purpose except, if the participation agreement so provides, for the purposes of Sections 3.5, 3.6, 11.11 and 11.21 but only to the extent that the cost of such benefits to Borrower does not exceed the cost which Borrower would have incurred in respect of the Lender granting such participation absent the participation, (iv) Borrower, the Administrative Agent and the other Lenders shall continue to deal solely and directly with such Lender in connection with such Lender's rights and obligations under this Agreement, (v) the participation interest shall be expressed as a percentage of the granting Lender's Pro Rata Share of the Revolving Facility as it then exists and shall not restrict an increase in the Revolving Facility (or the aggregate Commitments pertaining thereto), or in the granting Lender's rights and obligations hereunder, so long as the amount of the participation interest is not affected thereby and (vi) the consent of the holder of such participation interest shall not be required for amendments or waivers of provisions of the Loan Documents other than those which (A) extend any Amortization Date, any applicable Maturity Date or any other date upon which any payment of money is due to the Lenders, (B) reduce the rate of interest on the Notes, any fee or any other monetary amount payable to the Lenders, (C) reduce the amount of any installment of principal due under the Notes, or (D) release any Guarantor from its Guaranty.
- (f) Borrower agrees that upon the occurrence and during the continuance of any Event of Default, each Lender shall be entitled to assign its rights hereunder and under the Loan Documents, or grant participation interests in its rights under this Agreement and the Loan Documents, to any Person, in whole or in any part thereof, notwithstanding any provisions contained herein (including those set forth in subsection (b) above) or in any other Loan Document to the contrary, except that, other than (i) assignments by a Lender to an Affiliate of such Lender or to another Lender or (ii) pledges described in the last sentence of subsection (a) above, no assignment shall be made without the approval of the Administrative Agent.
- 11.9 <u>Right of Setoff</u>. If an Event of Default has occurred and is continuing, the Administrative Agent or any Lender (but in each case only with the consent of the Requisite Lenders) may exercise its rights under applicable Laws and, to the extent permitted by applicable Laws, apply any funds in any deposit account maintained with it by Borrower and/or any Property of Borrower in its possession against the Obligations.

11.10 Sharing of Setoffs. Each Lender severally agrees that if it, through the exercise of any right of setoff, banker's lien or counterclaim against Borrower, or otherwise, receives payment of the Obligations held by it that is ratably more than any other Lender, through any means, receives in payment of the Obligations held by that Lender, then, subject to applicable Laws: (a) the Lender exercising the right of setoff, banker's lien or counterclaim or otherwise receiving such payment shall purchase, and shall be deemed to have simultaneously purchased, from each of the other Lenders a participation in the Obligations held by the other Lenders and shall pay to the other Lenders a purchase price in an amount so that the share of the Obligations held by each Lender after the exercise of the right of setoff, banker's lien or counterclaim or receipt of payment shall be in the same proportion that existed prior to the exercise of the right of setoff, banker's lien or counterclaim or receipt of payment; and (b) such other adjustments and purchases of participations shall be made from time to time as shall be equitable to ensure that all of the Lenders share any payment obtained in respect of the Obligations ratably in accordance with each Lender's share of the Obligations immediately prior to, and without taking into account, the payment; provided that, if all or any portion of a disproportionate payment obtained as a result of the exercise of the right of setoff, banker's lien, counterclaim or otherwise is thereafter recovered from the purchasing Lender by Borrower or any Person claiming through or succeeding to the rights of Borrower, the purchase of a participation shall be rescinded and the purchase price thereof shall be restored to the extent of the recovery, but without interest. Each Lender that purchases a participation in the Obligations pursuant to this <u>Section 11.10</u> shall from and after the purchase have the right to give all notices, requests, demands, directions and other communications under this Agreement with respect to the portion of the Obligations purchased to the same extent as though the purchasing Lender were the original owner of the Obligations purchased. Borrower expressly consents to the foregoing arrangements and agrees that any Lender holding a participation in an Obligation so purchased pursuant to this Section 11.10 may exercise any and all rights of setoff, banker's lien or counterclaim with respect to the participation as fully as if the Lender were the original owner of the Obligation purchased.

11.11 <u>Indemnity by Borrower</u>. Borrower agrees to indemnify, save and hold harmless the Administrative Agent and each Lender and their respective directors, officers, agents, attorneys and employees (collectively the "Indemnitees") from and against: (a) any and all claims, demands, actions or causes of action (except a claim, demand, action, or cause of action for any amount excluded from the definition of "Taxes" in Section 3.10(e)) if the claim, demand, action or cause of action arises out of or relates to (i) any act or omission (or alleged act or omission) of Borrower, any Subsidiary or other Affiliate of Borrower or any partner, officer, director, stockholder, or other equity interest holder of Borrower relating to the Revolving Facility, (ii) the use or contemplated use of proceeds of any Borrowing, (iii) the relationship of Borrower and the Lenders under this Agreement, or (iv) the Loan Documents or the Revolving Facility in any other manner or aspect; (b) any administrative or investigative proceeding by any Governmental Agency arising out of or related to a claim, demand, action or cause of action described in clause (a) above; and (c) any and all liabilities, losses, reasonable costs or expenses (including reasonable attorneys' fees and the reasonably allocated costs of attorneys employed by any Indemnitee and disbursements of such attorneys and other professional services) that any Indemnitee suffers or incurs as a result of the assertion of any foregoing claim, demand, action or cause of action; provided that no Indemnitee shall be entitled to indemnification for any liability, loss, cost or expense caused by its own gross negligence or willful misconduct or for any liability, loss, cost or expense asserted against it by another Indemnitee. If any claim, demand, action or cause of action is asserted against any Indemnitee, such Indemnitee shall promptly notify Borrower, but the failure to so promptly notify Borrower shall not affect Borrower's obligations under this Section unless such failure materially prejudices Borrower's right to participate in the contest of such claim, demand, action or cause of action, as hereinafter provided. Such Indemnitee may (and shall, if requested by Borrower in writing) contest the validity, applicability and amount of such claim, demand, action or cause of action and shall permit Borrower to participate in such contest. Any Indemnitee that proposes to settle or compromise any claim or proceeding for which Borrower may be liable for payment of indemnity hereunder shall give Borrower written notice of the terms of such proposed settlement or compromise reasonably in advance of settling or compromising such claim or proceeding and shall obtain Borrower's prior consent (which shall not be unreasonably withheld or delayed). In connection with any claim, demand, action or cause of action covered by this Section 11.11 against more than one Indemnitee, all such Indemnitees shall be represented by the same legal counsel (which may be a law firm engaged by the Indemnitees or attorneys employed by an Indemnitee or a combination of the foregoing) selected by the Indemnitees, provided, that if such legal counsel determines in good faith that representing all such Indemnitees would or could result in a conflict of interest under Laws or ethical principles applicable to such legal counsel or that a defense or counterclaim is available to an Indemnitee that is not available to all such Indemnitees, then to the extent reasonably necessary to avoid such a conflict of interest or to permit unqualified assertion of such a defense or counterclaim, each affected Indemnitee shall be

entitled to separate representation by legal counsel selected by that Indemnitee, with all such legal counsel using reasonable efforts to avoid unnecessary duplication of effort by counsel for all Indemnitees; and <u>further provided</u> that the Administrative Agent (as an Indemnitee) shall at all times be entitled to representation by separate legal counsel (which may be a law firm or attorneys employed by the Administrative Agent or a combination of the foregoing). Any obligation or liability of Borrower to any Indemnitee under this <u>Section 11.11</u> shall survive the expiration or termination of this Agreement and the repayment of all Borrowings and the payment and performance of all other Obligations owed to the Lenders.

11.12 Nonliability of the Lenders. Borrower acknowledges and agrees that:

- (a) Any inspections of any Property of Borrower or any Subsidiary of Borrower made by or through the Administrative Agent or the Lenders are for purposes of administration of the Revolving Facility only and Borrower is not entitled to rely upon the same (whether or not such inspections are at the expense of Borrower);
- (b) By accepting or approving anything required to be observed, performed, fulfilled or given to the Administrative Agent or the Lenders pursuant to the Loan Documents, neither the Administrative Agent nor the Lenders shall be deemed to have warranted or represented the sufficiency, legality, effectiveness or legal effect of the same, or of any term, provision or condition thereof, and such acceptance or approval thereof shall not constitute a warranty or representation to anyone with respect thereto by the Administrative Agent or the Lenders;
- (c) The relationship between Borrower and the Administrative Agent and the Lenders is, and shall at all times remain, solely that of borrower and lenders; neither the Administrative Agent nor the Lenders shall under any circumstance be construed to be partners or joint venturers of Borrower, any Subsidiary of Borrower or any of their respective Affiliates; neither the Administrative Agent nor the Lenders shall under any circumstance be deemed to be in a relationship of confidence or trust or a fiduciary relationship with Borrower, any Subsidiary of Borrower or any of their respective Affiliates, or to owe any fiduciary duty to Borrower, any Subsidiary of Borrower or any of their respective Agent nor the Lenders undertake or assume any responsibility or duty to Borrower, any Subsidiary of Borrower or any of their respective Affiliates to select, review, inspect, supervise, pass judgment upon or inform Borrower, any Subsidiary of Borrower or any of their respective Affiliates of any matter in connection with their Property or the operations of Borrower, any Subsidiary of Borrower or any of their respective Affiliates; Borrower, Subsidiary of Borrower and their respective Affiliates shall rely entirely upon their own judgment with respect to such matters; and any review, inspection, supervision, exercise of judgment or supply of information undertaken or assumed by the Administrative Agent or the Lenders in connection with such matters is solely for the protection of the Administrative Agent and the Lenders and neither Borrower nor any other Person is entitled to rely thereon; and
- (d) The Administrative Agent and the Lenders shall not be responsible or liable to any Person for any loss, damage, liability or claim of any kind relating to injury or death to Persons or damage to Property caused by the actions, inaction or negligence of Borrower, any Subsidiary of Borrower and/or any of their respective Affiliates and Borrower hereby indemnifies and holds the Administrative Agent and the Lenders harmless on the terms set forth in Section 11.11 from any such loss, damage, liability or claim.
- 11.13 No Third Parties Benefited. This Agreement is made for the purpose of defining and setting forth certain obligations, rights and duties of Borrower, the Administrative Agent and the Lenders in connection with the Revolving Facility, and is made for the sole benefit of Borrower, the Administrative Agent and the Lenders, and the Administrative Agent's and the Lenders' successors and assigns. Except as provided in Sections 11.8 and 11.11, no other Person shall have any rights of any nature hereunder or by reason hereof.
- 11.14 Confidentiality. Each Lender agrees to hold any confidential information that it may receive from Borrower pursuant to this Agreement in confidence, except for disclosure: (a) to other Lenders or Affiliates of a Lender that have agreed to keep such information confidential to the same extent as if they were a party hereto; (b) to legal counsel and accountants for Borrower, any Subsidiary of Borrower or any Lender; (c) to other professional advisors to Borrower or any Subsidiary of Borrower or any Lender, provided that the recipient has accepted such information subject to a confidentiality agreement substantially similar to this Section 11.14; (d) to regulatory officials having jurisdiction over that Lender; (e) as required by Law or legal process, provided that each Lender agrees to notify Borrower of any such disclosures unless prohibited by applicable Laws, or in connection with any legal proceeding to which that Lender and Borrower or any Subsidiary of Borrower are adverse parties; and (f) to another financial institution in connection with a disposition or proposed disposition to that financial institution of all or part of that Lender's interests hereunder or a participation interest in its Note(s), provided that the recipient has accepted such information subject to a confidentiality

agreement substantially similar to this Section 11.14. For purposes of the foregoing, "confidential information" shall mean any information respecting Borrower or any Subsidiary of Borrower reasonably considered by Borrower to be confidential, other than (i) information previously filed with any Governmental Agency and available to the public, (ii) information previously published in any public medium from a source other than, directly or indirectly, that Lender, and (iii) information previously disclosed by Borrower or such Subsidiary of Borrower to any Person not associated with Borrower or such Subsidiary of Borrower or such Subsidiary of Borrower or which has not executed an appropriate confidentiality agreement with Borrower or such Subsidiary of Borrower. Nothing in this Section shall be construed to create or give rise to any fiduciary duty on the part of the Administrative Agent or the Lenders to Borrower or any Subsidiary of Borrower.

- 11.15 <u>Further Assurances</u>. Borrower shall, at its expense and without expense to the Lenders or the Administrative Agent, do, execute and deliver such further acts and documents as the Requisite Lenders or the Administrative Agent from time to time reasonably require for the assuring and confirming unto the Lenders or the Administrative Agent of the rights hereby created or intended now or hereafter so to be, or for carrying out the intention or facilitating the performance of the terms of any Loan Document.
- 11.16 <u>Integration</u>. This Agreement, together with the other Loan Documents and the letter agreements referred to in <u>Section 3.3</u>, comprises the complete and integrated agreement of the parties on the subject matter hereof and supersedes all prior agreements, written or oral, on the subject matter hereof. In the event of any conflict between the provisions of this Agreement and those of any other Loan Document, the provisions of this Agreement shall control and govern; <u>provided</u> that the inclusion of supplemental rights or remedies in favor of the Administrative Agent or the Lenders in any other Loan Document shall not be deemed a conflict with this Agreement. Each Loan Document was drafted with the joint participation of the respective parties thereto and shall be construed neither against nor in favor of any party, but rather in accordance with the fair meaning thereof.
- GOVERNED BY, AND CONSTRUED AND ENFORCED IN ACCORDANCE WITH, THE LAWS OF CALIFORNIA APPLICABLE TO CONTRACTS MADE AND PERFORMED IN CALIFORNIA. THE PARTIES AGREE THAT ALL ACTIONS OR PROCEEDINGS ARISING IN CONNECTION WITH THIS AGREEMENT SHALL BE TRIED AND LITIGATED ONLY IN A STATE OR FEDERAL COURT LOCATED IN THE STATE OF CALIFORNIA. THE PARTIES EXPRESSLY SUBMIT AND CONSENT IN ADVANCE TO SUCH JURISDICTION IN ANY ACTION OR PROCEEDING COMMENCED IN ANY SUCH COURT, AND THE PARTIES HEREBY WAIVE ANY OBJECTION THEY MAY HAVE BASED UPON LACK OF PERSONAL JURISDICTION AND HEREBY CONSENT TO THE GRANTING OF SUCH LEGAL OR EQUITABLE RELIEF AS IS DEEMED APPROPRIATE BY ANY SUCH COURT. FURTHERMORE, THE PARTIES HEREBY WAIVE, TO THE EXTENT PERMITTED UNDER APPLICABLE LAW, ANY RIGHT THEY MAY HAVE TO ASSERT THE DOCTRINE OF "FORUM NON CONVENIENS" OR TO OBJECT TO VENUE TO THE EXTENT ANY PROCEEDING IS BROUGHT IN ACCORDANCE WITH THIS SECTION 11.16.
- 11.18 <u>Severability of Provisions</u>. Any provision in any Loan Document that is held to be inoperative, unenforceable or invalid as to any party or in any jurisdiction shall, as to that party or jurisdiction, be inoperative, unenforceable or invalid without affecting the remaining provisions or the operation, enforceability or validity of that provision as to any other party or in any other jurisdiction, and to this end the provisions of all Loan Documents are declared to be severable.
- 11.19 <u>Headings</u>. Article and Section headings in this Agreement and the other Loan Documents are included for convenience of reference only and are not part of this Agreement or the other Loan Documents for any other purpose.
 - 11.20 <u>Time of the Essence</u>. Time is of the essence of the Loan Documents.
- 11.21 <u>Foreign Lenders and Participants</u>. Each Lender, and each holder of a participation interest herein, that is incorporated or otherwise organized under the Laws of a jurisdiction other than the United States of America or any State thereof or the District of Columbia shall deliver to Borrower (with a copy to the Administrative Agent), on or before the Closing Date (or on or

before accepting an assignment or receiving a participation interest herein pursuant to Section 11.8, if applicable) two duly completed copies, signed by a Responsible Official, of Form W-8BEN or W-8ECI (or other equivalent successor form) satisfactory to Borrower and the Administrative Agent that no withholding under the federal income tax laws is required with respect to such Person. Thereafter and from time to time, each such Person shall (a) promptly submit to Borrower (with a copy to the Administrative Agent), such additional duly completed and signed copies of such form (or such successor form(s) as shall be adopted from time to time by the relevant United States taxing authorities) as may then be available under then current United States laws and regulations to avoid, or such evidence as is satisfactory to Borrower and the Administrative Agent of any available exemption from, United States withholding taxes in respect of all payments to be made to such Person by Borrower pursuant to this Agreement and (b) take such steps as shall not be materially disadvantageous to it, in the reasonable judgment of such Person, and as may be reasonably necessary (including the re-designation of its Eurodollar Lending Office, if any) to avoid any requirement of applicable Laws that Borrower make any deduction or withholding for taxes from amounts payable to such Person. In the event that Borrower or the Administrative Agent becomes aware that a participation has been granted pursuant to Section 11.8(e) to a financial institution that is incorporated or otherwise organized under the Laws of a jurisdiction other than the United States of America, any State thereof or the District of Columbia, then, upon request made by Borrower or the Administrative Agent to the Lender that granted such participation, such Lender shall cause such participant financial institution to deliver the same documents and information to Borrower and the Administrative Agent as would be required under this Section if such financial institution were

- 11.22 Waiver of Right to Trial by Jury. EACH PARTY TO THIS AGREEMENT HEREBY EXPRESSLY WAIVES ANY RIGHT TO TRIAL BY JURY OF ANY CLAIM, DEMAND, ACTION OR CAUSE OF ACTION ARISING UNDER ANY LOAN DOCUMENT OR IN ANY WAY CONNECTED WITH OR RELATED OR INCIDENTAL TO THE DEALINGS OF THE PARTY HERETO OR ANY OF THEM WITH RESPECT TO ANY LOAN DOCUMENT, OR THE TRANSACTIONS RELATED THERETO, IN EACH CASE WHETHER NOW EXISTING OR HEREAFTER ARISING, AND WHETHER SOUNDING IN CONTRACT OR TORT OR OTHERWISE; AND EACH PARTY HEREBY AGREES AND CONSENTS THAT ANY SUCH CLAIM, DEMAND, ACTION OR CAUSE OF ACTION SHALL BE DECIDED BY COURT TRIAL WITHOUT A JURY, AND THAT ANY PARTY TO THIS AGREEMENT MAY FILE AN ORIGINAL COUNTERPART OR A COPY OF THIS SECTION WITH ANY COURT AS WRITTEN EVIDENCE OF THE CONSENT OF THE SIGNATORIES HERETO TO THE WAIVER OF THEIR RIGHT TO TRIAL BY JURY.
- 11.23 Purported Oral Amendments. BORROWER EXPRESSLY ACKNOWLEDGES THAT THIS AGREEMENT AND THE OTHER LOAN DOCUMENTS MAY ONLY BE AMENDED OR MODIFIED, OR THE PROVISIONS HEREOF OR THEREOF WAIVED OR SUPPLEMENTED, BY AN INSTRUMENT IN WRITING THAT COMPLIES WITH SECTION 11.2. BORROWER AGREES THAT IT WILL NOT RELY ON ANY COURSE OF DEALING, COURSE OF PERFORMANCE, OR ORAL OR WRITTEN STATEMENTS BY ANY REPRESENTATIVE OF THE ADMINISTRATIVE AGENT OR ANY LENDER THAT DOES NOT COMPLY WITH SECTION 11.2 TO EFFECT AN AMENDMENT, MODIFICATION, WAIVER OR SUPPLEMENT TO THIS AGREEMENT OR THE OTHER LOAN DOCUMENTS.
- Requisite Lenders, then Borrower may, at its sole expense and effort, upon notice to such Lender and the Administrative Agent, require such Lender to assign and delegate, without recourse (in accordance with and subject to the restrictions contained in, and consents required by, Section 11.8), all of its interests, rights and obligations under this Agreement and the related Loan Documents to an assignee that shall assume such obligations (which assignee may be another Lender, if a Lender accepts such assignment), provided that:
 - (a) Borrower shall have paid to the Administrative Agent the assignment fee specified in Section 11.8(d),
 - (b) such Lender shall have received payment of an amount equal to the outstanding principal of its Advances, accrued interest thereon, accrued fees and all other amounts payable to it hereunder and under the other Loan Documents from the assignee (to the extent of such outstanding principal and accrued interest and fees) or the Borrower (in the case of all other amounts); and
 - (c) such assignment does not conflict with applicable Laws.

No Lender shall be required to make such assignment or delegation if, prior thereto, as a result of a waiver by such Lender or otherwise, the circumstances entitling Borrower to require such assignment and delegation cease to apply.

11.25 <u>USA Patriot Act Notice</u>. Each Lender that is subject to the Patriot Act (as hereinafter defined) and the Administrative Agent (for itself and not on behalf of any Lender) hereby notifies the Borrower that pursuant to the requirements of the USA Patriot Act (Title III of Pub. L. 107-56 (signed into law October 26, 2001)) (the "Patriot Act"), it is required to obtain, verify and record information that identifies the Borrower, which information includes the name and address of the Borrower and other information that will allow such Lender or the Administrative Agent, as applicable, to identify the Borrower in accordance with the Patriot Act.

[THIS SPACE INTENTIONALLY LEFT BLANK - SIGNATURE PAGES TO FOLLOW]

IN WITNESS WHEREOF, the parties hereto have caused this Agreement to be duly executed as of the date first above written.

AMERICAN STATES WATER COMPANY,

a California corporation

By /s/ Floyd E. Wicks

Name: Floyd E. Wicks

Title: President and Chief Executive Officer

By /s/ Robert J. Sprowls

Name: Robert J. Sprowls

Title: Sr. Vice President-Finance, Chief Financial Officer, Treasurer

and Corporate Secretary

Address for Borrower:

630 East Foothill Boulevard San Dimas, California 91773-9016

Attn: Robert J. Sprowls Telecopier: (909) 394-1382 Telephone: (909) 394-3600 Website: http://www.aswater.com

With a copy to:

American States Water Company 630 East Foothill Boulevard San Dimas, California 91773-9016

Attn: Eva Tang

Telecopier: (909) 394-1382 Telephone: (909) 394-3600

WELLS FARGO BANK, NATIONAL ASSOCIATION, as Administrative Agent

By /s/ Deborah Moore

Name: Deborah Moore
Title: Administrative Agent

Address for notices to Administrative Agent for borrowings and payments:

Wells Fargo Bank, National Association

201 Third Street, 8th Floor

San Francisco, California 94103

Attn: Deborah Moore

Telecopier: (415) 512-9408 Telephone: (415) 477-5379

E-mail: mooredj@wellsfargo.com

With a copy to:

333 South Grand Avenue, Third Floor Los Angeles, California 90071

Attn: Julie Saavedra

Telecopier: (213) 628-1415 Telephone: (213) 253-6146

E-mail: saavedj@wellsfargo.com

WELLS FARGO BANK, NATIONAL ASSOCIATION, as a Lender

By /s/ John N. Cate

John N. Cate Vice President

Address:

333 South Grand Avenue, Third Floor Los Angeles, California 90071 Attn: American States Water Account Officer

Telecopier: (213) 687-3501 Telephone: (213) 253-6226

E-mail: john.n.cate@wellsfargo.com

COBANK, ACB,

as a Lender

By /s/ David Dombirer

Name: David Dombirer Title: Vice President

Address:

5500 South Quebec St.

Greenwood Village, Colorado 80111

Attn: David Dombirer Vice President

Telecopier: (303) 796-1473 Telephone: (303) 694-5848 E-mail: dombid@cobank.com

UNION BANK OF CALIFORNIA, N.A., as a Lender

By /s/ Susan K. Johnson

Name: Susan K. Johnson Title: Vice President

Address:

Energy Capital Services

445 South Figueroa St., 15th Floor

Los Angeles, California 90071

Attn: Susan K. Johnson Vice President

Telecopier: (213) 236-4096 Telephone: (213) 236-4125

E-mail: susan.johnson@uboc.com

COMERICA BANK, as a Lender

By /s/ Elise Walker

Name: Elise Walker Title: Vice President

Address:

611 Anton Blvd., MC 4462 Costa Mesa, California 92626

Attn: Elise Walker

Telecopier: (714) 433-3226 Telephone: (714) 433-3236

E-mail: emwalker@comerica.com

THE NORTHERN TRUST COMPANY, as a Lender

By /s/ John E. Burda

Name: John E. Burda Title: Vice President

Address:

50 S. LaSalle Street Chicago, Illinois 60675 Attn: John E. Burda

Telecopier: (312) 444-7028 Telephone: (312) 444-3455

E-mail: John_Burda@notes.ntrs.com

SCHEDULE 1.1 TO CREDIT AGREEMENT

LENDER COMMITMENTS/PRO RATA SHARES

Lender	Pro Rata Share		Amount of Revolving Commitment		
Wells Fargo Bank, National Association	32.941176	%\$	28,000,000		
CoBank, ACB	28.235294	%\$	24,000,000		
Union Bank of California, N.A.	18.823529	%\$	16,000,000		
Comerica Bank	14.117647	%\$	12,000,000		
The Northern Trust Company	5.882353	%\$	5,000,000		
Total	100	%\$	85,000,000		

CONSENT, WAIVER AND OMNIBUS AMENDMENT TO AMENDED AND RESTATED CREDIT AGREEMENT AND RELATED LOAN DOCUMENTS

This Consent, Waiver and Omnibus Amendment to Amended and Restated Credit Agreement and Related Loan Documents (this "Agreement"), dated as of October 11, 2005, is entered into with reference to the Amended and Restated Credit Agreement (as amended, supplemented or otherwise modified from time to time, the "Credit Agreement") dated as of June 3, 2005, among American States Water Company, a California corporation ("Borrower"), each lender from time to time a party thereto (each a "Lender" and collectively, the "Lenders") and Wells Fargo Bank, National Association, as Administrative Agent (in such capacity, the "Administrative Agent") and Lead Arranger. Capitalized terms not otherwise defined herein shall have the meanings set forth for such terms in the Credit Agreement. Section references herein are to sections of the Credit Agreement unless otherwise indicated.

RECITALS

This Agreement is made with reference to the following facts:

- A. Pursuant to the terms of the Credit Agreement, the Lenders have made certain credit facilities available to Borrower.
- B. Southern California Water Company, a California corporation and wholly-owned Subsidiary of Borrower ("SCW") has changed its name (the "Name Change") to Golden State Water Company, a California corporation ("GSW").
- C. Pursuant to the terms of that certain Note Purchase Agreement, dated as of October 11, 2005 (the "Note Purchase Agreement"), GSW desires to authorize, issue and sell to CoBank, ACB ("CoBank"), as purchaser, \$40,000,000 aggregate principal amount of GSW's 5.87% Senior Note due December 20, 2028 (the "GSW Note").
- D. CoBank is a Lender under the Credit Agreement and an agricultural cooperative bank subject to regulation by the Farm Credit Administration. In order to enter into the transactions contemplated by the Note Purchase Agreement, GSW will be required to, among other things, purchase non-voting participation certificates in CoBank (the "Participation Certificates") in such amounts and at such times as CoBank may require in accordance with its bylaws and 2005 Capital Plan, each as in effect on the Effective Date (the "Maximum Certificate Purchase Amount"). CoBank will have a statutory lien (the "Statutory Lien") on all of the Participation Certificates purchased by GSW to secure GSW's obligations under the GSW Note. Transactions contemplated by the Purchase Agreement, the GSW Note, the Participation Certificates and the Statutory Lien are hereinafter referred to as the "Note Purchase Transactions".
- E. Borrower and GSW now desire that the Lenders (i) amend the Credit Agreement and the other Loan Documents to reflect the Name Change and (ii) consent to the Note Purchase Transactions and waive the provisions of <u>Sections 6.9</u>, <u>6.10</u> and <u>6.14</u> of the Credit Agreement solely to the extent that such Sections would otherwise prohibit the Note Purchase Transactions.
- F. Subject to the terms and conditions set forth herein, the Lenders are willing to so amend the Loan Documents and so consent and waive such provisions of the Credit Agreement.
 - G. Borrower, GSW and the Administrative Agent, acting with the consent of the Lenders, hereby agree as follows:

Article 12. Omnibus Amendment to Credit Agreement and other the Loan Documents. Each of the parties hereto hereby agrees that (a) any and all references to Southern California Water Company, a California corporation, or SCW contained in the Credit Agreement and each of the other Loan Documents shall constitute references to Golden State Water Company and GSW, respectively, (b) the definition of SCW is hereby deleted from the Credit Agreement and each of the Loan Documents where applicable and (c) the following definition is hereby added to the Credit Agreement and each of the other Loan Documents where applicable:

"GSW" means Golden State Water Company, a California corporation, a wholly-owned Subsidiary of Borrower and the successor by name change to Southern California Water Company, a California corporation.

Article 13. Waiver and Consent. Each of the Lenders hereby (a) consents to the terms of the Note Purchase Transactions as described in the Note Purchase Agreement and the Note, each as in effect as of the date hereof (collectively, the "Note Purchase Transaction Documents") and (b) waives the provisions of Sections 6.9, 6.10 and 6.14 of the Credit Agreement solely to the extent that, absent such waiver, such provisions would prohibit the Note Purchase Transactions; provided, however, that none of the waivers or consents contained herein shall permit, or be deemed to permit, GSW to purchase Participation Certificates in an aggregate amount which exceeds the Maximum Certificate Purchase Amount. The waivers and consents set forth in this Agreement are one-time waivers only and shall be solely with respect to the Sections of the Credit Agreement and the limited transactions described in this Section 1.

Article 14. <u>Effectiveness</u>. This Agreement shall become effective on the date each of the conditions set forth below shall have been satisfied (the "Effective Date"):

- (a) <u>Documentation</u> The Administrative Agent shall have received each of the following in form and substance satisfactory to the Administrative Agent:
 - (i) duly executed counterparts of this Amendment executed by the parties hereto;
 - (ii) a Certificate of a Senior Officer of Borrower certifying that (A) each of the conditions to closing set forth in Section 3 of the Note Purchase Agreement have been satisfied in accordance with their respective terms, (B) attached thereto are true, correct, complete and duly executed copies of (1) the Note Purchase Transaction Documents, (2) the provisions of CoBank's bylaws setting forth the requirement that GSW purchase the Participation Certificates, (3) CoBank's 2005 Capital Plan and (4) a copy of the amendment to the Amended and Restated Articles of Incorporation of SCW effecting the Name Change, which amendment shall be in full force and effect not later than 10 days following the Effective Date; and
 - (iii) such other assurances, certificates, documents or consents related to the foregoing as the Administrative Agent or any Lender may reasonably request; and
- (b) <u>Fees and Expenses</u>. Borrower shall have reimbursed Administrative Agent and the Lenders for all of their respective reasonable costs and expenses (including reasonable attorney's fees and expenses) incurred in connection with the negotiation and drafting of this Agreement and the transaction contemplated hereby.

Article 15. Representations and Warranties. Except (i) for representations and warranties which expressly speak as of particular date or are no longer true and correct as a result of a change permitted by the Credit Agreement or the other Loan Documents or (ii) as disclosed by Borrower and approved in writing by the Requisite Lenders, Borrower hereby represents and warrants that each representation and warranty made by Borrower in Article 4 of the Credit Agreement (other than Sections 4.4, 4.6 (first sentence), 4.9 and 4.16) are true and correct as of the date hereof as though such representations and warranties were made on and as of the date hereof. Without in any way limiting the foregoing, Borrower represents and warrants to the Administrative Agent and the Lenders that no Default or Event of Default has occurred and remains continuing or will result from the waivers, consents or transactions set forth herein or contemplated hereby.

Article 16. <u>Confirmation</u>. In all respects, the terms of the Credit Agreement and the other Loan Documents, in each case as amended, waived or consented to hereby or herein, are hereby confirmed.

[THIS SPACE INTENTIONALLY LEFT BLANK - SIGNATURE PAGE TO FOLLOW]

IN WITNESS WHEREOF, Borrower, GSW and the Administrative Agent, acting with the consent of the Lenders, have executed this Agreement as of the date first set forth above by their duly authorized representatives.

By:

Name:

Title:

AMERICAN STATES WATER COMPANY, a California corporation

WELLS FARGO BANK, NATIONAL ASSOCIATION, as Administrative Agent and a Lender

/s/ John Cate

Vice President

/s/ John E. Burda

John E. Burda

John Cate

By: /s/ Floyd E. Wicks

> Floyd E.Wicks Name: Title: President & CEO

By: /s/ Robert J. Sprowls

> Name: Robert. J. Sprowls

Sr. Vice President-Finance, Chief Financial Officer, Title:

Treasurer & Corporate Secretary

THE NORTHERN TRUST COMPANY,

as a Lender

By:

GOLDEN STATE WATER COMPANY, a California corporation, successor by name change to Southern California Water Company, a California corporation

/s/ Robert J. Sprowls

Name: Robert J. Sprowls

Sr. Vice President-Finance, Chief Financial Officer, Treasurer & Corp. Secretary Title:

Title: Vice President

Name:

COBANK, ACB,

as a Lender

By:

By: /s/ David W. Dornbrier

> David W. Dornbrier Name:

Title: Vice President

UNION BANK OF CALIFORNIA, N.A.

By: /s/ Susan K. Johnson

Susan K. Johnson Name: Title: Vice President

COMERICA BANK,

as a Lender

By: /s/ Elise Walker

> Elise Walker Name: Vice President Title

SECOND AMENDMENT TO AMENDED AND RESTATED CREDIT AGREEMENT

THIS SECOND AMENDMENT TO AMENDED AND RESTATED CREDIT AGREEMENT (this "Amendment"), dated as of August 25, 2008, is entered into with reference to the Amended and Restated Credit Agreement (as amended, restated, supplemented or otherwise modified from time to time, the "Credit Agreement") dated as of June 3, 2005, by and among AMERICAN STATES WATER COMPANY, a California corporation ("Borrower"), each of the lenders party thereto (each an "Original Lender" and collectively, the "Original Lenders") and WELLS FARGO BANK, NATIONAL ASSOCIATION, as administrative agent for the Lenders (in such capacity, the "Administrative Agent") and lead arranger. Capitalized terms used herein and not otherwise defined shall have the meanings set forth for such terms in the Credit Agreement. Section references herein are to sections of the Credit Agreement unless otherwise stated.

RECITALS

WHEREAS, Borrower has requested that the Credit Agreement be modified as set forth in this Amendment, <u>including</u>, without limitation, that the amount of the Revolving Facility be increased by \$30,000,000 and that Borrower be provided with a one-time option to further increase the amount of the Revolving Facility up to \$15,000,000.

WHEREAS, the Administrative Agent, the Lenders and Borrower have agreed to make certain changes in the terms and conditions set forth in the Credit Agreement, and have agreed to amend the Credit Agreement to reflect said changes.

NOW, THEREFORE, for valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the parties hereto agree that the Credit Agreement shall be amended as follows:

- 1. <u>Section 1.1 Defined Terms (New)</u>. The following defined terms are hereby added to <u>Section 1.1</u> in the appropriate alphabetical place:
 - "Amendment No. 2" means the Second Amendment to Amended and Restated Credit Agreement, dated as of August 25, 2008, among Borrower, each of the Lenders party thereto and the Administrative Agent.
 - "Amendment No. 2 Effective Date" means the "Effective Date" as defined in Amendment No. 2.
- 2. <u>Section 1.1 Defined Term (Revision)</u>. The definition of the term "Commitment" is hereby amended to read in its entirety as follows:
 - "Commitment" means, with respect to each Lender, the commitment, if any, of such Lender to make Advances (expressed as the maximum aggregate amount of the Advances to be made by such Lender hereunder), as such commitment may be (a) reduced from time to time pursuant to Section 2.6, (b) increased pursuant to Section 2.10, and (c) reduced or increased from time to time pursuant to assignments by or to such Lender pursuant to Section 11.8. The initial amount of each Lender's Commitment is set forth on Schedule 1.1 or in the Assignment and Acceptance pursuant to which such Lender shall have assumed its Commitment, as applicable. The aggregate amount of the Lenders' Commitments on the Amendment No. 2 Effective Date is \$115,000,000.
- 3. Section 2.10. Section 2.10 is hereby added to the Credit Agreement and shall read in its entirety as follows:
 - 2.10 Optional Increase to the Commitments. Borrower may submit a one-time request in writing that the Revolving Facility be increased by increasing the then effective Commitments in an aggregate amount which does not result in the aggregate principal amount of the Commitments being greater than \$130,000,000 minus the amount of any permanent reductions to the Commitments which have then occurred prior to the date of any such request

pursuant to Section 2.6(a), provided that in connection with any such increase, the interest rates payable with respect to all Commitments (whether or not a part of the increased Commitments) shall be the same (and, if necessary, the Credit Agreement will be amended accordingly). Any request under this Section 2.10 shall be submitted by Borrower to the Lenders through the Administrative Agent not less than thirty (30) days prior to the proposed increase, specify the proposed effective date and amount of such increase and be accompanied by a Certificate signed by a Senior Officer of Borrower, stating that no Default or Event of Default exists as of the date of the request or will result from the requested increase. Borrower's right under this Section 2.10 to increase the aggregate Commitments shall terminate and be of no further effect if not exercised prior to the ninetieth (90th) day preceding the Maturity Date. The consent of the Lenders shall not be required for an increase in the amount of the Commitments pursuant to this Section 2.10; accordingly, this Section 2.10 shall supersede any provisions in Section 6.10 or 11.2 to the contrary.

- (a) Each Lender may approve or reject a request to participate in an increase in the amount of the Commitments in its sole and absolute discretion and, absent an affirmative written response within five (5) Banking Days after receipt of such request, shall be deemed to have rejected the request. The rejection of such a request by any number of Lenders shall not affect Borrower's right to increase the Commitments pursuant to this Section 2.10.
- (b) In responding to a request under this <u>Section 2.10</u>, each Lender that is willing to increase the amount of its Pro Rata Share of the increased Commitments shall specify the amount of the proposed increase which it is willing to assume. Each consenting Lender shall be entitled to participate ratably (based on its Pro Rata Share of the Commitments before such increase) in any resulting increase in the Commitments, subject to the right of the Administrative Agent to adjust allocations of the increased Commitments so as to result in the amounts of the Pro Rata Shares of the Lenders being in integral multiples of \$100,000.
- (c) If the aggregate principal amount offered to be assumed by the consenting Lenders is less than the amount requested or the aggregate principal amount offered to be assumed by the consenting Lenders is less than the amount requested without an increase in the interest rates payable with respect to all Commitments, Borrower may (i) reject the proposed increase in its entirety, (ii) accept the offered amounts, (iii) accept the offered amounts from consenting Lenders not requesting an increase in the interest rates payable with respect to all Commitments, or (iv) designate new lenders who qualify as Eligible Assignees and which are reasonably acceptable to the Administrative Agent as additional Lenders hereunder in accordance with clause (f) of this Section 2.10 (each, a "New Lender"), which New Lenders may assume the amount of the increase in the Commitments that has not been assumed by the consenting Lenders or the amount of the increase in the Commitments that has not been assumed by the consenting Lenders without an increase in the interest rates payable with respect to all Commitments.
- (d) After completion of the foregoing, the Administrative Agent shall give written notification to the Lenders and any New Lenders of the increase to the Commitments which shall thereupon become effective and in connection with such notification the Administrative Agent will distribute to Borrower and the Lenders a revised Schedule 1.1 reflecting the then applicable Pro Rata Shares of the Lenders.
- (e) Each New Lender shall become an additional party hereto as a Lender concurrently with the effectiveness of the proposed increase in the Commitments upon its execution of an instrument of joinder to this Agreement, which is in form and substance reasonably acceptable to the Administrative Agent and which, in any event, contains the representations, warranties, indemnities and other protections afforded to the Administrative Agent and the other Lenders which would be granted or made by an Eligible Assignee by means of the execution of an Assignment and Acceptance.
- (f) Subject to the foregoing, any increase to the Commitments requested under this <u>Section 2.10</u> shall be effective as of the date proposed by Borrower and shall be in the principal amount <u>equal to</u> (i) the amount which consenting Lenders are willing to assume as increases to their respective Commitments <u>plus</u> (ii) the amount offered

by any New Lenders. Upon the effectiveness of any such increase, each Borrowing outstanding shall be refinanced with new Advances reflecting the adjusted Pro Rata Shares of the Lenders in the Revolving Facility if there is any change thereto and Borrower shall:

- (i) issue replacement Notes to each affected Lender and new Notes to each New Lender (in each case, as may be requested by such Lender), and the percentage of Pro Rata Shares of each Lender will be adjusted to give effect to the increase in the Commitments;
- (ii) execute and deliver to the Administrative Agent such amendments to the Loan Documents as the Administrative Agent may reasonably request relating to such increase; and
- (iii) pay to the existing Lenders any breakage costs which are payable in connection with the refinancing of any Borrowings in the manner contemplated by Section 3.6.
- 4. Section 11.7. Section 11.7 is hereby amended and restated in its entirety and shall read as follows:
 - 11.7 Execution of Loan Documents. Unless the Administrative Agent otherwise specifies with respect to any Loan Document, (a) this Agreement and any other Loan Document may be executed in any number of counterparts and any party hereto or thereto may execute any counterpart, each of which when executed and delivered will be deemed to be an original and all of which counterparts of this Agreement or any other Loan Document, as the case may be, when taken together will be deemed to be but one and the same instrument and (b) execution of any such counterpart may be evidenced by telecopier or other electronic means of transmission of the signature of such party. The execution of this Agreement or any other Loan Document by any party hereto or thereto will not become effective until counterparts hereof or thereof, as the case may be, have been executed by all the parties hereto or thereto. Any party delivering an executed counterpart of this Agreement or any Loan Document by telecopier or other electronic means of transmission shall also deliver a manually executed counterpart of such document, but failure to do so shall not affect the validity, enforceability, or binding effect of such document.
- 5. <u>Schedule 1.1 (Lender Commitments/Pro Rata Shares)</u>. <u>Schedule 1.1</u> to the Credit Agreement is hereby deleted and replaced with <u>Schedule 1.1</u> to this Amendment.
- 6. Adjusting Purchase Payments. Revolving Credit Facility Usage as outstanding immediately prior to the effectiveness of this Amendment (the "Existing Usage") shall remain outstanding after giving effect to this Amendment. On the Effective Date, the applicable Lenders agree to purchase and sell undivided interests in the Existing Usage (such purchases and sales to be free and clear of Liens created, incurred, assumed or suffered to exist by, through or under the Lenders selling such undivided interests, but otherwise without recourse or warranty of any kind or nature whatsoever) by making or receiving Adjusting Purchase Payments as specified in Exhibit A to this Amendment (the "Adjusting Purchase Payments") so that the Existing Usage will be properly allocated and owing to the Lenders, as applicable, in accordance with the Pro Rata Shares specified in Exhibit A to this Amendment. Each Lender making an Adjusting Purchase Payment shall deliver it to the Administrative Agent and the Administrative Agent shall forward such Adjusting Purchase Payments to the Lenders entitled thereto promptly after receipt in accordance with the allocations specified in Exhibit A to this Amendment. As of the Effective Date, in addition to any other Advances that may be made, each Lender shall be deemed as having Advances outstanding in the amount of its Pro Rata Share of the Existing Usage consisting of outstanding Advances. In addition, as of the Effective Date, the Lenders shall hold participations in the outstanding Letters of Credit as provided in Section 2.5 in accordance with their Pro Rata Shares. As of the Effective Date, without giving effect to any Advances to be made on the Effective Date, (a) the outstanding principal balance of all Advances is [[\$55,750,000]], and (b) the Aggregate Effective Amount of all Letters of Credit is [[\$11,131,000]].
- 7. <u>Amendment Fee</u>. In consideration of the agreements set forth herein, Borrower hereby agrees to pay (a) to the Administrative Agent for the ratable accounts of the Original Lenders, an amendment fee (the "<u>Amendment Fee</u>") in the amount set

forth in a letter agreement heretofore entered into by and between Borrower and the Administrative Agent (the "Fee Letter"), which Amendment Fee shall be non-refundable when paid and is fully-earned as of (and due and payable on) the Effective Date, (b) to the Administrative Agent for the accounts of the Lenders who participate in the increase to the Revolving Facility (the "Revolver Increase"), a fee (the "Increase Fee") in the amount set forth in the Fee Letter, which Increase Fee shall be non-refundable when paid and is fully-earned as of (and due and payable on) the date of the Revolver Increase, and (c) all other fees set forth in the Fee Letter, as and when due under the terms of the Fee Letter.

- Effectiveness. This Amendment shall become effective on the date the Administrative Agent receives each of the following (the "Effective Date"):
 - (a) duly executed counterparts of this Amendment signed by the parties hereto;
 - (b) duly executed counterparts of the Fee Letter;
 - (c) duly executed Notes for each Lender with a revised or new Commitment, as applicable;
 - (d) the Amendment Fee; and
 - (e) the Increase Fee.
- Except as specifically provided herein, all terms and conditions of the Credit Agreement remain in full force and effect, without waiver or modification. This Amendment and the Credit Agreement shall be read together, as one document. This Amendment shall constitute a Loan Document.
- Borrower hereby remakes all representations and warranties contained in the Credit Agreement (except to the extent such representations and warranties expressly speak as of a prior date) and reaffirms all covenants set forth therein. Borrower further certifies that as of the date of, and after giving effect to, this Amendment, there exists no Event of Default as defined in the Credit Agreement, nor any condition, act or event which with the giving of notice or the passage of time or both would constitute any such Event of Default.

[Remainder of page intentionally left blank; signature pages follow.]

IN WITNESS WHEREOF, the parties hereto have caused this Amendment to be executed as of the day and year first above written.

BORROWER:

AMERICAN STATES WATER COMPANY,

a California corporation

By: /s/ Robert J. Sprowls

Name: Robert J. Sprowls

Executive Vice President – Finance, Chief Financial Officer, Treasurer & Corporate Secretary Title:

ADMINISTRATIVE AGENT:

WELLS FARGO BANK, NATIONAL ASSOCIATION,

as Administrative Agent

By: Name: /s/ DuVon G. Davis

DuVon G. Davis Title: Vice President

LENDERS:

WELLS FARGO BANK, NATIONAL ASSOCIATION,

as a Lender

/s/ DuVon G. Davis

By: /s/ DuVon G. Davis
Name: DuVon G. Davis Title: Vice President

COBANK, ACB,

as a Lender

By: /s/ David W. Dornbrier
Name: David W. Dornbrier
Title: Vice President

UNION BANK OF CALIFORNIA, N.A.,

as a Lender

By:

/s/ Harvey R. Horowitz Harvey R. Horowitz Vice President Name: Title:

COMERICA BANK,

as a Lender

By: /s/ Elise Walker
Name: Elise Walker
Title: Vice President

THE NORTHERN TRUST COMPANY,

as a Lender

By: /s/ John E. Burda
Name: John E. Burda

Title: Senior Vice President

EXHIBIT A

ADJUSTING PURCHASE PAYMENTS

[see attached]

SCHEDULE 1.1

LENDER COMMITMENTS/PRO RATA SHARES

Lender	Pro Rata Share	Commitment Amount		
Wells Fargo Bank, National Association	33.0434783	%\$	38,000,000	
CoBank, ACB	32.1739130	%\$	37,000,000	
Union Bank of California, N.A.	13.9130435	%\$	16,000,000	
Comerica Bank	10.4347826	%\$	12,000,000	
The Northern Trust Company	10.4347826	%\$	12,000,000	
	100	%\$	115,000,000	

THIRD AMENDMENT TO AMENDED AND RESTATED CREDIT AGREEMENT

THIS THIRD AMENDMENT TO AMENDED AND RESTATED CREDIT AGREEMENT (this "Amendment"), dated as of May 27, 2010, is entered into with reference to the Amended and Restated Credit Agreement (as amended, restated, supplemented or otherwise modified from time to time, the "Credit Agreement") dated as of June 3, 2005, by and among AMERICAN STATES WATER COMPANY, a California corporation ("Borrower"), each of the lenders party thereto, and WELLS FARGO BANK, NATIONAL ASSOCIATION, as administrative agent for the Lenders (in such capacity, the "Administrative Agent") and lead arranger. Capitalized terms used herein and not otherwise defined shall have the meanings set forth for such terms in the Credit Agreement. Section references herein are to sections of the Credit Agreement unless otherwise stated.

RECITALS

WHEREAS, Borrower has requested that the Credit Agreement be modified as set forth in this Amendment, <u>including</u>, without limitation, that the aggregate Commitments be reduced to \$100,000,000 and the maturity date of the Revolving Facility be extended to the date which is three (3) years following the Effective Date hereof.

WHEREAS, the Administrative Agent, the Lenders and Borrower have agreed to make certain changes in the terms and conditions set forth in the Credit Agreement, and have agreed to amend the Credit Agreement to reflect said changes.

NOW, THEREFORE, for valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the parties hereto agree that the Credit Agreement shall be amended as follows:

- 11. Section 1.1 Defined Terms (New). The following defined terms are hereby added to Section 1.1 in the appropriate alphabetical place:
- "Amendment No. 3" means the Third Amendment to Amended and Restated Credit Agreement, dated as of May 27, 2010, among Borrower, each of the Lenders party thereto and the Administrative Agent.
 - "Amendment No. 3 Effective Date" means the "Effective Date" as defined in Amendment No. 3.
 - "ASUS" means American States Utility Services, Inc., a California corporation and wholly-owned Subsidiary of Borrower.
 - "ASUS Water Sale Agreement" means the Water Sale Agreement, dated January 31, 2006, by and between Natomas and ASUS.
- "California Water Meter Indebtedness" means the Indebtedness evidenced by the Funding Agreement between the State of California Department of Public Health and Golden State Water Company, Project Number 3410015-006, Data Universal Number 009107873. As of the Amendment No. 3 Effective Date, the aggregate outstanding principal amount of such Indebtedness is \$0.00
 - "FRB" means the Board of Governors of the Federal Reserve System of the United States.
- "Initial Pricing Period" means the period commencing on the Amendment No. 3 Effective Date and ending on the first Pricing Occurrence to occur thereafter.

[&]quot;Lending Parties" means, collectively, the Lenders, any Swing Line Lender and any Issuing Lender.

[&]quot;Natomas" means Natomas Central Mutual Water Company, a California corporation.

[&]quot;NCSP True-Up Amount" has the meaning set forth in the definition of Net Cash Sales Proceeds.

"Net Cash Sales Proceeds" means, with respect to the Permitted Water Rights Disposition, the Cash proceeds received by or for the account of Borrower from such Disposition, including any cash payments received as deferred payment pursuant to a promissory note, receivable or otherwise (none of which shall defer payment in full, in cash, for a period of greater than one year and each of which shall be in form and substance reasonably satisfactory to the Administrative Agent), but only as and when received in Cash, the Cash proceeds received by or for the account of Borrower and its Subsidiaries from such Disposition, net of (a) any amount required to be paid to any Person owning an interest in the assets disposed of, (b) any amount applied to the repayment of Indebtedness secured by a Lien permitted under Section 6.9 on the asset disposed of, (c) any transfer, income or other taxes paid or reasonably estimated by Borrower to be payable as a result of such Disposition on any gain recognized in connection therewith, (d) professional fees and expenses, fees due to any Governmental Agency, broker's commissions and other out-of-pocket costs of sale actually paid on an arms-length basis to any Person attributable to such Disposition and (e) any reserves established with respect to liabilities in accordance with GAAP in connection with such Disposition; provided, however, that if the actual taxes due pursuant to clause (c) above (the "Actual Taxes") are less than the amount used by Borrower pursuant to clause (c) above to calculate the Net Cash Sales Proceeds paid to Borrower in respect of any Disposition, Net Cash Sales Proceeds shall also include an amount equal to (i) Net Cash Sales Proceeds calculated using Actual Taxes less (ii) the Net Cash Sales Proceeds actually paid to Borrower in respect of such Disposition (the "NCSP True-Up Amount").

"Permitted Water Rights Disposition" means the sale, by ASUS, of all or any portion of the water rights and Settlement Contract Base Supply (as defined in the ASUS Water Sale Agreement) entitlements to divert from the Sacramento River up to 5,000 acre-feet of water per year in years where Natomas receives one hundred percent (100%) of its Base Supply (as defined in the ASUS Water Sale Agreement), purchased, or to be purchased, by ASUS pursuant to the terms of the ASUS Water Rights Agreement; provided that (a) at the time such Disposition is consummated, no Default or Event of Default shall exist or shall result therefrom, (b) the sales price relating to any such Disposition shall be not less than seventy percent (70%) of the net book value (as determined in accordance with GAAP) of the assets sold pursuant to such Disposition, (c) Administrative Agent shall have received a pro forma Compliance Certificate to the effect that after giving pro forma effect to such Disposition (which Disposition shall be deemed to have occurred on the first day of the Rolling Period most recently completed prior to the date of such Disposition (the "Disposition Test Period"), Borrower would not have been in breach of any covenant set forth in Section 6.12 or Section 6.13 for the Disposition Test Period, (d) notwithstanding anything to the contrary in Section 3.1(e), the outstanding principal amount of Advances shall be prepaid (without a corresponding reduction to the Commitments) on or before the second Banking Day following receipt by Borrower or any Subsidiary of Net Cash Proceeds from such Disposition (or, in the case of any NCSP True-Up Amount, on or before the second Banking Day following a determination by Borrower or any Subsidiary that a NCSP True-Up Amount exists) by an amount equal to the lesser of seventy-five percent (75%) of such Net Cash Sales Proceeds and the outstanding principal amount of all Advances, and (e) such Disposition shall have been completed in accordance with all applicable Laws and upon such other ter

"Reserve Requirement" means the stated maximum rate (rounded upwards, as necessary, to the nearest 1/16th of one percent (0.0625%)), as in effect on any date of determination of all reserve requirements (including any marginal, emergency, supplemental, special or other reserves) applicable on such date to any member bank of the Federal Reserve System in respect of "Eurocurrency liabilities" as defined in Regulation D (or any successor category of liabilities under Regulation D) of the FRB as in effect on such day, whether or not applicable to any Lending Party.

12. Section 1.1 — Defined Term (Revision). The following defined terms contained in Section 1.1 of the Credit Agreement are hereby amended in full to read as follows:

"Alternate Base Rate" means for any day, the greatest of (a) the Prime Rate in effect on such day, (b) the Federal Funds Rate for such day plus one-half of one percent (0.50%) and (c) the One Month LIBOR Rate for such day (determined on a daily basis as set forth below) plus one percent (1.00%). As used in this definition of "Alternate Base Rate", "One Month LIBOR Rate" means, with respect to any interest rate calculation for a Loan or other Obligation bearing interest at the Alternate Base Rate, a rate per annum equal to the quotient (rounded upward if necessary to the nearest 1/16th of one percent (0.0625%)) of (i) the rate per annum referred to as the BBA (British Bankers Association) LIBOR RATE as reported on Reuters LIBOR page 1, or if not reported by Reuters, as reported by any service selected by Administrative Agent on the applicable day (provided that if such day is not a Eurodollar Banking Day for which a Eurodollar Rate is quoted, the next preceding Eurodollar Banking Day for which a Eurodollar Rate is quoted) at or about 11:00 a.m., London time (or as soon thereafter as practicable), for Dollar deposits being delivered in the London interbank eurodollar currency market for a term of one month commencing on such date of determination, divided by (ii) one minus the Reserve Requirement in effect on such day. If for any reason rates are not available as provided in clause (i) of the preceding sentence, the rate to be used in clause (i) shall be, at Administrative Agent's discretion (in each case, rounded upward if necessary to the nearest one-sixteenth (1/16) of one percent (0.0625%)), (1) the rate per annum at which Dollar deposits are offered to the Administrative Agent in the London interbank eurodollar currency market or (2) the rate at which Dollar deposits are offered to the Administrative Agent in, or by Wells Fargo to major banks in, any offshore interbank eurodollar market selected by Administrative Agent, in each case on the applicable day (provided that if such day is not a

Eurodollar Banking Day for which Dollar deposits are offered to Administrative Agent in the London interbank eurodollar currency market, the next preceding Eurodollar Banking Day for which Dollar deposits are offered to Administrative Agent in the London interbank eurodollar currency market) at or about 11:00 a.m., London time (or as soon thereafter as practicable) (for delivery on such date of determination) for a one month term. Each determination by Administrative Agent pursuant to this definition shall be conclusive absent manifest error.

"<u>Applicable Alternate Base Rate Margin</u>" means, with respect to any Alternate Base Rate Advance, for each Pricing Period, the interest rate margin set forth below (expressed in basis points per annum) opposite the Applicable Pricing Level for that Pricing Period:

Applicable Pricing	
Level	Margin
I	20.0
II	25.0
III	75.0
IV	150.0

"Applicable Commitment Fee Margin" means, for each Pricing Period, the margin set forth below (expressed in basis points per annum) opposite the Applicable Pricing Level for that Pricing Period:

Applicable Pricing	
Level	Margin
Ι	15.0
II	17.5
III	25.0
IV	35.0

"Applicable Eurodollar Rate Margin" means, with respect to any Eurodollar Rate Advance, for each Pricing Period, the interest rate margin set forth below (expressed in basis points per annum) opposite the Applicable Pricing Level for that Pricing Period:

Applicable Pricing	
Level	Margin
I	120.0
II	125.0
III	175.0
IV	250.0

"Applicable Pricing Level" means, (a) for the Initial Pricing Period, Pricing Level II and (b) thereafter, the pricing level set forth below opposite the Debt Rating achieved by Borrower as of the first day of that Pricing Period:

Pricing Level	Debt Rating								
I	Greater than or equal to A1 / A+								
II	Less than A1/A+ but greater than or equal to A2 /A								
III	Less than A2/A but greater than or equal to A3/A-								
IV	Less than A3/A-								

provided that in the event that the then prevailing Debt Ratings are "split ratings", Borrower will receive the benefit of the higher Debt Rating, unless the split is a "double split rating" (in which case the pricing level applicable to the middle Debt Rating will apply) or a "triple split rating" (in which case the pricing level applicable to the Debt Rating applicable to the lowest pricing level will apply). For purposes hereof, a Debt Rating is only a "split rating" if the Debt Rating applies to a different pricing level.

"Commitment" means, with respect to each Lender, the commitment, if any, of such Lender to make Advances (expressed as the maximum aggregate amount of the Advances to be made by such Lender hereunder), as such commitment may be (a) reduced from time to time pursuant to Section 2.6, (b) increased pursuant to Section 2.10, and (c) reduced or increased from time to time pursuant to assignments by or to such Lender pursuant to Section 11.8. The amount of each Lender's Commitment on the Amendment No. 3 Effective Date is set forth on Schedule 1.1 or in the Assignment and Acceptance pursuant to which such Lender shall have assumed its Commitment, as applicable. The aggregate amount of the Lenders' Commitments on the Amendment No. 3 Effective Date is \$100,000,000.

"Eurodollar Period" means, as to each Eurodollar Rate Loan, the period commencing on the date such Eurodollar Rate Loan is disbursed or converted to or continued as a Eurodollar Rate Loan and ending on the date one, two, three, six or, if made available by each of the Lenders making such Eurodollar Rate Loan, nine or twelve months thereafter, as selected by Borrower in its related Request for Borrowing; provided that (a) any Eurodollar Period that would otherwise end on a day that is not a Eurodollar Banking Day shall be extended to the next succeeding Eurodollar Banking Day unless such Eurodollar Banking Day falls in another calendar month, in which case such Eurodollar Period shall end on the next preceding Banking Day; (b) any Eurodollar Period that begins on the last Eurodollar Banking Day of a calendar month (or on a day for which there is no numerically corresponding day in the calendar month at the end of such Eurodollar Period) shall end on the last Eurodollar Banking Day of the calendar month at the end of such Eurodollar Period for any Advance shall extend beyond the Maturity Date.

"Eurodollar Rate" means for any Eurodollar Period, with respect to a Eurodollar Rate Advance, a rate per annum (rounded upwards, as necessary, to the nearest 1/16th of one percent (0.0625%)) obtained by dividing (a) the rate *per annum* determined by the Administrative Agent at approximately 11:00 a.m., London time, on the date that is two Eurodollar Banking Days prior to the beginning of such Eurodollar Period by reference to the British Bankers' Association "Interest Settlement Rates" for deposits in Dollars (as set forth by any service (including Bloomberg, Reuters and Thomson Financial) selected by the Administrative Agent that has been nominated by the British Bankers' Association as an authorized information vendor for the purpose of displaying such rates) in an amount approximately equal to the principal amount to which such Eurodollar Period applies (for delivery on the first day of such Eurodollar Period) with a term equivalent to such Eurodollar Period; provided that, if an interest rate is not ascertainable pursuant to the foregoing provisions of this definition, then "Eurodollar Rate" shall be the interest rate *per annum* determined by the Administrative Agent to be the average of the rates per annum at which deposits in Dollars in an amount approximately equal to the principal amount to which such Eurodollar Period applies (for delivery on the first day of such Eurodollar Period) with a term equivalent to such Eurodollar Period are offered for such Eurodollar Period by Wells Fargo to major banks in the London interbank offered market in London, England at approximately 11:00 a.m., London time, on the date that is two Eurodollar Banking Days prior to the beginning of such Eurodollar Period by (b) one minus the Reserve Requirement in effect on such date. Each determination by the Administrative Agent pursuant to this definition shall be conclusive absent manifest error.

"Federal Funds Rate" means for any day, the rate per annum equal to the weighted average of the rates on overnight Federal funds transactions with members of the Federal Reserve System arranged by Federal funds brokers on such day, as published by the Federal Reserve Bank of New York on the Banking Day next succeeding such day; provided that (a) if such day is not a Banking Day, then the Federal Funds Rate for such day shall be such rate on such transactions on the next preceding Banking Day as so published on the next succeeding Banking Day and (b) if no such rate is so published on such next succeeding Banking Day, then the Federal Funds Rate for such day shall be the average rate (rounded upward, if necessary, to a whole multiple of one one-hundredth of one percent (0.01%)) charged to Wells Fargo on such day on such transactions as determined by Administrative Agent.

"Lead Arranger" means Wells Fargo Securities, LLC.

"Maturity Date" means the earlier of (a) May 27, 2013 and (b) the termination or cancellation of the Revolving Facility (and all of the Commitments pertaining thereto) pursuant to the terms of this Agreement.

"Maximum Revolving Credit Amount" means \$100,000,000.

"<u>Permitted Capital Asset Indebtedness</u>" means Indebtedness of Borrower and its Subsidiaries consisting of Capital Lease Obligations, the California Water Meter Indebtedness or Indebtedness otherwise incurred to finance the purchase or construction of capital assets (which shall be deemed to exist if the Indebtedness is incurred at or within 90 days before or after the purchase or construction of the capital asset), or to refinance any such Indebtedness; provided that the aggregate principal amount of such Indebtedness shall not exceed \$15,000,000 at any one time outstanding (as determined in accordance with GAAP consistently applied).

"Pricing Period" means (a) the Initial Pricing Period and (b) each subsequent period commencing on the date of a Pricing Occurrence and ending on the next Pricing Occurrence to occur.

"Prime Rate" means the per annum rate of interest in effect for such day as publicly announced from time to time by Wells Fargo as its "Prime Rate," such rate being the rate of interest most recently announced within Wells Fargo at its principal office in San Francisco, California as its "Prime Rate," with the understanding that Wells Fargo's "Prime Rate" is one of Wells Fargo's base rates and serves as the basis upon which effective rates of interest are calculated for those loans making reference thereto, and is evidenced by the recording thereof after its announcement in such internal publication or publications as Wells Fargo may designate. Wells Fargo's "Prime Rate" is not intended to be the lowest rate of interest charged by Wells Fargo in connection with extensions of credit to borrowers. Any change in Wells Fargo's "Prime Rate" as announced by Wells Fargo shall take effect at the opening of business on the day specified in the public announcement of such change.

- 13. <u>Section 1.1 Defined Term (Deleted)</u>. The defined term "Eurodollar Reserve Percentage" is hereby deleted from the Credit Agreement and each of the other Loan Documents.
- 14. <u>Section 2.5(a)(ii)</u> <u>Letters of Credit (Letter of Credit Sublimit)</u>. The reference to "\$20,000,000" contained in <u>Section 2.5(a)(ii)</u> of the Credit Agreement is hereby amended in full to read "\$25,000,000".
- 15. <u>Section 2.10 Optional Increase to the Commitments</u>. The introductory paragraph of <u>Section 2.10</u> of the Credit Agreement is hereby amended in full to read as follows:
 - "2.10 Optional Increase to the Commitments. Borrower may submit a request in writing that the Revolving Facility be increased by increasing the then effective Commitments in an aggregate amount which does not result in the aggregate principal amount of the Commitments being greater than \$140,000,000 minus the amount of any permanent reductions to the Commitments which have then occurred prior to the date of any such request pursuant to Section 2.6(a), provided that (a) Borrower may make no more than two such requests, (b) the increase requested pursuant to the second such request shall not be effective less than twelve (12) months following the effectiveness of the increase granted pursuant to the first such request, (c) the first such request shall be in an amount of not less than \$20,000,000 and integral multiples of \$10,000,000 in excess thereof, (d) the second such request shall be in an amount of not less than \$10,000,000 and integral multiples of \$10,000,000 in excess thereof, and (e) in connection with any such increase, the interest rates payable with respect to all Commitments (whether or not a part of the increased Commitments) shall be the same (and, if necessary, the Credit Agreement will be amended accordingly). Any request under this Section 2.10 shall be submitted by Borrower to the Lenders through the Administrative Agent not less than thirty (30) days prior to the proposed increase, specify the proposed effective date and amount of such increase and be accompanied by a Certificate signed by a Senior Officer of Borrower, stating that no Default or Event of Default exists as of the date of the request or will result from the requested increase. Borrower's right under this Section 2.10 to increase the aggregate Commitments shall terminate and be of no further effect if not exercised prior to the ninetieth (90th) day preceding the Maturity Date. The consent of the Lenders shall not be required for an increase in the amount of the Commitments pursuant to this Section 2.10; accordingly, this Section
- 16. <u>Section 3.6 Eurodollar Costs and Related Matters</u>. Each reference to "Eurodollar Reserve Percentage" contained in <u>Sections 3.6(a)</u> and <u>3.6(b)(ii)</u> of the Credit Agreement are hereby amended in full to read "Reserve Requirement".
 - 17. Section 6.3 Dispositions of Property. Section 6.3 of the Credit Agreement is hereby amended in full to read as follows:
 - "6.3 <u>Disposition of Property.</u> Make any Disposition of its Property, whether now owned or hereafter acquired, except (a) Dispositions of obsolete Property or Property with no material remaining useful life, (b) Dispositions in an aggregate amount not to exceed \$5,000,000 in any Fiscal Year ending after the Closing Date or \$10,000,000 in the aggregate from and after the Closing Date to the Termination Date; provided that (i) at the time of any such Disposition pursuant to <u>clause (b)</u> only, no Default or Event of Default shall exist or shall result from such Disposition and (ii) the sales

price relating to a Disposition (pursuant to <u>clause (a)</u> or <u>(b)</u>) shall be paid in Cash and/or Indebtedness or other evidence of an Investment permitted pursuant to <u>Section 6.14(h)</u>, (c) Dispositions pursuant to any order of any Governmental Agency in an eminent domain proceeding and any settlement of any such proceeding, (d) a Permitted Water Rights Disposition and (e) as otherwise permitted pursuant to a letter agreement dated May 27, 2010, by and between the Administrative Agent and the Borrower.

- 18. Section 6.15 Operating Leases. Section 6.15 of the Credit Agreement is hereby amended in full to read as follows:
- "6.15 Operating Leases. Incur any obligation to pay rent under an operating lease in any Fiscal Year if to do so would result in the aggregate obligation of Borrower and its Subsidiaries to pay rent under all operating leases in that Fiscal Year to exceed \$6,000,000."
- 19. 11.22 Waiver of Jury Trial. Section 11.22 of the Credit Agreement is hereby amended in full to read as follows:
 - "11.22 Waiver of Jury Trial; Judicial Reference.
- (a) WAIVER OF JURY TRIAL. BORROWER, THE ADMINISTRATIVE AGENT AND EACH LENDER HEREBY WAIVES ANY RIGHT TO A TRIAL BY JURY IN ANY ACTION, PROCEEDING OR COUNTERCLAIM CONCERNING ANY RIGHTS UNDER THIS AGREEMENT OR THE OTHER LOAN DOCUMENTS, OR UNDER ANY AMENDMENT, WAIVER, CONSENT, INSTRUMENT, DOCUMENT OR OTHER AGREEMENT DELIVERED OR WHICH IN THE FUTURE MAY BE DELIVERED IN CONNECTION THEREWITH, OR ARISING FROM ANY FINANCING RELATIONSHIP EXISTING IN CONNECTION WITH THIS AGREEMENT, AND AGREES THAT ANY SUCH ACTION, PROCEEDINGS OR COUNTERCLAIM SHALL BE TRIED BEFORE A COURT AND NOT BEFORE A JURY. THE BORROWER HEREBY ACKNOWLEDGES THAT THIS PROVISION IS A MATERIAL INDUCEMENT FOR THE ADMINISTRATIVE AGENT AND THE LENDERS ENTERING INTO THIS AGREEMENT.
- JUDICIAL REFERENCE. TO THE EXTENT THAT THE WAIVER OF JURY TRIAL IN SECTION 11.22(a) IS (b) UNENFORCEABLE. THE PARTIES HERETO DESIRE THAT THEIR DISPUTES BE RESOLVED BY A JUDGE OR RETIRED JUDGE APPLYING THE APPLICABLE LAW. THEREFORE, THE PARTIES HERETO AGREE TO REFER, FOR A COMPLETE AND FINAL ADJUDICATION, ANY AND ALL ISSUES OF FACT OR LAW INVOLVED IN ANY LITIGATION OR PROCEEDING (INCLUDING ALL DISCOVERY AND LAW AND MOTION MATTERS, PRETRIAL MOTIONS, TRIAL MATTERS, AND POST-TRIAL MOTIONS (E.G., MOTIONS FOR RECONSIDERATION, NEW TRIAL AND TO TAX COSTS, ATTORNEY FEES AND PREJUDGMENT INTEREST)) UP TO AND INCLUDING FINAL JUDGMENT, BROUGHT TO RESOLVE ANY DISPUTE (WHETHER SOUNDING IN CONTRACT, TORT, UNDER ANY STATUTE, OR OTHERWISE) BETWEEN THE ADMINISTRATIVE AGENT OR ANY LENDER AND BORROWER ARISING OUT OF, CONNECTED WITH, OR RELATED OR INCIDENTAL TO THE RELATIONSHIP ESTABLISHED BETWEEN THE PARTIES IN CONNECTION WITH THIS AGREEMENT, THE LOAN DOCUMENTS OR THE TRANSACTIONS RELATED HERETO AND THERETO, TO A JUDICIAL REFEREE WHO SHALL BE APPOINTED UNDER A GENERAL REFERENCE PURSUANT TO CALIFORNIA CODE OF CIVIL PROCEDURE SECTION 638. THE REFEREE'S DECISION WOULD STAND AS THE DECISION OF THE COURT, WITH JUDGMENT TO BE ENTERED ON HIS STATEMENT OF DECISION IN THE SAME MANNER AS IF THE ACTION HAD BEEN TRIED BY THE COURT. THE ADMINISTRATIVE AGENT AND THE BORROWER SHALL SELECT A SINGLE NEUTRAL REFEREE, WHO SHALL BE A RETIRED STATE OR FEDERAL JUDGE WITH AT LEAST FIVE YEARS OF JUDICIAL EXPERIENCE IN CIVIL MATTERS. IN THE EVENT THAT THE AGENT AND THE BORROWER CANNOT AGREE UPON A REFEREE, THE REFEREE SHALL BE APPOINTED BY THE COURT. BORROWER SHALL BEAR THE FEES AND EXPENSES OF THE REFEREE UNLESS THE REFEREE OTHERWISE PROVIDES IN THE STATEMENT OF DECISION. EACH PARTY AGREES THAT ANY PARTY TO THIS AGREEMENT MAY FILE AN ORIGINAL COUNTERPART OR A COPY OF THIS SECTION 11.22 WITH ANY COURT AS WRITTEN EVIDENCE OF THE CONSENT OF THE SIGNATORIES HERETO TO THE REFERENCE TO A JUDICIAL REFEREE AS PROVIDED ABOVE."

- 20. <u>Schedule 1.1 (Lender Commitments/Pro Rata Shares)</u>. <u>Schedule 1.1</u> to the Credit Agreement is hereby amended in full to read as set forth on Annex I to this Amendment.
- 21. Adjusting Purchase Payments. Revolving Credit Facility Usage as outstanding immediately prior to the effectiveness of this Amendment (the "Existing Usage") shall remain outstanding after giving effect to this Amendment. On the Effective Date, the applicable Lenders agree to purchase and sell undivided interests in the Existing Usage (such purchases and sales to be free and clear of Liens created, incurred, assumed or suffered to exist by, through or under the Lenders selling such undivided interests, but otherwise without recourse or warranty of any kind or nature whatsoever) by making or receiving Adjusting Purchase Payments as specified in Annex II to this Amendment (the "Adjusting Purchase Payments") so that the Existing Usage will be properly allocated and owing to the Lenders, as applicable, in accordance with the Pro Rata Shares specified in such Annex II. Each Lender making an Adjusting Purchase Payment shall deliver it to the Administrative Agent and the Administrative Agent shall forward such Adjusting Purchase Payments to the Lenders entitled thereto promptly after receipt in accordance with the allocations specified such Annex II. As of the Effective Date, in addition to any other Advances that may be made, each Lender shall be deemed as having Advances outstanding in the amount of its Pro Rata Share of the Existing Usage consisting of outstanding Advances. In addition, as of the Effective Date, the Lenders shall hold participations in the outstanding Letters of Credit as provided in Section 2.5 in accordance with their Pro Rata Shares. As of the Effective Date, without giving effect to any Advances to be made on the Effective Date, (a) the outstanding principal balance of all Advances is \$19,000,000.00, and (b) the Aggregate Effective Amount of all Letters of Credit is \$11,081,000.00.
- 22. <u>Fees.</u> In consideration of the agreements set forth herein, Borrower hereby agrees to pay to the Administrative Agent and the Lead Arranger, either for their own account or for the ratable account of the Lenders, as applicable, each of the fees as shall be required by that certain letter agreement, dated April 26, 2010, entered into by and between Borrower, the Administrative Agent and the Lead Arranger (the "<u>Amendment Fee Letter</u>"), in each case, at the times and in the amounts set forth in such Amendment Fee Letter.
- 23. <u>Effectiveness</u>. This Amendment shall become effective on the date that each of the following conditions shall have been satisfied in a manner satisfactory to the Lenders in their sole and absolute discretion (the "<u>Effective Date</u>"):
- (a) The Administrative Agent shall have received all of the following in form and substance satisfactory to the Administrative Agent and its legal counsel:
 - (i) duly executed counterparts of this Amendment signed by the parties hereto;
 - (ii) duly executed Notes for each Lender with a revised or new Commitment, as applicable;
 - (iii) the favorable written legal opinion of O'Melveny & Myers LLP, special counsel to the Borrower; and
- (iv) such other assurances, certificates, documents, consents or opinions as the Administrative Agent and/or any Lender reasonably may require.
 - (b) Each of the fees set forth in the Amendment Fee Letter which are due and payable on the Effective Date shall have been paid.
- (c) Borrower shall have paid to the Administrative Agent, for the benefit of Lenders existing immediately prior to the Effective Date, interest in the amount of \$9,455.30 and accrued fees in the amount of \$12,559.39 owed to such Lenders on the Effective Date.

- (d) The reasonable costs and expenses of the Administrative Agent in connection with the preparation of this Agreement and each of the other Loan Documents prepared in connection therewith payable pursuant to <u>Section 11.3</u>, and invoiced to Borrower prior to the Effective Date, shall have been paid.
- (e) All legal matters relating to the Loan Documents shall be reasonably satisfactory to Bingham McCutchen LLP, special counsel to the Administrative Agent.
- 24. <u>Integration; Loan Document</u>. Except as specifically provided herein, all terms and conditions of the Credit Agreement remain in full force and effect, without waiver or modification. This Amendment and the Credit Agreement shall be read together, as one document. This Amendment shall constitute a Loan Document.
- 25. Representations and Warranties; No Defaults. Borrower hereby represents that all representations and warranties contained in Article 4 of the Credit Agreement (except to the extent such representations and warranties expressly speak as of a prior date, in which case they were true and correct in all material respects on and as of such earlier date) are and will be true and correct in all material respects on and as of the Amendment No. 3 Effective Date and reaffirms all covenants set forth therein. Borrower further certifies that as of the date of, and after giving effect to, this Amendment, there exists no Event of Default as defined in the Credit Agreement, nor any condition, act or event which with the giving of notice or the passage of time or both would constitute any such Event of Default.
- 26. Governing Law. THIS AMENDMENT SHALL BE GOVERNED BY, AND CONSTRUED AND ENFORCED IN ACCORDANCE WITH, THE LAWS OF CALIFORNIA APPLICABLE TO CONTRACTS MADE AND PERFORMED IN CALIFORNIA.
- 27. <u>Counterparts</u>. This Amendment may be executed in any number of counterparts and any party hereto may execute any counterpart, each of which when executed and delivered will be deemed to be an original and all of which counterparts of this Amendment, when taken together, will be deemed to be but one and the same instrument and execution of any such counterpart may be evidenced by telecopier or other electronic means of transmission of the signature of such party.

[Remainder of page intentionally left blank; signature pages follow.]

American States Water Company Attention: Robert J. Sprowls and Eva Tang May 27, 2010 Page

IN WITNESS WHEREOF, the parties hereto have caused this Amendment to be executed as of the day and year first above

written.

BORROWER:

AMERICAN STATES WATER COMPANY,

a California corporation

By: /s/ EVA G. TANG

Name: Eva G. Tang

Title: Senior Vice President-Finance, Chief Financial Officer, Corporate

Secretary and Treasurer

ADMINISTRATIVE AGENT:

WELLS FARGO BANK, NATIONAL ASSOCIATION,

as Administrative Agent

By: / s/ DuVon G. Davis

Name: DuVon G. Davis

Title: Vice President

LENDERS:

WELLS FARGO BANK, NATIONAL ASSOCIATION,

as a Lender

By: /s/ DuVon G. Davis

Name: DuVon G. Davis

Title: Vice PResident

COBANK, ACB,

as a Lender

By: /s/ David W. Dornbrier

Name: David W. Dornbrier

Title: Vice President

COMERICA BANK,

as a Lender

By: /s/ Mark C. Skrzynski

Name: Mark C. Skrzynski

Title: Corporate Banking Officer

U.S. BANK NATIONAL ASSOCIATION,

as a Lender

By: /s/ John I. Paul

Name: John I. Paul

Title: Portfolio Manager

ANNEX I

SCHEDULE 1.1

LENDER COMMITMENTS/PRO RATA SHARES

Lender	Pro Rata Share		Commitment Amount		
Wells Fargo Bank, National Association	38.0	%\$	38,000,000		
CoBank, ACB	35.0	%\$	35,000,000		
Comerica Bank	15.0	%\$	15,000,000		
U.S. Bank National Association	12.0	%\$	12,000,000		
	100.0	%\$	100,000,000		

ANNEX II

ADJUSTING PURCHASE PAYMENTS

Banks Making / (Receiving) Adjusting Purchase Payments	(Former Commitment	Former Pro Rata Share of Commitment		Share of Outstanding Advances	ľ	New Commitment	New Pro Rata Share of Commitment		Share of Outstanding Advances	Adjusting Purchase Payment to Pay / (Receive)
Wells Fargo Bank, National Association	\$	38,000,000	33.04347826 %	\$	6,278,260.86	\$	38,000,000	38.00000000	% \$	7,220,000	\$ 941,739.14
CoBank, ACB	\$	37,000,000	32.17391304 %	\$	6,113,043.48	\$	35,000,000	35.00000000	% \$	6,650,000	\$ 536,956.52
Union Bank, N.A.	\$	16,000,000	13.91304348 %	\$	2,643,478.26	\$	_	0.00000000 %	\$	_	\$ (2,643,478.26)
Comerica Bank	\$	12,000,000	10.43478261 %	\$	1,982,608.7	\$	15,000,000	15.00000000	% \$	2,850,000	\$ 867,391.3
The Northern Trust Company	\$	12,000,000	10.43478261 %	6\$	1,982,608.7	\$	_	0.00000000 %	\$	_	\$ (1,982,608.70)
U.S. Bank National Association	\$	_	0.00000000 %	\$	_	\$	12,000,000	12.00000000 %	\$	2,280,000	\$ 2,280,000
	\$	115,000,000	100.00000000 %	6 \$	19,000,000	\$	100,000,000	100.00000000	% \$	19,000,000	\$

Outstanding Advances = \$ 19,000,000

American States Water Company 630 East Foothill Boulevard San Dimas, California 91773-9016 Attn: Robert J. Sprowls and Eva Tang

Re: Permitted CCWC Disposition

Ladies and Gentlemen:

Reference is made to (a) that certain Amended and Restated Credit Agreement (as amended, restated. supplemented or otherwise modified from time to time, the "Credit Agreement"), dated as of June 3, 2005, currently by and among American States Water Company, a California corporation ("Borrower"), each of the lenders from time to time parties thereto (each a "Lender" and collectively, the "Lenders"), Wells Fargo Bank, National Association, as administrative agent for the Lenders (in such capacity, the "Administrative Agent"), and Wells Fargo Securities, LLC, as lead arranger, and (b) the Third Amendment to Amended and Restated Credit Agreement ("Amendment No. 3"), dated as of even date herewith, by and among Borrower, each of the Lenders listed therein and the Administrative Agent. Capitalized terms used herein and not otherwise defined shall have the meanings set forth for such terms in the Credit Agreement. Section references herein are to sections of the Credit Agreement unless otherwise stated.

This letter agreement is a Loan Document and is the letter agreement referred to in Section 6.3(e) of the Credit Agreement.

In connection with each of the transactions contemplated by Amendment No. 3 and for valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the parties hereto hereby agree as follows:

- 1. In connection with the Dispositions permitted pursuant to <u>Section 6.3</u>, Borrower shall be permitted to make the Permitted CCWC Disposition in accordance with the terms set forth herein and in the Credit Agreement.
 - 2. The following terms used herein shall have the meanings set forth below:

"CCWC" means Chaparral City Water Company, an Arizona corporation and wholly-owned Subsidiary of Borrower.

"CCWC Disposition Net Cash Sales Proceeds" means, with respect to the Permitted CCWC Disposition, the Cash proceeds received by or for the account of Borrower from such Disposition, including any cash payments received as deferred payment pursuant to a promissory note, receivable or otherwise (none of which shall defer payment in full, in cash, for a period of greater than one year and each of which shall be in form and substance reasonably satisfactory to the Administrative Agent), but only as and when received in Cash, the Cash proceeds received by or for the account of Borrower and its Subsidiaries from such Disposition, net of (a) any amount required to be paid to any Person owning an interest in the assets disposed of, (b) any amount applied to the repayment of Indebtedness secured by a Lien permitted under Section 6.9 on the asset disposed of, (c) any transfer, income or other taxes paid or reasonably estimated by Borrower to be payable as a result of such Disposition on any gain recognized in connection therewith, (d) professional fees and expenses, fees due to any Governmental Agency, broker's commissions and other out-of-pocket costs of sale actually paid on an arms-length basis to any Person attributable to such Disposition and (e) any reserves established with respect to liabilities in accordance with GAAP in connection with such Disposition; provided, however, that if the actual taxes due pursuant to clause (c) above (the "Actual Taxes") is less than the amount used by Borrower pursuant to clause (c) above to calculate the CCWC Disposition Net Cash Sales Proceeds paid to Borrower in respect of any Disposition, CCWC Disposition Net Cash Sales Proceeds salulated using Actual Taxes less (ii) the CCWC Disposition Net Cash Sales Proceeds actually paid to Borrower in respect of such Disposition (the "CCWC Disposition NCSP True-Up Amount").

"Permitted CCWC Disposition" means the sale of (a) all or substantially all of the assets of CCWC or (b) all or substantially all of the capital stock of CWCC, provided that (i) at the time of delivery of the CWCC Purchase Agreement referred to below and at the time such Disposition is consummated, no Default or Event of Default shall exist or shall result therefrom, (ii) the sales price relating to any such Disposition shall be not less than \$20,000,000, no less than fifty percent (50%) of which shall be paid in Cash, (iii) the sales agreement with respect to such Disposition (the "CCWC Purchase Agreement") shall have been executed and delivered (with a copy delivered to the Administrative Agent) on or prior to December 31, 2010, with such Disposition having been consummated not later than the date which is eighteen (18) months following such execution and delivery of the CCWC Purchase Agreement, (iv) Administrative Agent shall have received a pro forma Compliance Certificate to the effect that after giving pro forma effect to such Disposition (which Disposition shall be deemed to have occurred on the first day of the Rolling Period most recently completed prior to the date of such Disposition (the "Disposition Test Period"), Borrower would not have been in breach of any covenant set forth in Section 6.12 or Section 6.13 for the Disposition Test Period, (v) notwithstanding anything to the contrary in Section 3.1(e), the outstanding principal amount of all Advances shall be prepaid (without a corresponding reduction to the Commitments) on or before the second Banking Day following receipt by Borrower or any Subsidiary of Net Cash Proceeds from such Disposition (or, in the case of any CCWC Disposition NCSP True-Up Amount, on or before the second Banking Day following a determination by Borrower or any Subsidiary that a CCWC Disposition NCSP True-Up Amount exists) by an amount equal to the lesser of seventy-five percent (75%) of such CCWC Disposition Net Cash Sales Proceeds and the outstanding principal amount of

THIS LETTER AGREEMENT SHALL BE GOVERNED BY, AND CONSTRUED AND ENFORCED IN ACCORDANCE WITH, THE LAWS OF CALIFORNIA APPLICABLE TO CONTRACTS MADE AND PERFORMED IN CALIFORNIA.

This letter agreement may be executed in any number of counterparts and any party hereto may execute any counterpart, each of which when executed and delivered will be deemed to be an original and all of which counterparts of this letter agreement, when taken together, will be deemed to be but one and the same instrument and execution of any such counterpart may be evidenced by telecopier or other electronic means of transmission of the signature of such party.

[Signature page follows]

Very truly yours, WELLS FARGO BANK, NATIONAL ASSOCIATION

By: /s/ Duvon G. Davis

Name: Duvon G. Davis
Title: Vice President

ACKNOWLEDGED AND AGREED TO AS OF THIS 27th DAY OF MAY, 2010

AMERICAN STATES WATER COMPANY, a California corporation

By: /s/ Eva G. Tang

Name: Eva G. Tang

Title: SVP- Finance, CFO, Corporate Secretary & Treasurer

Third Amendment to Amended and Restated Credit Agreement

Fourth AMENDMENT TO AMENDED AND RESTATED CREDIT AGREEMENT

THIS FOURTH AMENDMENT TO AMENDED AND RESTATED CREDIT AGREEMENT (this "Amendment"), dated as of May 23, 2013, is entered into with reference to the Amended and Restated Credit Agreement (as amended, restated, supplemented or otherwise modified from time to time, the "Credit Agreement") dated as of June 3, 2005, by and among AMERICAN STATES WATER COMPANY, a California corporation ("Borrower"), each of the lenders party thereto, and WELLS FARGO BANK, NATIONAL ASSOCIATION, as administrative agent for the Lenders (in such capacity, the "Administrative Agent") and lead arranger. Capitalized terms used herein and not otherwise defined shall have the meanings set forth for such terms in the Credit Agreement. Section references herein are to sections of the Credit Agreement unless otherwise stated.

RECITALS

WHEREAS, Borrower has requested that the Credit Agreement be modified as set forth in this Amendment, <u>including</u>, without limitation, that the maturity date of the Revolving Facility be extended to the date which is five (5) years following the Effective Date hereof.

WHEREAS, the Administrative Agent, the Lenders and Borrower have agreed to make certain changes in the terms and conditions set forth in the Credit Agreement, and have agreed to amend the Credit Agreement to reflect said changes.

NOW, THEREFORE, for valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the parties hereto agree that the Credit Agreement shall be amended as follows:

- 1. Section 1.1 Defined Terms (New). The following defined terms are hereby added to Section 1.1 in the appropriate alphabetical place:
- "Amendment No. 4" means the Fourth Amendment to Amended and Restated Credit Agreement, dated as of May 23, 2013, among Borrower, each of the Lenders party thereto and the Administrative Agent.
 - "Amendment No. 4 Effective Date" means the "Effective Date" as defined in Amendment No. 4.
- 2. <u>Section 1.1 Defined Term (Revision)</u>. The following defined terms contained in <u>Section 1.1</u> of the Credit Agreement are hereby amended in full to read as follows:
 - "<u>Applicable Alternate Base Rate Margin</u>" means, with respect to any Alternate Base Rate Advance, for each Pricing Period, the interest rate margin set forth below (expressed in basis points per annum) opposite the Applicable Pricing Level for that Pricing Period:

Applicable Pricing	
Level	Margin
I	0.000
II	0.000
III	0.000
IV	0.000
V	0.000

"Applicable Commitment Fee Margin" means, for each Pricing Period, the margin set forth below (expressed in basis points per annum) opposite the Applicable Pricing Level for that Pricing Period:

Applicable Pricing	
Level	Margin
I	5.0
II	7.5
III	10.0
IV	12.5
V	15.0

"Applicable Eurodollar Rate Margin" means, with respect to any Eurodollar Rate Advance, for each Pricing Period, the interest rate margin set forth below (expressed in basis points per annum) opposite the Applicable Pricing Level for that Pricing Period:

Applicable Pricing Level	Margin
I	55.0
II	65.0
III	75.0
IV	85.0
V	95.0

"Applicable Pricing Level" means, for each Pricing Period, the pricing level set forth below opposite the Debt Rating achieved by Borrower as of the first day of that Pricing Period:

Pricing Level	Debt Rating
I	Greater than or equal to Aa3 / AA-
II	Less than Aa3 / AA- but greater than or equal to A1/A+ $$
III	Less than A1/A+ but greater than or equal to A2/A
IV	Less than A2/A but greater than or equal to A3/A-
V	Less than A3/A-

provided that in the event that the then prevailing Debt Ratings are "split ratings", Borrower will receive the benefit of the higher Debt Rating, unless the split is a "double split rating" (in which case the pricing level applicable to the middle Debt Rating will apply) or a "triple split rating" (in which case the pricing level applicable to the Debt Rating applicable to the lowest pricing level will apply). For purposes hereof, a Debt Rating is only a "split rating" if the Debt Rating applies to a different pricing level.

"Commitment" means, with respect to each Lender, the commitment, if any, of such Lender to make Advances (expressed as the maximum aggregate amount of the Advances to be made by such Lender hereunder), as such commitment may be (a) reduced from time to time pursuant to Section 2.6, (b) increased pursuant to Section 2.10, and (c) reduced or increased from time to time pursuant to assignments by or to such Lender pursuant to Section 11.8. The amount of each Lender's Commitment on the Amendment No. 4 Effective Date is set forth on Schedule 1.1 or in the Assignment and Acceptance pursuant to which such Lender shall have assumed its Commitment, as applicable. The aggregate amount of the Lenders' Commitments on the Amendment No. 4 Effective Date is \$100,000,000.

"Maturity Date" means the earlier of (a) May 23, 2018 and (b) the termination or cancellation of the Revolving Facility (and all of the Commitments pertaining thereto) pursuant to the terms of this Agreement.

- 3. <u>Section 2.10 Optional Increase to the Commitments</u>. The reference to "\$140,000,000" contained in the introductory paragraph of <u>Section 2.10</u> of the Credit Agreement is hereby amended in full to read "\$150,000,000".
 - 4. Section 4.9 Litigation. The first sentence of Section 4.9 of the Credit Agreement is deleted in its entirety and replaced with the following:

Except for (a) any matter fully covered as to subject matter and amount (subject to applicable deductibles and retentions) by insurance for which the insurance carrier has not asserted lack of subject matter coverage or reserved the right to do so, (b) any matter, or series of related matters, involving a claim against Borrower or any of its Subsidiaries, that is reasonably likely to result in a payment by Borrower or any such Subsidiary of less than \$1,000,000, (c) matters of an administrative nature involving a claim or charge against Borrower or any of its Subsidiaries of less than \$1,000,000, (d) matters before the California Public Utilities Commission involving a claim or charge against Borrower or any of its Subsidiaries which would not reasonably be expected to have a Material Adverse Effect, (e) matters of an administrative nature not involving a claim or charge against Borrower or any Subsidiary of Borrower and (f) matters set forth in Schedule 4.9, there are no actions, suits, proceedings or investigations pending as to which Borrower or any Subsidiary has been served or have received notice or, to the best knowledge of Borrower threatened against or affecting Borrower or any of its Subsidiaries or any Property of any of them before any Governmental Agency.

- 5. Section 6.1 Prepayment of Indebtedness. Section 6.1 of the Credit Agreement is amended by (a) deleting the "and" immediately preceding clause (c) therein and replacing it with a ",", (b) deleting the "." immediately following clause (c) therein and replacing it with an "and" and (c) inserting a new clause (d) as follows:
 - (d) the prepayment of Indebtedness in Cash without the issuance of additional Indebtedness; <u>provided</u> that no Default or Event of Default shall exist or shall result from such prepayment.
 - 6. Section 6.3 Disposition of Property. Section 6.3 of the Credit Agreement is deleted in its entirety and replaced with the following:

Make any Disposition of its Property, whether now owned or hereafter acquired, except (a) Dispositions of obsolete Property or Property with no material useful life, (b) Dispositions in an aggregate amount not to exceed \$5,000,000 in any Fiscal Year ending after the Closing Date or \$10,000,000 in the aggregate from and after the Amendment No.4 Effective Date to the Termination Date; provided that (i) at the time of any such Disposition pursuant to clause (b) only, no Default or Event of Default shall exist or shall result from such Disposition and (ii) the sales price relating to a Disposition (pursuant to clause (a) or (b)) shall be paid in Cash and/or Indebtedness or other evidence of an Investment permitted under Section 6.14(h), (c) Dispositions pursuant to any order of any Governmental Agency in an eminent domain proceeding and any settlement of any such proceeding, (d) a Permitted Water Rights Disposition, (e) as otherwise permitted pursuant to a letter agreement dated May 27, 2010, by and between the Administrative Agent and the Borrower, and (f) the sale of any or

all of the equity interests owned by Borrower or GSW in the Pomona Valley Protective Association so long as the consideration received for such sale is greater than or equal to the book value of such equity interests as of the Amendment No. 4 Effective Date.

- 7. <u>Sections 6.9, 6.10, 6.14</u> <u>Sections 6.9, 6.10</u> and <u>6.14</u> of the Credit Agreement are hereby amended by replacing the phrase "suffer to exist" contained in each such section and replacing it with "permit to exist".
- 8. Section 6.14 Investments and Acquisitions. Section 6.14 of the Credit Agreement is amended by (a) deleting the reference to \$500,000 contained in the proviso to clause (g) and replacing it with \$1,000,000, (b) deleting the reference to \$500,000 contained in clause (h) and replacing it with \$1,000,000, (c) deleting the "and" immediately following clause (h), (d) deleting the "." immediately following clause (i) and replacing it with a ";" and (e) inserting the new clauses (j), (k), (l), (m) and (n) as follow:
 - (j) advances in aid of construction, contributions in aid of construction or similar advances made by GSW in the ordinary course of business;
 - (k) other Investments made in the ordinary course of business in an amount not to exceed \$1,000,000 in any Fiscal Year; provided that at the time of making such Investment and after giving effect thereto, no Default or Event of Default shall exist or shall result from such Investment;
 - (1) Investments made by Borrower and GSW in the Pomona Valley Protective Association prior to June 3, 2005;
 - (m) Investments in the form of a promissory note received from Aerojet-General Corporation, as the maker, in favor of Borrower in settlement of litigation in the aggregate amount of \$8,000,000 payable ratable over a period of five years commencing in December 2009; and
 - (n) Investments in fixed income and equity securities made and to be made by Borrower or any of its Subsidiaries in a manner consistent with past practice and in accordance with Borrower's investment policy in a Rabbi Trust established to fund GSW's obligations under a Pension Restoration Plan for executive officers of Borrower and its Subsidiaries.
- 9. Section 9.1(1) Events of Default. Section 9.1(1) of the Credit Agreement is amended by deleting in its entirety the reference contained therein to "or Chapparal City Water Company".
- 10. <u>Schedule 1.1 (Lender Commitments/Pro Rata Shares)</u>. <u>Schedule 1.1</u> to the Credit Agreement is hereby amended in full to read as set forth on <u>Annex I</u> to this Amendment.
- 11. <u>Schedule 4.9 (Material Litigation)</u>. <u>Schedule 4.9</u> to the Credit Agreement is hereby amended in full to read as set forth on <u>Annex II</u> to this Amendment.
- 12. <u>Upfront Fee</u>. In consideration of the agreements set forth herein, Borrower hereby agrees to pay to the Administrative Agent, for the ratable account of the Lenders, an upfront fee in an aggregate amount equal to 0.155% of the commitments of each of the Lenders (\$155,000) (the "<u>Upfront Fee</u>"). The entire amount of the Upfront Fee will be fully earned and shall be due and payable in full in cash on the Effective Date.
- 13. <u>Effectiveness</u>. This Amendment shall become effective on the date that each of the following conditions shall have been satisfied in a manner satisfactory to the Lenders in their sole and absolute discretion (the "Effective Date"):

- (a) The Administrative Agent shall have received all of the following in form and substance satisfactory to the Administrative Agent and its legal counsel:
 - (i) duly executed counterparts of this Amendment signed by the parties hereto;
 - (ii) duly executed Note(s) for the Lender with a revised or new Commitment;
 - (iii) the favorable written legal opinion of Winston & Strawn LLP, special counsel to the Borrower;
- (iv) duly executed Assignment and Acceptances from each of CoBank, ACB, Comerica Bank and U.S. Bank National Association assigning their respective interests in the Obligations and Commitments to Wells Fargo Bank, National Association; and
- (v) such other assurances, certificates, documents, consents or opinions as the Administrative Agent and/or any Lender reasonably may require.
 - (b) The Upfront Fee which is due and payable on the Effective Date shall have been paid.
- (c) Borrower shall have paid to the Administrative Agent, for the benefit of Lenders existing immediately prior to the Effective Date, interest in the amount of \$0 and accrued fees in the amount of \$17,267.42 owed to such Lenders on the Effective Date.
- (d) The reasonable costs and expenses of the Administrative Agent in connection with the preparation of this Agreement and each of the other Loan Documents prepared in connection therewith payable pursuant to <u>Section 11.3</u>, and invoiced to Borrower prior to the Effective Date, shall have been paid.
- (e) All legal matters relating to the Loan Documents shall be reasonably satisfactory to Bingham McCutchen LLP, special counsel to the Administrative Agent.
- 14. <u>Integration; Loan Document</u>. Except as specifically provided herein, all terms and conditions of the Credit Agreement remain in full force and effect, without waiver or modification. This Amendment and the Credit Agreement shall be read together, as one document. This Amendment shall constitute a Loan Document.
- 15. Representations and Warranties; No Defaults. Borrower hereby represents that all representations and warranties contained in Article 4 of the Credit Agreement (except to the extent such representations and warranties expressly speak as of a prior date, in which case they were true and correct in all material respects on and as of such earlier date) are and will be true and correct in all material respects on and as of the Amendment No. 4 Effective Date and reaffirms all covenants set forth therein. Borrower further certifies that as of the date of, and after giving effect to, this Amendment, there exists no Event of Default as defined in the Credit Agreement, nor any condition, act or event which with the giving of notice or the passage of time or both would constitute any such Event of Default.
- 16. <u>Governing Law.</u> THIS AMENDMENT SHALL BE GOVERNED BY, AND CONSTRUED AND ENFORCED IN ACCORDANCE WITH, THE LAWS OF CALIFORNIA APPLICABLE TO CONTRACTS MADE AND PERFORMED IN CALIFORNIA.
- 17. <u>Counterparts</u>. This Amendment may be executed in any number of counterparts and any party hereto may execute any counterpart, each of which when executed and delivered will be deemed to be an original and all of which counterparts of this Amendment, when taken together, will be deemed to be but one and the same instrument and execution of any such counterpart may be evidenced by facsimile or other electronic means of transmission of the signature of such party.

[Remainder of page intentionally left blank; signature pages follow.]

IN WITNESS WHEREOF, the parties hereto have caused this Amendment to be executed as of the day and year first above

written.

BORROWER:

AMERICAN STATES WATER COMPANY,

a California corporation

By: /s/ Robert J. Sprowls

Name: Robert J. Sprowls

Title: President & Chief Executive Officer

Fourth Amendment to Amended and Restated Credit Agreement

ADMINISTRATIVE AGENT:

WELLS FARGO BANK, NATIONAL ASSOCIATION,

as Administrative Agent

By: /s/ Duvon G. Davis
Name: Duvon G. Davis

Name: Duvon G. Davi Title: Vice President

Fourth Amendment to Amended and Restated Credit Agreement

LENDERS:

WELLS FARGO BANK, NATIONAL ASSOCIATION,

as a Lender

By: /s/ Duvon G. Davis Name: Duvon G. Davis

Title: Vice President

Fourth Amendment to Amended and Restated Credit Agreement

ANNEX I

SCHEDULE 1.1

LENDER COMMITMENTS/PRO RATA SHARES

Lender	Pro Rata Share		Commitment Amount
Wells Fargo Bank, National Association	100.0	%\$	100,000,000
	100.0	%\$	100,000,000

LIMITED CONSENT

This LIMITED CONSENT (this "Consent") is entered into as of March 24, 2014 among AMERICAN STATES WATER COMPANY, a California corporation ("Borrower"), each of the lenders party to the Credit Agreement (defined below), and WELLS FARGO BANK, NATIONAL ASSOCIATION, as administrative agent for the Lenders (in such capacity, the "Administrative Agent") and lead arranger. Capitalized terms used herein and not otherwise defined shall have the meanings set forth for such terms in the Credit Agreement. Section references herein are to sections of the Credit Agreement unless otherwise stated.

RECITALS

- A. Borrower, the Lenders and the Administrative Agent are party to that Amended and Restated Credit Agreement dated as of June 3, 2005 (as amended through the date hereof and as further amended, restated, supplemented or otherwise modified from time to time, the "Credit Agreement"), pursuant to which, the Lenders have made certain credit facilities available to Borrower.
- B. Borrower has requested that the Lenders consent to Borrower being able to repurchase issued and outstanding common shares of Borrower (the "Borrower Securities Repurchase").
 - C. Subject to the terms and conditions set forth herein, the Lenders are willing to so consent to the Borrower Securities Repurchase.

NOW, THEREFORE, for valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the parties hereto agree that:

1. Limited Consent.

- (a) Notwithstanding any provision of the Credit Agreement and any other Loan Document to the contrary (including Sections 6.6, 6.11 and 6.14 of the Credit Agreement), the Requisite Lenders and the Administrative Agent hereby consent to the Borrower Securities Repurchase so long as:
 (i) at the time of any repurchase of Securities and after giving effect thereto, no Default or Event of Default has occurred and is continuing; (ii) all Securities purchased pursuant to the Borrower Securities Repurchase are acquired on or before June 30, 2016; and (iii) the aggregate amount of Securities purchased pursuant to the Borrower Securities Repurchase does not exceed the lesser of (A) \$50,000,000 and (B) 1,250,000 common shares of the Borrower.
- (b) The consent set forth in this Section 1 shall be limited precisely as written and shall not be deemed (i) to be a consent, waiver or amendment of compliance with any other term or condition of the Credit Agreement or any of the other Loan Documents or to prejudice any right or remedy which the Administrative Agent or any Lender may now have or may have in the future under or in connection with the Credit Agreement or the other Loan Documents; (ii) to be an amendment, consent or waiver of any other term or condition of the Credit Agreement or the other Loan Documents, to prejudice any right or remedy which the Administrative Agent or any Lender may now have or may have in the future under or in connection with the Credit Agreement or the other Loan Documents; or (iii) to be a consent to any future amendment, consent or waiver or departure from the terms and conditions of the Credit Agreement or the other Loan Documents. This Consent is a Loan Document and shall be construed in connection with and as part of the Loan Documents, and all terms, conditions, representations, warranties, covenants and agreements set forth in the Loan Documents, except as herein waived or amended, are hereby ratified and confirmed and shall remain in full force and effect. This Consent is a consent to the Credit Agreement and the other Loan Documents without any discharge, release or satisfaction of the existing obligations or indebtedness (or any guaranty or collateral security therefor), all of which obligations, indebtedness and security remain outstanding under the Loan Documents. The Borrower hereby reaffirms its obligations under each Loan Document to which it is a party.
- 2. <u>Representations and Warranties; No Defaults</u>. The Borrower hereby represents that all representations and warranties contained in Article 4 of the Credit Agreement (except to the extent such representations and warranties expressly speak as

of a prior date, in which case they were true and correct in all material respects on and as of such earlier date) are and will be true and correct in all material respects on and as of the date of this Consent and reaffirms all covenants set forth therein. The Borrower further certifies that as of the date of, and after giving effect to, this Consent, there exists no Event of Default as defined in the Credit Agreement, nor any condition, act or event which with the giving of notice or the passage of time or both would constitute any such Event of Default.

- 3. Effectiveness. This Consent shall become effective upon the satisfaction of each of the following conditions:
- (a) <u>Documentation</u> The Administrative Agent shall have received a fully executed copy of this Consent and such other certificates or documents related to the Borrower Securities Repurchase as the Administrative Agent may reasonably request.

- (b) Consent Fee. Receipt by the Administrative Agent of a consent fee of \$0, payable in cash, which fee will be fully earned and non-refundable once received.
- (c) <u>Fees and Expenses</u>. The reasonable costs and expenses of the Administrative Agent in connection with the preparation of this Consent and any other document prepared in connection herewith payable pursuant to <u>Section 11.3</u>, and invoiced to Borrower prior to the date hereof, shall have been paid.
- (d) Other Matters. All legal matters relating to this Consent and the other Loan Documents shall be reasonably satisfactory to Bingham McCutchen LLP, special counsel to the Administrative Agent.
- 4. Governing Law. THIS CONSENT SHALL BE GOVERNED BY, AND CONSTRUED AND ENFORCED IN ACCORDANCE WITH, THE LAWS OF CALIFORNIA APPLICABLE TO CONTRACTS MADE AND PERFORMED IN CALIFORNIA.
- 5. <u>Counterparts</u>. This Consent may be executed in any number of counterparts and any party hereto may execute any counterpart, each of which when executed and delivered will be deemed to be an original and all of which counterparts of this Consent, when taken together, will be deemed to be but one and the same instrument and execution of any such counterpart may be evidenced by facsimile or other electronic means of transmission of the signature of such party.

[remainder of page intentionally blank]

IN WITNESS WHEREOF, the parties hereto have caused this Consent to be executed as of the day and year first above written.

BORROWER:

AMERICAN STATES WATER COMPANY,

a California corporation

By: /s/ Eva G. Tang

Name: Eva G. Tang

Title: Senior Vice President — Finance, Chief Financial Officer, Corporate Secretary and Treasurer

ADMINISTRATIVE AGENT:

WELLS FARGO BANK, NATIONAL ASSOCIATION,

as Administrative Agent

By: /s/ Duvon G. Davis
Name: DuVon G. Davis
Title: Vice President

LENDERS:

WELLS FARGO BANK, NATIONAL ASSOCIATION,

as a Lender

By: /s/ Duvon G. Davis
Name: DuVon G. Davis
Title: Vice President

LIMITED CONSENT

This LIMITED CONSENT (this "Consent") is entered into as of May 20, 2015 among AMERICAN STATES WATER COMPANY, a California corporation ("Borrower"), each of the lenders party to the Credit Agreement (defined below), and WELLS FARGO BANK, NATIONAL ASSOCIATION, as administrative agent for the Lenders (in such capacity, the "Administrative Agent") and lead arranger. Capitalized terms used herein and not otherwise defined shall have the meanings set forth for such terms in the Credit Agreement. Section references herein are to sections of the Credit Agreement unless otherwise stated.

RECITALS

- A. Borrower, the Lenders and the Administrative Agent are party to that Amended and Restated Credit Agreement dated as of June 3, 2005 (as amended through the date hereof and as further amended, restated, supplemented or otherwise modified from time to time, the "Credit Agreement"), pursuant to which, the Lenders have made certain credit facilities available to Borrower.
- B. Borrower has requested that the Lenders consent to Borrower being able to repurchase issued and outstanding common shares of Borrower (the "Borrower Securities Repurchase").
 - C. Subject to the terms and conditions set forth herein, the Lenders are willing to so consent to the Borrower Securities Repurchase.

NOW, THEREFORE, for valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the parties hereto agree that:

6. Limited Consent.

- (a) Notwithstanding any provision of the Credit Agreement and any other Loan Document to the contrary (including Sections 6.6, 6.11 and 6.14 of the Credit Agreement), the Requisite Lenders and the Administrative Agent hereby consent to the Borrower Securities Repurchase so long as:
 (i) at the time of any repurchase of Securities and after giving effect thereto, no Default or Event of Default has occurred and is continuing; (ii) all Securities purchased pursuant to the Borrower Securities Repurchase are acquired on or before June 30, 2017; and (iii) the aggregate amount of Securities purchased pursuant to the Borrower Securities Repurchase does not exceed the lesser of (A) \$60,000,000 and (B) 1,200,000 common shares of the Borrower.
- (b) The consent set forth in this Section 1 shall be limited precisely as written and shall not be deemed (i) to be a consent, waiver or amendment of compliance with any other term or condition of the Credit Agreement or any of the other Loan Documents or to prejudice any right or remedy which the Administrative Agent or any Lender may now have or may have in the future under or in connection with the Credit Agreement or the other Loan Documents; (ii) to be an amendment, consent or waiver of any other term or condition of the Credit Agreement or the other Loan Documents, to prejudice any right or remedy which the Administrative Agent or any Lender may now have or may have in the future under or in connection with the Credit Agreement or the other Loan Documents; or (iii) to be a consent to any future amendment, consent or waiver or departure from the terms and conditions of the Credit Agreement or the other Loan Documents. This Consent is a Loan Document and shall be construed in connection with and as part of the Loan Documents, and all terms, conditions, representations, warranties, covenants and agreements set forth in the Loan Documents, except as herein waived or amended, are hereby ratified and confirmed and shall remain in full force and effect. This Consent is a consent to the Credit Agreement and the other Loan Documents without any discharge, release or satisfaction of the existing obligations or indebtedness (or any guaranty or collateral security therefor), all of which obligations, indebtedness and security remain outstanding under the Loan Documents. The Borrower hereby reaffirms its obligations under each Loan Document to which it is a party.
- 7. Representations and Warranties; No Defaults. The Borrower hereby represents that all representations and warranties contained in Article 4 of the Credit Agreement (except to the extent such representations and warranties expressly speak as of a prior date, in which case they were true and correct in all material respects on and as of such earlier date) are and will be true and

correct in all material respects on and as of the date of this Consent and reaffirms all covenants set forth therein. The Borrower further certifies that as of the date of, and after giving effect to, this Consent, there exists no Event of Default as defined in the Credit Agreement, nor any condition, act or event which with the giving of notice or the passage of time or both would constitute any such Event of Default.

- 8. Effectiveness. This Consent shall become effective upon the satisfaction of each of the following conditions:
- (a) <u>Documentation</u> The Administrative Agent shall have received a fully executed copy of this Consent and such other certificates or documents related to the Borrower Securities Repurchase as the Administrative Agent may reasonably request.
- (b) Consent Fee. Receipt by the Administrative Agent of a consent fee of \$0, payable in cash, which fee will be fully earned and non-refundable once received.
- (c) <u>Fees and Expenses</u>. The reasonable costs and expenses of the Administrative Agent in connection with the preparation of this Consent and any other document prepared in connection herewith payable pursuant to <u>Section 11.3</u>, and invoiced to Borrower prior to the date hereof, shall have been paid.
- (d) Other Matters. All legal matters relating to this Consent and the other Loan Documents shall be reasonably satisfactory to Morgan Lewis & Bockius LLP, special counsel to the Administrative Agent.
- 9. Governing Law. THIS CONSENT SHALL BE GOVERNED BY, AND CONSTRUED AND ENFORCED IN ACCORDANCE WITH, THE LAWS OF CALIFORNIA APPLICABLE TO CONTRACTS MADE AND PERFORMED IN CALIFORNIA.
- 10. <u>Counterparts</u>. This Consent may be executed in any number of counterparts and any party hereto may execute any counterpart, each of which when executed and delivered will be deemed to be an original and all of which counterparts of this Consent, when taken together, will be deemed to be but one and the same instrument and execution of any such counterpart may be evidenced by facsimile or other electronic means of transmission of the signature of such party.

[remainder of page intentionally blank]

IN WITNESS WHEREOF, the parties hereto have caused this Consent to be executed as of the day and year first above written.

BORROWER:

AMERICAN STATES WATER COMPANY,

a California corporation

By: /s/ Robert J. Sprowls

Name: Robert J. Sprowls

Title: President and Chief Executive Officer

ADMINISTRATIVE AGENT:

WELLS FARGO BANK, NATIONAL ASSOCIATION,

as Administrative Agent

By: /s/ DuVon G. Davis

Name: DuVon G. Davis
Title: Vice President

LENDERS:

WELLS FARGO BANK, NATIONAL ASSOCIATION,

as a Lender

By: /s/ DuVon G. Davis

Name: DuVon G. Davis
Title: Vice President

FIFTH AMENDMENT TO AMENDED AND RESTATED CREDIT AGREEMENT

THIS FIFTH AMENDMENT TO AMENDED AND RESTATED CREDIT AGREEMENT (this "Amendment"), dated as of October 26, 2016, is entered into with reference to the Amended and Restated Credit Agreement (as amended, restated, supplemented or otherwise modified from time to time, the "Credit Agreement") dated as of June 3, 2005, by and among AMERICAN STATES WATER COMPANY, a California corporation ("Borrower"), each of the lenders party thereto, and WELLS FARGO BANK, NATIONAL ASSOCIATION, as administrative agent for the Lenders (in such capacity, the "Administrative Agent") and lead arranger. Capitalized terms used herein and not otherwise defined shall have the meanings set forth for such terms in the Credit Agreement. Section references herein are to sections of the Credit Agreement unless otherwise stated.

RECITALS

WHEREAS, Borrower has requested that the Credit Agreement be modified as set forth in this Amendment, <u>including</u>, without limitation, that the amount of the Revolving Facility be increased by \$50,000,000 (to \$150,000,000) on the Effective Date hereof.

WHEREAS, the Administrative Agent, Wells Fargo, as the sole existing Lender as of the date hereof, and Borrower have agreed to make certain changes in the terms and conditions set forth in the Credit Agreement, and have agreed to amend the Credit Agreement to reflect said changes.

NOW, THEREFORE, for valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the parties hereto agree that the Credit Agreement shall be amended as follows:

- 1. Section 1.1 Defined Terms (New). The following defined terms are hereby added to Section 1.1 in the appropriate alphabetical place:
- "Amendment No. 5" means the Fifth Amendment to Amended and Restated Credit Agreement, dated as of October 26, 2016, among Borrower, each of the Lenders party thereto and the Administrative Agent.
 - "Amendment No. 5 Effective Date" means the "Effective Date" as defined in Amendment No. 5.
- 2. <u>Section 1.1 Defined Terms (Revisions)</u>. The following defined terms contained in <u>Section 1.1</u> of the Credit Agreement are hereby amended in full to read as follows:

"Commitment" means, with respect to each Lender, the commitment, if any, of such Lender to make Advances (expressed as the maximum aggregate amount of the Advances to be made by such Lender hereunder), as such commitment may be (a) reduced from time to time pursuant to Section 2.6, (b) increased pursuant to Section 2.10, and (c) reduced or increased from time to time pursuant to assignments by or to such Lender pursuant to Section 11.8. The amount of each Lender's Commitment on the Amendment No. 5 Effective Date is set forth on Schedule 1.1 or in the Assignment and Acceptance pursuant to which such Lender shall have assumed its Commitment, as applicable. The aggregate amount of the Lenders' Commitments on the Amendment No. 5 Effective Date is \$150,000,000.

"Eurodollar Rate" means for any Eurodollar Period, with respect to a Eurodollar Rate Advance, a rate per annum (rounded upwards, as necessary, to the nearest 1/16th of one percent (0.0625%)) obtained by dividing (a) the rate *per annum* determined by the Administrative Agent at approximately 11:00 a.m., London time, on the date that is two Eurodollar Banking Days prior to the beginning of such Eurodollar Period by reference to the British Bankers' Association "Interest Settlement Rates" for deposits in Dollars (as set forth by any service (including Bloomberg, Reuters and Thomson Financial)

selected by the Administrative Agent that has been nominated by the British Bankers' Association as an authorized information vendor for the purpose of displaying such rates) in an amount approximately equal to the principal amount to which such Eurodollar Period applies (for delivery on the first day of such Eurodollar Period) with a term equivalent to such Eurodollar Period; provided that, if an interest rate is not ascertainable pursuant to the foregoing provisions of this definition, then "Eurodollar Rate" shall be the interest rate *per annum* determined by the Administrative Agent to be the average of the rates per annum at which deposits in Dollars in an amount approximately equal to the principal amount to which such Eurodollar Period applies (for delivery on the first day of such Eurodollar Period) with a term equivalent to such Eurodollar Period are offered for such Eurodollar Period by Wells Fargo to major banks in the London interbank offered market in London, England at approximately 11:00 a.m., London time, on the date that is two Eurodollar Banking Days prior to the beginning of such Eurodollar Period by (b) one minus the Reserve Requirement in effect on such date. Each determination by the Administrative Agent pursuant to this definition shall be conclusive absent manifest error. Notwithstanding the foregoing, in no event shall the Eurodollar Rate be less than 0%.

"Federal Funds Rate" means for any day, the rate per annum equal to the weighted average of the rates on overnight Federal funds transactions with members of the Federal Reserve System arranged by Federal funds brokers on such day, as published by the Federal Reserve Bank of New York on the Banking Day next succeeding such day; provided that (a) if such day is not a Banking Day, then the Federal Funds Rate for such day shall be such rate on such transactions on the next preceding Banking Day as so published on the next succeeding Banking Day and (b) if no such rate is so published on such next succeeding Banking Day, then the Federal Funds Rate for such day shall be the average rate (rounded upward, if necessary, to a whole multiple of one one-hundredth of one percent (0.01%)) charged to Wells Fargo on such day on such transactions as determined by Administrative Agent. Notwithstanding the foregoing, if the Federal Funds Rate shall be less than 0%, such rate shall be deemed to be 0% for purposes of the Loan Documents.

- 3. <u>Schedule 1.1 (Lender Commitments/Pro Rata Shares)</u>. <u>Schedule 1.1</u> to the Credit Agreement is hereby amended in full to read as set forth on <u>Annex I</u> to this Amendment.
- 4. <u>Amendment Fee.</u> In consideration of the agreements set forth herein, Borrower hereby agrees to pay to the Administrative Agent, for the ratable account of the Lenders, an amendment fee in an aggregate amount equal to \$37,500 (which is 0.075% (7.5 basis points) on that portion of the increase in the Commitments of each of the Lenders made pursuant hereto) (the "<u>Amendment Fee</u>"). The entire amount of the Amendment Fee will be fully earned and shall be due and payable in full in cash on the Effective Date.
- 5. <u>Effectiveness</u>. This Amendment shall become effective on the date that each of the following conditions shall have been satisfied in a manner satisfactory to the Lenders in their sole and absolute discretion (the "<u>Effective Date</u>"):
- (a) The Administrative Agent shall have received all of the following in form and substance satisfactory to the Administrative Agent and its legal counsel:
 - (i) duly executed counterparts of this Amendment signed by the parties hereto;
 - (ii) duly executed Note(s) for the Lender with a revised or new Commitment;
 - (iii) the favorable written legal opinion of Winston & Strawn LLP, special counsel to Borrower; and
- (iv) such other assurances, certificates, documents, consents or opinions as the Administrative Agent and/or any Lender reasonably may require.
 - (b) The Amendment Fee which is due and payable on the Effective Date shall have been paid.
- (c) The reasonable costs and expenses of the Administrative Agent in connection with the preparation of this Agreement and each of the other Loan Documents prepared in connection therewith payable pursuant to <u>Section 11.3</u>, and invoiced to Borrower prior to the Effective Date, shall have been paid.
- (d) All legal matters relating to the Loan Documents shall be reasonably satisfactory to McGuireWoods LLP, special counsel to the Administrative Agent.
- 6. <u>Integration; Loan Document</u>. Except as specifically provided herein, all terms and conditions of the Credit Agreement remain in full force and effect, without waiver or modification. This Amendment and the Credit Agreement shall be read together, as one document. This Amendment shall constitute a Loan Document.

7. Representations and Warranties; No Defaults. Borrower hereby represents that all representations and warranties contained in Article 5 of the Credit Agreement (except to the extent such representations and warranties expressly speak as of a prior date, in which case they were true and correct in all material respects on and as of such earlier date) are and will be true and correct in all material respects on and as of the Amendment No. 5 Effective Date and reaffirms all covenants set forth therein. Borrower further certifies that as of the date of, and after giving effect to, this Amendment, there exists no Event of Default as defined in the Credit Agreement, nor any condition, act or event which with the giving of notice or the passage of time or both would constitute any such Event of Default.

- 8. Governing Law. THIS AMENDMENT SHALL BE GOVERNED BY, AND CONSTRUED AND ENFORCED IN ACCORDANCE WITH, THE LAWS OF CALIFORNIA APPLICABLE TO CONTRACTS MADE AND PERFORMED IN CALIFORNIA.
- 9. <u>Counterparts</u>. This Amendment may be executed in any number of counterparts and any party hereto may execute any counterpart, each of which when executed and delivered will be deemed to be an original and all of which counterparts of this Amendment, when taken together, will be deemed to be but one and the same instrument and execution of any such counterpart may be evidenced by facsimile or other electronic means of transmission of the signature of such party.

[Remainder of page intentionally left blank; signature pages follow.]

IN WITNESS WHEREOF, the parties hereto have caused this Amendment to be executed as of the day and year first above written.

BORROWER:

AMERICAN STATES WATER COMPANY, a California corporation

By: /s/ Eva G. Tang

Name: Eva G. Tang

Senior Vice President, Chief Financial Officer, Corporate Secretary and Treasurer

Fifth Amendment to Amended and Restated Credit Agreement

ADMINISTRATIVE AGENT:

WELLS FARGO BANK, NATIONAL ASSOCIATION, as Administrative Agent

By: /s/ DuVon G. Davis
Name: DuVon G. Davis
Title: Vice President

Fifth Amendment to Amended and Restated Credit Agreement

LENDERS:

WELLS FARGO BANK, NATIONAL ASSOCIATION, as a Lender

 By:
 /s/ DuVon G. Davis

 Name:
 DuVon G. Davis

 Title:
 Vice President

Fifth Amendment to Amended and Restated Credit Agreement

ANNEX I

SCHEDULE 1.1

LENDER COMMITMENTS/PRO RATA SHARES

Lender	Pro Rata Share		Commitment Amount
Wells Fargo Bank, National Association	100.0	%\$	150,000,000
	100.0	%\$	150,000,000

SIXTH AMENDMENT TO AMENDED AND RESTATED CREDIT AGREEMENT

THIS SIXTH AMENDMENT TO AMENDED AND RESTATED CREDIT AGREEMENT (this "Amendment"), dated as of May 23, 2018, is entered into with reference to the Amended and Restated Credit Agreement (as amended, restated, supplemented or otherwise modified from time to time, the "Credit Agreement") dated as of June 3, 2005, by and among AMERICAN STATES WATER COMPANY, a California corporation ("Borrower"), each of the lenders party thereto, and WELLS FARGO BANK, NATIONAL ASSOCIATION, as administrative agent for the Lenders (in such capacity, the "Administrative Agent") and lead arranger. Capitalized terms used herein and not otherwise defined shall have the meanings set forth for such terms in the Credit Agreement. Section references herein are to sections of the Credit Agreement unless otherwise stated.

RECITALS

WHEREAS, Borrower has requested that the Credit Agreement be modified as set forth in this Amendment.

WHEREAS, the Administrative Agent, Wells Fargo, as the sole existing Lender as of the date hereof, and Borrower have agreed to make certain changes in the terms and conditions set forth in the Credit Agreement, and have agreed to amend the Credit Agreement to reflect said changes.

NOW, THEREFORE, for valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the parties hereto agree that the Credit Agreement shall be amended as follows:

- 1. Section 1.1 Defined Terms (New). The following defined terms are hereby added to Section 1.1 in the appropriate alphabetical place:
- "Amendment No. 6" means the Sixth Amendment to Amended and Restated Credit Agreement, dated as of May 23, 2018, among Borrower, each of the Lenders party thereto and the Administrative Agent.
 - "Amendment No. 6 Effective Date" means the "Effective Date" as defined in Amendment No. 6.
- 2. <u>Section 1.1 Defined Terms (Revisions)</u>. The following defined terms contained in <u>Section 1.1</u> of the Credit Agreement are hereby amended in full to read as follows:
 - "Applicable Commitment Fee Margin" means, for each Pricing Period, the margin set forth below (expressed in basis points per annum) opposite the Applicable Pricing Level for that Pricing Period:

Applicable Pricing

Level	Margin
I	5.00
II	7.25
III	10.00
IV	12.50
V	15.00

"Applicable Eurodollar Rate Margin" means, with respect to any Eurodollar Rate Advance, for each Pricing Period, the interest rate margin set forth below (expressed in basis points per annum) opposite the Applicable Pricing Level for that Pricing Period:

Applicable Pricing

Level	Margin
I	52.50
II	62.50
III	72.50
IV	82.50
V	92.50

"<u>Applicable Pricing Level</u>" means, (a) for the Initial Pricing Period, Pricing Level II, and (b) thereafter, the pricing level set forth below opposite the Debt Rating achieved by Borrower as of the first day of that Pricing Period:

Pricing Level	Debt Rating
I	Greater than or equal to Aa3 / AA-
II	Less than Aa3 / AA- but greater than or equal to A1/A+
III	Less than A1/A+ but greater than or equal to A2/A
IV	Less than A2/A but greater than or equal to A3/A-
V	Less than A3/A-

provided that in the event that the then prevailing Debt Ratings are "split ratings", Borrower will receive the benefit of the higher Debt Rating, unless the split is a "double split rating" (in which case the pricing level applicable to the middle Debt Rating will apply) or a "triple split rating" (in which case the pricing level applicable to the Debt Rating applicable to the lowest pricing level will apply). For purposes hereof, a Debt Rating is only a "split rating" if the Debt Rating applies to a different pricing level.

"<u>Capital Lease</u>" means, as to any Person, a lease of any Property by that Person as lessee that is, or should be in accordance with GAAP (<u>including Financial Accounting Standards Board Statement No. 13</u>, as amended or superseded from time to time, and/or Accounting Standards Codification 842 "*Leases*"), recorded as a "capital lease" or as a "financing lease", as the case may be, on the balance sheet of that Person prepared in accordance with GAAP.

"Initial Pricing Period" means the period commencing on the Amendment No. 6 Effective Date and ending on the first Pricing Occurrence to occur thereafter.

"Maturity Date" means the earlier of (a) May 23, 2023 and (b) the termination or cancellation of the Revolving Facility (and all of the Commitments pertaining thereto) pursuant to the terms of this Agreement.

"Pricing Period" means (a) the Initial Pricing Period and (b) each subsequent period commencing on the date of a Pricing Occurrence and ending on the next Pricing Occurrence to occur.

- 3. <u>Section 2.10 Optional Increase to the Commitments</u>. The reference to "\$150,000,000" contained in the introductory paragraph of <u>Section 2.10</u> of the Credit Agreement is hereby amended in full to read "\$200,000,000".
- 4. <u>Effectiveness</u>. This Amendment shall become effective on the date that each of the following conditions shall have been satisfied in a manner satisfactory to the Lenders in their sole and absolute discretion (the "<u>Effective Date</u>"):
- (a) The Administrative Agent shall have received all of the following in form and substance satisfactory to the Administrative Agent and its legal counsel:
 - (i) duly executed counterparts of this Amendment signed by the parties hereto; and
- (ii) such other assurances, certificates, documents or consents as the Administrative Agent and/or any Lender reasonably may require.
- (b) The Upfront Fee described in that certain letter agreement dated as of May 17, 2018 between the Administrative Agent and Borrower shall have been paid.

- (c) The reasonable costs and expenses of the Administrative Agent in connection with the preparation of this Agreement and each of the other Loan Documents prepared in connection therewith payable pursuant to <u>Section 11.3</u>, and invoiced to Borrower prior to the Effective Date, shall have been paid.
- (d) All legal matters relating to the Loan Documents shall be reasonably satisfactory to McGuireWoods LLP, special counsel to the Administrative Agent.
- 5. <u>Integration; Loan Document</u>. Except as specifically provided herein, all terms and conditions of the Credit Agreement remain in full force and effect, without waiver or modification. This Amendment and the Credit Agreement shall be read together, as one document. This Amendment shall constitute a Loan Document.
- 6. Representations and Warranties; No Defaults. Borrower hereby represents that all representations and warranties contained in Article 5 of the Credit Agreement (except to the extent such representations and warranties expressly speak as of a prior date, in which case they were true and correct in all material respects on and as of such earlier date) are and will be true and correct in all material respects on and as of the Amendment No. 6 Effective Date and reaffirms all covenants set forth therein. Borrower further certifies that as of the date of, and after giving effect to, this Amendment, there exists no Event of Default as defined in the Credit Agreement, nor any condition, act or event which with the giving of notice or the passage of time or both would constitute any such Event of Default.
- 7. Governing Law. THIS AMENDMENT SHALL BE GOVERNED BY, AND CONSTRUED AND ENFORCED IN ACCORDANCE WITH, THE LAWS OF CALIFORNIA APPLICABLE TO CONTRACTS MADE AND PERFORMED IN CALIFORNIA.
- 8. <u>Counterparts</u>. This Amendment may be executed in any number of counterparts and any party hereto may execute any counterpart, each of which when executed and delivered will be deemed to be an original and all of which counterparts of this Amendment, when taken together, will be deemed to be but one and the same instrument and execution of any such counterpart may be evidenced by facsimile or other electronic means of transmission of the signature of such party.

[Remainder of page intentionally left blank; signature pages follow.]

IN WITNESS WHEREOF, the parties hereto have caused this Amendment to be executed as of the day and year first above written.

BORROWER:

AMERICAN STATES WATER COMPANY, a California corporation

By: /s/ Eva G. Tang

Name: Eva G. Tang

Senior Vice President, Chief Financial Officer, Corporate Secretary and Treasurer Title:

Sixth Amendment to Amended and Restated Credit Agreement

ADMINISTRATIVE AGENT:

WELLS FARGO BANK, NATIONAL ASSOCIATION, as Administrative Agent

By: /s/ DuVon G. Davis
Name: Du Von G. Davis
Title: Senior Vice President

Sixth Amendment to Amended and Restated Credit Agreement

LENDERS:

WELLS FARGO BANK, NATIONAL ASSOCIATION, as a Lender

By: /s/ DuVon G. Davis
Name: Du Von G. Davis

Title: Senior Vice President

Sixth Amendment to Amended and Restated Credit Agreement

SEVENTH AMENDMENT TO AMENDED AND RESTATED CREDIT AGREEMENT

THIS SEVENTH AMENDMENT TO AMENDED AND RESTATED CREDIT AGREEMENT (this "Amendment"), dated as of March 28, 2019, is entered into with reference to the Amended and Restated Credit Agreement (as amended, restated, supplemented or otherwise modified from time to time, the "Credit Agreement") dated as of June 3, 2005, by and among AMERICAN STATES WATER COMPANY, a California corporation ("Borrower"), each of the lenders party thereto, and WELLS FARGO BANK, NATIONAL ASSOCIATION, as administrative agent for the Lenders (in such capacity, the "Administrative Agent") and lead arranger. Capitalized terms used herein and not otherwise defined shall have the meanings set forth for such terms in the Credit Agreement. Section references herein are to sections of the Credit Agreement unless otherwise stated.

RECITALS

WHEREAS, Borrower has requested that the Credit Agreement be modified as set forth in this Amendment, <u>including</u>, without limitation, that the amount of the Revolving Facility be increased by \$50,000,000 (to \$200,000,000) on the Effective Date hereof.

WHEREAS, the Administrative Agent, Wells Fargo, as the sole existing Lender as of the date hereof, and Borrower have agreed to make certain changes in the terms and conditions set forth in the Credit Agreement, and have agreed to amend the Credit Agreement to reflect said changes.

NOW, THEREFORE, for valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the parties hereto agree that the Credit Agreement shall be amended as follows:

- 1. Section 1.1 Defined Terms (New). The following defined terms are hereby added to Section 1.1 in the appropriate alphabetical place:
- "Amendment No. 7" means the Seventh Amendment to Amended and Restated Credit Agreement, dated as of March 28, 2019, among Borrower, each of the Lenders party thereto and the Administrative Agent.
 - "Amendment No. 7 Effective Date" means the "Effective Date" as defined in Amendment No. 7.
- 2. <u>Section 1.1 Defined Terms (Revisions)</u>. The definition of "Commitment" contained in <u>Section 1.1</u> of the Credit Agreement is hereby amended to read as follows:

"Commitment" means, with respect to each Lender, the commitment, if any, of such Lender to make Advances (expressed as the maximum aggregate amount of the Advances to be made by such Lender hereunder), as such commitment may be (a) reduced from time to time pursuant to Section 2.6, (b) increased pursuant to Section 2.10, and (c) reduced or increased from time to time pursuant to assignments by or to such Lender pursuant to Section 11.8. The amount of each Lender's Commitment on the Amendment No. 7 Effective Date is set forth on Schedule 1.1 or in the Assignment and Acceptance pursuant to which such Lender shall have assumed its Commitment, as applicable. The aggregate amount of the Lenders' Commitments on the Amendment No. 7 Effective Date is \$200,000,000.

- 3. <u>Section 2.10 Optional Increase to the Commitments</u>. <u>Section 2.10</u> to the Credit Agreement is hereby amended to read as follows:
 - 2.10 [Reserved].

- 4. <u>Schedule 1.1 (Lender Commitments/Pro Rata Shares)</u>. <u>Schedule 1.1</u> to the Credit Agreement is hereby amended to read as set forth on <u>Annex I</u> to this Amendment.
- 5. <u>Effectiveness</u>. This Amendment shall become effective on the date that each of the following conditions shall have been satisfied in a manner satisfactory to the Lenders in their sole and absolute discretion (the "<u>Effective Date</u>"):
- (a) The Administrative Agent shall have received all of the following in form and substance satisfactory to the Administrative Agent and its legal counsel:
 - (i) duly executed counterparts of this Amendment signed by the parties hereto;

- (ii) a duly executed Note for Wells Fargo, as the sole Lender, reflecting its revised Commitment;
- (iii) updated resolutions of Borrower authorizing the increase in the commitment provided hereunder, the execution and delivery of this Amendment and the amended and restated Note described in the foregoing clause (ii), and such other matters relating hereto as Borrower's board of directors shall determine; and
- (iv) such other assurances, certificates, documents, consents or opinions as the Administrative Agent and/or any Lender reasonably may require.
- (b) An upfront fee described in that certain letter agreement dated as of March 28, 2019 between the Administrative Agent and Borrower shall have been paid.
- (c) The reasonable costs and expenses of the Administrative Agent in connection with the preparation of this Agreement and each of the other Loan Documents prepared in connection therewith payable pursuant to <u>Section 11.3</u>, and invoiced to Borrower prior to the Effective Date, shall have been paid.
- (d) All legal matters relating to the Loan Documents shall be reasonably satisfactory to McGuireWoods LLP, special counsel to the Administrative Agent.
- 6. <u>Integration; Loan Document</u>. Except as specifically provided herein, all terms and conditions of the Credit Agreement remain in full force and effect, without waiver or modification. This Amendment and the Credit Agreement shall be read together, as one document. This Amendment shall constitute a Loan Document.
- 7. Representations and Warranties; No Defaults. Borrower hereby represents that all representations and warranties contained in Article 5 of the Credit Agreement (except to the extent such representations and warranties expressly speak as of a prior date, in which case they were true and correct in all material respects on and as of such earlier date) are and will be true and correct in all material respects on and as of the Amendment No. 7 Effective Date and reaffirms all covenants set forth therein. Borrower further certifies that as of the date of, and after giving effect to, this Amendment, there exists no Event of Default as defined in the Credit Agreement, nor any condition, act or event which with the giving of notice or the passage of time or both would constitute any such Event of Default.
- 8. Governing Law. THIS AMENDMENT SHALL BE GOVERNED BY, AND CONSTRUED AND ENFORCED IN ACCORDANCE WITH, THE LAWS OF CALIFORNIA APPLICABLE TO CONTRACTS MADE AND PERFORMED IN CALIFORNIA.
- 9. <u>Counterparts</u>. This Amendment may be executed in any number of counterparts and any party hereto may execute any counterpart, each of which when executed and delivered will be deemed to be an original and all of which counterparts of this Amendment, when taken together, will be deemed to be but one and the same instrument and execution of any such counterpart may be evidenced by facsimile or other electronic means of transmission of the signature of such party.

[Remainder of page intentionally left blank; signature pages follow.]

IN WITNESS WHEREOF, the parties hereto have caused this Amendment to be executed as of the day and year first above written.

BORROWER:

AMERICAN STATES WATER COMPANY, a California corporation

By: /s/ Eva G. Tang Name:

Eva G. Tang

Senior Vice President-Finance, Chief Financial Officer, Corporate Secretary and Treasurer

Seventh Amendment to Amended and Restated Credit Agreement

ADMINISTRATIVE AGENT:

WELLS FARGO BANK, NATIONAL ASSOCIATION, as Administrative Agent

 By:
 /s/ DuVon G. Davis

 Name:
 DuVon G. Davis

 Title:
 Senior Vice President

Seventh Amendment to Amended and Restated Credit Agreement

LENDERS:

WELLS FARGO BANK, NATIONAL ASSOCIATION, as a Lender

 By:
 /s/ DuVon G. Davis

 Name:
 DuVon G. Davis

 Title:
 Senior Vice President

Seventh Amendment to Amended and Restated Credit Agreement

ANNEX I

SCHEDULE 1.1

LENDER COMMITMENTS/PRO RATA SHARES

Lender	Pro Rata Share		Commitment Amount
Wells Fargo Bank, National Association	100.0	%\$	200,000,000
	100.0	%\$	200,000,000

122769835_2

EIGHTH AMENDMENT TO AMENDED AND RESTATED CREDIT AGREEMENT

THIS EIGHTH AMENDMENT TO AMENDED AND RESTATED CREDIT AGREEMENT (this "Amendment"), dated as of October 31, 2019, is entered into with reference to the Amended and Restated Credit Agreement (as amended, restated, supplemented or otherwise modified from time to time, the "Credit Agreement") dated as of June 3, 2005, by and among AMERICAN STATES WATER COMPANY, a California corporation ("Borrower"), each of the lenders party thereto, and WELLS FARGO BANK, NATIONAL ASSOCIATION, as administrative agent for the Lenders (in such capacity, the "Administrative Agent") and lead arranger. Capitalized terms used herein and not otherwise defined shall have the meanings set forth for such terms in the Credit Agreement. Section references herein are to sections of the Credit Agreement unless otherwise stated.

RECITALS

WHEREAS, Borrower has requested that the Credit Agreement be modified as set forth in this Amendment, <u>including</u>, without limitation, that the amount of the Revolving Facility be temporarily increased by \$25,000,000 (to \$225,000,000) during the "Increased Commitment Period" (as defined below).

WHEREAS, the Administrative Agent, Wells Fargo, as the sole existing Lender as of the date hereof, and Borrower have agreed to make certain changes in the terms and conditions set forth in the Credit Agreement, and have agreed to amend the Credit Agreement to reflect said changes.

NOW, THEREFORE, for valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the parties hereto agree that the Credit Agreement shall be amended as follows:

1. <u>Section 1.1 – Defined Terms (New)</u>. The following defined terms are hereby added to <u>Section 1.1</u> in the appropriate alphabetical place:

"Amendment No. 8" means the Eighth Amendment to Amended and Restated Credit Agreement, dated as of October 31, 2019, among Borrower, each of the Lenders party thereto and the Administrative Agent.

"Amendment No. 8 Effective Date" means the "Effective Date" as defined in Amendment No. 8.

"Increased Commitment Period" means the period commencing on the Amendment No. 8 Effective Date and ending on June 30, 2020.

[Type here]

2. <u>Section 1.1 – Defined Terms (Revisions)</u>. The definitions of "Commitment" and "Maximum Revolving Credit Amount" contained in <u>Section 1.1</u> of the Credit Agreement are hereby amended to read as follows:

"Commitment" means, with respect to each Lender, the commitment, if any, of such Lender to make Advances (expressed as the maximum aggregate amount of the Advances to be made by such Lender hereunder), as such commitment may be (a) reduced from time to time pursuant to Section 2.6, (b) increased pursuant to Section 2.10, and (c) reduced or increased from time to time pursuant to assignments by or to such Lender pursuant to Section 11.8. The amount of each Lender's Commitment is set forth on Schedule 1.1 or in the Assignment and Acceptance pursuant to which such Lender shall have assumed its Commitment, as applicable. The aggregate amount of the Lenders' Commitments on the Amendment No. 8 Effective Date is \$225,000,000; provided, however, that such amount at all times other than during the Increased Commitment Period shall be \$200,000,000.

"Maximum Revolving Credit Amount" means (a) during the Increased Commitment Period, \$225,000,000, and (b) at all other times, \$200,000,000.

- 3. <u>Schedule 1.1 (Lender Commitments/Pro Rata Shares)</u>. <u>Schedule 1.1</u> to the Credit Agreement is hereby amended to read as set forth on <u>Annex I</u> to this Amendment.
- 4. <u>Effectiveness</u>. This Amendment shall become effective on the date that each of the following conditions shall have been satisfied in a manner satisfactory to the Lenders in their sole and absolute discretion (the "<u>Effective Date</u>"):
- (a) The Administrative Agent shall have received all of the following in form and substance satisfactory to the Administrative Agent and its legal counsel:
 - (i) duly executed counterparts of this Amendment signed by the parties hereto;
 - (ii) a duly executed Note for Wells Fargo, as the sole Lender, reflecting its revised Commitment;
- (iii) updated resolutions of Borrower authorizing the increase in the commitment provided hereunder, the execution and delivery of this Amendment and the amended and restated Note described in the foregoing clause (ii), and such other matters relating hereto as Borrower's board of directors shall determine; and
- (iv) such other assurances, certificates, documents, consents or opinions as the Administrative Agent and/or any Lender reasonably may require.

[Type here]

- (b) An upfront fee previously agreed to between the Administrative Agent and Borrower shall have been paid.
- (c) The reasonable costs and expenses of the Administrative Agent in connection with the preparation of this Agreement and each of the other Loan Documents prepared in connection therewith payable pursuant to <u>Section 11.3</u>, and invoiced to Borrower prior to the Effective Date, shall have been paid.
- (d) All legal matters relating to the Loan Documents shall be reasonably satisfactory to McGuireWoods LLP, special counsel to the Administrative Agent.
- 5. <u>Integration; Loan Document</u>. Except as specifically provided herein, all terms and conditions of the Credit Agreement remain in full force and effect, without waiver or modification. This Amendment and the Credit Agreement shall be read together, as one document. This Amendment shall constitute a Loan Document.
- 6. <u>Representations and Warranties; No Defaults.</u> Borrower hereby represents that all representations and warranties contained in Article 5 of the Credit Agreement (except to the extent such representations and warranties expressly speak as of a prior date, in which case they were true and correct in all material respects on and as of such earlier date) are and will be true and correct in all material respects on and as of the Amendment No. 8 Effective Date and reaffirms all covenants set forth therein. Borrower further certifies that as of the date of, and after giving effect to, this Amendment, there exists no Event of Default as defined in the Credit Agreement, nor any condition, act or event which with the giving of notice or the passage of time or both would constitute any such Event of Default.
- 7. <u>Governing Law.</u> THIS AMENDMENT SHALL BE GOVERNED BY, AND CONSTRUED AND ENFORCED IN ACCORDANCE WITH, THE LAWS OF CALIFORNIA APPLICABLE TO CONTRACTS MADE AND PERFORMED IN CALIFORNIA.
- 8. <u>Counterparts</u>. This Amendment may be executed in any number of counterparts and any party hereto may execute any counterpart, each of which when executed and delivered will be deemed to be an original and all of which counterparts of this Amendment, when taken together, will be deemed to be but one and the same instrument and execution of any such counterpart may be evidenced by facsimile or other electronic means of transmission of the signature of such party.

[Remainder of page intentionally left blank; signature pages follow.]

[Type here]

IN WITNESS WHEREOF, the parties hereto have caused this Amendment to be executed as of the day and year first above written.

BORROWER:

AMERICAN STATES WATER COMPANY, a California corporation

By: /s/ Eva G. Tang

Name: Eva G. Tang

Senior Vice President-Finance, Chief Financial Officer, Corporate Secretary and Treasurer Title:

S-1

Eighth Amendment to Amended and Restated Credit Agreement

ADMINISTRATIVE AGENT:

WELLS FARGO BANK, NATIONAL ASSOCIATION, as Administrative Agent

By: /s/ DuVon G. Davis
Name: DuVon G. Davis
Title: Senior Vice President

S-2

Eighth Amendment to Amended and Restated Credit Agreement

LENDERS:

WELLS FARGO BANK, NATIONAL ASSOCIATION, as a Lender

By: /s/ DuVon G. Davis
Name: DuVon G. Davis

Title: Senior Vice President

S-3

Eighth Amendment to Amended and Restated Credit Agreement

ANNEX I

SCHEDULE 1.1

LENDER COMMITMENTS/PRO RATA SHARES

<u>Lender</u>	Pro Rata Share	Commitment Amount
Wells Fargo Bank, National Association	100.0%	\$ 225,000,000 during the Increased Commitment Period
		\$200,000,000 at all other times
	100.0%	\$ 225,000,000 or \$200,000,000, as applicable

S--4 Eighth Amendment to Amended and Restated Credit Agreement

NINTH AMENDMENT TO AMENDED AND RESTATED CREDIT AGREEMENT

THIS NINTH AMENDMENT TO AMENDED AND RESTATED CREDIT AGREEMENT (this "Amendment"), dated as of March 16, 2020, is entered into with reference to the Amended and Restated Credit Agreement (as amended, restated, supplemented or otherwise modified from time to time, the "Credit Agreement") dated as of June 3, 2005, by and among AMERICAN STATES WATER COMPANY, a California corporation ("Borrower"), each of the lenders party thereto, and WELLS FARGO BANK, NATIONAL ASSOCIATION, as administrative agent for the Lenders (in such capacity, the "Administrative Agent") and lead arranger. Capitalized terms used herein and not otherwise defined shall have the meanings set forth for such terms in the Credit Agreement. Section references herein are to sections of the Credit Agreement unless otherwise stated.

RECITALS

WHEREAS, Borrower has requested that the Credit Agreement be modified as set forth in this Amendment, <u>including</u>, without limitation, that the Increased Commitment Period be extended to and including December 31, 2020 and the amount of the Revolving Facility be temporarily increased by \$35,000,000 (to \$260,000,000) during the Increased Commitment Period.

WHEREAS, the Administrative Agent, Wells Fargo, as the sole existing Lender as of the date hereof, and Borrower have agreed to make certain changes in the terms and conditions set forth in the Credit Agreement, and have agreed to amend the Credit Agreement to reflect said changes.

NOW, THEREFORE, for valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the parties hereto agree that the Credit Agreement shall be amended as follows:

- 1. <u>Section 1.1 Defined Terms (New)</u>. The following defined terms are hereby added to <u>Section 1.1</u> in the appropriate alphabetical place:
 - "Amendment No. 9" means the Ninth Amendment to Amended and Restated Credit Agreement, dated as of March 16, 2020, among Borrower, each of the Lenders party thereto and the Administrative Agent.
 - "Amendment No. 9 Effective Date" means the "Effective Date" as defined in Amendment No. 9.

"Benchmark Replacement" means the sum of: (a) the alternate benchmark rate (which may include Term SOFR) that has been selected by Administrative Agent and Borrower pursuant to Section 3.6(g) giving due consideration to (i) any selection or recommendation of a replacement rate or the mechanism for determining such a rate by the Relevant Governmental Body or (ii) any evolving or then-prevailing market convention for determining a rate of interest

as a replacement to LIBOR for U.S. dollar-denominated syndicated credit facilities and (b) the Benchmark Replacement Adjustment; provided that, if the Benchmark Replacement as so determined would be less than zero, the Benchmark Replacement will be deemed to be zero for the purposes of this Agreement.

"Benchmark Replacement Adjustment" means, with respect to any replacement of LIBOR with an Unadjusted Benchmark Replacement for each applicable Eurodollar Period, the spread adjustment, or method for calculating or determining such spread adjustment, (which may be a positive or negative value or zero) that has been selected by Administrative Agent and Borrower giving due consideration to (a) any selection or recommendation of a spread adjustment, or method for calculating or determining such spread adjustment, for the replacement of LIBOR with the applicable Unadjusted Benchmark Replacement by the Relevant Governmental Body or (b) any evolving or then-prevailing market convention for determining a spread adjustment, or method for calculating or determining such spread adjustment, for the replacement of LIBOR with the applicable Unadjusted Benchmark Replacement for U.S. dollar-denominated syndicated credit facilities at such time.

"Benchmark Replacement Conforming Changes" means, with respect to any Benchmark Replacement, any technical, administrative or operational changes (including changes to the definition of "Alternate Base Rate," the definition of "Eurodollar Period," timing and frequency of determining rates and making payments of interest and other administrative matters) that Administrative Agent decides may be appropriate to reflect the adoption and implementation of such Benchmark Replacement and to permit the administration thereof by Administrative Agent in a manner substantially consistent with market practice (or, if Administrative Agent decides that adoption of any portion of such market practice is not administratively feasible or if Administrative Agent determines that no market practice for the administration of the Benchmark Replacement exists, in such other manner of administration as Administrative Agent decides is reasonably necessary in connection with the administration of this Agreement).

"Benchmark Replacement Date" means the earlier to occur of the following events with respect to LIBOR:

- (a) in the case of clause (a) or (b) of the definition of "Benchmark Transition Event," the later of (i) the date of the public statement or publication of information referenced therein and (ii) the date on which the administrator of LIBOR permanently or indefinitely ceases to provide LIBOR; and
- (b) in the case of clause (c) of the definition of "Benchmark Transition Event," the date of the public statement or publication of information referenced therein.

"Benchmark Transition Event" means the occurrence of one or more of the following events with respect to LIBOR:

- (a) a public statement or publication of information by or on behalf of the administrator of LIBOR announcing that such administrator has ceased or will cease to provide LIBOR, permanently or indefinitely; <u>provided</u> that, at the time of such statement or publication, there is no successor administrator that will continue to provide LIBOR;
- (b) a public statement or publication of information by the regulatory supervisor for the administrator of LIBOR, the U.S. Federal Reserve System, an insolvency official with jurisdiction over the administrator for LIBOR, a resolution authority with jurisdiction over the administrator for LIBOR or a court or an entity with similar insolvency or resolution authority over the administrator for LIBOR, which states that the administrator of LIBOR has ceased or will cease to provide LIBOR permanently or indefinitely; <u>provided</u> that, at the time of such statement or publication, there is no successor administrator that will continue to provide LIBOR; or
- (c) a public statement or publication of information by the regulatory supervisor for the administrator of LIBOR announcing that LIBOR is no longer representative.

"Benchmark Transition Start Date" means (a) in the case of a Benchmark Transition Event, the earlier of (i) the applicable Benchmark Replacement Date and (ii) if such Benchmark Transition Event is a public statement or publication of information of a prospective event, the 90th day prior to the expected date of such event as of such public statement or publication of information (or if the expected date of such prospective event is fewer than 90 days after such statement or publication, the date of such statement or publication) and (b) in the case of an Early Opt-in Election, the date specified by Administrative Agent or the Requisite Lenders, as applicable, by notice to Borrower, Administrative Agent (in the case of such notice by the Requisite Lenders) and the Lenders.

"Benchmark Unavailability Period" means, if a Benchmark Transition Event and its related Benchmark Replacement Date have occurred with respect to LIBOR and solely to the extent that LIBOR has not been replaced with a Benchmark Replacement, the period (a) beginning at the time that such Benchmark Replacement Date has occurred if, at such time, no Benchmark Replacement has replaced LIBOR for all purposes hereunder in accordance with Section 3.6(g) and (b) ending at the time that a Benchmark Replacement has replaced LIBOR for all purposes hereunder pursuant to Section 3.6(g).

"Early Opt-in Election" means the occurrence of:

(a) (i) a determination by Administrative Agent or (ii) a notification by the Requisite Lenders to Administrative Agent (with a copy to Borrower) that the Requisite Lenders have determined that U.S. dollar-denominated syndicated credit facilities being executed at such time, or that include language similar to that contained in <u>Section</u>. 3.6(g)

are being executed or amended, as applicable, to incorporate or adopt a new benchmark interest rate to replace LIBOR, and

(b) (i) the election by Administrative Agent or (ii) the election by the Requisite Lenders to declare that an Early Opt-in Election has occurred and the provision, as applicable, by Administrative Agent of written notice of such election to Borrower and the Lenders or by the Requisite Lenders of written notice of such election to Administrative Agent.

"Federal Reserve Bank of New York's Website" means the website of the Federal Reserve Bank of New York at http://www.newyorkfed.org, or any successor source.

"LIBOR" means, subject to the implementation of a Benchmark Replacement in accordance with Section 3.6(g),

- (a) for any interest rate calculation with respect to a Eurodollar Rate Advance, the rate of interest per annum determined on the basis of the rate for deposits in Dollars for a period equal to the applicable Eurodollar Period as published by the ICE Benchmark Administration Limited, a United Kingdom company, or a comparable or successor quoting service approved by the Administrative Agent, at approximately 11:00 a.m. (London time) two (2) London Banking Days prior to the first day of the applicable Eurodollar Period. If, for any reason, such rate is not so published then "LIBOR" shall be determined by Administrative Agent to be the arithmetic average of the rate per annum at which deposits in Dollars would be offered by first class banks in the London interbank market to Administrative Agent at approximately 11:00 a.m. (London time) two (2) London Banking Days prior to the first day of the applicable Eurodollar Period for a period equal to such Eurodollar Period, and
- (b) for any interest rate calculation with respect to an Alternate Base Rate Advance, the rate of interest per annum determined on the basis of the rate for deposits in Dollars for a Eurodollar Period equal to one month (commencing on the date of determination of such interest rate) as published by ICE Benchmark Administration Limited, a United Kingdom company, or a comparable or successor quoting service approved by Administrative Agent, at approximately 11:00 a.m. (London time) on such date of determination, or, if such date is not a Business Day, then the immediately preceding Business Day. If, for any reason, such rate is not so published then "LIBOR" for such Alternate Base Rate Advance shall be determined by Administrative Agent to be the arithmetic average of the rate per annum at which deposits in Dollars would be offered by first class banks in the London interbank market to Administrative Agent at approximately 11:00 a.m. (London time) on such date of determination for a period equal to one month commencing on such date of determination.

Each calculation by Administrative Agent of LIBOR shall be conclusive and binding for all purposes, absent manifest error.

Notwithstanding the foregoing, (x) in no event shall LIBOR (including any Benchmark Replacement with respect thereto) be less than zero percent (0.00%) and (y) unless otherwise specified in any amendment to this Agreement entered into in accordance with Section 5.8(c), in the event that a Benchmark Replacement with respect to LIBOR is implemented then all references herein to LIBOR shall be deemed references to such Benchmark Replacement.

"LIBOR Rate" means a rate per annum determined by Administrative Agent pursuant to the following formula:

LIBOR Rate =	LIBOR
	1 00- Reserve Requirement

"Relevant Governmental Body" means the Federal Reserve Board and/or the Federal Reserve Bank of New York, or a committee officially endorsed or convened by the Federal Reserve Board and/or the Federal Reserve Bank of New York or any successor thereto.

"SOFR" with respect to any day means the secured overnight financing rate published for such day by the Federal Reserve Bank of New York, as the administrator of the benchmark, (or a successor administrator) on the Federal Reserve Bank of New York's Website.

"<u>Term SOFR</u>" means the forward-looking term rate based on SOFR that has been selected or recommended by the Relevant Governmental Body.

"<u>Unadjusted Benchmark Replacement</u>" means the Benchmark Replacement excluding the Benchmark Replacement Adjustment.

2. <u>Section 1.1 – Defined Terms (Revisions)</u>. The definitions of "Alternate Base Rate," "Commitment," "Eurodollar Rate," "Increased Commitment Period," "Maximum Revolving Credit Amount" and "Reserve Requirement" contained in <u>Section 1.1</u> of the Credit Agreement are hereby amended to read as follows:

"Alternate Base Rate" means, at any time, the highest of (a) the Prime Rate, (b) the Federal Funds Rate <u>plus</u> one-half of one percent (0.50%) and (c) the LIBOR Rate for a Eurodollar Period of one month <u>plus</u> one percent (1.00)%; each change in the Alternate Base Rate shall take effect simultaneously with the corresponding change or changes in the Prime Rate, the Federal Funds Rate or the One-Month LIBOR Rate, as applicable (<u>provided</u> that <u>clause (c)</u> shall not be applicable during any period in which LIBOR is unavailable or unascertainable).

"Commitment" means, with respect to each Lender, the commitment, if any, of such Lender to make Advances (expressed as the maximum aggregate amount of the Advances to be made by such Lender hereunder), as such commitment may be (a) reduced from time to time

pursuant to <u>Section 2.6</u>, (b) increased pursuant to <u>Section 2.10</u>, and (c) reduced or increased from time to time pursuant to assignments by or to such Lender pursuant to <u>Section 11.8</u>. The amount of each Lender's Commitment is set forth on <u>Schedule 1.1</u> or in the Assignment and Acceptance pursuant to which such Lender shall have assumed its Commitment, as applicable. The aggregate amount of the Lenders' Commitments on the Amendment No. 9 Effective Date is \$260,000,000; <u>provided</u>, <u>however</u>, that such amount at all times other than during the Increased Commitment Period shall be \$200,000,000.

"Eurodollar Rate" means the LIBOR Rate.

"Increased Commitment Period" means the period commencing on the Amendment No. 8 Effective Date and ending on December 31, 2020.

"Maximum Revolving Credit Amount" means (a) from the Amendment No. 8 Effective Date until the day immediately preceding the Amendment No. 9 Effective Date, \$225,000,000, (b) during the period commencing on the Amendment No. 9 Effective Date and continuing to and including the last day of the Increased Commitment Period, \$260,000,000, and (c) at all other times, \$200,000,000.

"Reserve Requirement" means, for any day, the percentage which is in effect for such day as prescribed by the FRB for determining the maximum reserve requirement (including any basic, supplemental or emergency reserves) in respect of eurocurrency liabilities or any similar category of liabilities for a member bank of the Federal Reserve System in New York City.

- 3. <u>Section 1.8 Rates.</u> The following new <u>Section 1.8</u> is added to the Credit Agreement.
- 1.8 <u>Rates</u>. Administrative Agent does not warrant or accept responsibility for, and shall not have any liability with respect to, the administration, submission or any other matter related to the rates in the definition of "LIBOR" or with respect to any rate that is an alternative or replacement for or successor to any such rate (including, without limitation, any Benchmark Replacement) or the effect of any of the foregoing, or of any Benchmark Replacement Conforming Changes.
- 4. <u>Section 3.6 (g) Benchmark Transition Event</u>. The following new subsection (g) is added to the Credit Agreement:
 - (g) Effect of Benchmark Transition Event.
 - (i) <u>Benchmark Replacement</u>. Notwithstanding anything to the contrary herein (including the foregoing subsections of this <u>Section 3.6</u>) or in any other Loan Document, upon the occurrence of a Benchmark Transition Event or an Early Opt-in Election, as applicable, Administrative Agent and Borrower may amend this Agreement to replace LIBOR with a Benchmark Replacement. Any such amendment with respect to a Benchmark Transition Event will become effective at 5:00 p.m. on the fifth (5th)

Banking Day after Administrative Agent has posted such proposed amendment to all Lenders and Borrower so long the Administrative Agent has not received, by such time, written notice of objection to such amendment from Lenders comprising the Requisite Lenders. Any such amendment with respect to an Early Opt-in Election will become effective on the date that Lenders comprising the Requisite Lenders have delivered to Administrative Agent written notice that such Requisite Lenders accept such amendment. No replacement of LIBOR with a Benchmark Replacement pursuant to this Section 3.6(g) will occur prior to the applicable Benchmark Transition Start Date.

- (ii) <u>Benchmark Replacement Conforming Changes</u>. In connection with the implementation of a Benchmark Replacement, Administrative Agent will have the right to make Benchmark Replacement Conforming Changes from time to time and, notwithstanding anything to the contrary herein or in any other Loan Document, any amendments implementing such Benchmark Replacement Conforming Changes will become effective without any further action or consent of any other party to this Agreement.
- (iii) Notices; Standards for Decisions and Determinations. Administrative Agent will promptly notify Borrower and the Lenders of (A) any occurrence of a Benchmark Transition Event or an Early Opt-in Election, as applicable, and its related Benchmark Replacement Date and Benchmark Transition Start Date, (B) the implementation of any Benchmark Replacement, (C) the effectiveness of any Benchmark Replacement Conforming Changes and (D) the commencement or conclusion of any Benchmark Unavailability Period. Any determination, decision or election that may be made by Administrative Agent or the Lenders pursuant to this Section 3.6(g), including any determination with respect to a tenor, rate or adjustment or of the occurrence or non-occurrence of an event, circumstance or date and any decision to take or refrain from taking any action, will be conclusive and binding absent manifest error and may be made in its or their sole discretion and without consent from any other party hereto, except, in each case, as expressly required pursuant to this Section 3.6(g).
- (iv) <u>Benchmark Unavailability Period</u>. Upon Borrower's receipt of notice of the commencement of a Benchmark Unavailability Period, Borrower may revoke any request for a Eurodollar Rate Advance of, conversion to or continuation of Eurodollar Rate Advances to be made, converted or continued during any Benchmark Unavailability Period and, failing that, Borrower will be deemed to have converted any such request into a request for a borrowing of or conversion to Alternate Base Rate Advances. During any Benchmark Unavailability Period, the component of the Alternate Base Rate based upon LIBOR will not be used in any determination of the Alternate Base Rate.
- 5. <u>Section 11.2 Amendments</u>. The following new paragraph is added to Section 11.2 of the Credit Agreement immediately before the last sentence thereof:

Notwithstanding the foregoing or anything contained herein or any other Loan Document to the contrary, Administrative Agent (and, if applicable, Borrower) may, without the consent of any Lender, enter into amendments or modifications to this Agreement or any of the other Loan Documents or to enter into additional Loan Documents in order to implement any Benchmark Replacement or any Benchmark Replacement Conforming Changes or otherwise effectuate the terms of Section 3.6(g) in accordance with the terms of Section 3.6(g).

- 6. <u>Schedule 1.1 (Lender Commitments/Pro Rata Shares)</u>. <u>Schedule 1.1</u> to the Credit Agreement is hereby amended to read as set forth on Annex I to this Amendment.
- 7. <u>Effectiveness</u>. This Amendment shall become effective on the date that each of the following conditions shall have been satisfied in a manner satisfactory to the Lenders in their sole and absolute discretion (the "<u>Effective Date</u>"):
- (i) The Administrative Agent shall have received all of the following in form and substance satisfactory to the Administrative Agent and its legal counsel:
 - (a) duly executed counterparts of this Amendment signed by the parties hereto;
 - (b) a duly executed Note for Wells Fargo, as the sole Lender, reflecting its revised Commitment;
- (c) updated resolutions of Borrower authorizing the increase in the commitment provided hereunder, the execution and delivery of this Amendment and the amended and restated Note described in the foregoing clause (ii), and such other matters relating hereto as Borrower's board of directors shall determine; and
- (d) such other assurances, certificates, documents, consents or opinions as the Administrative Agent and/or any Lender reasonably may require.
 - (ii) An upfront fee previously agreed to between the Administrative Agent and Borrower shall have been paid.
- (iii) The reasonable costs and expenses of the Administrative Agent in connection with the preparation of this Agreement and each of the other Loan Documents prepared in connection therewith payable pursuant to <u>Section 11.3</u>, and invoiced to Borrower prior to the Effective Date, shall have been paid.
- (iv) All legal matters relating to the Loan Documents shall be reasonably satisfactory to McGuireWoods LLP, special counsel to the Administrative Agent.
- 8. <u>Integration; Loan Document</u>. Except as specifically provided herein, all terms and conditions of the Credit Agreement remain in full force and effect, without waiver or modification. This

Amendment and the Credit Agreement shall be read together, as one document. This Amendment shall constitute a Loan Document.

- 9. <u>Representations and Warranties; No Defaults.</u> Borrower hereby represents that all representations and warranties contained in Article 5 of the Credit Agreement (except to the extent such representations and warranties expressly speak as of a prior date, in which case they were true and correct in all material respects on and as of such earlier date) are and will be true and correct in all material respects on and as of the Amendment No. 9 Effective Date and reaffirms all covenants set forth therein. Borrower further certifies that as of the date of, and after giving effect to, this Amendment, there exists no Event of Default as defined in the Credit Agreement, nor any condition, act or event which with the giving of notice or the passage of time or both would constitute any such Event of Default.
- 10. <u>Governing Law.</u> THIS AMENDMENT SHALL BE GOVERNED BY, AND CONSTRUED AND ENFORCED IN ACCORDANCE WITH, THE LAWS OF CALIFORNIA APPLICABLE TO CONTRACTS MADE AND PERFORMED IN CALIFORNIA.
- 11. <u>Counterparts</u>. This Amendment may be executed in any number of counterparts and any party hereto may execute any counterpart, each of which when executed and delivered will be deemed to be an original and all of which counterparts of this Amendment, when taken together, will be deemed to be but one and the same instrument and execution of any such counterpart may be evidenced by facsimile or other electronic means of transmission of the signature of such party.

[Remainder of page intentionally left blank; signature pages follow.]

	IN WITNESS	WHEREOF,	the parties	hereto	have	caused	this	Amendment	to be	e executed	as o	f the	day	and	year	first	above
written.																	

BORROWER:

AMERICAN STATES WATER COMPANY, a California corporation

By: <u>/s/ Eva G. Tang</u>_____

Name: Eva G. Tang

Title: SVP-Finance, CFO, Treasurer & Corporate Secretary

 $S\mbox{-}1$ Ninth Amendment to Amended and Restated Credit Agreement

ADMINISTRATIVE AGENT:

WELLS FARGO BANK, NATIONAL ASSOCIATION, as Administrative Agent

By: /s/ DuVon G. Davis

Name: DuVon G. Davis Title: Senior Vice President

S-2

Ninth Amendment to Amended and Restated Credit Agreement

LENDERS:

WELLS FARGO BANK, NATIONAL ASSOCIATION, as a Lender

By: /s/ DuVon G. Davis
Name: DuVon G. Davis
Title: Senior Vice President

S-3

Ninth Amendment to Amended and Restated Credit Agreement

ANNEX I

SCHEDULE 1.1

LENDER COMMITMENTS/PRO RATA SHARES

<u>Lender</u>	Pro Rata Share	Commitment Amount
Wells Fargo Bank, National Association	100.0%	\$260,000,000 during the Increased Commitment Period
		\$200,000,000 at all other times
	100.0%	\$ 260,000,000 or \$200,000,000, as applicable

S-1

Ninth Amendment to Amended and Restated Credit Agreement

TENTH AMENDMENT TO AMENDED AND RESTATED CREDIT AGREEMENT

THIS TENTH AMENDMENT TO AMENDED AND RESTATED CREDIT AGREEMENT (this "Amendment"), dated as of June 26, 2020, is entered into with reference to the Amended and Restated Credit Agreement (as amended, restated, supplemented or otherwise modified from time to time, the "Credit Agreement") dated as of June 3, 2005, by and among AMERICAN STATES WATER COMPANY, a California corporation ("Borrower"), each of the lenders party thereto, and WELLS FARGO BANK, NATIONAL ASSOCIATION, as administrative agent for the Lenders (in such capacity, the "Administrative Agent") and lead arranger. Capitalized terms used herein and not otherwise defined shall have the meanings set forth for such terms in the Credit Agreement. Section references herein are to sections of the Credit Agreement unless otherwise stated.

RECITALS

WHEREAS, Borrower has requested that the Credit Agreement be modified as set forth in this Amendment.

WHEREAS, the Administrative Agent, Wells Fargo, as the sole existing Lender as of the date hereof, and Borrower have agreed to make certain changes in the terms and conditions set forth in the Credit Agreement, and have agreed to amend the Credit Agreement to reflect said changes.

NOW, THEREFORE, for valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the parties hereto agree that the Credit Agreement shall be amended as follows:

- 1. <u>Section 1.1 Defined Terms (New)</u>. The following defined terms are hereby added to <u>Section 1.1</u> in the appropriate alphabetical place:
 - "Amendment No. 10" means the Tenth Amendment to Amended and Restated Credit Agreement, dated as of June 26, 2020, among Borrower, each of the Lenders party thereto and the Administrative Agent.
 - "Amendment No. 10 Effective Date" means the "Effective Date" as defined in Amendment No. 10.
 - "BVES" means Bear Valley Electric Service, Inc., a California corporation, a wholly-owned Subsidiary of Borrower.
 - "Consolidated Net Tangible Assets" means the total amount of Borrower's and its Subsidiaries' consolidated total assets (less applicable depreciation, amortization and other valuation reserves and other properly deductible items) after deducting: (a) all current liabilities, excluding (i) the Obligations and (ii) any current liabilities which are by their terms extendible or renewable at the option of the obligor on the liabilities to a time more than 12 months after the

time as of which the amount of current liabilities is being computed and (b) all goodwill, trade names, trademarks, patents and other like Intangible Assets, all as set forth on the most recent consolidated balance sheet of Borrower and its Subsidiaries and computed in accordance with GAAP.

2. <u>Section 1.1 – Defined Terms (Amended and Restated)</u>. The definitions of "Applicable Commitment Fee Margin," "Applicable Eurodollar Rate Margin," "Initial Pricing Period," and "LIBOR," contained in <u>Section 1.1</u> of the Credit Agreement are hereby amended to read as follows:

"<u>Applicable Commitment Fee Margin</u>" means, for each Pricing Period, the margin set forth below (expressed in basis points per annum) opposite the Applicable Pricing Level for that Pricing Period:

Applicable	
Pricing	
<u>Level</u>	<u>Margin</u>
I	10.00
II	12.50
III	15.00
IV	17.50
V	20.00

"<u>Applicable Eurodollar Rate Margin</u>" means, with respect to any Eurodollar Rate Advance, for each Pricing Period, the interest rate margin set forth below (expressed in basis points per annum) opposite the Applicable Pricing Level for that Pricing Period:

Applicable	
Pricing	
<u>Level</u>	<u>Margin</u>
I	87.50
II	100.00
III	112.50
IV	125.00
V	137.50

"<u>Initial Pricing Period</u>" means the period commencing on the Amendment No. 10 Effective Date and ending on the first Pricing Occurrence to occur thereafter.

"LIBOR" means, subject to the implementation of a Benchmark Replacement in accordance with Section 3.6(g),

(a) for any interest rate calculation with respect to a Eurodollar Rate Advance, the rate of interest per annum (rounded upwards, as necessary, to the nearest

1/16th of one percent (0.0625%)) determined on the basis of the rate for deposits in Dollars for a period equal to the applicable Eurodollar Period as published by the ICE Benchmark Administration Limited, a United Kingdom company, or a comparable or successor quoting service approved by the Administrative Agent, at approximately 11:00 a.m. (London time) two (2) Eurodollar Banking Days prior to the first day of the applicable Eurodollar Period. If, for any reason, such rate is not so published then "LIBOR" shall be determined by Administrative Agent to be the arithmetic average of the rate per annum at which deposits in Dollars would be offered by first class banks in the London interbank market to Administrative Agent at approximately 11:00 a.m. (London time) two (2) Eurodollar Banking Days prior to the first day of the applicable Eurodollar Period for a period equal to such Eurodollar Period, and

(b) for any interest rate calculation with respect to an Alternate Base Rate Advance, the rate of interest per annum (rounded upwards, as necessary, to the nearest 1/16th of one percent (0.0625%)) determined on the basis of the rate for deposits in Dollars for a Eurodollar Period equal to one month (commencing on the date of determination of such interest rate) as published by ICE Benchmark Administration Limited, a United Kingdom company, or a comparable or successor quoting service approved by Administrative Agent, at approximately 11:00 a.m. (London time) on such date of determination, or, if such date is not a Business Day, then the immediately preceding Business Day. If, for any reason, such rate is not so published then "LIBOR" for such Alternate Base Rate Advance shall be determined by Administrative Agent to be the arithmetic average of the rate per annum at which deposits in Dollars would be offered by first class banks in the London interbank market to Administrative Agent at approximately 11:00 a.m. (London time) on such date of determination for a period equal to one month commencing on such date of determination.

Each calculation by Administrative Agent of LIBOR shall be conclusive and binding for all purposes, absent manifest error.

Notwithstanding the foregoing, (x) in no event shall LIBOR (including any Benchmark Replacement with respect thereto) be less than zero percent (0.00%) and (y) unless otherwise specified in any amendment to this Agreement entered into in accordance with Section 5.8(c), in the event that a Benchmark Replacement with respect to LIBOR is implemented then all references herein to LIBOR shall be deemed references to such Benchmark Replacement.

- 3. <u>Section 1.1 Defined Term (Revised)</u>. The definition of "Permitted Encumbrances" contained in <u>Section 1.1</u> of the Credit Agreement is hereby amended to add a new subsection (s) to read as follows:
 - (s) Liens consisting of pledges or deposits to secure obligations under agreements with the Automated Power Exchange and other obligations under power purchase agreements and

statutory liens on all equity in CoBank, ACB required from time to time in connection with Indebtedness permitted pursuant to Section 6.10(f).

- 4. <u>Section 6.10 Indebtedness and Guaranty Obligations</u>. Section 6.10 of the Credit Agreement is hereby amended to read as follows:
 - 6.10 <u>Indebtedness and Guaranty Obligations</u>. Create, incur or assume any Indebtedness or Guaranty Obligation if an Event of Default has occurred and is continuing or if, after giving effect thereto, Borrower would not be in compliance with the provisions of <u>Section 6.12</u> or <u>Section 6.13</u> or an Event of Default would otherwise occur. Notwithstanding the foregoing, Borrower shall not permit any Subsidiary to create, incur, assume or permit to exist any Indebtedness or Guaranty Obligation, <u>except</u>:
 - (i) Permitted Acquisition Indebtedness,
 - (ii) Permitted Capital Asset Indebtedness,
 - (iii) existing Indebtedness set forth on Schedule 6.10(b),
 - (iv) Indebtedness owed to Borrower or a wholly-owned Subsidiary,
 - (v) unsecured term Indebtedness of GSW (<u>i.e.</u>, not revolving credit) that (i) either has a longer weighted average life than the Borrowings hereunder or satisfies the requirements of <u>Section 6.1</u>, (ii) to the extent that a Governmental Agency has regulatory jurisdiction over the issuance of such Indebtedness of GSW, the issuance of such Indebtedness is permitted by such regulatory jurisdiction, and (iii) is incurred in the ordinary course of business of GSW, is substantially consistent with the prior practices of GSW, and is provided by any Person or Governmental Agency under a credit agreement or facility substantially similar thereto,
 - (vi) unsecured Indebtedness of BVES that (i) BVES is permitted to incur by the Governmental Agency or Agencies, as applicable, having regulatory jurisdiction over BVES, (ii) is incurred in the ordinary course of business of BVES, and (iii) does not exceed at any time outstanding an aggregate amount of \$125,000,000 (and for purposes of computing compliance with the foregoing, all then undrawn commitments under any revolving credit facility shall be deemed incurred and outstanding Indebtedness); provided, however, that in no event shall the aggregate amount of Indebtedness incurred pursuant to this subsection (f) exceed at any time outstanding ten percent (10%) of Consolidated Net Tangible Assets, and
 - (vii) other unsecured Indebtedness in the aggregate principal amount not to exceed \$1,000,000.
- 5. <u>Effectiveness.</u> This Amendment shall become effective on the date that each of the following conditions shall have been satisfied in a manner satisfactory to the Lenders in their sole and absolute discretion (the "<u>Effective Date</u>"):

- (i) The Administrative Agent shall have received all of the following in form and substance satisfactory to the Administrative Agent and its legal counsel:
 - (a) duly executed counterparts of this Amendment signed by the parties hereto; and
- (b) such other assurances, certificates, documents, consents or opinions as the Administrative Agent and/or any Lender reasonably may require.
- (ii) The reasonable costs and expenses of the Administrative Agent in connection with the preparation of this Agreement and each of the other Loan Documents prepared in connection therewith payable pursuant to <u>Section 11.3</u>, and invoiced to Borrower prior to the Effective Date, shall have been paid.
- (iii) All legal matters relating to the Loan Documents shall be reasonably satisfactory to McGuireWoods LLP, special counsel to the Administrative Agent.
- 6. <u>Integration; Loan Document</u>. Except as specifically provided herein, all terms and conditions of the Credit Agreement remain in full force and effect, without waiver or modification. This Amendment and the Credit Agreement shall be read together, as one document. This Amendment shall constitute a Loan Document.
- 7. Representations and Warranties; No Defaults. Borrower hereby represents that all representations and warranties contained in Article 5 of the Credit Agreement (except to the extent such representations and warranties expressly speak as of a prior date, in which case they were true and correct in all material respects on and as of such earlier date) are and will be true and correct in all material respects on and as of the Amendment No. 10 Effective Date and reaffirms all covenants set forth therein. Borrower further certifies that as of the date of, and after giving effect to, this Amendment, there exists no Event of Default as defined in the Credit Agreement, nor any condition, act or event which with the giving of notice or the passage of time or both would constitute any such Event of Default.
- 8. <u>Governing Law.</u> THIS AMENDMENT SHALL BE GOVERNED BY, AND CONSTRUED AND ENFORCED IN ACCORDANCE WITH, THE LAWS OF CALIFORNIA APPLICABLE TO CONTRACTS MADE AND PERFORMED IN CALIFORNIA.
- 9. <u>Counterparts</u>. This Amendment may be executed in any number of counterparts and any party hereto may execute any counterpart, each of which when executed and delivered will be deemed to be an original and all of which counterparts of this Amendment, when taken together, will be deemed to be but one and the same instrument and execution of any such counterpart may be evidenced by facsimile or other electronic means of transmission of the signature of such party.

[Remainder of page intentionally left blank; signature pages follow.]

IN WITNESS WHEREOF, the parties hereto have caused this Amendment to be executed as of the day and year first above written.

BORROWER:

AMERICAN STATES WATER COMPANY, a California corporation

By: <u>/s/ Eva. G. Tang</u>

Name: Eva G. Tang

Title: SVP-Finance, CFO, Treasurer & Corporate Secretary

146221719

ADMINISTRATIVE AGENT:

WELLS FARGO BANK, NATIONAL ASSOCIATION, as Administrative Agent

By: /s/ DuVon G. Davis_____

Name: DuVon G. Davis Title: Senior Vice President

LENDERS:

WELLS FARGO BANK, NATIONAL ASSOCIATION, as a Lender

By: /s/ DuVon G. Davis_____

Name: DuVon G. Davis Title: Senior Vice President

ELEVENTH AMENDMENT TO AMENDED AND RESTATED CREDIT AGREEMENT

THIS ELEVENTH AMENDMENT TO AMENDED AND RESTATED CREDIT AGREEMENT (this "Amendment"), dated as of June 15, 2021, is entered into with reference to the Amended and Restated Credit Agreement (as amended, restated, supplemented or otherwise modified from time to time, the "Credit Agreement") dated as of June 3, 2005, by and among AMERICAN STATES WATER COMPANY, a California corporation ("Borrower"), each of the lenders party thereto, and WELLS FARGO BANK, NATIONAL ASSOCIATION, as administrative agent for the Lenders (in such capacity, the "Administrative Agent") and lead arranger. Capitalized terms used herein and not otherwise defined shall have the meanings set forth for such terms in the Credit Agreement. Section references herein are to sections of the Credit Agreement unless otherwise stated.

RECITALS

WHEREAS, Borrower has requested that the Credit Agreement be modified as set forth in this Amendment.

WHEREAS, the Administrative Agent, Wells Fargo, as the sole existing Lender as of the date hereof, and Borrower have agreed to make certain changes in the terms and conditions set forth in the Credit Agreement, and have agreed to amend the Credit Agreement to reflect said changes.

NOW, THEREFORE, for valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the parties hereto agree that the Credit Agreement shall be amended as follows:

1. <u>Section 1.1 – Defined Terms (New)</u>. The following defined terms are hereby added to <u>Section 1.1</u> in the appropriate alphabetical place:

"Amendment No. 11" means the Eleventh Amendment to Amended and Restated Credit Agreement, dated as of June 15, 2021, among Borrower, each of the Lenders party thereto and the Administrative Agent.

"Amendment No. 11 Effective Date" means the "Effective Date" as defined in Amendment No. 11.

2. <u>Section 1.1 – Defined Terms (Amended and Restated)</u>. The definitions of "Applicable Commitment Fee Margin," "Applicable Eurodollar Rate Margin," and "Initial Pricing Period" contained in <u>Section 1.1</u> of the Credit Agreement are hereby amended to read as follows:

"<u>Applicable Commitment Fee Margin</u>" means, for each Pricing Period, the margin set forth below (expressed in basis points per annum) opposite the Applicable Pricing Level for that Pricing Period:

Applicable	
Pricing	
<u>Level</u>	<u>Margin</u>
I	6.00
II	7.50
III	10.00
IV	12.50
V	15.00

"<u>Applicable Eurodollar Rate Margin</u>" means, with respect to any Eurodollar Rate Advance, for each Pricing Period, the interest rate margin set forth below (expressed in basis points per annum) opposite the Applicable Pricing Level for that Pricing Period:

Applicable	
Pricing	
<u>Level</u>	<u>Margin</u>
I	55.00
II	65.00
III	75.00
IV	85.00
V	100.00

"Initial Pricing Period" means the period commencing on the Amendment No. 11 Effective Date and ending on the first Pricing Occurrence to occur thereafter.

- 3. <u>Effectiveness</u>. This Amendment shall become effective on the date that each of the following conditions shall have been satisfied in a manner satisfactory to the Lenders in their sole and absolute discretion (the "<u>Effective Date</u>"):
- (a) The Administrative Agent shall have received all of the following in form and substance satisfactory to the Administrative Agent and its legal counsel:
 - (i) duly executed counterparts of this Amendment signed by the parties hereto; and

- (ii) such other assurances, certificates, documents, consents or opinions as the Administrative Agent and/or any Lender reasonably may require.
- (b) The reasonable costs and expenses of the Administrative Agent in connection with the preparation of this Agreement and each of the other Loan Documents prepared in connection therewith payable pursuant to <u>Section 11.3</u>, and invoiced to Borrower prior to the Effective Date, shall have been paid.
- (c) All legal matters relating to the Loan Documents shall be reasonably satisfactory to McGuireWoods LLP, special counsel to the Administrative Agent.
- 4. <u>Integration; Loan Document.</u> Except as specifically provided herein, all terms and conditions of the Credit Agreement remain in full force and effect, without waiver or modification. This Amendment and the Credit Agreement shall be read together, as one document. This Amendment shall constitute a Loan Document.
- 5. Representations and Warranties; No Defaults. Borrower hereby represents that all representations and warranties contained in Article 5 of the Credit Agreement (except to the extent such representations and warranties expressly speak as of a prior date, in which case they were true and correct in all material respects on and as of such earlier date) are and will be true and correct in all material respects on and as of the Amendment No. 11 Effective Date and reaffirms all covenants set forth therein. Borrower further certifies that as of the date of, and after giving effect to, this Amendment, there exists no Event of Default as defined in the Credit Agreement, nor any condition, act or event which with the giving of notice or the passage of time or both would constitute any such Event of Default.
- 6. <u>Governing Law.</u> THIS AMENDMENT SHALL BE GOVERNED BY, AND CONSTRUED AND ENFORCED IN ACCORDANCE WITH, THE LAWS OF CALIFORNIA APPLICABLE TO CONTRACTS MADE AND PERFORMED IN CALIFORNIA.
- 7. <u>Counterparts</u>. This Amendment may be executed in any number of counterparts and any party hereto may execute any counterpart, each of which when executed and delivered will be deemed to be an original and all of which counterparts of this Amendment, when taken together, will be deemed to be but one and the same instrument and execution of any such counterpart may be evidenced by facsimile or other electronic means of transmission of the signature of such party.

[Remainder of page intentionally left blank; signature pages follow.]

IN WITNESS WHEREOF, the parties hereto have caused this Amendment to be executed as of the day and year first above written.

BORROWER:

AMERICAN STATES WATER COMPANY, a California corporation

By: <u>/s/ Robert J. Sprowls</u>
Name: Robert J. Sprowls

Title: President and Chief Executive Officer

ADMINISTRATIVE AGENT:

WELLS FARGO BANK, NATIONAL ASSOCIATION, as Administrative Agent

By: /s/ DuVon G. Davis
Name: DuVon G. Davis
Title: Senior Vice President

LENDERS:

WELLS FARGO BANK, NATIONAL ASSOCIATION, as a Lender

By: /s/ DuVon G. Davis
Name: DuVon G. Davis
Title: Senior Vice President

TWELFTH AMENDMENT TO AMENDED AND RESTATED CREDIT AGREEMENT

THIS TWELFTH AMENDMENT TO AMENDED AND RESTATED CREDIT AGREEMENT (this "Amendment"), dated as of April 22, 2022, is entered into with reference to the Amended and Restated Credit Agreement (as amended, restated, supplemented or otherwise modified from time to time, the "Credit Agreement") dated as of June 3, 2005, by and among AMERICAN STATES WATER COMPANY, a California corporation ("Borrower"), each of the lenders party thereto, and WELLS FARGO BANK, NATIONAL ASSOCIATION, as administrative agent for the Lenders (in such capacity, the "Administrative Agent") and lead arranger. Capitalized terms used herein and not otherwise defined shall have the meanings set forth for such terms in the Credit Agreement. Section references herein are to sections of the Credit Agreement unless otherwise stated.

RECITALS

WHEREAS, Borrower has requested that the Credit Agreement be modified as set forth in this Amendment, <u>including</u>, without limitation, that the amount of the Revolving Facility be temporarily increased by \$80,000,000 (to \$280,000,000) during the Increased Commitment Period and LIBOR-based pricing for Revolving Credit Facility Usage be replaced with Term SOFR-based pricing.

WHEREAS, the Administrative Agent, Wells Fargo, as the sole existing Lender as of the date hereof, and Borrower have agreed to make certain changes in the terms and conditions set forth in the Credit Agreement, and have agreed to amend the Credit Agreement to reflect said changes.

NOW, THEREFORE, for valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the parties hereto agree that the Credit Agreement shall be amended as follows:

- 1. <u>Section 1.1 Defined Terms (New)</u>. The following defined terms are hereby added to <u>Section 1.1</u> in the appropriate alphabetical place:
- "<u>Adjusted Term SOFR</u>" means, for purposes of any calculation, the rate per annum equal to (a) Term SOFR for such calculation <u>plus</u> (b) the Term SOFR Adjustment; <u>provided</u> that if Adjusted Term SOFR as so determined shall ever be less than the Floor, then Adjusted Term SOFR shall be deemed to be the Floor.
- "Amendment No. 12" means the Twelfth Amendment to Amended and Restated Credit Agreement, dated as of April 22, 2022, among Borrower, each of the Lenders party thereto and the Administrative Agent.
 - "Amendment No. 12 Effective Date" means the "Effective Date" as defined in Amendment No. 12.
- "Applicable Adjusted Term SOFR Margin" means, with respect to any SOFR Advance, for each Pricing Period, the interest rate margin set forth below (expressed in basis points per annum) opposite the Applicable Pricing Level for that Pricing Period:

Applicable Pricing <u>Level</u>	<u>Margin</u>
I	55.00
II	65.00
III	75.00
IV	85.00
V	100.00

"Available Tenor" means, as of any date of determination and with respect to the then-current Benchmark, as applicable, (a) if such Benchmark is a term rate, any tenor for such Benchmark (or component thereof) that is or may be used for determining the length of an interest period pursuant to this Agreement or (b) otherwise, any payment period for interest calculated with reference to such Benchmark (or component thereof) that is or may be used for determining any frequency of making payments of interest calculated with reference to such Benchmark pursuant to this Agreement, in each case, as of such date and not including, for the avoidance of doubt, any tenor for such Benchmark that is then-removed from the definition of "Interest Period" pursuant to Section 3.5(c)(iv).

"Benchmark" means, initially, the Term SOFR Reference Rate; <u>provided</u> that if a Benchmark Transition Event has occurred with respect to the Term SOFR Reference Rate or the then-current Benchmark, then "Benchmark" means the applicable Benchmark Replacement to the extent that such Benchmark Replacement has replaced such prior benchmark rate pursuant to <u>Section 3.5(e)(i)</u>.

"Change in Law" means the occurrence, after the date of this Agreement, of any of the following: (a) the adoption or taking effect of any law, rule, regulation or treaty, (b) any change in any law, rule, regulation or treaty or in the administration, interpretation, implementation or application thereof by any Governmental Agency or (c) the making or issuance of any request, rule, guideline or directive (whether or not having the force of law) by any Governmental Agency; provided that notwithstanding anything herein to the contrary, (i) the Dodd-Frank Wall Street Reform and Consumer Protection Act and all requests, rules, guidelines, requirements or directives thereunder or issued in connection therewith or in implementation thereof and (ii) all requests, rules, guidelines, requirements or directives promulgated by the Bank for International Settlements, the Basel Committee on Banking Supervision (or any successor or similar authority) or the United States or foreign regulatory authorities, in each case pursuant to Basel III, shall in each case be deemed to be a "Change in Law", regardless of the date enacted, adopted, implemented or issued.

"Conforming Changes" means, with respect to either the use or administration of Term SOFR or the use, administration, adoption or implementation of any Benchmark Replacement, any technical, administrative or operational changes (including changes to the definition of "Alternate Base Rate," the definition of "Banking Day," the definition of "U.S. Government Securities Business Day," the definition of "Interest Period" or any similar or analogous definition (or the addition of a concept of "interest period"), timing and frequency of determining rates and making payments of interest, timing of borrowing requests or prepayment, conversion or continuation notices, the applicability and length of lookback periods, the applicability of Section 3.1(f) and other technical, administrative or operational matters) that the Administrative Agent decides may be appropriate to reflect the adoption and implementation of any such rate or to permit the use and administration thereof by the Administrative Agent in a manner substantially consistent with market practice (or, if the Administrative Agent decides that adoption of any portion of such market practice is not administratively feasible or if the

Administrative Agent determines that no market practice for the administration of any such rate exists, in such other manner of administration as the Administrative Agent decides is reasonably necessary in connection with the administration of this Agreement and the other Loan Documents).

"Connection Income Taxes" means Other Connection Taxes that are imposed on or measured by net income (however denominated) or that are franchise Taxes or branch profits Taxes.

"Excluded Taxes" means any of the following Taxes imposed on or with respect to a Recipient or required to be withheld or deducted from a payment to a Recipient, (a) Taxes imposed on or measured by net income (however denominated), franchise Taxes, and branch profits Taxes, in each case, (i) imposed as a result of such Recipient being organized under the laws of, or having its principal office or, in the case of any Lender, its applicable lending office located in, the jurisdiction imposing such Tax (or any political subdivision thereof) or (ii) that are Other Connection Taxes, (b) in the case of a Lender (including the Issuing Lender), U.S. federal withholding Taxes imposed on amounts payable to or for the account of such Lender with respect to an applicable interest in an Advance or Commitment pursuant to a law in effect on the date on which (i) such Lender acquires such interest in the Advance or Commitment (other than pursuant to an assignment request by the Borrower under Section 11.24) or (ii) such Lender changes its lending office, except in each case to the extent that, pursuant to Section 3.10(e), amounts with respect to such Taxes were payable either to such Lender's assignor immediately before such Lender became a party hereto or to such Lender immediately before it changed its lending office, (c) Taxes attributable to such Recipient's failure to comply with its backup withholding or information reporting requirements with respect to payments made under any Loan Document, and (d) any withholding Taxes imposed under FATCA.

"FATCA" means Sections 1471 through 1474 of the Code, as of the date of this Agreement (or any amended or successor version that is substantively comparable and not materially more onerous to comply with), any current or future regulations or official interpretations thereof, any agreements entered into pursuant to Section 1471(b)(1) of the Code and any fiscal or regulatory legislation, rules or practices adopted pursuant to any intergovernmental agreement, treaty or convention among Governmental Agencies entered into in connection with the implementation of the foregoing.

"Floor" means a rate of interest equal to 0.00%.

"Indemnified Taxes" means Taxes, other than Excluded Taxes, imposed on or with respect to any payment made by or on account of any obligation of the Borrower under any Loan Document.

"Interest Period" means, as to any SOFR Advance, the period commencing on the date such SOFR Advance is disbursed or converted to or continued as a SOFR Advance and ending on the date one (1), three (3) or six (6) months thereafter, in each case as selected by Borrower in its Request for Borrowing or Request for Continuation/Conversion and subject to availability; <u>provided</u> that:

- (a) the Interest Period shall commence on the date of advance of or conversion to any SOFR Advance and, in the case of immediately successive Interest Periods, each successive Interest Period shall commence on the date on which the immediately preceding Interest Period expires;
- (b) if any Interest Period would otherwise expire on a day that is not a Banking Day, such Interest Period shall expire on the next succeeding Banking Day; <u>provided</u> that if any Interest Period would otherwise expire on a day that is not a Banking Day but is a day of the month after which no

further Banking Day occurs in such month, such Interest Period shall expire on the immediately preceding Banking Day;

- (c) any Interest Period that begins on the last Banking Day of a calendar month (or on a day for which there is no numerically corresponding day in the calendar month at the end of such Interest Period) shall end on the last Banking Day of the relevant calendar month at the end of such Interest Period;
- (d) no Interest Period shall extend beyond the Maturity Date;
- (e) there shall be no more than five (5) Interest Periods in effect at any time; and
- (f) no tenor that has been removed from this definition pursuant to <u>Section 3.5(c)(iv)</u> shall be available for specification in any Request for Borrowing or Request for Continuation/Conversion.

"Other Connection Taxes" means, with respect to any Recipient, Taxes imposed as a result of a present or former connection between such Recipient and the jurisdiction imposing such Tax (other than connections arising from such Recipient having executed, delivered, become a party to, performed its obligations under, received payments under, received or perfected a security interest under, engaged in any other transaction pursuant to or enforced any Loan Document, or sold or assigned an interest in any Advance or Loan Document).

"Recipient" means (a) the Administrative Agent, (b) any Lender or (c) the Issuing Lender, as applicable.

"SOFR Advance" means any Advance bearing interest at a rate based on Adjusted Term SOFR as provided in Section 2.3.

"SOFR Administrator" means the Federal Reserve Bank of New York (or a successor administrator of the secured overnight financing rate).

"<u>Taxes</u>" means all present or future taxes, levies, imposts, duties, deductions, withholdings (including backup withholding), assessments, fees or other charges imposed by any Governmental Agency, including any interest, additions to tax or penalties applicable thereto.

"Term SOFR Adjustment" means a percentage equal to 0.10000% per annum.

"<u>Term SOFR Administrator</u>" means CME Group Benchmark Administration Limited (CBA) (or a successor administrator of the Term SOFR Reference Rate selected by the Administrative Agent in its reasonable discretion).

"Term SOFR Reference Rate" means the forward-looking term rate based on SOFR.

"U.S. Government Securities Business Day" means any day except for (a) a Saturday, (b) a Sunday or (c) a day on which the Securities Industry and Financial Markets Association recommends that the fixed income departments of its members be closed for the entire day for purposes of trading in United States government securities; provided, that for purposes of notice requirements in Sections 2.1(b), 2.3(a), 2.4(c), and 3.1(e), in each case, such day is also a Banking Day.

2. <u>Section 1.1 – Defined Terms (Revisions)</u>. The following definitions contained in <u>Section 1.1</u> of the Credit Agreement are hereby amended to read as follows:

"Alternate Base Rate" means, for any day, the highest of (a) the Prime Rate in effect on such day, (b) the Federal Funds Rate in effect on such day <u>plus</u> 0.50% and (c) Adjusted Term SOFR for a one-month tenor in effect on such day <u>plus</u> 1.00%; each change in the Alternate Base Rate shall take

effect simultaneously with the corresponding change or changes in the Prime Rate, the Federal Funds Rate or Adjusted Term SOFR, as applicable (<u>provided</u> that <u>clause (c)</u> shall not be applicable during any period in which Adjusted Term SOFR is unavailable or unascertainable).

"<u>Applicable Letter of Credit Fee Rate</u>" means, as of any date of determination, the then effective Applicable Adjusted Term SOFR Margin.

"Banking Day" means any day that (a) is not a Saturday, Sunday or other day on which the Federal Reserve Bank of New York is closed and (b) is not a day on which commercial banks in California or New York are closed.

"Benchmark Replacement" means, with respect to any Benchmark Transition Event, the sum of: (a) the alternate benchmark rate that has been selected by the Administrative Agent and Borrower giving due consideration to (i) any selection or recommendation of a replacement benchmark rate or the mechanism for determining such a rate by the Relevant Governmental Body or (ii) any evolving or then-prevailing market convention for determining a benchmark rate as a replacement to the then-current Benchmark for Dollar-denominated syndicated credit facilities at such time and (b) the related Benchmark Replacement Adjustment; provided that, if such Benchmark Replacement as so determined would be less than the Floor, such Benchmark Replacement will be deemed to be the Floor for the purposes of this Agreement and the other Loan Documents.

"Benchmark Replacement Adjustment" means, with respect to any replacement of the then-current Benchmark with an Unadjusted Benchmark Replacement for any applicable Available Tenor, the spread adjustment, or method for calculating or determining such spread adjustment, (which may be a positive or negative value or zero) that has been selected by the Administrative Agent and Borrower giving due consideration to (a) any selection or recommendation of a spread adjustment, or method for calculating or determining such spread adjustment, for the replacement of such Benchmark with the applicable Unadjusted Benchmark Replacement by the Relevant Governmental Body or (b) any evolving or then-prevailing market convention for determining a spread adjustment, or method for calculating or determining such spread adjustment, for the replacement of such Benchmark with the applicable Unadjusted Benchmark Replacement for Dollar-denominated syndicated credit facilities at such time.

"Benchmark Replacement Date" means the earliest to occur of the following events with respect to the then-current Benchmark:

- (a) in the case of clause (a) or (b) of the definition of "Benchmark Transition Event," the later of (i) the date of the public statement or publication of information referenced therein and (ii) the date on which the administrator of such Benchmark (or the published component used in the calculation thereof) permanently or indefinitely ceases to provide all Available Tenors of such Benchmark (or such component thereof); or
- (b) in the case of clause (c) of the definition of "Benchmark Transition Event," the first date on which such Benchmark (or the published component used in the calculation thereof) has been determined and announced by the regulatory supervisor for the administrator of such Benchmark (or such component thereof) to be non-representative; <u>provided</u> that such non-representativeness will be determined by reference to the most recent statement or publication referenced in such clause (c) and even if any Available Tenor of such Benchmark (or such component thereof) continues to be provided on such date.

For the avoidance of doubt, the "Benchmark Replacement Date" will be deemed to have occurred in the case of clause (a) or (b) with respect to any Benchmark upon the occurrence of the applicable event or events set forth therein with respect to all then-current Available Tenors of such Benchmark (or the published component used in the calculation thereof).

"Benchmark Transition Event" means the occurrence of one or more of the following events with respect to the then-current Benchmark:

- (a) a public statement or publication of information by or on behalf of the administrator of such Benchmark (or the published component used in the calculation thereof) announcing that such administrator has ceased or will cease to provide all Available Tenors of such Benchmark (or such component thereof), permanently or indefinitely; <u>provided</u> that, at the time of such statement or publication, there is no successor administrator that will continue to provide any Available Tenor of such Benchmark (or such component thereof);
- (b) a public statement or publication of information by the regulatory supervisor for the administrator of such Benchmark (or the published component used in the calculation thereof), the FRB, the Federal Reserve Bank of New York, an insolvency official with jurisdiction over the administrator for such Benchmark (or such component) or a court or an entity with similar insolvency or resolution authority over the administrator for such Benchmark (or such component), which states that the administrator of such Benchmark (or such component) has ceased or will cease to provide all Available Tenors of such Benchmark (or such component thereof) permanently or indefinitely; provided that, at the time of such statement or publication, there is no successor administrator that will continue to provide any Available Tenor of such Benchmark (or such component thereof); or
- (c) a public statement or publication of information by the regulatory supervisor for the administrator of such Benchmark (or the published component used in the calculation thereof) announcing that all Available Tenors of such Benchmark (or such component thereof) are not, or as of a specified future date will not be, representative.

For the avoidance of doubt, a "Benchmark Transition Event" will be deemed to have occurred with respect to any Benchmark if a public statement or publication of information set forth above has occurred with respect to each then-current Available Tenor of such Benchmark (or the published component used in the calculation thereof).

"Benchmark Transition Start Date" means, in the case of a Benchmark Transition Event, the earlier of (a) the applicable Benchmark Replacement Date and (b) if such Benchmark Transition Event is a public statement or publication of information of a prospective event, the 90th day prior to the expected date of such event as of such public statement or publication of information (or if the expected date of such prospective event is fewer than 90 days after such statement or publication, the date of such statement or publication).

"Benchmark Unavailability Period" means the period (if any) (x) beginning at the time that a Benchmark Replacement Date has occurred if, at such time, no Benchmark Replacement has replaced the then-current Benchmark for all purposes hereunder and under any Loan Document in accordance with Section 3.5(c) and (y) ending at the time that a Benchmark Replacement has replaced the then-current Benchmark for all purposes hereunder and under any Loan Document in accordance with Section 3.5(c).

"Commitment" means, with respect to each Lender, the commitment, if any, of such Lender to make Advances (expressed as the maximum aggregate amount of the Advances to be made by such Lender hereunder), as such commitment may be (a) reduced from time to time pursuant to Section 2.6, and (b) reduced or increased from time to time pursuant to assignments by or to such Lender pursuant to Section 11.8. The amount of each Lender's Commitment is set forth on Schedule 1.1 or in the Assignment and Acceptance pursuant to which such Lender shall have assumed its Commitment, as applicable. The aggregate amount of the Lenders' Commitments on the Amendment No. 12 Effective Date is \$280,000,000; provided, however, that such amount at all times other than during the Increased Commitment Period shall be \$200,000,000.

"Continuation," "Continue" and "Continued" each refers to a continuation of SOFR Advances from one Interest Period to the next Interest Period pursuant to Section 2.4.

"Increased Commitment Period" means the period commencing on the Amendment No. 12 Effective Date and ending on May 23, 2023.

"Maximum Revolving Credit Amount" means (a) during the period commencing on the Amendment No. 12 Effective Date and continuing to and including the last day of the Increased Commitment Period, \$280,000,000, and (b) at all other times, \$200,000,000.

"Relevant Governmental Body" means the FRB or the Federal Reserve Bank of New York, or a committee officially endorsed or convened by the FRB or the Federal Reserve Bank of New York, or any successor thereto.

"SOFR" means a rate equal to the secured overnight financing rate as administered by the SOFR Administrator.

"Term SOFR" means,

- (a) for any calculation with respect to a SOFR Advance, the Term SOFR Reference Rate for a tenor comparable to the applicable Interest Period on the day (such day, the "Periodic Term SOFR Determination Day") that is two (2) U.S. Government Securities Business Days prior to the first day of such Interest Period, as such rate is published by the Term SOFR Administrator; provided, however, that if as of 5:00 p.m. (Eastern time) on any Periodic Term SOFR Determination Day the Term SOFR Reference Rate for the applicable tenor has not been published by the Term SOFR Administrator and a Benchmark Replacement Date with respect to the Term SOFR Reference Rate has not occurred, then Term SOFR will be the Term SOFR Reference Rate for such tenor as published by the Term SOFR Administrator on the first preceding U.S. Government Securities Business Day for which such Term SOFR Reference Rate for such tenor was published by the Term SOFR Administrator so long as such first preceding U.S. Government Securities Business Day is not more than three (3) U.S. Government Securities Business Days prior to such Periodic Term SOFR Determination Day, and
- (b) for any calculation with respect to an Alternate Base Rate Advance on any day, the Term SOFR Reference Rate for a tenor of one month on the day (such day, the "Alternate Base Rate Term SOFR Determination Day") that is two (2) U.S. Government Securities Business Days prior to such day, as such rate is published by the Term SOFR Administrator; provided, however, that if as of 5:00 p.m. (Eastern time) on any Alternate Base Rate Term SOFR Determination Day the Term SOFR Reference Rate for the applicable tenor has not been published by the Term SOFR Administrator and a Benchmark Replacement Date with respect to the Term SOFR Reference Rate has not occurred, then Term SOFR will be the Term SOFR Reference Rate for such tenor as published by the Term SOFR Administrator on the

first preceding U.S. Government Securities Business Day for which such Term SOFR Reference Rate for such tenor was published by the Term SOFR Administrator so long as such first preceding U.S. Government Securities Business Day is not more than three (3) U.S. Government Securities Business Days prior to such Alternate Base Rate SOFR Determination Day.

"Type" refers to the distinction between Advances bearing interest at the Alternate Base Rate and Advances bearing interest based on Adjusted Term SOFR.

"<u>Unadjusted Benchmark Replacement</u>" means the applicable Benchmark Replacement excluding the related Benchmark Replacement Adjustment.

- 3. <u>Section 1.1 Defined Terms (Deleted)</u>. The following definitions contained in <u>Section 1.1</u> of the Credit Agreement are deleted from the Credit Agreement: "Applicable Eurodollar Rate Margin," "Benchmark Replacement Conforming Changes," "Designated Eurodollar Market," "Early Opt-in Election," "Eurodollar Banking Day," "Eurodollar Base Rate," "Eurodollar Lending Office," "Eurodollar Market," "Eurodollar Period," "Eurodollar Rate," "Eurodollar Rate Advance," "Eurodollar Reference Lender," "LIBOR," "LIBOR Rate," "Reserve Requirement" and "Special Eurodollar Circumstance."
 - 4. <u>Section 1.8 Rates. Section 1.8 of the Credit Agreement is hereby amended to read as follows:</u>

1.8 <u>Rates</u>. The Administrative Agent does not warrant or accept any responsibility for, and shall not have any liability with respect to, (a) the continuation of, administration of, submission of, calculation of or any other matter related to the Term SOFR Reference Rate, Adjusted Term SOFR or Term SOFR, or any component definition thereof or rates referred to in the definition thereof, or with respect to any alternative, successor or replacement rate thereto (including any Benchmark Replacement), including whether the composition or characteristics of any such alternative, successor or replacement rate (including any Benchmark Replacement), as it may or may not be adjusted pursuant to Section 3.5(c), will be similar to, or produce the same value or economic equivalence of, or have the same volume or liquidity as, the Term SOFR Reference Rate, Adjusted Term SOFR, Term SOFR or any other Benchmark prior to its discontinuance or unavailability, or (b) the effect, implementation or composition of any Conforming Changes. The Administrative Agent and its Affiliates or other related entities may engage in transactions that affect the calculation of the Term SOFR Reference Rate, Adjusted Term SOFR, Term SOFR, any alternative, successor or replacement rate (including any Benchmark Replacement) or any relevant adjustments thereto and such transactions may be adverse to Borrower. The Administrative Agent may select information sources or services in its reasonable discretion to ascertain the Term SOFR Reference Rate, Adjusted Term SOFR or Term SOFR, or any other Benchmark, any component definition thereof or rates referred to in the definition thereof, in each case pursuant to the terms of this Agreement, and shall have no liability to Borrower, any Lender or any other person or entity for damages of any kind, including direct or indirect, special, punitive, incidental or consequential damages, costs, losses or expenses (whether in tort, contract or otherwise and whether at law or in equity), for any error or calculation of any such rate (or component thereof) provided by any such information source or service.

- 5. <u>Subsections 2.1(b), (c), (d) and (g) Advances (Generally</u>). <u>Subsections 2.1(b), (c), (d)</u> and (g) of the Credit Agreement are hereby amended to read as follows:
- (b) Subject to the next sentence, each Borrowing shall be made pursuant to a Request for Borrowing which shall specify (i) the date of such requested Borrowing, (ii) the Type of Advances comprising such Borrowing, (iii) the amount of such Borrowing, and (iv) in the case of a

Borrowing consisting of SOFR Advances, the Interest Period therefor. Unless the Administrative Agent has notified, in its sole and absolute discretion, Borrower to the contrary not less than three (3) days prior to the date of any Borrowing, a Borrowing may be requested by telephone by a Responsible Official of Borrower, in which case Borrower shall confirm such request by promptly delivering a Request for Borrowing (conforming to the preceding sentence) in person or by telecopier to the Administrative Agent. The Administrative Agent shall incur no liability whatsoever hereunder in acting upon any telephonic request for a Borrowing purportedly made by a Responsible Official of Borrower, and Borrower hereby agrees to indemnify the Administrative Agent from any loss, cost, expense or liability as a result of so acting.

- (c) Promptly following receipt of a Request for Borrowing, the Administrative Agent shall notify each Lender by telephone or telecopier (and if by telephone, promptly confirmed by telecopier) of the date of the requested Borrowing, the Type of Advances comprising such Borrowing, the Interest Period (if applicable), and the amount corresponding to that Lender's ratable share of the Borrowing. Not later than 1:00 p.m., California time, on the date specified for any Borrowing (which must be a Banking Day), each Lender shall make its ratable share of the Borrowing in immediately available funds available to the Administrative Agent at the Administrative Agent's Office. Upon satisfaction or waiver of the applicable conditions set forth in Article 8, all Advances shall be credited on that date in immediately available funds to the Designated Deposit Account.
- (d) Anything in paragraph (b) above to the contrary notwithstanding, Borrower may not (a) request Alternate Base Rate Advances for any Borrowing if the aggregate amount of such Borrowing is less than \$500,000 (and any such Borrowing exceeding such minimum amount shall be in an integral multiple of \$100,000), provided that the foregoing minimum amount shall not apply to an Alternate Base Rate Advance that causes the aggregate amount borrowed under the Revolving Facility to equal the full amount available for Advances hereunder or Advances pursuant to Section 2.5, or (b) elect SOFR Advances for any Borrowing (i) if the aggregate amount of such Borrowing is less than \$1,000,000 (and any such Borrowing exceeding such minimum amount shall be in an integral multiple of \$500,000) or (ii) if the obligation of the Lenders to make SOFR Advances shall then be suspended pursuant to Section 2.4 or 3.5.
- (g) The Administrative Agent, on behalf of the Lenders, is hereby authorized to make Borrowings available to Borrower upon fulfillment of the applicable conditions set forth in <u>Article 8</u>. Upon fulfillment of such applicable conditions, the proceeds of Borrowings shall either be credited in immediately available funds to the Designated Deposit Account or remitted directly to one or more third parties, as directed by Borrower and approved by the Administrative Agent. The proceeds of any Borrowing consisting of SOFR Advances shall be so credited or remitted on the first day of the applicable Interest Period for such Borrowings.
- 6. <u>Sections 2.2 (Alternate Base Rate Advances), 2.3 (SOFR Advances) and 2.4 (Conversion and Continuation of Advances).</u>
 Sections 2.2, 2.3, and 2.4 of the Credit Agreement are hereby amended to read as follows:
- 2.2 <u>Alternate Base Rate Advances</u>. Each request by Borrower for a Borrowing comprised of Alternate Base Rate Advances shall be made pursuant to a Request for Borrowing (or telephonic or other request for Borrowing referred to in the second sentence of <u>Section 2.1(b)</u>, if applicable) received by the Administrative Agent, at the Administrative Agent's Office, not later than 11:00 a.m. California time, at least one (1) Banking Day before the requested Borrowing. All Advances

shall constitute Alternate Base Rate Advances unless properly designated as a SOFR Advance pursuant to Section 2.3 or 2.4.

2.3 SOFR Advances.

- (a) Each request by Borrower for a Borrowing comprised of SOFR Advances shall be made pursuant to a Request for Borrowing (or telephonic or other request for Borrowing referred to in the second sentence of Section 2.1(b), if applicable) received by the Administrative Agent, at the Administrative Agent's Office, not later than 9:00 a.m., California time, at least three (3) U.S. Government Securities Business Days before the first day of the applicable Interest Period.
- (b) On the date which is two (2) U.S. Government Securities Business Days before the first day of the applicable Interest Period, the Administrative Agent shall confirm its determination of the applicable Adjusted Term SOFR (which determination shall be conclusive in the absence of manifest error) and promptly shall give notice of the same to Borrower and the Lenders by telephone or telecopier (and if by telephone, promptly confirmed by telecopier).
- (c) Unless the Administrative Agent and the Requisite Lenders otherwise consent, SOFR may not be outstanding under more than five (5) separate Interest Periods at any one time.
- (d) No Borrowing comprised of SOFR Advances may be requested during the continuation of a Default or Event of Default.
- (e) <u>Term SOFR Conforming Changes</u>. In connection with the use or administration of Term SOFR, the Administrative Agent will have the right to make Conforming Changes from time to time and, notwithstanding anything to the contrary herein or in any other Loan Document, any amendments implementing such Conforming Changes will become effective without any further action or consent of any other party to this Agreement or any other Loan Document. The Administrative Agent will promptly notify Borrower and the Lenders of the effectiveness of any Conforming Changes in connection with the use or administration of Term SOFR.

2.4 Conversion and Continuation.

(a) Optional Conversions. Borrower may on any Banking Day, upon notice given to the Administrative Agent not later than 9:00 a.m. (California time) on the third U.S. Government Securities Business Day prior to the date of a proposed Conversion if the Conversion is into SOFR Advances, or one Banking Day prior to the date of a proposed Conversion if the Conversion is into Alternate Base Rate Advances, and subject to the provisions of Section 3.5, Convert all or any portion of the Advances of one Type outstanding under the Revolving Facility (and, in the case of SOFR Advances, having the same Interest Period) into Advances of the other Type under the Revolving Facility; provided that any Conversion of SOFR Advances into Alternate Base Rate Advances on other than the last day of an Interest Period for such SOFR Advances shall be subject to Section 3.1(f), any Conversion of Alternate Base Rate Advances into SOFR Advances shall be in an amount not less than \$1,000,000 or integral multiples of \$500,000 in excess thereof and no Conversion of any Advances shall result in more than five (5) separate Interest Periods being outstanding under the Revolving Facility. Each such notice of Conversion shall be made pursuant to a Request for Continuation/Conversion and shall, within the restrictions specified above, specify (i) the date of such Conversion, (ii) the aggregate amount and Type of the Advances (and, in the case of SOFR Advances, the Interest Period therefor) to be Converted and

(iii) if such Conversion is into SOFR Advances, the duration of the initial Interest Period for such Advances. Each request for Conversion shall be irrevocable and binding on Borrower.

(b) Certain Mandatory Conversions.

- (i) On the date on which the aggregate unpaid principal amount of SOFR Advances comprising any Borrowing shall be reduced, by payment or prepayment or otherwise, to less than \$1,000,000 such Advances shall automatically Convert into Alternate Base Rate Advances.
- (ii) If Borrower shall fail to select the duration of any Interest Period for any outstanding SOFR Advances in accordance with the provisions contained in <u>Section 2.1(b)</u> and in <u>clause (a)</u> or <u>(c)</u> of this <u>Section 2.4</u>, each such SOFR Advance will automatically, on the last day of the then existing Interest Period therefor, Convert into an Alternate Base Rate Advance.
- (iii) Upon the occurrence and during the continuance of any Event of Default and upon notice from the Administrative Agent to Borrower at the request of the Requisite Lenders, (x) each SOFR Advance will automatically, on the last day of the then existing Interest Period therefor, Convert into an Alternate Base Rate Advance and (y) the obligation of the Lenders to make, or to Convert Advances into, or to Continue, SOFR Advances shall be suspended.
- (c) <u>Continuations</u>. Borrower may, on any U.S. Government Securities Business Day, upon notice given to the Administrative Agent not later than 9:00 a.m. (California time) on the third U.S. Government Securities Business Day prior to the date of the proposed Continuation and subject to the provisions of <u>Section 3.5</u>, Continue all or any portion of the SOFR Advances having the same Interest Period; <u>provided</u> that any such Continuation shall be made only on the last day of an Interest Period for such SOFR Advances, no Continuation of SOFR Advances shall be in an amount less than \$1,000,000 and no Continuation of any SOFR Advances shall result in more than five (5) separate Interest Periods being outstanding under the Revolving Facility. Each such notice of Continuation shall be made pursuant to a Request for Continuation/Conversion and shall, within the restrictions specified above, specify (i) the date of such Continuation, (ii) the aggregate amount and category of, and the Interest Period for, the Advances being Continued and (iii) the duration of the initial Interest Period for the SOFR Advances subject to such Continuation. Each notice of Continuation shall be irrevocable and binding on Borrower.
- 7. <u>Subsections 3.1(c) and (e) Principal and Interest on SOFR Advances</u>. <u>Subsections 3.1(c)</u> and <u>(e)</u> of the Credit Agreement are hereby amended to read as follows:
- (c) Interest accrued on each SOFR Advance which is for a term of three months or less shall be due and payable on the last day of the related Interest Period. Interest accrued on each other SOFR Advance shall be due and payable on the date which is three months after the date such SOFR Advance was made and on the last day of the related Interest Period. Except as otherwise provided in Section 3.7, the unpaid principal amount of any SOFR Advance shall bear interest at a rate per annum equal to Adjusted Term SOFR for that SOFR Advance plus the Applicable Adjusted Term SOFR Margin.
- (e) The principal Indebtedness evidenced by the Notes may, at any time and from time to time, voluntarily be paid or prepaid in whole or in part without premium or penalty, <u>except</u> that with respect to any voluntary prepayment under this subsection, (i) any partial prepayment shall be not less than \$1,000,000 and shall be an integral multiple of \$500,000, <u>except</u> as provided in <u>Section 2.8(a)</u>, (ii) the Administrative Agent shall have received written notice of any prepayment by 9:00 a.m. California time on the date that is (x) in the case of a Adjusted Term SOFR Advance three (3) U.S.

Government Securities Business Days before the date of prepayment unless the prepayment is of a SOFR Advance to be made at the end of its applicable Interest Period and (y) in the case of an Alternate Base Rate Advance or a prepayment of a SOFR Advance made at the end of its applicable Interest Period, one (1) U.S. Government Securities Business Day before the date of prepayment, which notice shall identify the date and amount of the prepayment and the Advance(s) being prepaid, (iii) each prepayment of principal on any SOFR Advance shall be accompanied by payment of interest accrued to the date of payment on the amount of principal paid, and (iv) any payment or prepayment of all or any part of any SOFR Advance on a day other than the last day of the applicable Interest Period shall be subject to Section 3.6(e).

8. <u>Section 3.1(f) – Indemnity.</u> The following new subsection (f) is hereby added to <u>Section 3.1</u> of the Credit Agreement:

(f) Indemnity. Borrower hereby indemnifies each of the Lenders against any loss, cost or expense (including any loss, cost or expense arising from the liquidation or reemployment of funds or from any fees payable) which may arise, be attributable to or result due to or as a consequence of (a) any failure by Borrower to make any payment when due of any amount due hereunder in connection with a SOFR Advance, (b) any failure of Borrower to borrow or continue a SOFR Advance or convert to a SOFR Advance on a date specified therefor in a Request for Borrowing or Request for Continuation/Conversion, (c) any failure of Borrower to prepay any SOFR Advance on a date specified therefor in any notice of prepayment (regardless of whether any such notice of prepayment may be revoked, or (d) any payment, prepayment or conversion of any SOFR Advance on a date other than the last day of the Interest Period therefor (including as a result of an Event of Default). A certificate of such Lender setting forth the basis for determining such amount or amounts necessary to compensate such Lender shall be forwarded to Borrower through the Administrative Agent and shall be conclusively presumed to be correct save for manifest error. Borrower shall pay such Lender the amount shown as due on any such certificate within 10 Business Days after receipt thereof. All of the obligations of the Credit Parties under this Section 3.5 shall survive the resignation or replacement of the Administrative Agent or any assignment of rights by, or the replacement of, a Lender, the termination of the Commitments and the repayment, satisfaction or discharge of all obligations under any Loan Document.

9. <u>Section 3.5 – Changed Circumstances</u>. <u>Section 3.5</u> of the Credit Agreement is hereby amended to read as follows:

3.5 Changed Circumstances.

(a) <u>Circumstances Affecting Benchmark Availability</u>. Subject to subsection (c) below, in connection with any request for a SOFR Advance or a conversion to or continuation thereof or otherwise, if for any reason (i) the Administrative Agent shall determine (which determination shall be conclusive and binding absent manifest error) that reasonable and adequate means do not exist for ascertaining Adjusted Term SOFR for the applicable Interest Period with respect to a proposed SOFR Advance on or prior to the first day of such Interest Period or (ii) the Requisite Lenders shall determine (which determination shall be conclusive and binding absent manifest error) that Adjusted Term SOFR does not adequately and fairly reflect the cost to such Lenders of making or maintaining such Advances during such Interest Period and, in the case of clause (ii), the Requisite Lenders have provided notice of such determination to the Administrative Agent, then, in each case, the Administrative Agent shall promptly give notice thereof to Borrower. Upon notice thereof by the Administrative Agent to Borrower, any obligation of the Lenders to make SOFR Advances, and any right of Borrower to convert any

Advance to or continue any Advance as a SOFR Advance, shall be suspended (to the extent of the affected SOFR Advances or the affected Interest Periods) until the Administrative Agent (with respect to clause (ii), at the instruction of the Requisite Lenders) revokes such notice. Upon receipt of such notice, (A) Borrower may revoke any pending request for a borrowing of, conversion to or continuation of SOFR Advances (to the extent of the affected SOFR Advances or the affected Interest Periods) or, failing that, Borrower will be deemed to have converted any such request into a request for a borrowing of or conversion to Alternated Base Rate Advances in the amount specified therein and (B) any outstanding affected SOFR Advances will be deemed to have been converted into Alternate Base Rate Advances at the end of the applicable Interest Period. Upon any such prepayment or conversion, Borrower shall also pay accrued interest on the amount so prepaid or converted, together with any additional amounts required pursuant to Section 3.1(f).

(b) Laws Affecting SOFR Availability. If, after the date hereof, the introduction of, or any change in, any Applicable Law or any change in the interpretation or administration thereof by any Governmental Agency, central bank or comparable agency charged with the interpretation or administration thereof, or compliance by any of the Lenders (or any of their respective Lending Offices) with any request or directive (whether or not having the force of law) of any such Governmental Agency, central bank or comparable agency, shall make it unlawful or impossible for any of the Lenders (or any of their respective Lending Offices) to honor its obligations hereunder to make or maintain any SOFR Advance, or to determine or charge interest based upon SOFR, the Term SOFR Reference Rate, Adjusted Term SOFR or Term SOFR, such Lender shall promptly give notice thereof to the Administrative Agent and the Administrative Agent shall promptly give notice to Borrower and the other Lenders (an "Illegality Notice"). Thereafter, until each affected Lender notifies the Administrative Agent and the Administrative Agent notifies Borrower that the circumstances giving rise to such determination no longer exist, (i) any obligation of the Lenders to make SOFR Advances, and any right of Borrower to convert any Advance to a SOFR Advance or continue any Advance as a SOFR Advance, shall be suspended and (ii) if necessary to avoid such illegality, the Administrative Agent shall compute the Alternate Base Rate without reference to clause (c) of the definition of "Alternate Base Rate". Upon receipt of an Illegality Notice, Borrower shall, if necessary to avoid such illegality, within five (5) days after demand of any Lender (with a copy to the Administrative Agent), prepay or, if applicable, convert all SOFR Advances to Alternate Base Rate Advances (in each case, if necessary to avoid such illegality, the Administrative Agent shall compute the Alternate Base Rate without reference to clause (c) of the definition of "Alternate Base Rate"), on the last day of the Interest Period therefor, if all affected Lenders may lawfully continue to maintain such SOFR Advances to such day, or immediately, if any Lender may not lawfully continue to maintain such SOFR Advances to such day. Upon any such prepayment or conversion, Borrower shall also pay accrued interest on the amount so prepaid or converted, together with any additional amounts required pursuant to Section 3.1(f) sufficient to compensate such Lender in light of such circumstances, to the extent reasonably allocable to such obligations under this Agreement, provided that Borrower shall not be obligated to pay any such amount which arose prior to the date which is 180 days preceding the date of such demand or is attributable to periods prior to the date which is 180 days preceding the date of such demand.

(c) Benchmark Replacement Setting.

(i) <u>Benchmark Replacement</u>. Notwithstanding anything to the contrary herein or in any other Loan Document, upon the occurrence of a Benchmark Transition Event, the Administrative Agent and Borrower may amend this Agreement to replace the then-current

Benchmark with a Benchmark Replacement. Any such amendment with respect to a Benchmark Transition Event will become effective at 5:00 p.m. on the fifth (5th) Banking Day after the Administrative Agent has posted such proposed amendment to all affected Lenders and Borrower so long as the Administrative Agent has not received, by such time, written notice of objection to such amendment from Lenders comprising the Requisite Lenders. No replacement of a Benchmark with a Benchmark Replacement pursuant to this Section 3.6(c)(i) will occur prior to the applicable Benchmark Transition Start Date.

(ii) <u>Benchmark Replacement Conforming Changes</u>. In connection with the use, administration, adoption or implementation of a Benchmark Replacement, the Administrative Agent will have the right to make Conforming Changes from time to time and, notwithstanding anything to the contrary herein or in any other Loan Document, any amendments implementing such Conforming Changes will become effective without any further action or consent of any other party to this Agreement or any other Loan Document.

(iii) Notices; Standards for Decisions and Determinations. The Administrative Agent will promptly notify Borrower and the Lenders of (A) the implementation of any Benchmark Replacement and (B) the effectiveness of any Conforming Changes in connection with the use, administration, adoption or implementation of a Benchmark Replacement. The Administrative Agent will promptly notify Borrower of the removal or reinstatement of any tenor of a Benchmark pursuant to Section 3.5(c)(iv) and the commencement of any Benchmark Unavailability Period. Any determination, decision or election that may be made by the Administrative Agent or, if applicable, any Lender (or group of Lenders) pursuant to this Section 3.5(c), including any determination with respect to a tenor, rate or adjustment or of the occurrence or non-occurrence of an event, circumstance or date and any decision to take or refrain from taking any action or any selection, will be conclusive and binding absent manifest error and may be made in its or their sole discretion and without consent from any other party to this Agreement or any other Loan Document, except, in each case, as expressly required pursuant to this Section 3.5(c).

(iv) <u>Unavailability of Tenor of Benchmark</u>. Notwithstanding anything to the contrary herein or in any other Loan Document, at any time (including in connection with the implementation of a Benchmark Replacement), (A) if the then-current Benchmark is a term rate (including the Term SOFR Reference Rate) and either (1) any tenor for such Benchmark is not displayed on a screen or other information service that publishes such rate from time to time as selected by the Administrative Agent in its reasonable discretion or (2) the regulatory supervisor for the administrator of such Benchmark has provided a public statement or publication of information announcing that any tenor for such Benchmark is not or will not be representative, then the Administrative Agent may modify the definition of "Interest Period" (or any similar or analogous definition) for any Benchmark settings at or after such time to remove such unavailable or non-representative tenor and (B) if a tenor that was removed pursuant to clause (A) above either (1) is subsequently displayed on a screen or information service for a Benchmark (including a Benchmark Replacement) or (2) is not, or is no longer, subject to an announcement that it is not or will not be representative for a Benchmark (including a Benchmark Replacement), then the Administrative Agent may modify the definition of "Interest Period" (or any similar or analogous definition) for all Benchmark settings at or after such time to reinstate such previously removed tenor.

(v) <u>Benchmark Unavailability Period</u>. Upon Borrower's receipt of notice of the commencement of a Benchmark Unavailability Period, (A) Borrower may revoke any

pending request for a borrowing of, conversion to or continuation of SOFR Advances to be made, converted or continued during any Benchmark Unavailability Period and, failing that, Borrower will be deemed to have converted any such request into a request for a borrowing of or conversion to Alternated Base Rate Loans and (B) any outstanding affected SOFR Advances will be deemed to have been converted to Alternate Base Rate Advances at the end of the applicable Interest Period. During any Benchmark Unavailability Period or at any time that a tenor for the then-current Benchmark is not an Available Tenor, the component of the Alternate Base Rate based upon the then-current Benchmark or such tenor for such Benchmark, as applicable, will not be used in any determination of the Alternate Base Rate.

10. <u>Section 3.6 – Increased Costs.</u> <u>Section 3.6</u> of the Credit Agreement is hereby amended to read as follows:

3.6 Increased Costs.

- (a) Increased Costs Generally. If any Change in Law shall:
- (i) impose, modify or deem applicable any reserve (including pursuant to regulations issued from time to time by the FRB for determining the maximum reserve requirement (including any emergency, special, supplemental or other marginal reserve requirement) with respect to eurocurrency funding (currently referred to as "Eurocurrency liabilities" in Regulation D of the FRB, as amended and in effect from time to time)), special deposit, compulsory loan, insurance charge or similar requirement against assets of, deposits with or for the account of, or advances, loans or other credit extended or participated in by, any Lender or the Issuing Lender;
- (ii) subject any Recipient to any Taxes (other than (a) Taxes described in clauses (b) through (d) of the definition of Excluded Taxes, (b) Connection Income Taxes, and (c) Indemnified Taxes), on its loans, loan principal, letters of credit, commitments, or other obligations, or its deposits, reserves, other liabilities or capital attributable thereto: or
- (iii) impose on any Lender or the Issuing Lender any other condition, cost or expense (other than Taxes) affecting this Agreement or Advances made by such Lender or any Letter of Credit or participation therein;

and the result of any of the foregoing shall be to increase the cost to such Lender, the Issuing Lender or such other Recipient of making, converting to, continuing or maintaining any Advance (or of maintaining its obligation to make any such Advance), or to increase the cost to such Lender, the Issuing Lender or such other Recipient of participating in, issuing or maintaining any Letter of Credit (or of maintaining its obligation to participate in or to issue any Letter of Credit), or to reduce the amount of any sum received or receivable by such Lender, the Issuing Lender or such other Recipient hereunder (whether of principal, interest or any other amount) then, upon written request of such Lender, the Issuing Lender or other Recipient, the Borrower shall promptly pay to any such Lender, the Issuing Lender or other Recipient, as the case may be, such additional amount or amounts as will compensate such Lender, the Issuing Lender or other Recipient, as the case may be, for such additional costs incurred or reduction suffered.

(b) <u>Capital Requirements</u>. If any Lender or the Issuing Lender determines that any Change in Law affecting such Lender or the Issuing Lender or any lending office of such Lender or such Lender's or the Issuing Lender's holding company, if any, regarding capital or liquidity requirements, has or would have the effect of reducing the rate of return on such Lender's or the Issuing

Lender's capital or on the capital of such Lender's or the Issuing Lender's holding company, if any, as a consequence of this Agreement, the Commitment of such Lender or the Advances made by, or participations in Letters of Credit or Swingline Loans held by, such Lender, or the Letters of Credit issued by the Issuing Lender, to a level below that which such Lender or the Issuing Lender's or the Issuing Lender's holding company could have achieved but for such Change in Law (taking into consideration such Lender's or the Issuing Lender's policies and the policies of such Lender's or the Issuing Lender's holding company with respect to capital adequacy and liquidity), then from time to time upon written request of such Lender or the Issuing Lender Borrower shall promptly pay to such Lender or the Issuing Lender, as the case may be, such additional amount or amounts as will compensate such Lender or the Issuing Lender's or the Issuing Lender's holding company for any such reduction suffered.

- (c) <u>Certificates for Reimbursement</u>. A certificate of a Lender, or the Issuing Lender or such other Recipient setting forth the amount or amounts necessary to compensate such Lender or the Issuing Lender, such other Recipient or any of their respective holding companies, as the case may be, as specified in subsection (a) or (b) of this Section and delivered to Borrower, shall be conclusive absent manifest error. Borrower shall pay such Lender or the Issuing Lender or such other Recipient, as the case may be, the amount shown as due on any such certificate within ten (10) days after receipt thereof.
- (d) <u>Delay in Requests</u>. Failure or delay on the part of any Lender or the Issuing Lender or such other Recipient to demand compensation pursuant to this Section shall not constitute a waiver of such Lender's or the Issuing Lender's or such other Recipient's right to demand such compensation; <u>provided</u> that Borrower shall not be required to compensate any Lender or the Issuing Lender or any other Recipient pursuant to this Section for any increased costs incurred or reductions suffered more than nine (9) months prior to the date that such Lender or the Issuing Lender or such other Recipient, as the case may be, notifies the Borrower of the Change in Law giving rise to such increased costs or reductions, and of such Lender's or the Issuing Lender's or such other Recipient's intention to claim compensation therefor (except that if the Change in Law giving rise to such increased costs or reductions is retroactive, then the nine-month period referred to above shall be extended to include the period of retroactive effect thereof).
- (e) <u>Survival</u>. All of the obligations of Borrower under this <u>Section 3.6</u> shall survive the termination of the Commitments and the repayment, satisfaction or discharge of all obligations under any Loan Document.
 - 11. <u>Section 3.10(e) Taxes. Section 3.10(e) of the Credit Agreement is hereby amended to read as follows:</u>
- (e) Any and all payments by or on account of any obligation of the Borrower under any Loan Document shall be made without deduction or withholding for any Taxes, except as required by applicable law. If any applicable law (as determined in the good faith discretion of an applicable withholding agent) requires the deduction or withholding of any Tax from any such payment by a withholding agent, then the applicable withholding agent shall be entitled to make such deduction or withholding and shall timely pay the full amount deducted or withheld to the relevant Governmental Agency in accordance with applicable law and, if such Tax is an Indemnified Tax, then the sum payable by the Borrower shall be increased as necessary so that after such deduction or withholding has been made (including such deductions and withholdings applicable to additional sums payable under this

Section) the applicable Recipient receives an amount equal to the sum it would have received had no such deduction or withholding been made. If and when receipt of such payment results in an excess payment or credit to a Lender on account of such Taxes, that Lender shall promptly refund such excess to Borrower.

12. <u>Section 11.2 – Amendments</u>. The paragraph contained in <u>Section 11.2</u> of the Credit Agreement immediately before the last sentence thereof is hereby amended to read as follows:

Notwithstanding the foregoing or anything contained herein or any other Loan Document to the contrary, Administrative Agent (and, if applicable, Borrower) may, without the consent of any Lender, enter into amendments or modifications to this Agreement or any of the other Loan Documents or to enter into additional Loan Documents in order to implement any Benchmark Replacement or any Conforming Changes or otherwise effectuate the terms of <u>Section 3.5</u>, in each case in accordance with the terms of <u>Section 3.5</u>.

- 13. Continuation of Eurodollar Rate Advances. Notwithstanding anything in this Amendment to the contrary, if on the Effective Date (as defined below) there is any outstanding Eurodollar Rate Advance under the Credit Agreement that is bearing interest at a fixed rate determined in relation to the Eurodollar Rate (i.e., the LIBOR Rate) for a Eurodollar Period (as each of those terms are defined in Credit Agreement prior to the effectiveness of this Amendment), then such outstanding Eurodollar Rate Advance shall continue to bear interest at such fixed rate in accordance with the Credit Agreement as in effect immediately prior to the Effective Date until the end of its applicable Eurodollar Period (such Eurodollar Rate Advance shall continue to be subject to the Benchmark Replacement provisions and the prepayment fee provisions of the Credit Agreement as in effect immediately prior to the Effective Date, and for the purpose of the application of payments provision of the Credit Agreement, payments of principal shall be applied last to such Eurodollar Rate Advance). After the end of its applicable Eurodollar Period, such outstanding Eurodollar Rate Advance that had been bearing interest at a fixed rate determined in relation to the Eurodollar Rate shall bear interest in accordance with the interest rate provisions of the Credit Agreement as amended by this Amendment (including the provisions therein that require Borrower to provide the Administrative Agent with prior notice in the event Borrower wishes to select a SOFR Advance thereunder).
- 14. <u>Schedule 1.1 (Lender Commitments/Pro Rata Shares)</u>. <u>Schedule 1.1</u> to the Credit Agreement is hereby amended to read as set forth on <u>Annex I</u> to this Amendment
- 15. <u>Exhibit F -Request for Borrowing and Exhibit G Request for Continuation/Conversion</u>. Exhibit F to the Credit Agreement (Request for Borrowing) and Exhibit G to the Credit Agreement (Request for Continuation/Conversion) are hereby amended and restated in their entirety to read as set forth in <u>Annex 2</u> and <u>Annex 3</u> hereto, respectively.
- 16. <u>Effectiveness</u>. This Amendment shall become effective on the date that each of the following conditions shall have been satisfied in a manner reasonably satisfactory to the Lenders (the "<u>Effective Date</u>"):
- (a) The Administrative Agent shall have received all of the following in form and substance reasonably satisfactory to the Administrative Agent and its legal counsel:
 - (i) duly executed counterparts of this Amendment signed by the parties hereto;
- (ii) a duly executed Seventh Amended and Restated Note for Wells Fargo, as the sole Lender, reflecting its revised Commitment; and

- (iii) resolutions of Borrower authorizing the increase in the commitment provided hereunder, the execution and delivery of this Amendment and the Seventh Amended and Restated Note described in the foregoing clause (ii), and such other matters relating hereto as Borrower's board of directors shall determine.
- (b) An upfront fee with respect to the Amendment previously agreed to between the Administrative Agent and Borrower shall have been paid.
- (c) The reasonable costs and expenses of the Administrative Agent in connection with the preparation of this Agreement and each of the other Loan Documents prepared in connection therewith payable pursuant to Section 11.3 of the Credit Agreement, and invoiced to Borrower prior to the Effective Date, shall have been paid.
- (d) All legal matters relating to the Loan Documents shall be reasonably satisfactory to McGuireWoods LLP, special counsel to the Administrative Agent.
- 17. <u>Integration; Loan Document</u>. Except as specifically amended by this Amendment, all terms and conditions of the Credit Agreement remain in full force and effect, without waiver or modification. This Amendment and the Credit Agreement shall be read together, as one document. This Amendment shall constitute a Loan Document.
- Representations and Warranties; No Defaults. Borrower hereby represents that all representations and warranties contained in Article 4 of the Credit Agreement (except to the extent such representations and warranties expressly speak as of a prior date, in which case they were true and correct in all material respects on and as of such earlier date) are and will be true and correct in all material respects on and as of the Effective Date and reaffirms all covenants set forth therein. Borrower further certifies that as of the Effective Date and after giving effect to this Amendment, there exists no Default or Event of Default, each as defined in the Credit Agreement.
- 19. <u>Governing Law.</u> THIS AMENDMENT SHALL BE GOVERNED BY, AND CONSTRUED AND ENFORCED IN ACCORDANCE WITH, THE LAWS OF CALIFORNIA APPLICABLE TO CONTRACTS MADE AND PERFORMED IN CALIFORNIA.
- 20. <u>Counterparts</u>. This Amendment may be executed in any number of counterparts and any party hereto may execute any counterpart, each of which when executed and delivered will be deemed to be an original and all of which counterparts of this Amendment, when taken together, will be deemed to be but one and the same instrument and execution of any such counterpart may be evidenced by facsimile or other electronic means of transmission of the signature of such party.

[Remainder of page intentionally left blank; signature pages follow.]

IN WITNESS WHEREOF, the parties hereto have caused this Amendment to be executed as of the day and year first above written.

BORROWER:

AMERICAN STATES WATER COMPANY, a California corporation

By: /s/ Eva G. Tang

Name: Eva G. Tang

Title: Senior Vice President - Finance, Chief Financial

Officer, Corporate Secretary and Treasurer

ADMINISTRATIVE AGENT:

WELLS FARGO BANK, NATIONAL ASSOCIATION, as Administrative Agent

By: <u>/s/ Jonathan Berns</u>

Name: Jonathan Berns Title: Senior Vice President

LENDERS:

WELLS FARGO BANK, NATIONAL ASSOCIATION, as a Lender

By: /s/ Jonathan Berns
Name: Jonathan Berns

Title: Senior Vice President

ANNEX I

SCHEDULE 1.1

LENDER COMMITMENTS/PRO RATA SHARES

<u>Lender</u>	Pro Rata Share	Commitment Amount
Wells Fargo Bank, National Association	100.0%	\$280,000,000 during the Increased Commitment Period
		\$200,000,000 at all other times
Total	100.0%	\$280,000,000 or \$200,000,000, as applicable

ANNEX II

EXHIBIT F

REQUEST FOR BORROWING

[see attached]

EXHIBIT F

REQUEST FOR BORROWING

- 1. This REQUEST FOR BORROWING is executed and delivered by AMERICAN STATES WATER COMPANY, a California corporation ("Borrower"), to WELLS FARGO BANK, NATIONAL ASSOCIATION, as Administrative Agent, pursuant to that certain Amended and Restated Credit Agreement (as amended, modified or extended, the "Agreement") dated as of June 3, 2005, among Borrower, the Lenders that are parties thereto, and Wells Fargo Bank, National Association, as Administrative Agent and Arranger. Any terms used herein and not defined herein shall have the meanings set forth for such terms in the Agreement.
 - 2. Borrower hereby requests a Borrowing pursuant to the Agreement as follows:
 - (a) AMOUNT OF REQUESTED BORROWING: \$
 - (b) DATE OF REQUESTED BORROWING:
 - (c) TYPE OF ADVANCES COMPRISING REQUESTED BORROWINGS (Check one box only):

 □ ALTERNATE BASE RATE ADVANCES
 - ☐ SOFR ADVANCES FOR AN INTEREST PERIOD OF MONTHS[1]
 - 3. In connection with the request, Borrower certifies that:
- (a) Now and as of the date of the requested Borrowing, except (i) for representations and warranties which expressly speak as of a particular date or are no longer true and correct as a result of a change permitted by the Agreement or (ii) as disclosed by Borrower and approved in writing by the Requisite Lenders, each representation and warranty made by Borrower in Article 4 of the Agreement (other than Sections 4.4, 4.6 (first sentence), 4.9 and 4.16) will be true and correct in all material respects, both immediately before and after giving effect to such Borrowing, as though such representations and warranties were made on and as of that date;
 - (b) No circumstance or event has occurred that constitutes a Material Adverse Effect since the Closing Date; and
- (c) Other than matters described in Schedule 4.9 to the Agreement or not required as of the Closing Date to be described therein or disclosed by Borrower and approved in writing by the Requisite Lenders, there is no action, suit, proceeding or investigation pending or, to the best knowledge of Borrower (on behalf of Borrower and its Subsidiaries), threatened against or affecting Borrower or Subsidiary of Borrower or any Property of any of them before any Governmental Agency that constitutes a Material Adverse Effect.

[1] Specify whether 1, 3 or 6-month Interest Period.

4. This Request for Borrowing is executed on, hereby certifies, on behalf of Borrower, each and every matter.	, by a Responsible Official of Borrower. The undersigned, in such capacity, er contained herein to be true and correct.
	BORROWER:
	AMERICAN STATES WATER COMPANY, a California corporation
	By: Name: Title:
	-41-

ANNEX III

EXHIBIT G

REQUEST FOR CONTINUATION/CONVERSION

[see attached]

EXHIBIT G

REQUEST FOR CONTINUATION/CONVERSION

1. This REQUEST FOR CONTINUATION/CONVERSION is executed and delivered by AMERICAN STATES WATER
COMPANY, a California corporation ("Borrower"), to WELLS FARGO BANK, NATIONAL ASSOCIATION, as Administrative Agent,
pursuant to that certain Amended and Restated Credit Agreement (as amended, modified or extended, the "Agreement") dated as of June 3,
2005, among Borrower, the Lenders that are parties thereto, and Wells Fargo Bank, National Association, as Administrative Agent and
Arranger. Any terms used herein and not defined herein shall have the meanings set forth for such terms in the Agreement.
2. Pursuant to <u>Section 2.4</u> of the Credit Agreement, Borrower hereby irrevocably requests the [continuation] [conversion] of the Advances specified herein, as follows:
(a) The date of [continuation] [conversion] is, which is a [Banking Day] [U.S. Government Securities
Business Day which is the last day of the applicable Interest Period].
(b) The amount of the Advances to be [converted] [continued] is \$
(c) Type of Conversion/Continuation being requested (check all boxes that apply):

Period of [one] [three] [six] month[s].

[] SOFR Advance of \$_____ with an Interest Period of [one] [three] [six] month[s] [converted to] [continued as] a SOFR Advance with an Interest Period of [one] [three] [six] month[s].

[] Alternate Base Rate Advance of \$_____ converted to a SOFR Advance for an Interest

3. In connection with the request, Borrower certifies that no event has occurred and is continuing that is a Default or Event of Default.

[] SOFR Advance of \$_____ converted to an Alternate Base Rate Advance.

indersigned, in such capacity, hereby certific	es, on behalf of Borrower, each and every matter contained herein to be true and correct. BORROWER:	
	AMERICAN STATES WATER COMPANY, a California corporation	
	By: Name: Title:	

THIRTEENTH AMENDMENT TO AMENDED AND RESTATED CREDIT AGREEMENT

THIS THIRTEENTH AMENDMENT TO AMENDED AND RESTATED CREDIT AGREEMENT (this "<u>Amendment</u>"), dated as of May 8, 2023, is entered into with reference to the Amended and Restated Credit Agreement (as amended, restated, supplemented or otherwise modified from time to time, the "<u>Credit Agreement</u>") dated as of June 3, 2005, by and among AMERICAN STATES WATER COMPANY, a California corporation ("<u>Borrower</u>"), each of the lenders party thereto, and WELLS FARGO BANK, NATIONAL ASSOCIATION, as administrative agent for the Lenders (in such capacity, the "<u>Administrative Agent</u>") and lead arranger. Capitalized terms used herein and not otherwise defined shall have the meanings set forth for such terms in the Credit Agreement. Section references herein are to sections of the Credit Agreement unless otherwise stated.

RECITALS

WHEREAS, Borrower has requested a two-month extension of the scheduled maturity date of the Revolving Facility.

WHEREAS, the Administrative Agent, Wells Fargo, as the sole existing Lender as of the date hereof, and Borrower have agreed to extend the Maturity Date and have agreed to such two-month extension and to amend the Credit Agreement to reflect such extension.

NOW, THEREFORE, for valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the parties hereto agree that the Credit Agreement shall be amended as follows:

- 1. <u>Section 1.1 Defined Term.</u> The definitions of "Increased Commitment Period" and "Maturity Date" contained in <u>Section 1.1</u> of the Credit Agreement are hereby amended to read as follows:
 - "Increased Commitment Period" means the period commencing on the Amendment No. 12 Effective Date and ending on July 23, 2023.
 - "Maturity Date" means the earlier of (a) July 23, 2023 and (b) the termination or cancellation of the Revolving Facility (and all of the Commitments pertaining thereto) pursuant to the terms of this Agreement.
- 2. <u>Effectiveness</u>. This Amendment shall become effective on the date the Administrative Agent shall have received duly executed counterparts of this Amendment signed by the parties hereto (such date, the "<u>Effective Date</u>").
- 3. <u>Integration; Loan Document</u>. Except as specifically amended by this Amendment, all terms and conditions of the Credit Agreement remain in full force and effect, without waiver or modification. This Amendment and the Credit Agreement shall be read together, as one document. This Amendment shall constitute a Loan Document.
- 4. <u>Representations and Warranties; No Defaults</u>. Borrower hereby represents that all representations and warranties contained in Article 4 of the Credit Agreement (except to the extent such representations and warranties expressly speak as of a prior date, in which case they were true and correct

in all material respects on and as of such earlier date) are and will be true and correct in all material respects on and as of the Effective Date and reaffirms all covenants set forth therein. Borrower further certifies that as of the Effective Date and after giving effect to this Amendment, there exists no Default or Event of Default, each as defined in the Credit Agreement.

- 5. <u>Governing Law.</u> THIS AMENDMENT SHALL BE GOVERNED BY, AND CONSTRUED AND ENFORCED IN ACCORDANCE WITH, THE LAWS OF CALIFORNIA APPLICABLE TO CONTRACTS MADE AND PERFORMED IN CALIFORNIA.
- 6. <u>Counterparts</u>. This Amendment may be executed in any number of counterparts and any party hereto may execute any counterpart, each of which when executed and delivered will be deemed to be an original and all of which counterparts of this Amendment, when taken together, will be deemed to be but one and the same instrument and execution of any such counterpart may be evidenced by facsimile or other electronic means of transmission of the signature of such party.

[Remainder of page intentionally left blank; signature pages follow.]

IN WITNESS WHEREOF, the parties hereto have caused this Amendment to be executed as of the day and year first above written.

BORROWER:

AMERICAN STATES WATER COMPANY, a California corporation

By: /s/ Eva G. Tang

Name: Eva G. Tang

Title: Senior Vice President - Finance, Chief Financial

Officer, Corporate Secretary and Treasurer

ADMINISTRATIVE AGENT:

WELLS FARGO BANK, NATIONAL ASSOCIATION, as Administrative Agent

By: /s/ Jonathan Berns

Name: Jonathan Berns

Title: Director

LENDERS:

WELLS FARGO BANK, NATIONAL ASSOCIATION, as a Lender

By: <u>/s/ Jonathan Berns</u> Name: Jonathan Berns

Title: Director

Exhibit 31.1

Certification pursuant to Section 302 of the Sarbanes-Oxley Act of 2002 for AWR

I, Robert J. Sprowls, certify that:

- 1) I have reviewed this quarterly report on Form 10-Q for the period ended March 31, 2023 of American States Water Company (referred to as "the Registrant");
- Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the Registrant as of, and for, the periods presented in this report;
- The Registrant's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the Registrant and
 - designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that a) material information relating to the Registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
 - designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
 - evaluated the effectiveness of the Registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
 - disclosed in this report any change in the Registrant's internal control over financial reporting that occurred during the Registrant's most recent fiscal quarter (the Registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the Registrant's internal control over financial reporting.
- The Registrant's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the Registrant's auditors and the audit committee of Registrant's board of directors (or persons performing the equivalent function):
 - all significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the Registrant's ability to record, process, summarize and report financial information; and
 - any fraud, whether or not material, that involves management or other employees who have a significant role in the Registrant's internal controls over financial reporting.

May 10, 2023 /s/ ROBERT J. SPROWLS Dated:

Robert J. Sprowls

President and Chief Executive Officer

Exhibit 31.1.1

Certification pursuant to Section 302 of the Sarbanes-Oxley Act of 2002 for GSWC

I, Robert J. Sprowls, certify that:

- 1) I have reviewed this quarterly report on Form 10-Q for the period ended March 31, 2023 of Golden State Water Company (referred to as "GSWC");
- 2) Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3) Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of GSWC as of, and for, the periods presented in this report;
- 4) GSWC's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for GSWC and have:
 - a) designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that
 material information relating to GSWC, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the
 period in which this report is being prepared;
 - designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to
 provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance
 with generally accepted accounting principles;
 - c) evaluated the effectiveness of GSWC's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
 - d) disclosed in this report any change in GSWC's internal control over financial reporting that occurred during GSWC's most recent fiscal quarter (GSWC's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, GSWC's internal control over financial reporting.
- 5) GSWC's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the GSWC's auditors and the audit committee of GSWC's board of directors (or persons performing the equivalent function):
 - a) all significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the GSWC's ability to record, process, summarize and report financial information; and
 - b) any fraud, whether or not material, that involves management or other employees who have a significant role in GSWC's internal controls over financial reporting.

Dated: May 10, 2023 By: /s/ ROBERT J. SPROWLS

Robert J. Sprowls President and Chief Executive Officer

Exhibit 31.2

Certification pursuant to Section 302 of the Sarbanes-Oxley Act of 2002 for AWR

I, Eva G. Tang, certify that:

- 1) I have reviewed this quarterly report on Form 10-Q for the period ended March 31, 2023 of American States Water Company (referred to as "the Registrant");
- 2) Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3) Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the Registrant as of, and for, the periods presented in this report;
- 4) The Registrant's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the Registrant and have:
 - a) designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that
 material information relating to the Registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during
 the period in which this report is being prepared;
 - b) designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
 - c) evaluated the effectiveness of the Registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
 - d) disclosed in this report any change in the Registrant's internal control over financial reporting that occurred during the Registrant's most recent fiscal quarter (the Registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the Registrant's internal control over financial reporting.
- 5) The Registrant's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the Registrant's auditors and the audit committee of Registrant's board of directors (or persons performing the equivalent function):
 - a) all significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the Registrant's ability to record, process, summarize and report financial information; and
 - b) any fraud, whether or not material, that involves management or other employees who have a significant role in the Registrant's internal controls over financial reporting.

Dated: May 10, 2023 By: /s/ EVA G. TANG

Eva G. Tang Senior Vice President-Finance, Chief Financial Officer, Corporate Secretary and Treasurer

Exhibit 31.2.1

Certification pursuant to Section 302 of the Sarbanes-Oxley Act of 2002 for GSWC

I, Eva G. Tang, certify that:

- 1) I have reviewed this quarterly report on Form 10-Q for the period ended March 31, 2023 of Golden State Water Company (referred to as "GSWC");
- 2) Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3) Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of GSWC as of, and for, the periods presented in this report;
- 4) GSWC's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for GSWC and have:
 - a) designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that
 material information relating to GSWC, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the
 period in which this report is being prepared;
 - designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to
 provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance
 with generally accepted accounting principles;
 - c) evaluated the effectiveness of GSWC's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
 - d) disclosed in this report any change in GSWC's internal control over financial reporting that occurred during GSWC's most recent fiscal quarter (GSWC's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, GSWC's internal control over financial reporting.
- 5) GSWC's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to GSWC's auditors and the audit committee of GSWC's board of directors (or persons performing the equivalent function):
 - a) all significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect GSWC's ability to record, process, summarize and report financial information; and
 - any fraud, whether or not material, that involves management or other employees who have a significant role in GSWC's internal controls over financial reporting.

Dated: May 10, 2023 By: /s/ EVA G. TANG

Eva G. Tang Senior Vice President-Finance, Chief Financial Officer and Secretary

Exhibit 32.1

Certification pursuant to Section 906 of the Sarbanes-Oxley Act of 2002 (18 U.S.C. Section 1350)

In connection with the Quarterly Report of American States Water Company and Golden State Water Company (the "Registrant") on Form 10-Q for the quarter ended March 31, 2023, as filed with the Securities and Exchange Commission on the date hereof (the "Report"), I, Robert J. Sprowls, Chief Executive Officer of the Registrant, certify, pursuant to 18 U.S.C. § 1350, as adopted pursuant to § 906 of the Sarbanes-Oxley Act of 2002, that, to the best of my knowledge:

- (1) The Report fully complies with the requirements of section 13(a) or 15(d) of the Securities Exchange Act of 1934; and
- (2) The information contained in the Report fairly presents, in all material respects, the financial condition and results of operations of the Registrant.

/s/ ROBERT J. SPROWLS

Robert J. Sprowls

President and Chief Executive Officer

Date: May 10, 2023

Exhibit 32.2

Certification pursuant to Section 906 of the Sarbanes-Oxley Act of 2002 (18 U.S.C. Section 1350)

In connection with the Quarterly Report of American States Water Company and Golden State Water Company (the "Registrant") on Form 10-Q for the quarter ended March 31, 2023, as filed with the Securities and Exchange Commission on the date hereof (the "Report"), I, Eva G. Tang, Chief Financial Officer of the Registrant, certify, pursuant to 18 U.S.C. § 1350, as adopted pursuant to § 906 of the Sarbanes-Oxley Act of 2002, that, to the best of my knowledge:

- (1) The Report fully complies with the requirements of section 13(a) or 15(d) of the Securities Exchange Act of 1934; and
- (2) The information contained in the Report fairly presents, in all material respects, the financial condition and results of operations of the Registrant.

/s/ EVA G. TANG

Eva G. Tang

Senior Vice President-Finance, Chief Financial Officer,

Corporate Secretary and Treasurer

Date: May 10, 2023